### OAK PARK AND RIVER FOREST HIGH SCHOOL DISTRICT 200

### BOARD OF EDUCATION MEETING 201 North Scoville Avenue, Room 213, Oak Park, Illinois 60302 Thursday, February 24, 2011 – 7:00 p.m. Revised Meeting Agenda

7:00 p.m.	I.	Call to Order, Pledge of Allegiance, and Roll Call	Dr. Dietra D. Millard
7:05 p.m.	II.	Changes to the Agenda	Dr. Dietra D. Millard
7:10 p.m.	III.	Introductions	Dr. Dietra D. Millard
		A. Introduction of Visitors	
7:20 p.m.	IV.	Public Comment	Dr. Dietra D. Millard
7:30 p.m.	V.	Board of Education Members	Dr. Dietra D. Millard
		<ul><li>A. Status of F.O.I.A. Requests</li><li>B. Board of Education Comments</li></ul>	
7:45 p.m.	VI.	School Reports and Student Life	Dr. Dietra D. Millard
		<ul> <li>A. Student Council Report</li> <li>B. Principal's Report</li> <li>C. Certification of Mid-Year Graduates</li> <li>D. Student Discipline</li> </ul>	Katrina Vogel Nathaniel L. Rouse Nathaniel L. Rouse Action Nathaniel L. Rouse Action
8:00 p.m.	VII.	District, Community and State Reports	Dr. Dietra D. Millard
		<ul> <li>A. Internal District Committees/Liaison Representative Citizens' Council PTO Huskies Boosters' Club Alumni Association Tradition of Excellence Concert Tour Association APPLE Faculty Senate Executive Committee</li> <li>B. External Liaison Reports</li> <li>C. Superintendent's Report</li> </ul>	ve Reports Co-Chairs, Terry Finnegan Amy McCormack Amy McCormack Amy McCormack Sharon Patchak-Layman Sharon Patchak-Layman Dr. Ralph H. Lee James Paul Hunter Board Members Dr. Steven T. Isoye
8:20 p.m.	VIII.	Consent Items	Dr. Dietra D. Millard
		<ul> <li>A. Approval of the Check Disbursements and Financial dated February 24, 2010</li> <li>B. Approval of the Monthly Financial Reports</li> <li>C. Approval of the Treasurer's Report</li> <li>D. Approval of Special Education Bus Bid</li> <li>E. Approval of Field Trip and Activity Bus Bid</li> <li>F. Approval of PTAB Resolution</li> </ul>	al Resolutions Action Action Action Action Action Action

		Adiom	rnment Dr. Dietra D.	Millard
		student of	ve to enter closed session for the purpose of discussinglitigati discipline,collective bargaining and/or negotiations, and employment and/or dismissal of personnel.	
9:45 p.m.	XV.	Closed	Session Dr. Dietra D.	. Millard
		B.	Declaration that the audio tapes dated May 2009 be destroyed Discussion of Non-agenda Items	Information
		А.	Approval of Open Minutes and Closed Session Minutes of January 27, February 8, and February 15, 2011, and a	Action
9:40 p.m.	XIV.	Other	Dr. Dietra D.	. Millard
9:35 p.m.	XIII.	<b>Negoti</b> A.	ations John C. Allen, IV/Amy Mc Approval of B&G Letters of Agreement	Cormack Action
9:20 p.m.	XII.	<b>Instru</b> A.	ctionDr. Ralph H. Lee/Terry Finnegan/PhilipApproval of DVR Intergovernmental Agreement	<b>M. Prale</b> Action
		A. B.	Approval of Personnel Recommendations Approval of Sabbatical Leaves	Action Action
9:10 p.m.	XI.	Huma	n Resources John C. Allen/Amy McCormack/J	ohn C. Allen
		А. В.	Acceptance of Gifts and Donations Approval of Construction Management Contract	Action Action
8:50 p.m.	Х.	Finan		
		D.	Amendment of Policy 4122, Substitute Teachers	Action
		D. C.	Approval of Policy 4105, Equal Opportunity and Minority Recruitment, for First Reading	Action
		А. В.	Approval of Policy 3310, Contracts/Purchasing, for First Reading Approval of Policy 3820, Energy Management, for First Reading	Action Action
8:35 p.m.	IX.	Policy	Dr. Dietra D. Millard/Sharon Patchak-Layman/Dr. Ste	ven T. Isoye
		I.	Approval of Student Fees 2011-12	Action
		G. H.	Approval of Architectural Services Contract Approval of Office Supply Contract Extension	Action Action

### Oak Park and River Forest High School District 200 201 North Scoville Avenue • Oak Park, IL 60302-2296

TO:	OPRF District 200 Board of Education	
FROM:	Principal	
DATE:	February 24, 2011	
SUBJECT:	Certification of January Graduates	ACTION

The students whose names appear below have satisfied graduation requirements of the Consolidated High School District #200, Oak Park and River Forest High School. Your approval is requested to certify their completion of the requirements for diplomas as mid-year graduates.

Richard Luis Berger	110422
Mary E. Bixler	100195
Imani Jade Kendra Branch	110580
Alexandria Olivia Burnley	100370
Wendi A. Castrejon	116471
Arianna Douchand	111308
Dionte Raphael Evans	115999
Marita Anne Farruggia	116735
Mauricio Fernandez	100840
Danecia Talon Fife	116777
Meghan Gardner	115988
Samantha A. Gogo	082595
Robert L. Gholston	100960
Michael R. Hazinski	101147
Taja-Nia Horne	116633
Ryan J. Makely	116278
Melissa Marti	113155
Paige McCulloch-Burton	113280
Veronica Camilla Quam	116714
Jamie Raisor	116185

Motion:

Move to certify the January 2011 Graduates as presented.

Roll call vote.

Agenda Item No. VI. C.

201 North Scoville Avenue • Oak Park, IL 60302-2296

TO: Board of Education

FROM: Cheryl L. Witham

DATE: February 24, 2011

RE: Approval of Check Disbursements and Financial Resolutions

### BACKGROUND

It is a requirement that the Board of Education accepts and approves the check disbursements.

### SUMMARY OF FINDINGS

Attached are the check disbursement lists for February 24, 2011.

### **RECOMMENDATIONS (OR FUTURE DIRECTIONS)**

MOTION: To approve the February 24, 2011 check disbursement listing as presented.

ROLL CALL VOTE

AGENDA ITEM VIII. A.

### RESOLUTION RATIFYING AND CONFIRMING EXECUTION OF CERTAIN VOUCHERS AND PAYMENT OF CERTAIN BILLS AND EXPENSES

Be it resolved by the Board of Education of the Oak Park and River Forest High School, District Number 200, Cook County, Illinois, as follows:

- SECTION 1: That this Board of Education does hereby ratify and confirm the execution of the vouchers for this date of February 24, 2011 by the President and Secretary of this Board of Education, copies of which are attached hereto.
- SECTION 2: That this Board of Education does hereby ratify and confirm that the payment of the bills and expenses were covered by the vouchers attached hereto.
- SECTION 3: This resolution shall be in full force and effect upon its adoption.

ADOPTED this 24<sup>th</sup> day of February, 2011

President of the Board of Education

Secretary of the Board of Education

3frdt101.p 04.10.12.00.02-010071

### Oak Park, IL HARRIS CHECKS (Dates: 02/01/11 - 02/24/11)

2:29 PM

02/16/11 PAGE: 1

	CHECK		INVOICE		AMOUNT
NUMBER		VENDOR	DESCRIP	accrual	50.00
		ACCESS CREDIT UNION	-	accrual	145.66
		AMERIPRISE FINANCIAL SERVICES	-	accrual	85.56
		AMERIPRISE FINANCIAL SERVICES	-	accrual	60.00
		AMERIPRISE FINANCIAL SERVICES	-	accrual	97.30
		BLATT, HASENMILLER, LEIBSKER&MOORE		accrual	125.08
		CHEREDNYK, OLEH	-		164.25
		CLERK OF COURT,	· ·	accrual	554.00
		CONNECTICUT - CCSPC		acerual	624.55
		DIVERSIFIED COLLECTION SERVICES, IN			2,009.36
	02/18/2011			accrual	6,804.20
	02/18/2011		-	accrual	
		FIDELITY INVESTMENTS		accrual	2,554.70
		FIDELITY INVESTMENTS	-	accrual	11,366.80
		FIDELITY INVESTMENTS	-	accrual	65.32
		FIRST INVESTORS		accrual	349.26
		FIRST INVESTORS	-	accrual	1,300.00
		FREEDMAN ANSELMO LINDBERG AND RAPPE			444.75
152049	02/18/2011	GENERAL REVENUE CORPORATION	-	accrual	394.56
152050	02/18/2011	GLENN STEARNS CHAPTER 13 TRUSTEE	Payrol1	accrual	357.01
152051	02/18/2011	GREAT AMERICAN LIFE INS CO	Payroll	accrual	92.22
152051	02/18/2011	GREAT AMERICAN LIFE INS CO	Payroll	accrual	400.00
152052	02/18/2011	HUNTER, CARLA	Payroll	accrual	280.00
152053	02/18/2011	IRS			0.00
152054	02/18/2011	IRS	Payroll	accrual	8.91
152054	02/18/2011	I R S	Payroll	accrual	6.21
152054	02/18/2011	IRS	Payroll	accrual	3.08
152054	02/18/2011	IRS	Payroll	accrual	13.15
152054	02/18/2011	IRS	Payroll	accrua1	3.08
152054	02/18/2011	IRS	Payroll	accrual	8.91CR
152054	02/18/2011	IRS	Payroll	accrual	6.21CR
152054	02/18/2011	IRS	Payroll	accrual	3.08CR
152054	02/18/2011	IRS	Payroll	accrual	13.15CR
152054	02/18/2011	IRS	Payroll	accrual	3.08CR
152054	02/18/2011	IRS	Payroll	accrual	16,208.95
152054	02/18/2011	IRS	Payroll	accrual	3,904.39
152054	02/18/2011	IRS	Payroll	accrual	498.35
	02/18/2011		Payroll	accrual	146,160.06
152054	02/18/2011	IRS	Payroll	accrual	52.80
	02/18/2011		- Payroll	accrual	0.00
	02/18/2011		Payroll	accrual	17,481.34
	02/18/2011			accrual	210.31
	02/18/2011		-	accrual	23,927.54
	02/18/2011		-	accrual	17,481.34
	02/18/2011		-	accrual	210.31
			Payroll		159.66
	-		Payroll		16,292.21
			Payroll		0.00
			Payroll		37,942.81
			Payroll		2,123.13
			-	accrual	153.21
			Payroll		353.21
		LINCOLN INVESTMENT PLANNING, INC.	-		250.00
			Payroll		9,137.13
			Payroll		30,985.50
			Payroli Payroll		30,985.50
152061			Payroil Payroll	auuruar	2,549.46

### 3frdt101.p

### Oak Park, 1L HARRIS CHECKS (Dates: 02/01/11 - 02/24/11)

2:29 PM

PAGE:

2

02/16/11

CHECK	CHECK		INVOICE	
NUMBER	DATE	VENDOR	DESCRIPTION	AMOU
1,52061	02/18/2011	M G TRUST COMPANY	Payroll accrual	167.71
152062	02/18/2011	METLIFE	Payroll accrual	193.46
152062	02/18/2011	METLIFE	Payroll accrual	375.00
152063	02/18/2011	NCPERS GROUP LIFE INS (#1985)	Payroll accrual	232.00
152064	02/18/2011	OPRFHS/FACULTY SENATE DUES	Payroll accrual	7,976.52
152065	02/18/2011	PACIFIC LIFE	Payroll accrual	509.46
152065	02/18/2011	PACIFIC LIFE	Payroll accrual	2,089.61
152066	02/18/2011	RIVER FOREST COMMUNITY CENTER	Payroll accrual	9,820.00
152067	02/18/2011	SEIU LOCAL 73	Payroll accrual	42.00
152067	02/18/2011	seiu local 73	Payroll accrual	2,468.49
152068	02/18/2011	STATE OF ILLINOIS	Payroll accrual	6.36
152068	02/18/2011	STATE OF ILLINOIS	Payroll accrual	6.36
152068	02/18/2011	STATE OF ILLINOIS	Payroll accrual	394.99
152068	02/18/2011	STATE OF ILLINOIS	Payroll accrual	36.96
152068	02/18/2011	STATE OF ILLINOIS	Payroll accrual	52,768.48
		STATE OF ILLINOIS	Payroll accrual	36.96
		STATE OF ILLINOIS	Payroll accrual	25.00
	02/18/2011		Payroll accrual	7,918.33
	02/18/2011		Payroll accrual	19.15
	02/18/2011		Payroll accrual	480.17
	02/18/2011		Payroll accrual	5,940.64
	02/18/2011		Payroll accrual	14.36
	02/18/2011		- Payroll accrual	16.89
	02/18/2011		Payroll accrual	401.17
		TEACHERS RETIREMENT SYSTEM	Payroll accrual	12.62
		TEACHERS RETIREMENT SYSTEM	Payroll accrual	85,641.88
		TEACHERS RETIREMENT SYSTEM	Payroll accrual	4,552.38
		TEACHERS RETIREMENT SYSTEM	Payroll accrual	204.51
		TEACHERS RETIREMENT SYSTEM	- Payroll accrual	5,295.51
		TEACHERS RETIREMENT SYSTEM	Payroll accrual	280.90
		TEACHERS RETIREMENT SYSTEM	Payroll accrual	360.77
		TOM VAUGHN STANDING TRUSTEE	Payroli accrual	129.50
		U.S. DEPART. OF EDUCATION	Payroll accrual	83.39
		U.S. DEPART. OF EDUCATION	Payroll accrual	118.61
		WADDELL & REED	Payrol1 accrual	706.68
		WADDELL & REED	Payroll accrual	1,004.75
		A R T STUDIO CLAY CO	Cinnamonstone	595.00
		AIRGAS NORTH CENTRAL	INVOICE DATE 1/31/11	103.00
		ALARM DETECTION SYSTEMS, INC.		186.00
192077	02/24/2011	ALMER DIRECTION OIDTING, 1961	Alarm Monitoring QRTLY MAR -	
			MAY	
160077	02/24/2011	ALARM DETECTION SYSTEMS, INC.	2010/11 - Open P.O Fire	186.00
196011	02/24/2011	ABRAN DEFECTION DEDEMAND, INC.	Alarm Monitoring QRTLY MAR -	
			MAY	
150070	00/04/0011	AMERICAN PAYROLL ASSOCIATION	1 YR RENEWAL A.W.	219.00
		APPLAUSE LEARNING RESOURCES	WLD Italian Resource	63.85
192019	02/24/2011	APPLAGE DEARNING RESOURCES	Materials	
150000	00/06/0051	AQUA PURE ENTERPRISES, INC.		563.41
132080	V717819011	AGOU LOUR BUILVEUTORS' THC.	Chemicals	5.551.44
100001	00/08/001	ADAMADY	2010/11 - Maintenance &	1,415.52
102081	02/24/2011	ANARANIN	Custodial Uniforms JAN 2011	ميدين من من من من من من
			INVOICES	
1 50000	03/01/0014	רס מסועגם אווה המעגם	CARLSON BOOK ORDER BAKER AND	410.35
152082	02/24/2011	BAKER AND TAYLOR CO	TAYLOR	420.33
		DAUGD MUD DAVI OF CO	LOPEZ OPEN BOOK ORDER	1,003.88
152082	02/24/2011	BAKER AND TAYLOR CO	STEEL MAT DRAG FOR GROUNDS	261.32

3frdt101.p

### Cak Park, IL HARRIS CHECKS (Dates: 02/01/11 - 02/24/11)

2:29 PM

02/16/11 PAGE:

3

CHECK	CHECK		INVOICE	
	DATE	VENDOR	DESCRIPTION	AMOUI
		BEACON ATHLETICS	STEEL MAT DRAG FOR GROUNDS	114.91
		BLICK ART MATERIALS	Misc. Clay and Wheelthrowing	150.49
133004	00,00,000		supplies	
152085	02/24/2011	BOSCO, JOHN	B B-BALL 2/9/11	47.00
		BROWN, JAKE	B B-BALL 2/11/11	86.00
		CALUMET PHOTO/CPI	Enlarger timer and PC cord	224,64
		CALUMET PHOTO/CPI	Developer	34.19
		CALUMET PHOTO/CPI	Developer	280.50
		CAMBIUM LEARNING INC	Classroom Supplies REWARDS	168.14
102000	0212412011	CANDIGH IDAMAANO ING	SUPPLY FOR SPED	
1 6 2 4 9 9	02/24/2011	CARLSEN'S ELEVATOR SERVICE	2010/11 - Open Purchase Order	472.50
192009	02/24/2013	CHILDER O DESTRUCT SERVICE	- Elevator Repairs &	
			Maintenance	
1 = 2 0 0 0	00/06/0011	CENTER FOR PSYCHOLOGICAL SERVICES	CONSULTATION 2/4/11 SPED	500.00
			Camera lens repair	260.00
		CENTRAL CAMERA CHENEY, ELIZABETH	SECTIONAL INSA	90.00
152092	02/24/2011	CHENEY, BEISABEIN	JUDGING/TOURNEY & DOWNERS	
			GROVE SO	
			B B-BALL 2/5 & 9/11	70,00
		CLARK, TOM	B B-BALL 2/11/11	70.00
		CLARK, TOM	BTRACK 2/8/11	57.00
		CLEMENT, EARL	JUDGE FOR CONCERT BAND	200.00
152095	02/24/2011	COFFMAN, TIM	AUDITIONS	
			DRIVER'S ED SILVER CHEVY	279,73
152096	02/24/2011	CURRIE MOTORS CHEVROLET		275173
			IMPALA TIRE ROTATION, ETC	
			1/26/11 WHITE IMPALA & SILVER IMPALA	954.92
152096	02/24/2011	CURRIE MOTORS CHEVROLET		239.22
			REPAIRS DRIVER'S ED SILVER CHEVY TIRE	354.22
152096	02/24/2011	CURRIE MOTORS CHEVROLET		104.02
			REPAIR 1/11/11	57.00
152097	02/24/2013	DAVELIS, CRAIG	B B-BALL 2/11/11	49.90
152098	02/24/2011	DEMCO	SPINE LABELS (MCKENZIE	49.30
			REQUEST)	214 5
152098	02/24/2011	DEMCO	BOOK LAMINATE (MCKENZIE	214.62
			REQUEST)	60.4 O
152099	02/24/2011	DUNBAR ARMORED, INC.	INVOICE DATE 2/1/11 JAN	604.0
			EXCESS & FEB MONTHLY FEE	
152100	02/24/2011	EYE ON EDUCATION	WLD Supplies for DiVerdi	35.95
152101	. 02/24/2013	FIELDTURF USA, INC.	SWEEPRIGHT DEBRIS SWEEPER	1,175.00
152102	02/24/2011	FORREST, MIKE	G B-BALL 2/5/11	86.00
152103	8 02/24/201:	FRANCZEK RADELET & ROSE	GENERAL SCHOOL WORK &	6,107.50
			LABOR/EMPLOYMENT THROUGH	
			12/31/10	
152104	1 02/24/2013	L GEPPETTO'S	SPED 2/25/11 SPED TEAM	262.04
			BASKETBALL GAME REFRESHMENTS	
152105	5 02/24/2013	L GIBBS, MIKE	BTRACK 2/12/11	87.00
		L GILMER, DAVID	SPOKEN WORD/SOPH SLAM AND	354.0
			MENTORING SERVICES	
152107	7 02/24/201	l grainger		0.0
	3 02/24/201			0.0
	02/24/201		OPEN PO FOR PLUMBING PARTS	91.8
= * •			FROM GRAINGER	
15210	9 02/24/201	1 GRAINGER	OPEN PO FOR PLUMBING PARTS	46.4
, J & & J .			FROM GRAINGER	
150104	9 02/24/201	3 GRAINGER	2010-11 Open P.O Lights &	14.8
T 7 7 7 0 ;	, valasiavi		Lamps	

3frdt101.p 04.10.12.00.02-010071

2:29 PM

4

02/16/11

PAGE:

	CHECK		INVOICE	
	DATE		DESCRIPTION	AMOUN1
152109	02/24/2011	GRAINGER	2010-11 Open P.O Lights &	141.73
150100	00/04/0011		Lamps 2010-11 Open P.O Heating &	38.56
152109	02/24/2011	GRAINGER	Ventilating	50.50
150100	00 (04/0011		2010-11 Open P.O Heating &	110.41
122103	02/24/2011	GRAINGER	Ventilating	770.41
1 5 9 1 9 9	00/04/0011		OPEN PO FOR MAINTENANCE	915.58
127108	02/24/2011	GRAINGER	SUPLIES	540.00
150100	00/04/0011	(D. ). TM() TD	MOBILE CHAIR TRUCK	563.86
	02/24/2011		MOBILE CHAIR TRUCK	563.86CF
	02/24/2011		OPEN PO FOR PLUMBING PARTS	47.80
122103	02/24/2011	GRAINGER	FROM GRAINGER	41.00
151100	02/24/2011	CD 1 THODD	2010-11 Open P.O Heating &	173.04
197109	UZ/Z4/ZUIL	GRAINGER	Ventilating	712.01
160100	02/24/2011	CDATNCED	2010-11 Open P.O Heating &	6.98
192109	02/24/4011	GRAINGER	Ventilating	0,00
160110	00/04/0011	GREEN, DAN	B B-BALL 2/11/11	47.00
		HARPER, FELIX	B B-BALL 2/11/11	57.00
		HARFER, FELIA HARRIS, DONALD	B B-BALL 2/9/11	47.00
			B B-BALL 2/5/11	57.00
			ANNUAL PO FOR OCE 4511 COPIER	813.28
104114	02/24/2011	RARKIS DAWK WAPERVILLUS	LEASE PAYMENTS-INVOICE 2/1/11	010100
150115	01/04/0011	HIGHWATER CLAYS	BOOKSTORE SUPPLIES	128.84
		HOUSE, WILLIE	B B-BALL 2/5/11	57.00
		ILLINOIS BATTERY SPECIALISTS	UPS BATTERIES FOR B&G	720.00
		INTERFACE FLOR	Carpet replacements for Julie	
177110	02/24/2011	TITLERING LEON	Freys office, Math Directors	-,
			office, 225 and 298C	
150110	02/24/2011	JOE RIZZA OF NORTH RIVERSIDE	DRIVERS ED TAN FORD FUSION	32.34
,1 ,7 , 1 , 1 , 1	02/24/2021	bol Killan of Kokta Ketakeral	OIL CHANGE 1/31/11	
152120	02/24/2011	JOHNSTON, SHARON	SECTIONAL IHSA	90.00
1, 14, 12, 0	02/24/2022		JUDGING/TOURNEY & DOWNERS	
			GROVE SO	
150101	02/24/2011	JONES, MICHAEL	REIMB FOR DRIVER'S ED FOR	175.00
170101	02/23/2011		К.J.	
152122	02/24/2011	KAGAN PUBLISHING		87.00
			480 pages, BKCL ? \$44 Dr.	
			Spencer Kagan & Miguel	
152123	02/24/2011	KENNEDY, FRANK	MONTHLY MILEAGE & CELL REIMB	100.00
100100			PER CONTRACT - FEB 2011	
152124	02/24/2011	KIDD, SIERRA	SPOKEN WORD AND MOTIVATIONAL	188.00
150101			SERVICES	
152125	02/24/2011	KIMP, JOSE	B B-BALL 2/11/11	47.00
		KONICA-MINOLTA BUSINESS SOLUTIONS U		321.91
			FY2011 INVOICE DATED 2/7/11	
152126	02/24/2011	KONICA-MINOLTA BUSINESS SOLUTIONS U	MONTHLY MAINTENANCE AND	1,215.00
190100			OVERAGE FY2011 INVOICE DATE	
			2/6/11	
152127	02/24/2011	KRANZ INC.	SALT FOR SNOW REMOVAL	829.80
		KRANZ INC.	UPRIGHT VACUUM CLEANER FOR	1,293.50
			3RD SHIFT CUSTODIANS	
152128	02/24/2011	LOPINA, BILL	BTRACK 2/9/11	86.00
			B B-BALL 2/11/11	57.00
			B B-BALL 2/11/11	47.00
		MONTGOMERY, LARRY	G B-BALL 2/5/11	86.00
		NELSON, GREGORY	B B-BALL 2/5/11	57.00

3frdt101.p

04.10.12.00.02-010071

2:29 PM

5

02/16/11

PAGE:

INVOICE CHECK CHECK DESCRIPTION AMOUNT NUMBER DATE VENDOR 47.00 152133 02/24/2011 NIELSEN, GLEN G B-BALL 2/5/11 57.00 G B-BALL 2/5/11 152134 02/24/2011 OLIVER, PAMELA 152135 02/24/2011 PRICE, JIM 57.00 G B-BALL 2/8/11 G B-BALL 2/5/11 57.00 152136 02/24/2011 RAMSEYER, STEPHEN 152137 02/24/2011 RAYA, ALBERT G B-BALL 2/5/11 57.00 B B-BALL 2/8/11 57.00 152138 02/24/2011 SCHMELZER, GREG 86.00 152139 02/24/2011 SYLVESTER, STEVE B B-BALL 2/9/11 47.00 152140 02/24/2011 TURNBULL, LEONARD G B-BALL 2/5/11 152141 02/24/2011 VANDERMOON, SCOTT B B-BALL 2/11/11 47.00 152142 02/24/2011 VELARDE, VINCENTE G B-BALL 2/5/11 47.00 B B-BALL 2/11/11 86.00 152143 02/24/2011 VISCONTI, FRANK 47.00 152144 02/24/2011 WINKLER SR., KIRK G B-BALL 2/5/11 152145 02/24/2011 MC KINNEY, MILTON SPOKEN WORD & MENTORSHIP 280.00 SERVICES 26.30 SPED 152146 02/24/2011 MEREDITH CULLIGAN WATER CO. 35.00 152147 02/24/2011 NATIONAL FORENSIC LEAGUE 3 STUDENT MEMBERSHIPS LESS \$10 CREDIT ON ACCOUNT ACCT#905548 GROUP 5998 FROM 711.55 152148 02/24/2011 NTCOR GAS 1/1-2/1/11 ACCT#503548 GROUP 5315 FROM 4,602.85 152148 02/24/2011 NICOR GAS 1/1 - 2/1/11ACCT#115548 GROUP 5515 FROM 518.48 152148 02/24/2011 NICOR GAS 1/1-2/1/11 152149 02/24/2011 O'CONNOR, MARK JUDGE FOR CONCERT BAND 200.00 AUDITIONS 1.224.00 152150 02/24/2011 OCE DATA PROCESSING FROM 2/1 -4/30/2011 & BILLABLE USAGE FROM 11/1 - 1/31 152151 02/24/2011 PETTY CASH, OPRFHSD #200 REIMB BUSINESS OFFICE PETTY 879.52 CASH 105.80 CUSTODIAL SUPPLIES 152152 02/24/2011 PIKE SYSTEMS 335.98 152152 02/24/2011 PIKE SYSTEMS CUSTODIAL SUPPLIES 152153 02/24/2011 PIONEER PRESS AD #'S 21362 AND 21433 51.20 JUDGE FOR CONCERT BAND 200.00 152154 02/24/2011 PODJASEK, BRANDON AUDITIONS 87.46 152155 02/24/2011 PORTER'S CAMERA STORE Thermometers Quill Brand Wire Step File 36.79 152156 02/24/2011 QUILL CORP. Quill Office Supply Order 22.65 152156 02/24/2011 QUILL CORP. 4.61 152156 02/24/2011 OUILL CORP. Quill Office Supply Order 152156 02/24/2011 QUILL CORP. Quill Brand permanent markers 5.84 Quill Office Supply Order 32.96 152156 02/24/2011 QUILL CORP. 152156 02/24/2011 QUILL CORP. MATH DEPT MARKERS & 23.36 HIGHLIGHTERS 10/15/10 ENGLISH/HISTORY ART 215.00 152157 02/24/2011 R & D BUS COMPANY, INC. INSTITUTE 12/8/10 ENGLISH/SPECIAL ED 260.00 152157 02/24/2011 R & D BUS COMPANY, INC. RED ORCHARD 152158 02/24/2011 RIDDELL/ALL AMERICAN FOOTBALL 15,264.94 RECONDITIONING/REPLACEMENT EQUIPMENT 152159 02/24/2011 ROBBINS SCHWARTZ NICHOLAS LIFTON BILLING THROUGH 1/31/11 11.857.50 152160 02/24/2011 SALKELD SPORTS INC, DIV OF KESSLER' BOYS TENNIS EQUIPMENT 889.40 152160 02/24/2011 SALKELD SPORTS INC, DIV OF KESSLER' Polo shirts for students at 328.60 Rush Oak Park Hospital work program 681.02 152161 02/24/2011 SCHOOL SPECIALTY INC/CPO Poster Paper and paint

3frdt101.p 04.10.12.00.02-010071 2:29 PM

PAGE: 6

02/16/11

CHECK	CHECK		INVOICE	
NUMBER	DATE	VENDOR	DESCRIPTION	AMOUNT
152162	02/24/2011	SIMPLEXGRINNELL LP	SERVICE #19471046 REPAIRS	336.00
			DURING SUMMER CONSTRUCTION	
			8/31/10	
152163	02/24/2011	TARRANT, TOM	REIMB FOR VARIOUS GIRLS TRACK	53.91
			EQT	
152164	02/24/2011	THOMAS PUMP COMPANY	CUST ID# OAK160 PACO MECH	196.00
			SEAL KIT ~ B&G	
152165	02/24/2011	TYPE-TECH, INC.	B&G IBM SERVICE 1/8/11	135.00
152166	02/24/2011	VANGUARD ENERGY SERVICES, LLC	ACCT#400024 SERVICE PERIOD	45,109.39
			1/1-1/31/11	
152167	02/24/2011	VAN DYKE, LINDA	JUDGE FOR CONCERT BAND	200.00
			AUDITIONS	
152168	02/24/2011	VILLAGE OF OAK PARK	CUST#32 NOV & DEC 2010	2,772.83
			VARIOUS DEPT'S GASOLINE	
			EXPENSE	
152168	02/24/2011	VILLAGE OF OAK PARK	CUST#32 JAN 2011 DRIVERS ED	187.86
			GASOLINE EXPENSE	
152169	02/24/2011	WIGHT & COMPANY	PROJECT 05-4825-36 ASBESTOS	774.12
			3-YEAR RE-INSPECTION & 6 MCS	
			SURVEILLANCES - FOR PERIOD	
			1/1 - 1/21/11	
152170	02/24/2011	WILLIAMS, DAN	JUDGE FOR CONCERT BAND	200.00
			AUDITIONS	
152171	02/24/2011	WOODWARD, JUSTIN	JUDGE FOR CONCERT BAND	200.00
			AUDITIONS	
152172	02/24/2011	WORKPLUS OCCUPATIONAL HEALTH-RHC	JAN 2011 SPED PHYSICALS R.M.,	385.00
			T.N. & C.S.	
01000057	02/01/2011	ANDERSON TRAVELS	Wire transfer for India trip	15,000.00
			payment	
01000058	02/09/2011	ILLINOIS DEPARTMENT OF REVENUE	REMITTANCE OF SALES TAX	4,173.00
			COLLECTED IN BKST JANUARY	
			2011	

Totals for checks 685,607.16

PAGE: 7

02/16/11

### FUND SUMMARY

FUND	DESCRIPTION	BALANCE SHEET	REVENUE	EXPENSE	TOTAL
10	EDUCATION FUND	426,303.74	177.20	35,407.26	461,888.20
14	CAFETERIA FUND	8,852.50	0.00	410.84	9,263.34
15	BOOKSTORE FUND	1,571.00	4,173.00	128.84	5,872.84
20	OPERATIONS & MAINTENANCE	28,905.75	0.00	65,876.18	94,781.93
40	TRANSPORTATION FUND	0.00	0.00	2,545.03	2,545.03
50	ILL MUN RET FUND	79,562.00	0.00	0.00	79,562.00
80	TORT IMMUNITY FUND	0.00	0.00	16,039.06	16,039.06
84	ACTIVITY FUND	15,654.76	0.00	0.00	15,654.76
*** F	'und Summary Totals ***	560,849.75	4,350.20	120,407.21	685,607.16

Г	Report	13	Stmnt Date	
Oak Fark, IL	Voided Check Report	Cash Posting	Date	
0	voi	υ	Check Amount Date	
			T Check Date	
			Check # Payee Key Payee Name	ОК ОАК РАВК РТИ
	023		Payee Key	ANTZ OT OAL
3aprpt03.p	04.10.12.00.00-010023		Check #	THAS VITINIIMMOD TO BANK

# CEACT COMMUNITY BANK OF OAK PARK RIV

02/16/2011	02/16/2011			
\$-80.00 02/16/2011 02/16/2011	\$~590.00 02/16/2011 02/16/2011	\$-670.00	\$670.00	Total \$\$ \$-670.00
V 02/16/2011	S V 02/16/2011	2	N	Totals: Bank CBACT
BROWNWELDOZ BROWN, MELINDA	INTERNATOO2 INTERNATIONAL MUSEUM OF S V 02/16/2011	Number Of Checks:	Total Checks:	1
BROWNMEL002	INTERNAT002			
6596	6601			

\* Bud of report \*\*\*\*\*\*\*\*\*\*\*\*\*\*

2:31 PM

### RESOLUTION RATIFYING AND CONFIRMING EXECUTION OF CERTAIN VOUCHERS AND PAYMENT OF CERTAIN BILLS AND EXPENSES

Be it resolved by the Board of Education of the Oak Park and River Forest High School, District Number 200, Cook County, Illinois, as follows:

- SECTION 1: That this Board of Education does hereby ratify and confirm the execution of the vouchers from the Imprest Account for February 24, 2011 by the President and Secretary of this Board of Education, copies of which are attached hereto.
- SECTION 2: That this Board of Education does hereby ratify and confirm that the payment of the bills and expenses were covered by the vouchers attached hereto.
- SECTION 3: This resolution shall be in full force and effect upon its adoption.

ADOPTED this February 24, 2011

President of the Board of Education

Secretary of the Board of Education

04.10.12.00.02-010071

### Oak Park, IL IMPREST CHECKS (Dates: 02/10/11 - 02/16/11)

2:26 PM

PAGE: 1

02/16/11

IMPREST		CHECK		INVOICE
CHECK #	VENDOR	DATE	AMOUNT	DESCRIPTION
31096	CHENEY, PATRICIA	02/16/2011	650.00	IHSA STATE SPEECH COMPETITION
				TRAVEL ADVANCE - 4 COACHES/5
				STUDENTS
31097	PEORIA CIVIC CENTER	02/16/2011	40.00	IHSA STATE SPEECH COMPETITION
				ENTRY FEE
31098	T-MOBILE	02/16/2011	104.09	CURRENT CHARGES FROM 12/25/10
				- 1/ 24/11
31098	T-MOBILE	02/16/2011	129.05	SPED CELL PHONE BILL FROM 1/
				- 2/7/11
31099	VERIZON WIRELESS	02/16/2011	108.98	ACCT#28639289-00001 FOR 1/27
				- 2/26/11
		Totals for checks	1,032.12	

02/16/11 Page:

2

2

### FUND SUMMARY

FUND	DESCRIPTION	BALANCE SHEET	REVENUE	EXPENSE	TOTAL
1.0	EDUCATION FUND	0.00	0.00	819.05	819.05
20	OPERATIONS & MAINTENANCE	0.00	0.00	213.07	213.07
*** F	und Summary Totals ***	0.00	0.00	1,032.12	1,032.12

### RESOLUTION RATIFYING AND CONFIRMING EXECUTION OF CERTAIN VOUCHERS AND PAYMENT OF CERTAIN BILLS AND EXPENSES

Be it resolved by the Board of Education of the Oak Park and River Forest High School, District Number 200, Cook County, Illinois, as follows:

- SECTION 1: That this Board of Education does hereby ratify and confirm the execution of the vouchers from the Student Activity Accounts for February 24, 2011 by the President and Secretary of this Board of Education, copies of which are attached hereto.
- SECTION 2: That this Board of Education does hereby ratify and confirm that the payment of the bills and expenses were covered by the vouchers attached hereto.
- SECTION 3: This resolution shall be in full force and effect upon its adoption.

ADOPTED this February 24, 2011

President of the Board of Education

Secretary of the Board of Education

### 04.10.12.00.02-010071

ACTIVITY

CHECK # VENDOR

GE: 1

.p			Oak Park, IL	2:23 PM	02/16/11
2.00.02-0	10071	ACTIVITY CHECKS	(Dates: 02/04/11	- 02/16/11)	PAGE :
CTIVITY		CHECK		INVOICE	
CHECK #	VENDOR	DATE	AMOUNT	DESCRIPTION	
6586	CHICAGO AUTOMOBILE TRADE ASSOC	02/11/2011	392.00	ADMISSION FEE TO AUTO SHOW 56	
				STUDENTS FIELD TRIP NM & RVH	
6587	COVAL, KEVIN	02/11/2011	300.00	SPOKEN WORD WINTER SHOWCASE	
				FEATURE POET	
6588	DAWSON, LEKEJA	02/11/2011	60.00	SPOKEN WORD ASSISTANCE	
6589	FROSCH TRAVEL INTERNATIONAL	02/11/2011	22,893.00	AIRLINE TICKET FOR ITALIAN	
				EXCHANGE	
6590	LEE, BRENDAN	02/11/2011	1,035.14	REIMB FOR IRELAND TRIP COACH	
				HIRE EXPENSE	
6591	OSAIGBOVO, ITOHAN	02/11/2011	200.00	SPOKEN WORD CONTEST	
				COORDINATOR	
6592	SANTANA, CHRISTINA	02/11/2011	75.00	SPOKEN WORD PROFESSIONAL	
				SERVICES	
6593	TEEN REACH INC	02/11/2011	700.00	MOTIVATIONAL SPEAKER FOR	
				SNOWBALL WEEKEND 2/20/11	

6588	DAWSON, LEKEJA	02/11/2011	60.00	SPOKEN WORD ASSISTANCE
6589	FROSCH TRAVEL INTERNATIONAL	02/11/2011	22,893.00	AIRLINE TICKET FOR ITALIAN
				EXCHANGE
6590	LEE, ERENDAN	02/11/2011	1,035.14	REIMB FOR IRELAND TRIP COACH
				HIRE EXPENSE
6591	OSAIGBOVO, ITOHAN	02/11/2011	200.00	SPOKEN WORD CONTEST
				COORDINATOR
6592	SANTANA, CHRISTINA	02/11/2011	75.00	SPOKEN WORD PROFESSIONAL
				SERVICES
6593	TEEN REACH INC	02/11/2011	700.00	MOTIVATIONAL SPEAKER FOR
				SNOWBALL WEEKEND 2/20/11
6594	WESTGATE FLOWERS	02/11/2011	135.48	Freshmen Class Flower Sale
6595	BATTAGLIA, CHRISTINE	02/16/2011	330.37	REIMB FOR CHEERLEADING END OF
				SEASON PARTY & AWARDS
6596	BROWN, MELINDA	02/16/2011	80.00	REIMB FOR CHEER LUNCH @ STATE
				COMPETITION
6597	DAWSON, LEKEJA	02/16/2011	60.00	SPOKEN WORD WINTER SHOWCASE
				PREP
6598	GOLDEN TOUCH LIMOUSINE	02/16/2011	391.00	ART HISTORY TRIP WASH DC TOUR
				BUS FOR MAR 12
6599	GRAPHIC EDGE	02/16/2011	760.80	BOYS TRACK SPIRIT GEAR
6600	INDUSTRIAL WHOLESALE LUMBER	02/16/2011	411.78	Lumber WINTER MUSICAL
6601	INTERNATIONAL MUSEUM OF SURGICAL SC	02/16/2011	590.00	SCIENCE FIELD TRIP 2/17/11
6602	MIDWEST SERVICE & INSTALLATION	02/16/2011	8,023.52	Wellness purchase for adapted
				душ
6603	NOVEL-TEES	02/16/2011	152.00	BOYS SWIMMING T-SHIRTS
6604	PALOMBI, COLIN	02/16/2011	620.00	SHOOT & EDIT SPOKEN WORD'S
				WINTER SHOWCASE
6605	R & M SPECIALTIES	02/16/2011	2,030.00	UNITY T-SHIRTS (RESPECT)
				SPORTS TEAMS
6606	SANTANA, CHRISTINA	02/16/2011	150.00	SPOKEN WORD WINTER SHOWCASE
				PREP
6607	SUPER CDA	02/16/2011	200.00	JV & V 2011 CHEER CAMP
				DEPOSIT JULY 26-28
6608	XEROX CORP	02/16/2011	126.81	BASE CHARGE JAN 2011 PTO
6609	BROWN, MELODY	02/16/2011	80.00	REIMB FOR CHEER LUNCH @ STATE
				COMPETITION
6610	INTERNATIONAL MUSEUM OF SURGICAL SC	02/16/2011	413.00	SCIENCE FIELD TRIP ON 2/17
				REDUCED ADMISSION FEE

Totals for checks 40,209.90

2

02/16/11

PAGE:

### FUND SUMMARY

FUND DESCRIPTION	BALANCE SHEET	REVENUE	EXPENSE	TOTAL
84 ACTIVITY FUND	40,209.90	0.00	0.00	40,209.90
*** Fund Summary Totals ***	40,209.90	0.00	0.00	40,209.90

### RESOLUTION AUTHORIZING EXECUTION OF CERTAIN VOUCHERS FOR THE MONTH OF JANUARY, 2011

Be it resolved by the Board of Education of the Oak Park and River Forest High School, District Number 200, Cook County, Illinois as follows:

- Section 1: That this Board of Education has approved and does hereby approve the voucher used by its School Treasurer, all pursuant to the powers granted under the Illinois School Code.
- Section 2: That the President and Secretary of this Board of Education be and are hereby authorized to execute and sign on behalf of this Board of Education vouchers with all required information for the following expenditures during the Month of MARCH, 2011:
  - a) Payroll for the employees of this District not to exceed \$4,500,000 for said month.
  - b) Contractual fringe benefits for the employees of this District not to exceed \$700,000 for said month.

Provided however, that all such vouchers to be signed by the President and Secretary of the Board of Education shall be approved as accurate and due and owing by the Chief Financial Officer (or other designated officer) prior to the signing of such vouchers.

Further provided, however, that all such vouchers shall contain information as required by law in order that the School Treasurer can make the appropriate disbursements and entries into the records.

Section 3: This resolution shall be in full force and effect upon its adoption.

ADOPTED this 24<sup>TH</sup> day of February, 2011

President of the Board of Education

Secretary of the Board of Education

201 North Scoville Avenue • Oak Park, IL 60302-2296

TO:	Board of Education
FROM:	Cheryl Witham
DATE:	February 24, 2011
RE:	Financial Reports

### BACKGROUND

It is a requirement that the Board of Education accepts and approves the monthly Financial Reports.

### SUMMARY OF FINDINGS

Attached are the Financial Reports for January, 2011.

### **RECOMMENDATIONS (OR FUTURE DIRECTIONS)**

MOTION: To accept the January, 2011 Financial Reports as presented.

ROLL CALL VOTE

AGENDA ITEM VIII. B.

### Education Fund

	Audited 2009-2010	Fiscal to Date January 31 <u>2010</u>	%	Original Budget 2010-2011	Fiscal to Date January 31 2011	<u>%</u>
Receipts						
Property Taxes	47,315,686	23,780,982	50.3%	43,378,031	20,011,147	46.1%
Other Local Sources	3,617,642	1,881,710	52.0%	2,036,844	2,170,936	106.6%
State Sources	2,834,759	1,734,729	61.2%	1,535,232	1,917,409	124.9% 1
Federal Sources	2,607,278	1,490,079	57.2%	1,199,587	509,543	42,5%
	56,375,365	28,887,500	51.2%	48,149,694	24,609,035	51.1%
Expenditures						
General Instruction	19,919,270	8,740,326	43.9%	21,480,792	9,466,101	44.1%
Special Education	5,555,934	2,452,751	44.1%	5,109,106	2,251,245	44.1%
Adult Education	20,282	6,000	29.6%	20,539	-	0.0%
Vocational Programs	335,859	191,959	57.2%	378,059	219,032	57.9%
Interscholastic Programs	1,963,819	961,910	49.0%	2,189,182	1,031,114	47.1%
Summer School	287,451	164,877	57.4%	313,566	155,526	49.6%
Drivers Education	757,147	327,289	43.2%	765,563	327,587	42.8%
Other Instructional	2,801,472	1,264,845	45.1%	3,006,379	1,044,657	34.7% 2
Support Srvs Pupil	6,607,062	3,083,843	46.7%	6,754,759	3,045,359	45.1%
Support Srvs Admin.	4,625,192	2,439,252	52.7%	4,663,602	2,374,615	50.9%
	42,873,488	19,633,052	45.8%	44,681,547	19,915,236	44.6%
Other Sources/(Uses)						
Transfers fr. Other Funds	1,139,202	-	0.0%	-	-	N/A
Transfers to Other Funds	(1,000,000)	-	0.0%	-	-	N/A
	139,202	_	0.0%			
Change in Fund Balance	13,641,079	9,254,448		3,468,147	4,693,799	
Beginning Balance	52,572,102	52,572,102		66,213,181	66,213,181	
Ending Balance	66,213,181	61,826,550		69,681,328	70,906,980	

1. State aid budget was significantly reduced for fiscal 2011 due to uncertainty of the state actually paying Districts. The majority of the actual collections relates to the prior year grants and claims that the state was late in paying to Districts.

2. District started the EAC program this school year and has reduced our out placed special education students and thus lowered the amount it pays for tuition.

### **Bookstore Fund**

	Audited 2009-2010	Fiscal to Date January 31 <u>2010</u>	°/a	Original Budget 2010-2011	Fiscal to Date January 31 <u>2011</u>	<u>%</u>
Receipts						
Other Local Sources	812,920	622,495	76.6%	850,003	704,423	82.9%
	812,920	622,495	76.6%	850,003	704,423	82.9%
Expenditures						~~ ~~ ~
Support Srvs Other	809,034	695,457	86.0%	850,003	754,305	88.7%
	809,034	695,457	86.0%	850,003	754,305	88.7%
Change in Fund Balance	3,886	(72,962)		-	(49,882)	
Beginning Balance	695,840	695,840		699,726	699,726	
Ending Balance	699,726	622,878		699,726	649,844	

### Cafeteria Fund

	Audited 2009-2010	Fiscal to Date January 31 <u>2010</u>	%	Original Budget 2010-2011	Fiscal to Date January 31 <u>2011</u>	<sup>9</sup> /a
Receipts						
Other Local Sources	2,030,684	1,117,851	55.0%	2,021,070	1,080,893	53.5%
State Sources	7,644	6,649	87.0%	13,987	6,141	43.9%
Federal Sources	193,456	83,936	43.4%	185,828	98,895	53.2%
	2,231,784	1,208,436	54.1%	2,220,885	1,185,929	53.4%
Expenditures						
Support Srvs, - Admin.	2,168,698	1,003,762	46.3%	2,219,788	1,010,004	45.5%
	2,168,698	1,003,762	46.3%	2,219,788	1,010,004	45.5%
Change in Fund Balance	63,086	204,674		1,097	175,925	
Beginning Balance	274,282	274,282		337,368	337,368	
Ending Balance	337,368	478,956		338,465	513,293	

### **Operations and Maintenance Fund**

	Audited 2009-2010	Fiscal to Date January 31 <u>2010</u>	ª/0	Original Budget 2010-2011	Fiscal to Date January 31 <u>2011</u>	%
Receipts						
Property Taxes	6,136,075	2,884,314	47.0%	5,990,710	2,735,706	45.7%
Other Local Sources	2,297,496	873,242	38.0%	962,216	981,799	102.0%
	8,433,571	3,757,556	44.6%	6,952,926	3,717,505	53.5%
Expenditures						
Support Srvs Admin.	8,823,475	3,756,326	42.6%	5,368,583	2,658,748	49.5%
	8,823,475	3,756,326	42.6%	5,368,583	2,658,748	49.5%
Other Sources/(Uses)						
Transfers	1,035,354	1,000,000	96.6%	26,210	-	0.0%
Transfers		-	N/A	(1,610,000)	(1,610,000)	100.0% 1
	1,035,354	1,000,000	96.6%	(1,583,790)	(1,610,000)	101.7%
Change in Fund Balance	645,450	1,001,230		553	(551,243)	
Beginning Balance	9,044,358	9,044,358		9,689,808	9,689,808	
Ending Balance	9,689,808	10,045,588		9,690,361	9,138,565	

1. Transfer to Capital Projects fund per the budget.

### Life Safety Fund

	Audited 2009-2010	Fiscal to Date January 31 2010	<del>0</del> /a	Original Budget 2010-2011	Fiscal to Date January 31 <u>2011</u>	<sup>0</sup> /0
Receipts	2003 2010	-		w		
Property Taxes	1,883.021	632,198	33.6%	2,090,611	1,044,045	49.9% 1
Other Local Sources	3,697	513	13.9%	301,670	73	0.0%
	1,886,718	632,711	33.5%	2,392,281	1,044,118	43.6%
Expenditures						
Support Srvs Business	1,474,581	904,409	61.3%	1,701,822	1,641,980	96.5%
	1,474,581	904,409	61.3%	1,701,822	1,641,980	96.5%
Other Sources/(Uses)						
Transfers	(618,263)	(618,263)	100.0%	(614,263)	-	0.0%
	(618,263)	(618,263)		(614,263)	- -	
Change in Fund Balance	(206,126)	(889,961)		76,196	(597,862)	
Beginning Balance	355,137	355,137		149,011	149,011	
Ending Balance	149,011	(534,824)		225,207	(448,851)	

1. Life Safety levy was increased for the 2009 levy compared to the 2008 levy and therefore the increase in collections in the current year.

### **Bond and Interest Fund**

	Audited 2009-2010	Fiscal to Date January 31 <u>2010</u>	%	Original Budget 2010-2011	Fiscal to Date January 31 2011	%
Receipts						
Property Taxes	3,052,500	1,553,706	50.9%	2,868,619	1,354,486	47.2%
Other Local Sources	35,354	25,491	72.1%	26,210	3,772	14.4%
	3,087,854	1,579,197	51.1%	2,894,829	1,358,258	46.9%
Expenditures						
Debt Service	4,787,112	4,431,631	92.6%	3,500,790	3,142,653	89.8%
	4,787,112	4,431,631	92.6%	3,500,790	3,142,653	89.8%
Other Sources/(Uses)						
Principal on Bonds Sold	10,810,000	10,810,000	100.0%	-	-	N/A
Premium on Bonds Sold	801,095	801,095	100.0%	-	-	N/A
Payment to Escrow	(11, 468, 408)	(11,468,408)	100.0%	-	-	N/A
Transfers	618,263	618,263	100.0%	614,263		0.0%
Transfers	(35,354)		0.0%	(26,210)		0.0%
	725,596	760,950	104.9%	588,053		0.0%
Change in Fund Balance	(973,662)	(2,091,484)		(17,908)	(1,784,395)	
Beginning Balance	2,468,889	2,468,889		1,495,227	1,495,227	
Ending Balance	1,495,227	377,405		1,477,319	(289,168)	

### **Transportation Fund**

	Audited 2009-2010	Fiscal to Date January 31 <u>2010</u>	°/8	Original Budget 2010-2011	Fiscal to Date January 31 2011	<u>%</u>
Receipts		-				
Property Taxes	938,197	460,654	49.1%	842,131	401,334	47.7% 1
Other Local Sources	45,423	30,834	67.9%	22,016	7,505	34.1%
State Sources	850,067	592,479	69.7%	645,379	425,846	66.0%
	1,833,687	1,083,967	59.1%	1,509,526	834,685	55.3%
Expenditures						
Support Srvs Business	1,417,211	651,859	46.0%	1,375,537	645,744	46.9%
	1,417,211	651,859	46.0%	1,375,537	645,744	46.9%
Other Sources/(Uses)						
Other source	31,000	-	0.0%	-	-	N/A
Guler source	31,000			-		
Change in Fund Balance	447,476	432,108		133,989	188,941	
Beginning Balance	2,114,846	2,114,846		2,562,322	2,562,322	
Ending Balance	2,562,322	2,546,954		2,696,311	2,751,263	

1. Property tax bills were not sent out until mid November and collections are still behind last year. Several large distributions have been received in January.

### Illinois Municipal Retirement/Social Security Fund

	Andited _2009-2010	Fiscal to Date January 31 <u>2010</u>	<sup>0</sup> /0	Original Budget 2010-2011	Fiscal to Date January 31 2011	<u>%</u>
Receipts						
Property Taxes	2,362,524	1,134,543	48.0%	2,314,608	1,032,644	44.6%
Other Local Sources	106,550	18,282	17.2%	79,730	5,546	7.0%
	2,469,074	1,152,825	46.7%	2,394,338	1,038,190	43.4%
Expenditures						
General Instruction	336,245	148,174	44.1%	414,745	174,965	42.2%
Special Education	190,309	80,106	42.1%	224,800	88,376	39.3%
Vocational Programs	26,406	13,949	52.8%	25,273	9,825	38.9%
Interscholastic Programs	109,477	52,887	48.3%	121,161	56,742	46.8%
Summer School	9,143	5,364	58.7%	9,844	5,477	55.6%
Drivers Education	5,560	2,463	44.3%	9,239	2,928	31.7%
Other Instructional	1,241	536	43.2%	1,232	534	43.3%
Support Srvs Pupil	336,130	151,871	45.2%	393,359	169,269	43.0%
Support Srvs Admin.	824,926	447,194	54.2%	898,271	460,943	51.3%
	1,839,437	902,544	49.1%	2,097,924	969,059	46.2%
Change in Fund Balance	629,637	250,281		296,414	69,131	
Beginning Balance	1,323,641	1,323,641		1,953,278	1,953,278	
Ending Balance	1,953,278	1,573,922		2,249,692	2,022,409	

### **Capital Projects Fund**

	Audited 2009-2010	Fiscal to Date January 31 <u>2010</u>	°⁄a	Original Budget 2010-2011	Fiscal to Date January 31 2011	°/0
Receipts				Martin -		
Other Local Sources	<del>_</del>		N/A	300,000	527	0.2%
		-	N/A	300,000	527	0.2%
Expenditures						
Support Srvs Business	-	-	N/A	1,910,000	1,430,248	74.9%
			N/A	1,910,000	1,430,248	74.9%
Other Sources/(Uses) Transfers	<u></u>	<u> </u>	N/A	1,610,000	1,610,000	100.0% 1
		u		1,610,000	1,610,000	
Change in Fund Balance	-	-		-	180,279	
Beginning Balance	<u> </u>					
Ending Balance				-	180,279	

1. Transfer from O&M fund to establish the Capital Projects fund per the budget.

### Working Cash Fund

	Audited 2009-2010	Fiscal to Date January 31 <u>2010</u>	<sup>0</sup> /0	Original Budget 2010-2011	Fiscal to Date January 31 2011	
Receipts						
Property Taxes	1,091,083	535,987	49.1%	1,045,305	527,572	50.5%
Other Local Sources	129,470	88,392	68.3%	69,485	22,132	31.9%
	1,220,553	624,379	51.2%	1,114,790	549,704	49.3%
Expenditures						
Transfers			N/A	<u> </u>	<u> </u>	N/A
			N/A			N/A.
Other Sources/(Uses)						
Principal on Bonds Sold	1,000,000	1,000,000	100.0%	-	-	N/A
Transfers	(1,000,000)	(1,000,000)	100.0%		-	N/A
	<u> </u>					
Change in Rund Polonee	1,220,553	624.379		1,114,790	549,704	
Change in Fund Balance	1,220,305	024,577		1,117,770	5 (5,10)	
Beginning Balance	5,300,950	5,300,950		6,521,503	6,521,503	
Ending Balance	6,521,503	5,925,329		7,636,293	7,071,207	

### Tort Immunity Fund

	Audited 2009-2010	Fiscal to Date January 31 <u>2010</u>	<u> </u>	Original Budget 2010-2011	Fiscal to Date January 31 <u>2011</u>	º⁄o
Receipts						
Property Taxes	1,263,583	632,689	50.1%	1,104,552	527,035	47.7%
Other Local Sources	38,162	23,653	62.0%	20,076	5,966	29.7%
	1,301,745	656,342	50.4%	1,124,628	533,001	47.4%
Expenditures						
Support Srvs Admin.	869,427	706,790	81.3%	1,121,112	714,235	63.7%
	869,427	706,790	81.3%	1,121,112	714,235	63.7%
Change in Fund Balance	432,318	(50,448)		3,516	(181,234)	
Beginning Balance Ending Balance	1,917,776 2,350,094	1,917,776 1,867,328		<u>2,350,094</u> 2,353,610	2,350,094 2,168,860	

### Dental Self Insurance Fund

	Audited 2009-2010	Fiscal to Date January 31 <u>2010</u>	%	Original Budget 2010-2011	Fiscal to Date January 31 2011	<u>%</u>
Receipts						
Insurance Premiums	409,548	197,102	48.1%	452,853	195,014	43.1%
Other Local Sources	3,680	2,509	68.2%	2,000	511	25.6%
	413,228	199,611	48.3%	454,853	195,525	43.0%
Expenditures						
Staff Services	407,364	186,947	45.9%	452,853	204,409	45.1%
Change in Fund Balance	5,864	12,664		2,000	(8,884)	
Beginning Balance	143,399	143,399		149,263	149,263	
Ending Balance	149,263	156,063		151,263	140,379	

### Medical Self Insurance Fund

	Audited 2009-2010	Fiscal to Date January 31 <u>2010</u>	<sup>0</sup> /0	Original Budget 2010-2011	Fiscal to Date January 31 2011	<u>%</u>
Receipts						
Insurance Premiums	4,732,751	2,567,088	54.2%	5,577,698	3,311,567	59.4%
Other Local Sources	52,274	35,277	67.5%	20,000	9,501	47.5%
	4,785,025	2,602,365	54.4%	5,597,698	3,321,068	59.3%
Expenditures Staff Services	4,541,907	1,863,575	41.0%	5,577,698	2,393,629	42.9%
Change in Fund Balance	243,118	738,790		20,000	927,439	
Beginning Balance Ending Balance	1,594,968 1,838,086	1,594,968		1,838,086	1,838,086 2,765,525	

### Self-Insurance Workers' Comp Fund

	Audited 2009-2010	Fiscal to Date January 31 <u>2010</u>	<sup>5</sup> ⁄0	Original Budget 2010-2011	Fiscal to Date January 31 2011	<u> </u>
Receipts						
Insurance Premiums	-	-	N/A	-	-	N/A
Other Local Sources	-	**	N/A	-		N/A
Transfers	-	-	N/A	-	-	N/A
		**		-	-	
<i>Expenditures</i> Staff Services			N/A	*	~	N/A
Change in Fund Balance		-		-	-	
Beginning Balance	15,857	15,857		15,857	15,857	
Ending Balance	15,857	15,857		15,857	15,857	

201 North Scoville Avenue • Oak Park, IL 60302-2296

TO:	Board of Education
FROM:	Cheryl Witham
DATE:	February 24, 2011
RE:	Treasurer's Reports

### BACKGROUND

It is a requirement that the Board of Education accepts and approves the monthly Treasurer's Reports.

### SUMMARY OF FINDINGS

Attached is the Treasurer Report for January, 2011.

### **RECOMMENDATIONS (OR FUTURE DIRECTIONS)**

MOTION: To accept the January 2011, Treasurer Report as presented.

### ROLL CALL VOTE

AGENDA ITEM VIII. C.

	Funds	Opening Cash Balance <u>01/01/11</u>	Cash <u>Receipts</u>	Cash <u>Disbursements</u>	Adjustments to Cash (JE's)	Ending Cash Balance <u>01/31/11</u>	% of Total
15 15 15	Education Food Service Book Store	67,267,351.14 350,586.04 283,122,38	5,449,390.87 268,092.87 60,117.92	(3,036,081.87) (126,886.67) (48.796.62)	(441,730.13) (8,870.49) (1,580.03)	69,238,930.01 482,921.75 292.863.65	72.29% 0.50% 0.31%
		67,901,059.56	5,777,601.66	(3,211,765.16)	(452,180.65)	70,014,715.41	73.10%
20	Operations, Building & Maintenance	8,602,902.81	760,661.45	(402,694.08)	(50,298.72)	8,910,571.46	9.30%
30	Bond & Interest Fund	(605,192.05)	300,095.93	(1,335.00)	3	(306,431.12)	-0.32%
40	Transportation Fund	2,348,483.16	89,118.68	(114,991.76)	(1,740.00)	2,320,870.08	2.42%
50	IMRF & SS Fund	1,934,960.39	229,064.25	(153,435.91)	30.54	2,010,619.27	2.10%
60	Site & Construction	317,731.66	29.52	(137,482.60)	1	180,278.58	0.19%
70	Working Cash	6,946,079.36	117,987.14	r	ı	7,064,066.50	7.38%
80	Tort Immunity	2,096,409.93	116,678.23	(45,531.97)	(5,725.45)	2,161,830.74	2.26%
81	Dental Self Insurance	164,768.04	27.10	(37,251.12)	37,966.76	165,510.78	0.17%
82	Medical Self Insurance	3,059,223.69	507.42	(437,206.79)	476,292.94	3,098,817.26	3.24%
83	Workers' Comp Self Insurance	15,857.04	•	·	ı	15,857.04	0.02%
84 84	Harris - PMA Community Bank	331,908.67 243.714.28	52,332.10 51.165.10	(5,930.45) (61.289.71)	(4,345.42)	373,964,90 233 589 67	0.39%
	Total - Activity Funds	575,622.95	103,497.20	(67,220.16)	(4,345.42)	607,554.57	0.63%
06	Fire Prevention & Safety	(697,784.63)	234,653.12			(463,131.51)	-0,48%
	Total - All Funds	\$ 92,660,121.91 \$	\$ 7,729,921.70 \$	(4,608,914.55) \$	\$	95,781,129.06	100.00%

Summary of adjustments to cash: Reclassification of food service chargebacks. Reclassification of bookstore chargebacks.

Reclassification of expenditures PPO/Pharmacy reclassification.

# Oak Park & River Forest High School District 200 Treasurers Report January 31, 2011

	Account <u>Balance</u>	Treasurer's <u>Control</u>	% of <u>Total</u>
Harris Bank Comingled Account (treas ofc.) Statement CTTO Less: Outstanding Checks Plus: Deposits in Transit	126,015.94	126,015.94	0.13%
Adjusted Harris ISDLAF Account (Liquid & Max) Statement Balance Less: Outstanding Checks	<u>126,015.94</u> 95,815,846.98 (568,158.64)	120,013.94	0.1378
Plus: Deposits in Transit Adjusted	109,537.37 95,357,225.71		99.56%
Community Bank Student Activity Account Statement Balance Less: Outstanding Checks Plus: Deposits in Transit	265,929.74 (32,387.07) 47.00		0.24%
Adjusted           Community Bank Imprest Account           Statement Balance	<u>233,589.67</u> 18,538.17		0.24/0
Less: Outstanding Checks Plus: Deposits in Transit Adjusted	(1,640.43) 		0.02%
Petty Cash	7,400.00		0.01%
Workers Compensation Escrow	40,000.00		0.04%
Total Cash and Investments	\$ 95,781,129.06 \$	126,015.94	100.00%

Note: Petty cash number includes \$2,000 that is in the Athletic Imprest account maintained by the Athletic Department.

			Oak Park & Riv Sci	Oak Park & River Forest High School District 200 Schedule of Investments January 31, 2011
	Average	Investment		Prior Month
	Interest	Value	% of	% of
Bv Financial Institution	Rate *	01/31/11	Total	Total
Harris ISDI AF - Liquid MM	0.02%	1,267,589.77	1.32%	1.25%
Harris ISDLAF - Max MM	0.06%	969,512.92	1.01%	16.56%
Harris ISDLAF - SDA	0.15%	2,705,986.78	2.82%	2.92%
Harris ISDI AF - CD's	0.44%	65,042,869.47	67.79%	73.46%
Harris ISDLAF - Gov't Securities	0.43%	3,929,888.04	4.10%	5.67%
Harris ISDLAF - Term Series	0.27%	21,900,000.00	22.83%	0.00%
Harris - CTTO MM	*	126,015.94	0.13%	0.14%
Total All Investments by Institution		95,941,862.92	100.00%	100.00%
	Average Interest	Investment Value	% of	Prior Month % of

Harris - CTTO MM	*	126,015.94	0.13%	0.14%
Total All Investments by Institution	10	95,941,862.92	100.00%	100.00%
	Average	Investment		Prior Month
	Interest	Value	% of	% of
Bv investment Tvpe	Rate *	01/31/11	Total	Total
CD's	0.44%	65,042,869.47	67.79%	73.46%
Government Securities	0.43%	3,929,888.04	4.10%	5.67%
Term Series	0.27%	21,900,000.00	22.83%	0.00%
Money Market	0.08%	5,069,105.41	5.28%	20.86%
Total All Investments by Type		95,941,862.92	100.00%	100.00%
	Average	Investment		Prior Month
	Interest	Value	% of	% of
	Rate *	01/31/11	Total	Total
By Maturity Age				
1 month	0.24%	4,799,285.61	5.00%	6.04%
2 months	0.78%	4,400,000.00	4.59%	5.18%
3 months	0.56%	3,700,000.00	3.86%	4.75%
4-6 months	0.51%	17,593,500.00	18.34%	17.36%
7-9 months	0.35%	42,531,045.00	44.33%	25.18%
10-12 months	0.25%	4,330,369.47	4.51%	6.06%
1 year +	0.49%	13,518,557.43	14.09%	14.58%
2 vears +	0.00%	ł	0.00%	0.00%
Mature on demand	0.08%	5,069,105.41	5.28%	20.86%
Total Investments		95,941,862.92	100.00%	100.00%

\* The rate of interest is not known for funds invested with the Trustee of the former CTTO.

Comparative Interest Rate Information (as of February 7)

Fixed Income Type of Security	60	06	120	180	270	<del>~~</del>	18	2
	Days	Days	Days	Days	Days	Year	Months	Year
CD's	0.00% - 0.10%	0.00% - 0.15%	0.00% - 0.15% 0.	0.09% - 0.25% N/A	0.15% - 0.50% 0.03%	0.2(	0.25% - 0.60% 0.48%	0.25% - 0.60% 0.40% - 0.75% 0.48% 0.68%
Government Agency Government Treasury	N/A	A/N	A/N	0.04%	0.109	0.20%	0.51%	0.57%
<mark>Liquid Asset Funds</mark> Liquid Class Max Class	7 Day Effective 0.02% 0.09%							

201 North Scoville Avenue • Oak Park, IL 60302-2296

RE:	Executive Summary for Special Education Transportation RFP 2011-2014
DATE:	February 24, 2011
FROM:	Cheryl L. Witham
TO:	Board of Education

### BACKGROUND

On February 1, 2011, a Request for Proposals was published for Special Education Transportation. The current contract with Grand Prairie Transit expires at the conclusion of this school year and all contract extensions have been exhausted.

### SUMMARY OF FINDINGS

Proposals were received from five firms, All-Ways, First Student, Grand Prairie, Illinois Central and Lakeview Bus Lines. A summary of the costing results can be found below.

	2011-2012	2012-2013	2013-2014	3-Year
Vendor	Total Cost	Total Cost	Total Cost	Cost
Illinois Central	\$ 707,653.54	\$ 728,894.38	\$ 750,745.54	\$ 2,187,293,46
Lakeview	\$ 836,438.00	\$ 836,438.00	\$ 857,444.00	\$ 2,530,320.00
Grand Prairie	\$ 847,412.56	\$ 868,597.59	\$ 894,657.95	\$ 2,610,668.10
First Student	\$ 896,833.82	\$ 919,257.63	\$ 942,234.06	\$ 2,758,325.51
All-Town	\$ 903,811.41	\$ 928,651.52	\$ 954,191.44	\$ 2,786,654.37

A check of references (Chicago Public Schools, Oswego District 308 and Forest Ridge District 142) netted positive feedback. The referenced school districts indicated that Illinois Central was a reliable provider, offering excellent safety, on-time service and timely accommodation of any scheduling deviations.

### **RECOMMENDATIONS (OR FUTURE DIRECTIONS)**

MOTION: To accept the RFP from Illinois Central for the District's Special Education transportation services.

ROLL CALL VOTE

AGENDA ITEM VIII.D.

201 North Scoville Avenue • Oak Park, IL 60302-2296

TO: Board of Education

FROM: Cheryl L. Witham

DATE: February 24, 2011

### RE: Executive Summary for Field Trip/Extracurricular Transportation Services Bid BACKGROUND

On February 1, 2011, bids were solicited for Field Trip/Extracurricular Transportation Services. The current contract with R&D Bus Company is set to expire at the conclusion of this school year.

### SUMMARY OF FINDINGS

Proposals were received from six firms, All-Ways, First Student, Grand Prairie, Illinois Central and Lakeview Bus Lines. Vendors were asked to provide a full tariff schedule as well as pricing for six commonly-travelled OPRFHS trips as determined by their proposed tariff schedule. Below is a summary of the costing results.

	<u>R&amp;D</u>	All-Ways	Grand Prairie	Illinois Central	First Student	<b>Lakeview</b>
Lindberg Park	135.00	139.00	309.00	144.00	275.63	175.00
Downers North	320.00	225.00	363.00	180.00	287.81	185.00
Volo Bog	650.00	435.00	531.00	252.00	447.86	265.00
Lyons Twp.	215.00	225.00	363.00	180.00	287.81	185.00
Museum of S&I	265.00	315.00	531.00	180.00	447.86	265.00
Starved Rock	750.00	685.00	411.00	252.00	412.86	265.00
Total	2,335.00	2,024.00	2,508.00	1,188.00	2,159.83	1,340.00

A check of references netted positive feedback. The reference school districts (Chicago Public Schools, Oswego District 308 and Forest Ridge District 142) indicated that Illinois Central was a reliable provider, offering excellent safety, on-time service and timely accommodation of any scheduling deviations.

### **RECOMMENDATIONS (OR FUTURE DIRECTIONS)**

MOTION: To accept the bid with Illinois Central for the District's field trip/extracurricular transportation Services during SY 2011 – 2012.

ROLL CALL VOTE

AGENDA ITEM VIII. E.

WEB: www.oprfhs.org

TTY/TDD: (708) 524-5500

201 North Scoville Avenue • Oak Park, IL 60302-2296

RE:	Approval of PTAB Resolution
DATE:	February 24, 2011
FROM:	Cheryl L. Witham
TO:	Board of Education

### BACKGROUND

The administration is asking the Board of Education to consider this resolution giving the administration, in conjunction with our law firm Franczek, Radelet P.C. authority to file a Request to Intervene in Appeal Proceedings for the re-assessment of property within our District.

### SUMMARY OF FINDINGS

These PTAB requests have a negative effect on our property tax collection in subsequent years and the revenue cannot be recaptured. This request is necessary because the District must intervene within 30 days of notice. This is often too short a time period to bring the issue before the Board of Education.

This resolution must be approved by the Board of Education on an annual basis.

### **RECOMMENDATIONS (OR FUTURE DIRECTIONS)**

MOTION: To approve the PTAB resolution as presented.

ROLL CALL VOTE

AGENDA ITEM VIII. F.

### Resolution Authorizing Intervention in Proceedings before the State of Illinois Property Tax Appeal Board

Whereas, an owner or manager of a parcel or parcels of real property located within the boundaries of the State of Illinois has the right to file an appeal challenging the assessed value of the parcel or parcels of real property with the State of Illinois Property Tax Appeal Board ("PTAB"); and

Whereas, a reduction in the assessed value of a parcel or parcels granted by the PTAB will lead to the issuance of a real estate tax refund from the Board of Education of Oak Park & River Forest High School District No. 200 ("Board"); and

Whereas, a taxing district has the right to intervene in proceedings before the PTAB in order to protect the taxing district's revenue interest in the assessed value of a parcel or parcels; and

Whereas, the time period during which a taxing district may intervene is 60 days after the taxing district's receipt, from the Cook County Board of Review, of notice of the filing of an appeal by an owner or manager of a parcel or parcels of real property; and

Whereas, the Board has determined that it is necessary, desirable, advantageous, and in the public interest to defend the Board's real property tax base by intervening in PTAB appeals filed on parcels within the boundaries of the Board.

**NOW THEREFORE, BE IT HEREBY RESOLVED,** by the Board of Education of Oak Park & River Forest High School District No. 200, Cook County, Illinois, as follows:

- 1. The Board finds that all of the recitals contained above are true and correct, and that the same are hereby incorporated herein by reference.
- 2. The Board hereby authorizes Franczek Radelet P.C., as its legal representative, to: a) file a Request to Intervene in Appeal Proceedings in all 2011 assessment-year appeals filed at the PTAB for which the Board receives notice from the Cook County Board of Review, and b) represent the Board's interests in those proceedings.
- 3. All motions and resolutions or parts thereof in conflict with the provisions of this Resolution are, to the extent of such conflict, hereby repealed.
- 4. If any section, paragraph, clause, or provision of this Resolution shall be held invalid, the invalidity of such section, paragraph, clause, or provision shall not affect any of the other sections, paragraphs, clauses, or provisions of this Resolution.
- 5. This Resolution shall be in force and effect upon its adoption.

After a full and complete discussion thereof, Member	moved that the
foregoing Resolution be adopted and Member	seconded the motion. The
President directed the Secretary to call the roll for a vote upon	the motion to adopt this
Resolution. Upon a roll call vote taken, the Board of Education voted	l as follows:

YES:	
AYS:	
bstaining:	
RESENT:	
bsent from Meeting:	

The President declared the motion carried and the Resolution duly adopted.

......

Dated:\_\_\_\_\_

By:\_\_\_\_\_ President, Board of Education

ATTEST

Secretary, Board of Education

#### Oak Park and River Forest High School District 200

201 North Scoville Avenue • Oak Park, IL 60302-2296

TO:	Board of Education
FROM:	Cheryl L. Witham
DATE:	February 24, 2011
RE:	Office Supplies Contract Renewal

#### BACKGROUND

On July 1, 2008, the District signed a one-year contract with Quill Corporation, the lowest responsible bidder from the 2008 office supplies bid. The bid contract allows for up to three one-year extensions.

#### SUMMARY OF FINDINGS

Quill Corporation has agreed to extend the district's 2008 - 2009 contract pricing and to continue to offer free shipping on all orders. A competitive bid would most likely yield a price increase. More importantly, Quill is the only firm in the past two bid processes that offers free next-day shipping with no minimum order amount, allowing for a more responsible "just-in-time" supply inventory.

#### **RECOMMENDATIONS (OR FUTURE DIRECTIONS)**

**MOTION:** To extend the contract with Quill for the allowable one-year term, July 1, 2011 through June 30, 2012.

ROLL CALL VOTE

AGENDA ITEM VIII. H.

QUILL

100 Schelter Road Lincolnshire, IL 60069 Phone: 1-800-634-4809 Fax; 1-800-789-2016 www.Quill.com



Best experience in office products

January 26, 2011

Tim Keeley Oak Park River Forest School District 200 Purchasing Department 201 N. Scoville Avenue Oak Park, IL 60302

Fax No.: 708-434-3910 Page 1 of 3

Quill Bid Number: 10-9469 Customer Bid Info Office Supply Bid Renewal Account Number: 2527265

Dear Tim Keeley,

Thank you for your recent bid request renewal. We appreciate your interest in Quill and are pleased to extend special pricing on the enclosed bid. Please note that the <u>bid prices</u> and <u>free shipping</u> are good through <u>February 28, 2012.</u>

In addition, for items not on your enclosed list, we're pleased to offer your organization an additional <u>10%</u> column discount\* to our already discounted prices in the catalog which are up to 80% off list. This extra savings is good through <u>February 28, 2012</u>.

Below are a few guidelines that we ask for you to follow to ensure the proper pricing and handling of your purchase orders:

- Please use the Quill bid number above, Quill item numbers and prices that are referenced on your bid when placing purchase orders for easy and accurate processing. Orders will ship and invoice the same day, if orders must invoice and/or ship on or after a certain date this must be indicated clearly on your purchase order
- D Please fax or email your tax exempt form to <u>tax.exempt@quill.com</u> or fax # 800-499-8805
- □ Quill offers easy online ordering with visibility to your special item pricing as well as your extra discount. <u>www.Quill.com</u>

Thank you once again for choosing Quill. We look forward to working with you in the future.

#### Sincerely, Quill's Bid Department

- The 10% discount does not apply to office machines (printers, copiers, fax machines, vending machines, etc), technology products, computer hardware and software, or any item in Quill's Special Order Catalog. Purniture is also excluded from this offer, but we will, however, pay all of the shipping charges related to standard tailgate delivery when you purchase furniture from our full-line catalog. Installation or set up not included.
- Certain coupons, sales or promotions may not be used with this offer.

# Quill Bid Response Form

10-9469	Custamer Bld#: OFFICE SUPPLIES	Organization Name: OAK PARK RIVER FORFET S 0 200	10725672	TIM KEELEY	7084343266	Both
Quill Bld#:	Gustamer Bld#:	Organization Name:	Enterprise#: 10725672	Contact Name: TIM KEELEY	Contact Phone#: 7084343266	Bid Type: Both

Bid Open Date: 1/22/2010 Bld Open Time: 12:00 AM

Item Expiration Date: 3/1/2012 Catalog Expiration Date: 3/1/2012 Total Items: 40 ( 40 Won, 0 Lost ) Total: \$ 115.80 ( 115.80 Won, 0.00 Lost )

31     EA       33     EA       56     EA       55     DZ       11     DZ       15     18X=1 DZ       16     DZ       17     DZ       18     EA=1 ST       19     DZ       11     DZ       12     EA=1 ST       13     18X=1 DZ       14     EA=1 ST       15     18X=1 DZ       16     DZ       17     DZ       18     18X=1 DZ       10     DZ       18X     12 EA=1 DZ       10     DZ       11     DZ       12     EA=1 DZ       13     BX       14     12 EA=1 DZ       15     12 EA=1 DZ       16     12 EA=1 DZ       17     12 EA=1 BX       18     12 EA=1 BX       17     12 EA=1 BX	Line#	Line# Item#	Description	Color Make	Make	Otv Price		A MAIL		CaseExt	Ext		
2010.2011 WALL CALENDAR         MEAU         ME	-	E1700	#17 CALENDAR BASE			?					Total		Page
OB CARTEMP FULL STRIP         DACK         1         3.37         EA         48         3.27         Disconstructure         58         Disconstructure <thdisconstructure< th=""> <thdisconstruc< td=""><td>5</td><td>SK1616</td><td>2010-2011 WALL CALENDAR</td><td>-</td><td>MEAU</td><td></td><td>- 1</td><td>ΕA</td><td></td><td>Т</td><td>R 34</td><td></td><td>,</td></thdisconstruc<></thdisconstructure<>	5	SK1616	2010-2011 WALL CALENDAR	-	MEAU		- 1	ΕA		Т	R 34		,
FERAMOLE         BLACK         1         3.66         E.A         3.66         Own Brand           FERAMOLE         BOOK 80/SH3         WHITE         7         94         7         2         94           BOOK 80/SH3         BOOK 80/SH3         MHITE         FANOLA         1         55         36         0.01         800           FERAMINE         MHITE         CRAYOLA         1         44         82         72         94           CRAYOLAMIL DUSTENER, MAGIC         ONLL RENDATIONAL         1         38         EA         12         96           CRAYOLAMIL WITE         INTERNATIONAL         1         38         EA         14         37           CRAYOLAMITE         INTERNATIONAL         1         38         EA         14         33           CRAYOLAMITE         INTERNATIONAL         1         38         EA         14         33           MOSTENER BOTTLE, WEDGE         ONLL RETONATIONAL         1         38         EA         77         Oull Bend           OULL MEDUM PONT PEN         BLACK         1         102         18         14         33           OULL SECONOMY PENCILS, RE         BUUE         SAMFORD         1         102	3	796040	OB CONTEMP FULL STRIP		AI-A-ULANCE		4	EA		48	102	Dice-Art THOAD	
BOCUCUCED STEND         WHTE         1         EB         FA         72         56           FERMMERT GLUE STC         WHTE         TAPE VUIDE         WHTE         F         A         BX         72         56           FERMMERT GLUE STC         WHTE         CRAYOLA MATI-DUST CHAIX         WHTE         CRAYOLA MATI-DIST         CRAYOLA MATI-DIST         CRAYOLA MATI-DIST         CRAYOLA         CRAYOLA MATI-DIST         CRAY			STAPLER	BLACK		<del></del>		EA		96	00 6	0107711720.000	Ŕ
PERAVMENT GLUE STIC         WHITE         CRAYOLA         1         2         5         5         59           TARE VANDENTIGUE STIC         MHITE         CRAYOLA         1         77         6         72         59           TARE VANDESPERSER, MAGIC         DEMORATE         DEMORATE         1         77         6         72         59           TARE VANDESPERSER, MAGIC         DEMORATE         DEMORATE         1         77         1         72         59           TARE VANDESPERSER, MAGIC         DEMORATE         1         77         12         144         73         55         20         71         20         25         20         71         20         25         20         71         20         12         25         21         20         25         20         71         20         12         25         21         20         25         20         71         20         12         25         21         20         25         20         71         20         12         25         21         20         25         20         71         20         25         21         20         25         21         20         25         21	*	25774	BOOK, BO/SHTS	WHITE		+	1			3	00'0	UUIII Brand	ő
CGAYCIA ANTI-BUIST CHAIK         WHITE         CRAYOLA         1         94         EA         72         94         52         94         52         94         53         54         77         53         54         77         53         54         77         55         94         55         55         94         55         94         57         54         77         54         77         55         94         57         56         77         94         55         73         94         57         56         71         01011         86         77         70         72         55         94         77         94         77         70         73         73         73         71         94         77         70         71         94         77         72         94         77         71         94         77         72         130         71         94         77         94         77         94         77         94         77         94         77         94         77         94         77         94         77         94         77         94         77         94         77         94         77         94	0	00196	PERMANENT GLUE STIC	WHITE			ŧ		i	22	69		
Type wonspensesType wonspenseType w	6	501402	CRAYOLA ANTI-DRIST CHAIN			+		۲		72	6		
MOISTEMER BOTILE. WEDGE         OFFICEMATE         1         77         EA         144         77         5         7         10         2         18         5         5         7         10         2         18         5         5         7         10         2         18         2         10         2         11         2         18         2         13         2         10         2         10         2         10         2         10         2         10         2         10         2         10         2 <th2< th="">         2         <th2< th=""></th2<></th2<>	2	3M105	TAPE WHOISPENSER MAGIC		CHAYOLA	*		ЗХ		77			
TYPE         UTTLENATIONAL         1         35         EA         144         33           QUILL SECONOMY PENCILS #2         INTERNATIONAL         1	æ	97800	MOISTENER BOTTLE. WFINGF				1	A		142	77		ž
Outline         Construct Section         Res         1         55         DZ         120         1300         130		17140	ТҮРЕ		ULTICEMALE INTERNATIONAL	<del>.</del>		5		144			
QUIL MEDIUM POINT PEN         BLUE         1         71         0.2         1 BX = 1.02         .55         Quill Brand           QUIL MEDUM POINT PEN         RED         1         71         0.2         1 BX = 1.02         35         71         Quill Brand           QUIL MEDUM POINT PEN         RED         1         71         0.2         1 BX = 1.02         36         71         Quill Brand           WET EASE MARKERS, FINE         SAMFORD         1         1.9         0.2         1 BX = 1.02         36         71         Quill Brand           WET EASE MARKERS, FINE         SAMFORD         1         1.9         0.0         1.8         4.6         71         Quill Brand           VIS A-VIS WET EASE FINE BPK ASTD SAMFORD         1         1.9         0.0         1.8         4.6         71         Quill Brand           VIS A-VIS WET EASE FINE BPK ASTD SAMFORD         1         1.90         DZ         4.6 = 1.5T         72         1.90           VIS A-VIS WET EASE FINE BPK ASTO SAMFORD         1         1.00         DZ         1.8 = 1.0Z         72         1.00           MEDUUM         REDU         1         1.00         DZ         1.8 = 1.0Z         72         1.00           MEDUUM	0	7961485	UNIT'S ECONOMY PENCILS, #2					5		ł	55-		520
QUILL MEDIUM POINT PEN         RED         1         71         DZ         1 BX = 1 DZ         36         71 Quil Brand           WITE-OUT, OUTX DRY         BLACK         1         71         DZ         1 BX = 1 DZ         36         71 Quil Brand           WITE-OUT, OUTX DRY         WHTE FINE         SANFORD         1         98         EA         144         .50           WITE-OUT, OUTX DRY         WHTE FINE         SANFORD         1         98         51         4 EA = 1 ST         72         1 39           VIS-AVIS WET-ERASE FINE SPK ASSTD SANFORD         1         98         51         4 EA = 1 ST         72         1 39           VIS-AVIS WET-ERASE FINE SPK ASSTD SANFORD         1         9.80         17         2 8 EA = 1 ST         72         1 39           VIS-AVIS WET-ERASE FINE SPK ASSTD SANFORD         1         1.00         DZ         1 BX = 1 DZ         72         1 39           AUBEDIUM         PAPERMATE OFFICE PEN         BLUE         BLACK         NREDUM         1 1.00         72         1 30           PAPERMATE OFFICE PEN         BLACK         SANFORD         1         1 40         DZ         1 22         1 30           PAPERMATE OFFICE PEN         BLUE         BLACK         S	1	790480	UNIT MEDIUM POINT PEN	BI.UE		-		22		120	-55	Quill Brand	205
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	12	70/10/	UUILL MEDIUM POINT PEN	RED		. 4		-	ZO L = Y9	36	11.	Quill Brand	265
WITE-OUT_OUCK DRYWATEBIC1 $102$ $1 \text{ BX} = 102$ $36$ $71$ Quill BrandWET ERASE MARKERS, FINEMET ERASE MARKERS, FINESAMFORD1 $199$ $57$ $4 \text{ EA} = 151$ $72$ $199$ VIS-AVIS WET-ERASE FINE-BPK ASSTDSAMFORD1 $4 \text{ B9}$ $51$ $8 \text{ EA} = 151$ $72$ $4 \text{ B9}$ SUFE STARPIE MARKERBLACKSAMFORD1 $4 \text{ B9}$ $51$ $8 \text{ EA} = 151$ $72$ $4 \text{ B9}$ SUFERMATE OFFICE PENBLUENLENL $100$ DZ $18 \times 102$ $72$ $100$ PAPERMATE OFFICE PENBLACKSAMFORD1 $100$ DZ $18 \times 102$ $72$ $100$ PAPERMATE OFFICE PENBLACKBLACK $1100$ DZ $72$ $100$ $72$ $100$ MEDIUMEXPO FRASEBLACKSAMFORD1 $110$ DZ $72$ $100$ $74$ MEDIUMEXPO FRASEBLACKSAMFORD1 $110$ DZ $72$ $100$ MEDIUMEXPO FRASEBLACKSAMFORD1 $110$ $27$ $12$ $74$ MARKER BKGUUIL STANDARD STAPLESBLACKSAMFORD1 $74$ $70$ MARKER BKGUUIL STANDARD STAPLESBLACKSAMFORD1 $74$ $70$ MARKER BKGUUIL STANDARD STAPLESBLACKSAMFORD1 $74$ $74$ MARKER BKGUUIL STANDARD STAPLESBLACKSAMFORD1 $74$ $74$ MARKER BK			QUILL MEDIUM POINT PEN	BLACK		- +	E		<u>1 BX = 1 DZ</u>	36	.7.	Quill Brand	756
WET ERASE MARKERS, FINE         SAMFORD         1		1 ror uurz	WITE-OUT, QUICK DRY	WHITE	AIC.			_	•~ i	36	11.	Quil Brand	246
WIS-AVIS WET-ERASE FINE-BPK ASSTD       VIS-AVIS WET-ERASE FINE-BPK ASSTD       1.159       ST       4.6.4       ST       7.2       1.9.9         SUPER SHAPTIE MARKER       BLACK       BLACK       A.NORD       1       4.8.9       ST       8.6.4.50       2.8.0         NEDRUM       BLUE       BLUE       1.0       DZ       1.8.1.0       1.2       9.8.0         PAPERMATE OFFICE PEN.       BLUE       1.00       DZ       1.8.1.0       7.2       1.00         PAPERMATE OFFICE PEN.       BLACK       AND       1       1.00       DZ       1.8.1.0.2       7.2       1.00         MEDIUM       MEDIUM       RED       1       1.00       DZ       1.8.1.0.2       7.2       1.00         MEDIUM       REDIUM       1       1.00       DZ       1.8.1.0.2       7.2       1.00         MEDIUM       REDIUM       1       1.00       DZ       1.8.1.0.2       7.2       1.00         MEDIUM       REDIUM       RED       1       1.00       DZ       7.2       1.00         MEDIUM       REDIUM       1       1.19       EA       7.2       1.00       7.4         MEDIUM       EXPOLERER       SANFORD       1	- 4	100/4	WET ERASE MARKERS, FINE		SANEODO	-	1	_		144	80		
SUPER SHARPLE MARKER       BLACK       SANFORD       1       4.89       ST       8 EA = 1 ST       72       4.89         PAPERMATE OFFICE PEN.       BLUE       1       9.80       DZ       12       9.80         PAPERMATE OFFICE PEN.       BLUE       1       1.00       DZ       1 BX = 1 DZ       72       9.80         PAPERMATE OFFICE PEN.       BLACK       SANFORD       1       1.00       DZ       72       1.00         PAPERMATE OFFICE PEN.       BLACK       SANFORD       1       1.00       DZ       72       1.00         MEDIUM       EXPO ERASER       BLACK       SANFORD       1       1.10       DZ       72       1.00         MEDIUM       BLACK       SANFORD       1       1.10       DZ       72       1.00         MEDIUM       EXPO ERASER       BLACK       SANFORD       1       1.19       Z2       1.00         MEDIUM       EXPO ERASER       BLACK       SANFORD       1       1.100       Z2       1.100         MEDIUM       EXPO ERASER       BLACK       SANFORD       1       7.47       Z4       Z4         MEDIUM       EXPO ERASE       BLACK       SANFORD       1	0 4	16078	VIS-A-VIS WET-ERASE FINE-8PK	ASSTD	SANFODD	-	1		IEA=1ST	72	1.99		
PAPERMATE OFFICE PEN.         BLUE $$ $$ $$ $$ $$ $$ $$ $$ $$ $$ $$ $$ $$ $$ $  $		10055	SUPER SHARPIE MARKER	RI ACK	ANE ODD			_	EA = 1	2	4.80		
WEEDUM         BLUE         1         1.00         DZ         1 BX = 1 DZ         72         1.00           PAPERMATE OFFICE PEN.         RED         1         1.00         DZ         1 BX = 1 DZ         72         1.00           PAPERMATE OFFICE PEN.         BLACK         RED         1         1.00         DZ         72         1.00           PAPERMATE OFFICE FEN.         BLACK         SANFORD         1         1.19         EA         72         1.00           MEDIUM         EXPO ERASER         BLACK         SANFORD         1         1.19         EA         72         1.00           MEDIUM         EXPO ERASER         BLACK         SANFORD         1         1.19         EA         72         1.19           MARKER-BK         BLACK         SANFORD         1         7.47         DZ         12         7.47           MARKER-BK         BLACK         SANFORD         1         7.47         DZ         12         7.47           GUILL STANDARD STAPLES         ELW         1         7.47         DZ         12         7.47           GUILL STANDARD STAPLES         ELW         SANFORD         1         7.47         DZ         1.44         7.0	1	331	PAPERMATE OFFICE PEN.		UNU MAR	•	1	27		12	9.80		102
PAPERMATE OFFICE PEN.REDT11.00721.00PAPERMATE OFFICE PEN.BLACKRED11.00DZ721.00PAPERMATE OFFICE PEN.BLACKSANFORD11.10DZ721.00MEDIUM.EXPO FEASERBLACKSANFORD11.19EA721.00MEDIUM.EXPO FEASERBLACKSANFORD11.19EA721.00MEDIUM.EXPO FEASERBLACKSANFORD17.47DZ121.41MARKER-BKBLACKSANFORD17.47DZ127.47MARKER-BKGUILL STANDARD STAPLESBLACKSANFORD17.47DZGOUCTLUQUD ACCENT HIGHLIGHTERFLWSANFORD17.477.47SOUCCTLUQUD ACCENT HIGHLIGHTERFLWSANFORD17.0EA12SOUCTLOUID ACCENT HIGHLIGHTERFLWSANFORD1.14MI BrandGUILL STANDARD STAPLES1.14BX121214.70SOUCTCUILL STANDARD STAPLES1.14BX12500.14.14GUILL STANDARD STAPLESD1.16BX12AS500.14GUILL STANDARD STAPLESD1.14BX7.0.14.14GUILL STANDARD STAPLESD1.16BX12AS500GUIS STEEL BINDER CLIPSD1 <td>ſ</td> <td></td> <td>WIE DIUM</td> <td>BLUE</td> <td></td> <td></td> <td></td> <td></td> <td>BX = 1 07</td> <td></td> <td></td> <td></td> <td>507</td>	ſ		WIE DIUM	BLUE					BX = 1 07				507
PAPERMATE OFFICE PEN.BLACKBLACKI.00DZ721.00MEDIUMEXPO EPASERSANFORD11.10DZ721.00EXPO EPASERSANFORD11.19EA121.99EXPO EPASERBLACKSANFORD17.47DZ121.99MARKER-BKBLACKSANFORD17.47DZ127.47MARKER-BKBLACKSANFORD17.47DZ127.47GUILL STANDARD STAPLES1-49BK5000 EA = 1 BXBD.49GUILL STANDARD STAPLES1-49BK5000 EA = 1 BXBD.49GUILL STANDARD STAPLES1-14BK5000 EA = 1 BXBD.49GUILL STANDARD STAPLES1-14BK500 EA = 1 BXBD.49GUILL STANDARD STAPLES1-14BK12 EA = 1 BX70.14GUIL STANDARD STAPLES1-14BK12 EA = 1 BX30.14GUIL STANDER CLIPS1-14BK12 EA = 1 BX50.14GUIS STEEL BINDER CLIPS1-18BX12 EA = 1 BX50.14GUIS STEEL BINDER CLIPS1-18BX12 EA = 1 BX50.14GUIS STEEL BINDER CLIPS1-18BX12 EA = 1 BX50.14GUIS STEEL BINDER CLIPS1-1812-1830.1450GUIS STEEL BINDER CLIPS-18-18 <td>80</td> <td>332</td> <td></td> <td>RED</td> <td></td> <td>+</td> <td>ł</td> <td></td> <td></td> <td>7,</td> <td>1.00</td> <td></td> <td></td>	80	332		RED		+	ł			7,	1.00		
meculum         BLACK         BLACK         I.00         D2         72         1.00           EXPO ERASER         SANFORD         1         1.19         EA         12         1.19           MARKER-BK         BLACK         SANFORD         1         7.47         2         1.00           MARKER-BK         BLACK         SANFORD         1         7.47         2         7.47           MARKER-BK         BLACK         SANFORD         1         7.47         2         7.47           CUILL STANDARD STAPLES         BLACK         SANFORD         1         7.47         7         7           G000LL STANDARD STAPLES         BLACK         SANFORD         1         -49         BX         500         -49         Oull Brand           G000LL STANDARD STAPLES         FLYW         SANFORD         1         -49         BX         500         -13         2           J34X348         GUB STEEL BINDER CLIPS, 2X1*         1         -14         BX         -12         A         -14         -14         -14         -14         -14         -14         -14         -14         -14         -14         -14         -14         -14         -14         -14         -14 <td></td> <td>333</td> <td>OFFICE PEN.</td> <td></td> <td></td> <td>-</td> <td></td> <td>2</td> <td></td> <td>72</td> <td>1.00</td> <td></td> <td></td>		333	OFFICE PEN.			-		2		72	1.00		
FINE POINT DRY ERASE         SANFORD         1         1.19         EA         12         1.19           MARKER-BIX         BLACK         SANFORD         1         7.47         DZ         12 EA=1 DZ         12         1.19           MARKER-BIX         BLACK         SANFORD         1         7.47         DZ         12 EA=1 DZ         12         7.47           GUILL STANDARD STAPLES         BLACK         SANFORD         1         -49         BX         5000 EA=1 BX         80         -49         Quill Brand           GOUGUT         LIQUID ACCENT HIGHLIGHTER         FLYW         SANFORD         1         -70         EA         12 EA=1 DZ         14         -70           AX308         34X308         1        10         BX         12 EA=1 DZ         14         -70           AB STEEL BINDER CLIPS, 2X1*         1        10         BX         12 EA=1 BX         300        14 Guilt Brand           AB STEEL BINDER CLIPS, 2X1*         1        10         BY         12 EA=1 BX         50        70         BA		81505				¥		N		72	1.00		
MARKER.BK         BLACK         SANFORD         1         7.47         DZ         12 EA=1 DZ         12         7.47           GUILL STANDARD STAPLES         BLACK         SANFORD         1         7.47         DZ         12 EA=1 DZ         12         7.47           5000CT         LIQUID ACCENT HIGHLIGHTER         FLYW         SANFORD         1         .49         BX         5000 EA=1 BX         B0         .49         Quill Brand           QB STEEL BINDER CLIPS         T         .70         EA         12 EA=1 DZ         14         .70           34X348         1         .34         BX         12 EA=1 BX         300         .14 Guill Brand           CB STEEL BINDER CLIPS, 2X1*         1         .36         BX         12 EA=1 BX         50         .39         Guill Brand		RADO 1 DU	FINE POINT NEV EBACT		SANFORD	1		8		¢₽			
QUIL STANDARD STAPLES       1       49       BX       LEA=1DZ       12       7.47         5000CT       LIQUID ACCENT HIGHLIGHTER       FLW       SANFORD       1       49       BX       5000 EA = 1 BX       B0       .49       Quill Brand         1       LIQUID ACCENT HIGHLIGHTER       FLW       SANFORD       1       .70       EA = 1 DZ       144       .70         308 STEEL BINDER CLIPS       1       .14       BX       12 EA = 1 BX       300       .14       .14       uill Brand         0B STEEL BINDER CLIPS       1       .13       BX       12 EA = 1 BX       300       .14       .09       .018       Brand		Values		BLACK	SANFORD					2	1.19		207
LTOUID ACCENT HIGHLIGHTER         Ft YW         SANFORD         1         -49         BX         5000 £A = 1 BX         B0         -49         Quill Brand           QB STEEL BINDER CLIPS         Ft YW         SANFORD         1         .70         EA         12 EA = 1 DZ         14         .70           QB STEEL BINDER CLIPS, 2X1*         1         .14         BX         12 EA = 1 DZ         14         .70           QB STEEL BINDER CLIPS, 2X1*         1         .14         BX         12 EA = 1 BX         300         .14 Guilt Brand		791123	ARD STAPLES						2 EA= 1 DZ	12	7.47		283
QB STEEL BINDER CLIPS     TU     SANF URD     T     T0     EA     T2 EA = 1 DZ     T4A     T0       3/4X3/8     3/4X3/8     1     .14     BX     12 EA = 1 BX     300     .14     Oill Brand       QB STEEL BINDER CLIPS, 2X1*     1     .14     BX     12 EA = 1 BX     300     .14     Oill Brand       QB STEEL BINDER CLIPS, 2X1*     1	17	24425	VT HIGHI ICHTED	. 1		-			10 EA = 1 BX	80	49	Oull Brand	300
34X308         1         .14         BX         12 EA = 1 BX         300         .14         Oull Brand           OB STEEL BINDER CLIPS, 2X1*         1         .89         BX         12 EA = 1 BX         300         .14         Oull Brand		720200			ANFURD	4		┥─┤	EA = 1 DZ	141	02.		22
1 .89 BX 12 EA = 1 BX 60 .39 Quill Brand		721000	3/4X3/8 OB STEEL BRIDED STEEL			÷			: EA = 1 BX	300	14	Juill Brand	č
			A C C C C C C C C C C C C C C C C C C C			-		+-	FAELAY	0			244
						 				B	62.	Juil Brand	244

-	384	209	200	607	236	236	2	490	207		207	207		207	auc C		244	T	244	431	1.40
	J. Salquil Brand	2	5		1.14 Quill Brand	2.19 Quill Brand			5.78 Quill Brand		5.78 Quill Brand	5.78 Quill Brand		5.78 Quill Brand			1.59 Quill Brand		J. OPP. Cluin Brand	5.41 Quill Brand	54 Oult Reard
Ö r	7	2.22	3.25		1.1	2.1	10 6/		5.78		5.76	5.76		5.78	5.59		1.59	64 6	3.08	5.41	52
<u>د</u>		57	12		£	18		_	5	ŗ	ž	12		7	12		2		2	9	144
100 EA = 1 BX	10 EA - 4 DA	701 - 47 71	12 EA = 1 DZ		701 = 43 71	12 EA = 1 DZ	250 EA = 1 BX								12 EA = 1 DZ		Yani - vui	1  PK = 10  BX		1 EA = 1 BX	
3.99 BX	20 02	1	51 DZ	114 17	- 1	20 6	N BX	1	77 0	B D2	1	ZOR	8 22	-	ZO	x a a		PX A		XN T	ž
3.5	222		3.25			2.19	10.60	1, 10	5	5.78	1	5.78	5.78		5.59	1.50		3.69	L. L.	0.41	Sd
	-				·   ·		+		•	<b>*</b>		-		-		Q.14		¥	-	-	=
444-447-7742-00-477-877-877-877-877-877-877-877-877-877	WHITE AMPAD																				
MNILA	WHITE			YELLW	YELLW		NHAP1	BLACK		KED	BIDF		GREEN	Murre					WHITE		
QB FILE FOLDER, LTR, 1/3-CUT  MNILA  SCRATCH PADS 3725-100	SHEETS	SCRATCH PADS,4"X6",100-	OB SELE EVICK MOTES	1/2X2-	<b>OB SELF-STICK NOTES, 3X3-</b>	QB ENVELOPE OPEN-END,	9-X12"	UUILL DRY ERASE MARKER CHS TIP			QUILL DRY ERASE MARKER CHS TIP		CHS TIP	RECYCLED WRITING PADS, 5X		SMOOTH STREAMAN	QB PAPER CLIPS, JUMBO	HIDDA	QB LASER LBLS 1X 2 5/8, N	SIBLE 3/4X1705	
740137 21420	05912	21431			[7384YW	OE91228		787139	787137		787140	787130		20152Q		C 17.5	P1JG		710786	765002	
26 27	2	28	04	5	B	31		32	33		34	35		36	17	5	38		39	40	

1

5

&u&b&d Page 2/2

#### Oak Park and River Forest High School District 200

201 North Scoville Avenue • Oak Park, IL 60302-2296

TO:Board of EducationFROM:Cheryl L. WithamDATE:February 24, 2011RE:Student Fees for Fiscal Year 2011 - 2012

#### BACKGROUND

Each year the administration reviews the student fees.

#### SUMMARY OF FINDINGS

Last year the athletic Pay to Play fee was increase by \$5.00, and all other fees remained the same.

Description	Amount 2010	Amount 2011	mount 2012
Freshman	\$85.00	\$85.00	\$85.00
Sophomore	\$245.00	\$245.00	\$245.00
Juniors	\$60.00	\$60.00	\$60.00
Seniors	\$95.00	\$95.00	\$95.00
Pay to Play	\$50.00	\$55.00	\$55.00

#### **RECOMMENDATIONS (OR FUTURE DIRECTIONS)**

MOTION: To approve no increase in the student registration fees for the 2011 – 2012 School Year.

ROLL CALL VOTE

AGENDA ITEM VIII. I.

#### Oak Park and River Forest High School District 200

201 North Scoville Avenue 
Oak Park, IL 60302-2296

RE:	Architect Contract
DATE:	February 24, 2011
FROM:	Cheryl L. Witham
TO:	Board of Education

#### BACKGROUND

We have incorporated Ms. Patchak-Layman's request to add language to section 3.2.5.1. We have carefully reviewed Mr. Allen's request to insure diversity in the labor contracts and have added language to the Construction Contract.

#### SUMMARY OF FINDINGS

Attorney Ken Florey presented and explained the contracts at the February 15<sup>th</sup> Finance Committee meeting.

#### **RECOMMENDATIONS (OR FUTURE DIRECTIONS)**

MOTION: To approve the Architect Contract as presented.

ROLL CALL VOTE

AGENDA ITEM VIII. G.

## AIA° Document B101<sup>™</sup> – 2007

#### Standard Form of Agreement Between Owner and Architect

AGREEMENT made as of the day of (In words, indicate day, month and year.)year)

in the year

**BETWEEN** the Architect's client identified as the Owner: (Name, legal status, address and other information)

Oak Park and River Forest High School District 200

and the Architect: (Name, legal status, address and other information)

Legat Architects 2015 Spring Road, Suite 175 Oak Brook, Illinois 60523

1

for the following Project:Project (Name, location and detailed description)

Blanket Agreement for all work directed by Owner and as more specifically described per project on AIA 6802 – 2007. The Owner and Architect agree as follows. This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

AlA Document B101™ – 2007 (formerly B151™ – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for resale.

#### TABLE OF ARTICLES

- 1 INITIAL INFORMATION
- 2 ARCHITECT'S RESPONSIBILITIES
- 3 SCOPE OF ARCHITECT'S BASIC SERVICES
- 4 ADDITIONAL SERVICES
- 5 OWNER'S RESPONSIBILITIES
- 6 COST OF THE WORK
- 7 COPYRIGHTS AND LICENSES
- 8 CLAIMS AND DISPUTES
- 9 TERMINATION OR SUSPENSION
- 10 MISCELLANEOUS PROVISIONS
- 11 COMPENSATION
- 12 SPECIAL TERMS AND CONDITIONS
- 13 SCOPE OF THE AGREEMENT

#### **EXHIBIT A INITIAL INFORMATION**

#### ARTICLE 1 INITIAL INFORMATION

§ 1.1 This Agreement is based on the Initial Information set forth in this Article 1 and in optional approved <u>Amendment example</u> Exhibit A, Initial Information:

(Complete Exhibit A, Initial Information, and incorporate it into the Agreement at Section 13.2, or state below Initial Information such as details of the Project's site and program, Owner's contractors and consultants, Architect's consultants, Owner's budget for the Cost of the Work, authorized representatives, anticipated procurement method, and other information relevant to the Project.)

§ 1.2 The Owner's anticipated dates for commencement of construction and Substantial Completion of the Work are set forth below:

.1 Commencement of construction date:

See attached Exhibit A which is incorporated into this Agreement.

.2 Substantial Completion date:

See attached Exhibit A which is incorporated into this Agreement. § 1.3 The Owner and Architect may rely on the Initial Information. Both parties, however, recognize that such information may materially change and, in that event, the Owner and the Architect shall appropriately adjust the schedule, the Architect's services and the Architect's compensation.

**§ 1.3** The Owner and Architect may rely on the Initial Information. Both parties, however, recognize that such information may materially change and, in that event, the Owner and the Architect shall appropriately adjust the schedule, the Architect's services and the Architect's compensation.

Init.

AlA Document B101<sup>TM</sup> – 2007 (formerly B151<sup>TM</sup> – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for resale.

§ 1.4 The Owner is engaging a Construction Manager in an "at risk" capacity. All references in this Agreement to "Contractor" shall also mean "Construction Manager."

#### ARTICLE 2 ARCHITECT'S RESPONSIBILITIES

§ 2.1 The Architect shall provide the professional services as set forth in this Agreement.

§ 2.2 The Architect shall perform its services consistent with the professional skill and care ordinarily provided by architects practicing in the same or similar locality under the same or similar circumstances. circumstances performing architectural services for school districts. The Architect shall perform its services as expeditiously as is consistent with such professional skill and care and the orderly progress of the Project. No other provision of this Agreement (or any document incorporated herein, or of any document binding upon the parties) shall be effective to the extent that it abridges or abrogates the standard of care set forth in this Section, regardless of the apparent specificity of the modifying provision.

§ 2.2.1 The Architect shall engage sub-consultants for portions of the Services as deemed necessary by the Architect. All services, whether performed by the Architect directly, or by sub-consultants shall be performed with the same standard of care provided in Section 2.2 of this Agreement.

§ 2.2.2 The Architect shall perform all Services in accordance with the professional standard of care relative to all applicable federal and state laws, as well as rules, regulations and specifications adopted by the Illinois State Board of Education, the Illinois Department of Public Health, and any other agency with authority over the construction or rehabilitation of school facilities.

§ 2.3 The Prior to commencement of services hereunder, the Architect shall identify a representative authorized to act on behalf of the Architect with respect to the Project.

§ 2.4 Except with the Owner's knowledge and consent, the Architect shall not engage in any activity, or accept any employment, interest or contribution that would reasonably appear to compromise the Architect's professional judgment with respect to this Project.

§ 2.5 The Architect shall maintain the following insurance for the duration of this Agreement. If any of the requirements set forth below exceed the types and limits the Architect normally maintains, the Owner shall reimburse the Architect for any additional cost: Agreement:

(Identify types and limits of insurance coverage, and other insurance requirements applicable to the Agreement, if any.)

.1 General Liability

See attached Certificate of Insurance, EXHIBIT B

.2 Automobile Liability

See attached Certificate of Insurance, EXHIBIT B

.3 Workers' Compensation

See attached Certificate of Insurance, EXHIBIT B

.4 Professional Liability

See attached Certificate of Insurance, EXHIBIT C § 2.5.1 Additional insurance requirements are found in Article 10 of this Agreement.

#### SCOPE OF ARCHITECT'S BASIC SERVICES ARTICLE 3

§ 3.1 The Architect's Basic Services consist of those described in Article 3 and include usual and customary structural, mechanical, and electrical engineering services. Services not set forth in this electrical engineering, and other services necessary for the complete design of, and preparation of construction documents for, the Project. Services not set forth in Article 3 are Additional Services.

AlA Document B101<sup>™</sup> – 2007 (formerly B151<sup>™</sup> – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of Init. this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for resale User Notes:

§ 3.1.1 The Architect shall manage the Architect's services, consult with the Owner, research applicable design criteria, attend Project meetings, communicate with members of the Project team and report progress to the Owner.

§ 3.1.2 The Architect shall coordinate its services with those services provided by the Owner and the Owner's consultants. consultants, if any. The Architect shall be entitled to rely on the accuracy and completeness of services and information furnished by the Owner and the Owner's consultants. The Architect shall provide prompt written notice to the Owner if the Architect becomes If the Architect, in the exercise of the applicable standard of care, should reasonably have become aware of any error, omission or inconsistency in such services or information.

§ 3.1.3 As soon as practicable after the date of this Agreement, the Architect shall submit for the Owner's approval a schedule for the performance of the Architect's services. The schedule initially shall include anticipated dates for the commencement of construction and for Substantial Completion and Final Completion of the Work as set forth in the Initial Information. The schedule shall include allowances for periods of time required for the Owner's review, for the performance of the Owner's consultants, consultants (if any), and for approval of submissions by authorities having jurisdiction over the Project. Once approved by the Owner, time limits established by the schedule shall not, except for reasonable cause, be exceeded by the Architect or Owner. With the Owner's approval, the Architect shall adjust the schedule, if necessary, necessary as the Project proceeds until the commencement of construction.

§ 3.1.4 The Architect shall not be responsible for an Owner's directive or substitution made without the Architect's approval.

§ 3.1.5 The Architect shall, at appropriate times, contact the governmental authorities required to approve the Construction Documents and the entities providing utility services to the Project. In designing the Project, the Architect shall respond to applicable design requirements imposed by such governmental authorities and by such entities providing utility services. Architect shall prepare the Construction Documents in compliance with all applicable building codes, as well as all federal, state, and local laws. As necessary, Architect shall review each agreement between utility companies and the Owner to confirm that the proposed utility service capacity, equipment, entrance location(s) and routing path has been provided to meet the requirements of the project design.

§ 3.1.7 Upon becoming aware of any error or omission in the Construction Documents (whether during the bidding phase or construction phase) Architect shall promptly perform the necessary professional service to issue an addendum or change order, as the case may be, to correct or clarify errors, omissions, or ambiguities. Such service shall be performed by Architect without additional compensation, unless the error or omission was not the fault of the Architect or their Consultants.

§ 3.1.8 If Architect key personnel are identified in Architect's proposal, such personnel shall not be removed or replaced for services performed for the Project without Owner's prior express written Consent,

§ 3.1.6 The Architect shall assist the Owner in connection with the Owner's responsibility for filing documents required for the approval of governmental authorities having jurisdiction over the Project.

#### § 3.2 SCHEMATIC DESIGN PHASE SERVICES

§ 3.2.1 The Architect shall review the program and other information furnished by the Owner, if any, and shall review laws, codes, and regulations applicable to the Architect's services.

§ 3.2.2 The Architect shall prepare a preliminary evaluation of the Owner's program, schedule, budget for the Cost of the Work, Project site, and the proposed procurement or delivery method and other Initial Information, if any, each in terms of the other, to ascertain the requirements of the Project. The Architect shall notify the Owner of (1) any inconsistencies discovered in the information, and (2) other information or consulting services that may be reasonably needed for the Project.

§ 3.2.3 The Architect shall present its preliminary evaluation to the Owner and shall discuss with the Owner alternative approaches to design and construction of the Project, including the feasibility of incorporating environmentally responsible design approaches. The Architect shall reach an understanding with the Owner regarding the requirements of the Project.

Init.

AIA Document B101<sup>TM</sup> - 2007 (formerly B151<sup>TM</sup> - 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under this AIA the law. This document was produced by AIA software at 16:31:20 on 02/11/2011 under Order No.3032139990 1 which expires on 08/16/2011, and is not for resale. User Notes:

§ 3.2.4 Based on the Project's requirements agreed upon with the Owner, the Architect shall prepare and present for the Owner's approval a preliminary design illustrating the scale and relationship of the Project components.

§ 3.2.5 Based on the Owner's approval of the preliminary design, design and comments thereupon, the Architect shall prepare Schematic Design Documents for the Owner's approval. The Schematic Design Documents shall consist of drawings and other documents including a site plan, if appropriate, and preliminary building plans, sections and elevations; and may include some combination of study models, perspective sketches, or digital modeling. Preliminary selections of major building systems and construction materials shall be noted on the drawings or described in writing.

§ 3.2.5.1 The Architect shall consider environmentally responsible design alternatives, such as material choices and building orientation, together with other considerations based on program and aesthetics, in developing a design that is consistent with the Owner's program, schedule and budget for the Cost of the Work. The Owner may obtain other environmentally responsible design services under Article 4.

§ 3.2.5.2 The Architect shall consider the value of alternative materials, building systems and equipment, together with other considerations based on program and aesthetics, aesthetics in developing a design for the Project that is consistent with the Owner's program, schedule and budget for the Cost of the Work.

§ 3.2.6 The Architect shall submit to the Owner an estimate of the Cost of the Work prepared in accordance with Section 6.3.

§ 3.2.7 The Architect shall submit the Schematic Design Documents to the Owner, and request the Owner's approval. The Owner may approve the Schematic Design Documents subject to comments. The Architect shall consider and evaluate the comments for incorporation into subsequent portions of the of the Services hereunder.

#### § 3.3 DESIGN DEVELOPMENT PHASE SERVICES

§ 3.3.1 Based on the Owner's approval of the Schematic Design Documents, and on the Owner's written authorization of any adjustments in the Project requirements and the budget for the Cost of the Work, the Architect shall prepare Design Development Documents for the Owner's approval. The Design Development Documents shall illustrate and describe the development of the approved Schematic Design Documents and shall consist of drawings and other documents including plans, sections, elevations, typical construction details, and diagrammatic layouts of building systems to fix and describe the size and character of the Project as to architectural, structural, mechanical and electrical systems, and such other elements as may be appropriate. The Design Development Documents shall also include outline specifications that identify major materials and systems and establish in general their quality levels.

§ 3.3.2 The Architect shall update the estimate of the Cost of the Work.

§ 3.3.3 The Architect shall submit the Design Development Documents documents to the Owner, advise the Owner of any adjustments to the estimate of the Cost of the Work, and request the Owner's approval. The Owner may approve the Design Development Documents subject to comments. The Architect shall consider and evaluate the comments for incorporation into subsequent portions of the Services hereunder.

#### § 3.4 CONSTRUCTION DOCUMENTS PHASE SERVICES

§ 3.4.1 Based on the Owner's approval of the Design Development Documents, and on the Owner's written authorization of any adjustments in the Project requirements and the budget for the Cost of the Work, the Architect shall prepare Construction Documents for the Owner's approval. The Construction Documents shall illustrate and describe the further development of the approved Design Development Documents and shall consist of Drawings and Specifications setting forth in detail the quality levels of materials and systems and other requirements for the construction of the Work. The Owner and Architect acknowledge that in order to construct the Work the Contractor will provide additional information, including Shop Drawings, Product Data, Samples and other similar submittals, which the Architect shall review in accordance with Section 3.6.4.

§ 3.4.2 The Architect shall incorporate into the Construction Documents the design requirements of governmental authorities having jurisdiction over the Project.

§ 3.4.3 During the development of the Construction Documents, the Architect shall assist the Owner in the development and preparation of (1) bidding and procurement information that describes the time, place and conditions

User Notes:

Init.

1

AIA Document B101<sup>TM</sup> – 2007 (formerly B151<sup>TM</sup> – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for resale

of bidding, including bidding or proposal forms; (2) the form of agreement between the Owner and Contractor; and (3) the Conditions of the Contract for Construction (General, Supplementary and other Conditions). The Architect shall also compile a project manual that includes the Conditions of the Contract for Construction and Specifications and may include bidding requirements and sample forms.

§ 3.4.4 The Architect shall update the estimate for the Cost of the Work.

**§ 3.4.5** The Architect shall submit the Construction Documents to the Owner, advise the Owner of any adjustments to the estimate of the Cost of the Work, take any action required under Section 6.5, and request the Owner's approval.

#### § 3.5 BIDDING OR NEGOTIATION PHASE SERVICES

#### § 3.5.1 GENERAL

The Architect shall assist the Owner in establishing a list of prospective contractors. Following the Owner's approval of the Construction Documents, the Architect shall assist the Owner in (1) obtaining either competitive bids or negotiated proposals; (2) confirming responsiveness of bids or proposals; (3) determining the successful bid or proposal, if any; and, (4) awarding and preparing contracts for construction.

§ 3.5.2 The Architect shall submit to Owner's attorneys for review and approval all agreements and conditions prepared for publication to bidders prior to publication and advertisement for bids.

#### § 3.5.2 COMPETITIVE BIDDING

§ 3.5.2.1 Bidding Documents shall consist of bidding requirements and proposed Contract Documents.

§ 3.5.2.2 The Architect shall assist the Owner in bidding the Project by

- .1 procuring the reproduction of Bidding Documents for distribution to prospective bidders;
- .2 distributing the Bidding Documents to prospective bidders, requesting their return upon completion of the bidding process, and maintaining a log of distribution and retrieval and of the amounts of deposits, if any, received from and returned to prospective bidders;
- .3 organizing and conducting a pre-bid conference for prospective bidders;
- .4 preparing responses to questions from prospective bidders and providing clarifications and interpretations of the Bidding Documents to all prospective bidders in the form of addenda; and
- .5 organizing and conducting the opening of the bids, and subsequently documenting and distributing the bidding results, as directed by the Owner.

§ 3.5.2.3 The Architect shall consider requests for substitutions, if the Bidding Documents permit substitutions, and shall prepare and distribute addenda identifying approved substitutions to all prospective bidders.

#### § 3.5.3 NEGOTIATED PROPOSALS

§ 3.5.3.1 It is anticipated that all construction work shall be publicly bid. Where public bidding is not required, Architect shall assist with negotiated proposals. Proposal Documents shall consist of proposal requirements and proposed Contract Documents.

§ 3.5.3.2 The Architect shall assist the Owner in obtaining proposals by

- .1 procuring the reproduction of Proposal Documents for distribution to prospective contractors, and requesting their return upon completion of the negotiation process;
- .2 organizing and participating in selection interviews with prospective contractors; and
- .3 participating in negotiations with prospective contractors, and subsequently preparing a summary report of the negotiation results, as directed by the Owner.

§ 3.5.3.3 The Architect shall consider requests for substitutions, if the Proposal Documents permit substitutions, and shall prepare and distribute addenda identifying approved substitutions to all prospective contractors.

#### § 3.6 CONSTRUCTION PHASE SERVICES

#### § 3.6.1 GENERAL

§ 3.6.1.1 The Architect shall provide administration of the Contract between the Owner and the Contractor as set forth below and in AIA Document A201<sup>TM</sup>-2007, General Conditions of the Contract for Construction. <u>Construction as amended</u>. If the Owner and Contractor modify AIA Document A201-2007, <u>A201-2007</u>, <u>A201-2007</u> subsequent to contract award,

Init.

1

AIA Document B101<sup>TM</sup> – 2007 (formerly B151<sup>TM</sup> – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for resale.

those modifications shall not affect the Architect's services under this Agreement unless the Owner and the Architect amend this Agreement.

§ 3.6.1.2 The Architect shall advise and consult with the Owner during the Construction Phase Services. The Architect shall have authority to act on behalf of the Owner only to the extent provided in this Agreement. The Architect shall not have control over, charge of, or responsibility for the construction means, methods, techniques, sequences or procedures, or for safety precautions and programs in connection with the Work, nor shall the Architect be responsible for the Contractor's failure to perform the Work in accordance with the requirements of the Contract Documents. The Architect shall be responsible for the Architect's negligent acts or omissions, but shall not have control over or charge of, and shall not be responsible for, acts or omissions of the Contractor or of any other persons or entities performing portions of the Work.

§ 3.6.1.3 Subject to Section 4.3, the Architect's responsibility to provide Construction Phase Services commences with the award of the Contract for Construction and terminates on the date the Architect issues the final Certificate for Payment.

§ 3.6.1.4 The Architect shall promptly make corrections to the Construction Documents, at no cost to the Owner, if any errors, omissions, or ambiguities are discovered during the Construction Phase, unless the error or omission was not the fault of the Architect or their Consultants.

§ 3.6.1.5 The Architect shall conduct progress periodic review and payment review meetings which shall include the Owner, the Architect, and appropriate prime contractors. The Architect shall record the meetings in written meeting minutes, which shall be distributed for review no more than two days following each meeting. Comments on the published meeting minutes shall be discussed as part of the following progress review and payment meeting. Each progress review and payment meeting shall cover, at a minimum, progress, new and old business, and quality schedule. Notwithstanding, it is anticipated that the Contractor shall have separate meetings for its own purposes of planning and coordinating its own work.

#### § 3.6.2 EVALUATIONS OF THE WORK

§ 3.6.2.1 The Architect Architect, and its sub-consultants, shall visit the site at intervals appropriate to the stage of construction, or as otherwise required in Section 4.3.3, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine, in general, if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect shall not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. On the basis of the site visits, the Architect shall keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and report to the Owner (1) known deviations from the Contract Documents and from the most recent construction schedule submitted by the Contractor, approved construction schedule, and (2) defects and deficiencies observed in the Work.

§ 3.6.2.1.1 On the basis of Architect's on-site construction observation, the Architect shall endeavor to keep the Owner informed of the progress and quality of the Work, and shall endeavor to guard the Owner against defects and deficiencies in the Work. The Architect shall promptly notify the Owner of any defects or deficiencies in the Work observed by the Architect. The Architect shall submit a field report to the Owner for each week of site visits or for each site visit, whichever the Architect deems appropriate, and shall submit a copy to the Contractor.

§ 3.6.2.2 The Architect has the authority and responsibility to reject Work that does not conform to the Contract Documents. Whenever the Architect considers it necessary or advisable, the Architect shall have the authority to require inspection or testing of the Work in accordance with the provisions of the Contract Documents, whether or not such Work is fabricated, installed or completed. However, neither this authority of the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Contractor, Subcontractors, material and equipment suppliers, their agents or employees or other persons or entities performing portions of the Work.

§ 3.6.2.3 The Architect shall interpret and decide matters concerning performance under, and requirements of, the Contract Documents on written request of either the Owner or Contractor. The Architect's response to such requests shall be made in writing within any time limits agreed upon or otherwise with reasonable promptness.promptness so as not to delay the progress of construction,

AIA Document B101<sup>TM</sup> – 2007 (formerly B151<sup>TM</sup> – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights Init. reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for resale User Notes:

§ 3.6.2.4 Interpretations and decisions of the Architect shall be consistent with the intent of and reasonably inferable from the Contract Documents and shall be in writing or in the form of drawings. When making such interpretations and decisions, the Architect shall endeavor to secure faithful performance by both Owner and Contractor, shall not show partiality to either, and shall not be liable for results of interpretations or decisions rendered in good faith. The Architect's decisions on matters relating to aesthetic effect shall be final if consistent with the intent expressed in the Contract Documents.

§ 3.6.2.5 Unless the Owner and Contractor designate another person to serve as an Initial Decision Maker, as that term is defined in AIA Document A201–2007, the Architect shall render initial decisions on Claims between the Owner and by the Contractor as provided in the Contract Documents.

#### § 3.6.3 CERTIFICATES FOR PAYMENT TO CONTRACTOR

§ 3.6.3.1 The Architect shall review and certify the amounts due the Contractor and shall issue certificates in such amounts. The Architect's certification for payment shall constitute a representation to the Owner, based on the Architect's evaluation of the Work as provided in Section 3.6.2 and on the data comprising the Contractor's Application for Payment, that, to the best of the Architect's knowledge, information and belief, the Work has progressed to the point indicated and that the quality of the Work is in accordance with the Contract Documents. Documents, and that the Contractor is entitled to the amount sought in the application for payment. The foregoing representations are subject (1) to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, (2) to results of subsequent tests and inspections, (3) to correction of minor deviations from the Contract Documents prior to completion, and (4) to specific qualifications expressed by the Architect.

§3.6.3.1.1 The Construction Documents shall provide that each Prime Contractor submit a schedule of values prior to commencing its Work. The schedule of values, among other things, shall be used by the Architect as a basis for evaluating the applications for payment.

§3.6.3.1.2 The Architect shall determine whether Contractor's contractual prerequisites to payment have been satisfied for each Application for Payment; such as submission of Certified Payrolls, submission of Schedule of Values and submission of Lien Waivers executed by Subcontractors, among other things.

§ 3.6.3.2 The issuance of a Certificate for Payment shall not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work, (2) reviewed construction means, methods, techniques, sequences or procedures, (3) reviewed copies of requisitions received from Subcontractors and material suppliers and other data requested by the Owner to substantiate the Contractor's right to payment, or (4) ascertained how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.

§ 3.6.3.3 The Architect shall maintain a record of the Applications and Certificates for Payment.

#### § 3.6.4 SUBMITTALS

§ 3.6.4.1 The Architect shall review the Contractor's submittal schedule and shall not unreasonably delay or withhold approval. The Architect's action in reviewing submittals shall be taken in accordance with the approved submittal schedule or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time in the Architect's professional judgment to permit adequate review.

**§ 3.6.4.2** In accordance with the Architect-approved submittal schedule, the Architect shall review and approve or take other appropriate action upon the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. Review of such submittals is not for the purpose of determining the accuracy and completeness of other information such as dimensions, quantities, and installation or performance of equipment or systems, which are the Contractor's responsibility. The Architect's review shall not constitute approval of safety precautions or, unless otherwise specifically stated by the Architect, of any construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.

Init.

AIA Document B101<sup>TM</sup> – 2007 (formerly B151<sup>TM</sup> – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for resale.

§ 3.6.4.3 If the Contract Documents specifically require the Contractor to provide professional design services or certifications by a design professional related to systems, materials or equipment, the Architect shall specify the appropriate performance and design criteria that such services must satisfy. The Architect shall review Shop Drawings -shop drawings and other submittals related to the Work designed or certified by the design professional retained by the Contractor that bear such professional's seal and signature when submitted to the Architect. The Architect shall be entitled to rely upon the adequacy, accuracy and completeness of the services, certifications and approvals performed or provided by such design professionals.

§ 3.6.4.4 Subject to the provisions of Section 4.3, the Architect shall review and respond to requests for information about the Contract Documents. The Architect shall set forth in the Contract Documents the requirements for requests for information shall include, at a minimum, a detailed written statement that indicates the specific Drawings or Specifications in need of clarification and the nature of the clarification requested. The Architect's response to such requests shall be made in writing within any time limits agreed upon, or otherwise with reasonable promptness. promptness in compliance with the Contractors reasonable requested amount of review time for the Architects response. If appropriate, the Architect shall prepare and issue supplemental Drawings and Specifications in response to requests for information.

§ 3.6.4.5 The Architect shall maintain a record of submittals and copies of submittals supplied by the Contractor Contractor, and the Architect's responses thereto, in accordance with the requirements of the Contract Documents. The Architect shall also maintain a record of requests for information and responses thereto, and copies of the same.

#### § 3.6.5 CHANGES IN THE WORK

§ 3.6.5.1 The Architect may authorize minor changes in the Work that are consistent with the intent of the Contract Documents and do not involve an adjustment in the Contract Sum or an extension of the Contract Time. Subject to the provisions of Section 4.3, the Architect shall prepare Change Orders and Construction Change Directives for the Owner's approval and execution in accordance with the Contract Documents.

§ 3.6.5.2 The Architect shall maintain records relative to changes in the Work.

#### § 3.6.6 PROJECT COMPLETION

§ 3.6.6.1 The Architect shall conduct inspections to determine the date or dates of Substantial Completion and the date of final completion; issue Certificates of Substantial Completion; receive from the <u>Contractor Contractor, review for</u> <u>compliance with the Contract Documents</u>, and forward to the Owner, for the Owner's review and records, written warranties and related documents required by the Contract Documents and assembled by the Contractor; and issue a final Certificate for Payment based upon a final inspection indicating the Work complies with the requirements of the Contract Documents. <u>However, nothing in this paragraph shall obligate the Architect to make legal determinations regarding the adequacy of language or signatures in the documents provided.</u>

§ 3.6.6.2 The Architect's inspections shall be conducted with the Owner to check conformance of the Work with the requirements of the Contract Documents and to verify the accuracy and completeness of the list submitted by the Contractor of Work to be completed or corrected. punch list created after Substantial Completion.

§ 3.6.6.3 When the Work is found to be substantially complete, the Architect shall inform the Owner about the balance of the Contract Sum remaining to be paid the Contractor, including the amount to be retained from the Contract Sum, if any, for final completion or correction of the Work.

§ 3.6.6.4 The Architect shall forward to the Owner the following information received from the Contractor: (1) consent of surety or sureties, if any, to reduction in or partial release of retainage or the making of final payment; (2) affidavits, receipts, releases and waivers of liens or bonds indemnifying the Owner against liens; and (3) any other documentation required of the Contractor under the Contract Documents.

§ 3.6.6.5 Upon request of the Owner, and prior to the expiration of one year from the date of Substantial Completion, the Architect shall, without additional compensation, conduct a meeting with the Owner to review the facility operations and performance.

AlA Document B101<sup>™</sup> – 2007 (formerly B151<sup>™</sup> – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AlA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AlA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AlA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for resale.

§3.6.6.6 The Architect shall periodically review the Contractor's marked up as-built drawings to determine whether the Contractor is recording information as required in the Contract Documents. Before Final Completion, the Architect shall require the Contractor to transfer the as-built drawings to the Owner.

#### **ARTICLE 4** ADDITIONAL SERVICES

§ 4.1 Additional Services listed below are not included in Basic Services but may be required for the Project. The Architect shall provide the listed Additional Services only if specifically designated in the table below as the Architect's responsibility, and the Owner shall compensate the Architect as provided in Section 11.2. (Designate the Additional Services the Architect shall provide in the second column of the table below. In the third column indicate whether the service description is located in Section 4.2 or in an attached exhibit. If in an exhibit, identify the exhibit.)

The Owner may from time to time, without invalidating this Agreement, direct changes in the Architect's Services. The changes may include additions, deletions, or modifications to the Architect's Services. Upon receipt of a Direction for Change in Services, the Architect shall proceed promptly to enact the Change in Services. Concurrently, the Owner and Architect shall agree on an equitable adjustment in the Architect's Basis of Payment.

	(Architect, Owner or Not Provided)	Location of Service Description (Section 4.2 below or in an exhibit attached to this document and identified below)
§ 4.1.1 Programming		
§ 4.1.2 Multiple preliminary designs		
§ 4.1.3 Measured drawings		
§ 4.1.4 Existing facilities surveys		
§ 4.1.5 Site Evaluation and Planning (B203™ 2007)		
§ 4.1.6 Building information modeling		
4.1.7 Civil engineering		
4.1.8 Landscape design		
4.1.9 Architectural Interior Design (B252TM 2007)		
4.1.10 Value Analysis (B204™ 2007)		
4.1.11 Detailed cost estimating		
4.1.12 On-site project representation		
4.1.13 Conformed construction documents		
4.1.14 As-Designed Record drawings		
4.1.15 As Constructed Record drawings		
4.1.16 Post occupancy evaluation		
4.1.17 Facility Support Services (B210 <sup>™</sup> 2007)		
4.1.18 Tenant related services		
4.1.19 Coordination of Owner's consultants		
4.1.20 Telecommunications/data design		
4.1.21 Security Evaluation and Planning (B206™ 2007)		
4.1.22 Commissioning (B211™ 2007)		
4.1.23 Extensive environmentally responsible design		÷.
4.1.24 LEED <sup>®</sup> Certification (B214 <sup>™</sup> 2007)		
4.1.25 Fast-track design services		
4.1.26 Historic Preservation (B205™ 2007)		
4.1.27 Furniture, Furnishings, and Equipment Design (B253™ 2007)		

Init.

AIA Document B101™ - 2007 (formerly B151™ - 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of 10 Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under this AIA the law. This document was produced by AIA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for resale. User Notes:

**§ 4.2** Insert a description of each Additional Service designated in Section 4.1 as the Architect's responsibility, if not further described in an exhibit attached to this document.

<u>§4.2 If the Owner issues any direction to the Architect which the Architect deems to be a Direction for Change to this</u> <u>Agreement warranting amendment to the Architect's basis of payment on contract price, as the case may be, the</u> <u>Architect shall promptly notify the Owner.</u>

§ 4.3 Additional Services may be provided requested by the Owner after execution of this Agreement, without invalidating the Agreement. Except for services required due to the fault of the Architect, any Additional Services provided in accordance with this Section 4.3 shall entitle the Architect to compensation pursuant to Section 11.3 and an appropriate adjustment in the Architect's schedule.

**§ 4.3.1** Upon recognizing the need to perform the following Additional Services, the Architect shall notify the Owner with reasonable promptness and explain the facts and circumstances giving rise to the need. The Architect shall not proceed to provide the following any amended services until the Architect receives the Owner's written authorization:

- Services necessitated by a change in the Initial Information, previous instructions or approvals given by the Owner, or a material change in the Project including, but not limited to, size, quality, complexity, the Owner's schedule or budget for Cost of the Work, or procurement or delivery method;
- .2 Services necessitated by the Owner's request for extensive environmentally responsible design alternatives, such as unique system designs, in depth material research, energy modeling, or LEED® certification;
- 3 Changing or editing previously prepared Instruments of Service necessitated by the enactment or revision of codes, laws or regulations or official interpretations;
- 4 Services necessitated by decisions of the Owner not rendered in a timely manner or any other failure of performance on the part of the Owner or the Owner's consultants or contractors;
- 5 Preparing digital data for transmission to the Owner's consultants and contractors, or to other Owner authorized recipients;
- .6 Preparation of design and documentation for alternate bid or proposal requests proposed by the Owner;
- .7 Preparation for, and attendance at, a public presentation, meeting or hearing;
- .8 Preparation for, and attendance at a dispute resolution proceeding or legal proceeding, except where the Architect is party thereto;
- .9 Evaluation of the qualifications of bidders or persons providing proposals;
- .10 Consultation concerning replacement of Work resulting from fire or other cause during construction; or
- .11 Assistance to the Initial Decision Maker, if other than the Architect.authorization.

**§ 4.3.2** To avoid delay in the Construction Phase, the Architect shall provide the following Additional Services, notify the Owner with reasonable promptness, and explain the facts and circumstances giving rise to the need. If the Owner subsequently determines that all or parts of those services are not required, the Owner shall give prompt written notice to the Architect, and the Owner shall have no further obligation to compensate the Architect for those services:

- .1 Reviewing a Contractor's submittal out of sequence from the submittal schedule agreed to by the Architect;
- .2 Responding to the Contractor's requests for information that are not prepared in accordance with the Contract Documents or where such information is available to the Contractor from a careful study and -comparison of the Contract Documents, field conditions, other Owner-provided information, Contractor prepared coordination drawings, or prior Project correspondence or documentation;
- .3 Preparing Change Orders and Construction Change Directives that require evaluation of Contractor's proposals and supporting data, or the preparation or revision of Instruments of Service;
- .4 Evaluating an extensive number of Claims as the Initial Decision Maker; or
- .5 Evaluating substitutions proposed by the Owner or Contractor and making subsequent revisions to Instruments of Service resulting therefrom; or

AlA Document B101<sup>™</sup> – 2007 (formerly B151<sup>™</sup> – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for resale.

.6 To the extent the Architect's Basic Services are affected, providing Construction Phase Services 60 days after (1) the date of Substantial Completion of the Work or (2) the anticipated date of Substantial Completion identified in Initial Information, whichever is earlier.

§ 4.3.3 The Architect shall provide Construction Phase Services exceeding the limits set forth below as Additional Services. When the limits below are reached, the Architect shall notify the Owner:

- .1 <u>Two</u> (2) reviews of each Shop Drawing, Product Data item, sample and similar submittal of the Contractor
- .2 Twelve (12) visits weekly to the site by the Architect and appropriate Architect's Consultants over the duration of the Project during construction
- .3 <u>Three (3) inspections for any portion of the Work to determine whether such portion of the Work</u> is substantially complete in accordance with the requirements of the Contract Documents
- .4 Three (3) inspections for any portion of the Work to determine final completion
- .5 Architect shall provide all site visits and inspections necessitated to resolve errors or conflicts in the Architect's and Architect's Consultants Construction Documents, as part of Basic Services and in addition to the amounts identified in 4.3.3.1, .2, .3, and 4.

§ 4.3.4 If the services covered by this Agreement have not been completed within ( ) months of the date of this Agreement, time contemplated in Section 1.2, through no fault of the Architect, extension of the Architect's services beyond that time shall be compensated as Additional Services.

#### **OWNER'S RESPONSIBILITIES** ARTICLE 5

§ 5.1 Unless otherwise provided for under this Agreement, the Owner shall shall, in consultation with the Architect, provide information in a timely manner regarding requirements for and limitations on the Project, including a written program which shall set forth the Owner's objectives, schedule, constraints and criteria, including space requirements and relationships, flexibility, expandability, special equipment, systems and site requirements. Within 15 days after receipt of a written request from the Architect, the Owner shall furnish the requested information as necessary and relevant for the Architect to evaluate, give notice of or enforce lien rights.

§ 5.2 The Owner shall shall, in consultation with the Architect, establish and periodically update the Owner's budget for the Project, including (1) the budget for the Cost of the Work as defined in Section 6.1; (2) the Owner's other costs; and, (3) reasonable contingencies related to all of these costs. If the Owner significantly increases or decreases the Owner's budget for the Cost of the Work, the Owner shall notify the Architect. The Owner and the Architect shall thereafter agree in writing to a corresponding change in the Project's scope and quality.

§ 5.3 The Owner shall identify a representative authorized to act on the Owner's behalf with respect to the Project. The Owner shall render decisions and approve the Architect's submittals in a timely manner in order to avoid unreasonable delay in the orderly and sequential progress of the Architect's services.

§ 5.4 The At the Architect's request, the Owner shall furnish surveys to describe physical characteristics, legal limitations and utility locations for the site of the Project, and a written legal description of the site. The surveys and legal information shall include, as applicable, grades and lines of streets, alleys, pavements and adjoining property and structures; designated wetlands; adjacent drainage; rights-of-way, restrictions, easements, encroachments, zoning, deed restrictions, boundaries and contours of the site; locations, dimensions and necessary data with respect to existing buildings, other improvements and trees; and information concerning available utility services and lines, both public and private, above and below grade, including inverts and depths. All the information on the survey shall be referenced to a Project benchmark.

§ 5.5 The Owner shall furnish services of geotechnical engineers, which may include but are not limited to test borings, test pits, determinations of soil bearing values, percolation tests, evaluations of hazardous materials, seismic evaluation, ground corrosion tests and resistivity tests, including necessary operations for anticipating subsoil conditions, with written reports and appropriate recommendations.

§ 5.6 The Owner Architect shall coordinate the services of its own Owner's consultants with those services provided by the Architect. Upon the Architect's request, the Owner shall furnish copies of the scope of services in the contracts between the Owner and the Owner's consultants. The Owner shall furnish the services of consultants other than those

AIA Document B101<sup>TM</sup> - 2007 (formerly B151<sup>TM</sup> - 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of 12 Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under this AIA' the law. This document was produced by AIA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for resale. User Notes:

designated in this Agreement, or authorize the Architect to furnish them as an Additional Service, when the Architect requests such services and demonstrates that they are reasonably required by the scope of the Project. The Owner shall require that its consultants maintain professional liability insurance as appropriate to the services provided.

§ 5.7 The Owner shall shall, upon request of the Architect, furnish tests, inspections and reports required by law or the Contract Documents, such as structural, mechanical, and chemical tests, tests for air and water pollution, and tests for hazardous materials.

**§ 5.8** The Owner shall furnish all legal, insurance and accounting services, including auditing services, that may be reasonably necessary at any time for the Project to meet the Owner's needs and interests.

§ 5.9 The Owner shall provide prompt written notice to the Architect if the Owner becomes aware of any fault or defect in the Project, including errors, omissions or inconsistencies in the Architect's Instruments of Service. Failure of the Owner to discover such defects, or to so notify the Architect, shall not relieve the Architect of any of its obligations hereunder.

§ 5.10 Except as otherwise provided in this Agreement, or when direct communications have been specially authorized, the Owner shall endeavor to communicate with the Contractor and the Architect's consultants through the Architect about matters arising out of or relating to the Contract Documents. The Owner shall promptly notify the Architect of any direct communications that may affect the Architect's services.

§ 5.11 Before executing the Contract for Construction, the <u>Owner Architect</u> shall coordinate the Architect's duties and responsibilities set forth in the Contract for Construction with the Architect's services set forth in this Agreement. The Owner shall provide the Architect a copy of the executed agreement between the Owner and Contractor, including the General Conditions of the Contract for Construction.

§ 5.12 The Owner shall provide the Architect access to the Project site prior to commencement of the Work and shall obligate the Contractor to provide the Architect access to the Work wherever it is in preparation or progress.

#### ARTICLE 6 COST OF THE WORK

§ 6.1 For purposes of this Agreement, the Cost of the Work shall be the total cost to the Owner to construct all elements of the Project designed or specified by the Architect and shall include contractors' general conditions costs, overhead and profit. The Cost of the Work does not include the compensation of the Architect, the pre-construction related costs of the Construction Manger, the costs of the land, rights-of-way, financing, contingencies for changes in the Work or other costs that are the responsibility of the Owner.

§ 6.2 The Owner's budget for the Cost of the Work is provided in Initial Information, will be established in each approved Amendment issued, and may be adjusted throughout the Project as required under Sections 5.2, 6.4 and 6.5. -5.2. Evaluations of the Owner's budget for the Cost of the Work, the preliminary estimate of the Cost of the Work and updated estimates of the Cost of the Work prepared by the Architect, represent the Architect's judgment as a design professional. It is recognized, however, that neither the Architect nor the Owner has control over the cost of labor, materials or equipment; the Contractor's methods of determining bid prices; or competitive bidding, market or negotiating conditions. Accordingly, the Architect cannot and does not warrant or represent that bids or negotiated prices will not vary from the Owner's budget for the Cost of the Work or from any estimate of the Cost of the Work or evaluation prepared or agreed to by the Architect. the Architect, except as provided elsewhere in the Article.

**§ 6.3** In preparing estimates of the Cost of Work, the <u>Architect Construction Manager</u> shall be permitted to include contingencies for design, bidding and price escalation; to determine what materials, equipment, component systems and types of construction are to be included in the Contract Documents; to make reasonable adjustments in the program and scope of the Project; and to include in the Contract Documents alternate bids as may be necessary to adjust the estimated Cost of the Work to meet the Owner's budget for the Cost of the Work. The <u>Architect's Construction Manager's</u> estimate of the Cost of the Work shall be based on current area, volume or similar conceptual estimating techniques. If the Owner requests detailed cost estimating services, the Architect shall provide such services as an Additional Service under Article 4.

Init.

AlA Document B101<sup>™</sup> – 2007 (formerly B151<sup>™</sup> – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AlA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AlA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AlA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for

**§ 6.4** If the Bidding or Negotiation Phase has not commenced within 90 days after the Architect submits the Construction Documents to the Owner, through no fault of the Architect, the Owner's budget for the Cost of the Work shall be adjusted to reflect changes in the general level of prices in the applicable construction market.

**§ 6.5** If at any time the Architect's estimate of the Cost of the Work exceeds the Owner's budget for the Cost of the Work, the Architect shall make appropriate recommendations to the Owner to adjust the Project's size, quality or budget for the Cost of the Work, and the Owner shall cooperate with the Architect in making such adjustments.

**§ 6.6** If the Owner's budget for the Cost of the Work at the conclusion of the Construction Documents Phase Services - is exceeded by the lowest bona fide bid or negotiated proposal, the Owner shall

- .1 give written approval of an increase in the budget for the Cost of the Work;
- .2 authorize rebidding or renegotiating of the Project within a reasonable time;
- .3 terminate in accordance with Section 9.5;
- .4 in consultation with the Architect, revise the Project program, scope, or quality as required to reduce the Cost of the Work; or
- .5 implement any other mutually acceptable alternative.

**§ 6.7** If the Owner chooses to proceed under Section 6.6.4, the Architect, without additional compensation, shall modify the Construction Documents as necessary to comply with the Owner's budget for the Cost of the Work at the conclusion of the Construction Documents Phase Services, or the budget as adjusted under Section 6.6.1. The Architect's modification of the Construction Documents shall be the limit of the Architect's responsibility under this Article 6.

#### ARTICLE 7 COPYRIGHTS AND LICENSES

§ 7.1 The Architect and the Owner warrants warrants that in transmitting Instruments of Service, or any other information, the transmitting party the Architect is the copyright owner of such information or has permission from the copyright owner to transmit such information for its use on the Project. If the Owner and Architect intend Architect intend Architect intends to transmit Instruments of Service or any other information or documentation in digital form, they shall endeavor to establish necessary protocols governing such transmissions.

§ 7.2 The Architect and the Architect's consultants shall be deemed the authors and owners of their respective Instruments of Service, including the Drawings and Specifications, and shall retain all common law, statutory and other reserved rights, including copyrights. Submission or distribution of Instruments of Service to meet official regulatory requirements or for similar purposes in connection with the Project is not to be construed as publication in derogation of the reserved rights of the Architect and the Architect's consultants.Provided that the Owner is not in default under this Agreement, all studies, reports, drawings, specifications, and other documents, whether in hardcopy or electronic form, prepared by Architect as part of the Services shall be deemed property of Owner.

<u>.1</u> The architect shall remove from the electronic versions of the documents the title block, Professional seals, Professional stamp and signatures on all electronic versions of the Construction Documents provided to Owner as Owner's property;

.2 The Architect shall maintain rights to use limited images for marketing and promotional purposes.

**§ 7.3** Upon execution of this Agreement, the Architect grants to the Owner a nonexclusive license to use the Architect's Instruments of Service solely and exclusively for purposes of constructing, using, maintaining, altering and adding to the Project, provided that the Owner substantially performs its obligations, including prompt payment of all sums when due, under this Agreement. The Architect shall obtain similar nonexclusive licenses from the Architect's consultants consistent with this Agreement. The license granted under this section permits the Owner to authorize the Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers, as well as the Owner's consultants and separate contractors, to reproduce applicable portions of the Instruments of Service solely and exclusively for use in performing services or construction for the Project. If the Architect rightfully terminates this Agreement for cause as provided in Section 9.4, the license granted in this Section 7.3 shall terminate.

**§ 7.3.1** In the event the Owner uses the Instruments of Service without retaining the author of the Instruments of Service, the Owner releases the Architect and Architect's consultant(s) from all claims and causes of action arising from such uses. The Owner, to the extent permitted by law, further agrees to indemnify and hold harmless the

AlA Document B101<sup>™</sup> – 2007 (formerly B151<sup>™</sup> – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AlA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AlA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AlA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for

Architect and its consultants from all costs and expenses, including the cost of defense, related to claims and causes of action asserted by any third person or entity to the extent such costs and expenses arise from the Owner's use of the Instruments of Service under this Section 7.3.1. The terms of this Section 7.3.1 shall not apply if the Owner rightfully terminates this Agreement for cause under Section 9.4.

**§ 7.4** Except for the licenses granted in this Article 7, no other license or right shall be deemed granted or implied under this Agreement. The Owner shall not assign, delegate, sublicense, pledge or otherwise transfer any license granted herein to another party without the prior written agreement of the Architect. Any unauthorized use of the Instruments of Service shall be at the Owner's sole risk and without liability to the Architect and the Architect's consultants.

ARTICLE 8 CLAIMS AND DISPUTES § 8.1 GENERAL

**§ 8.1.1** The Owner and Architect shall commence all claims and causes of action, whether in contract, tort, or otherwise, against the other arising out of or related to this Agreement in accordance with the requirements of the method of binding dispute resolution selected in this Agreement within the period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Architect waive all claims and causes of action not commenced in accordance with this Section 8.1.1.

**§ 8.1.2** To the extent damages are covered by property insurance, the Owner and Architect waive all rights against each other and against the contractors, consultants, agents and employees of the other for damages, except such rights -as they may have to the proceeds of such insurance as set forth in AIA Document A201–2007, General Conditions of -the Contract for Construction. The Owner or the Architect, as appropriate, shall require of the contractors, consultants, agents and employees in favor of the other parties enumerated herein.

**§ 8.1.3** The Architect and Owner waive consequential damages for claims, disputes or other matters in question arising out of or relating to this Agreement. This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination of this Agreement, except as specifically provided in Section 9.7.

#### § 8.2 MEDIATION

**§ 8.2.1** Any claim, dispute or other matter in question arising out of or related to this Agreement shall be subject to mediation as a condition precedent to binding dispute resolution. If such matter relates to or is the subject of a lien arising out of the Architect's services, the Architect may proceed in accordance with applicable law to comply with the lien notice or filing deadlines prior to resolution of the matter by mediation or by binding dispute resolution.

**§ 8.2.2** The Owner and Architect shall-may endeavor to resolve claims, disputes and other matters in question between them by mediation which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Mediation Procedures in effect on the date of the Agreement. mediation. A request for mediation shall be made in writing, delivered to the other party to the Agreement, and filed with the person or entity administering the mediation. The request may be made concurrently with the filing of a complaint or other appropriate demand for binding dispute resolution but, in such event, mediation shall proceed in advance of binding dispute resolution proceedings, which shall be stayed pending mediation for a period of 60 days from the date of filing, unless stayed for a longer period by agreement of the parties or court order. If an arbitration proceeding is stayed pursuant to this section, the parties may nonetheless proceed to the selection of the arbitrator(s) and agree upon a schedule for later proceedings.

§ 8.2.3 The parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction thereof.

AlA Document B101<sup>TM</sup> – 2007 (formerly B151<sup>TM</sup> – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AlA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AlA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AlA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for resale.

**§ 8.2.4** If the parties do not resolve a dispute through mediation pursuant to this Section 8.2, the method of binding dispute resolution shall be the following:

(Check the appropriate box. If the Owner and Architect do not select a method of binding dispute resolution below, or do not subsequently agree in writing to a binding dispute resolution method other than litigation, the dispute will be resolved in a court of competent jurisdiction.)

[-] Arbitration pursuant to Section 8.3 of this Agreement

-] Litigation in a court of competent jurisdiction

-] Other (Specify)

#### § 8.3 ARBITRATION

§ 8.3.1 If the parties have selected arbitration as the method for binding dispute resolution in this Agreement, any claim, dispute or other matter in question arising out of or related to this Agreement subject to, but not resolved by, mediation shall be subject to arbitration which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Arbitration Rules in effect on the date of this Agreement. A demand for arbitration shall be made in writing, delivered to the other party to this Agreement, and filed with the person or entity administering the arbitration.

At the Agreement of the Owner and Architect, all claims, disputes and other matters in question between any of the Architect, Owner, Contractor, Subcontractor or any material supplier arising out of, or relating to, agreements to which two or more of said parties are bound, or the Contract Documents or the breach thereof, except with respect to the Architect's decisions on matters relating to aesthetic effect, shall be decided by arbitration in accordance with the Construction Industry Arbitration Rules of the American Arbitration Association then obtaining at the time of filing. as modified herein, unless the parties mutually agree otherwise. In any such arbitration, the arbitrator shall make separate findings as to liability and the amount of damages with respect to each party to the arbitration to the extent any liability or responsibility for damages exists. All parties which have an interest in the dispute shall be joined as parties to the arbitration. The Architect's sub-consultant agreements shall require such joinder. The arbitrator shall have authority to decide all issues between the parties including but not limited to procedural matters, claims for extras, delay and liquidated damages, matters involving defects in the Work, right to payment, whether matters decided by the Architect involve aesthetic effect and whether the necessary procedures for arbitration have been followed. The foregoing agreement to arbitrate and any other agreement to arbitrate with an additional person or persons, duly consented to by the parties, shall be specifically enforceable under the prevailing arbitration law. The award rendered by the arbitrator shall be final, and judgment may be entered upon it in accordance with applicable law in any court having jurisdiction thereof.

**§ 8.3.1.1** A demand for arbitration shall be made no earlier than concurrently with the filing of a request for mediation, but in no event shall it be made after the date when the institution of legal or equitable proceedings based on the claim, dispute or other matter in question would be barred by the applicable statute of limitations. For statute of limitations purposes, receipt of a written demand for arbitration by the person or entity administering the arbitration shall constitute the institution of legal or equitable proceedings based on the claim, dispute or other matter in question.

§ 8.3.2 The foregoing agreement to arbitrate and other agreements to arbitrate with an additional person or entity duly consented to by parties to this Agreement shall be specifically enforceable in accordance with applicable law in any court having jurisdiction thereof demand for arbitration shall be made within a reasonable time after the claim, dispute or other matter in question has arisen, and in no event shall it be made after the date when institution of legal or equitable proceedings based on such claim, dispute or other matter in question would be barred by an applicable statute of limitations.

§ 8.3.3 The award rendered by the arbitrator(s) shall be final, and judgment may be entered upon it in accordance with -applicable law in any court having jurisdiction thereof. All parties shall carry on their respective obligations, and perform their duties during any arbitration proceedings.

§ 8.3.4 <u>CONSOLIDATION OR JOINDERIn addition to the other rules of the American Arbitration Association</u> applicable to any arbitration hereunder, the following shall apply:

AlA Document B101<sup>TM</sup> – 2007 (formerly B151<sup>TM</sup> – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for resale.

(a) Promptly upon the empanelling of the arbitrator, the arbitrator shall direct all parties to set forth in writing and to serve upon each other party a detailed statement of its contentions of fact and law.

(b) All parties to the arbitration shall be entitled to reasonable discovery procedures to be established by the arbitrator.

(c) The arbitration shall be commenced and conducted as expeditiously as possible consistent with affording reasonable discovery as provided herein. Similarly, the scope of discovery, and the extent of proceedings hereunder relating to discovery, shall be consistent with the parties' intent that the arbitration be conducted as expeditiously as possible.

**§ 8.3.4.1** Either party, at its sole discretion, may consolidate an arbitration conducted under this Agreement with any other arbitration to which it is a party provided that (1) the arbitration agreement governing the other arbitration permits consolidation; (2) the arbitrations to be consolidated substantially involve common questions of law or fact; and (3) the arbitrations employ materially similar procedural rules and methods for selecting arbitrator(s).

**§ 8.3.4.2** Either party, at its sole discretion, may include by joinder persons or entities substantially involved in a common question of law or fact whose presence is required if complete relief is to be accorded in arbitration, provided -that the party sought to be joined consents in writing to such joinder. Consent to arbitration involving an additional person or entity shall not constitute consent to arbitration of any claim, dispute or other matter in question not described in the written consent.

**§ 8.3.4.3** The Owner and Architect grant to any person or entity made a party to an arbitration conducted under this Section 8.3, whether by joinder or consolidation, the same rights of joinder and consolidation as the Owner and Architect under this Agreement.

**§8.3.5** In the event of any litigation or arbitration between the parties hereunder, all reasonable attorneys' fees and other costs incurred shall be awarded to the prevailing party.

#### ARTICLE 9 TERMINATION OR SUSPENSION

§ 9.1 If the Owner fails to make payments to the Architect in accordance with this Agreement, such failure shall be considered substantial nonperformance and cause for termination or, at the Architect's option, cause for suspension of performance of services under this Agreement. If the Architect elects to suspend services, the Architect shall give seven days' written notice to the Owner before suspending services. In the event of a suspension of services, the Architect shall have no liability to the Owner for delay or damage caused the Owner because of such suspension of services. Before resuming services, the Architect shall be paid all sums due prior to suspension and any expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.

§ 9.2 If the Owner suspends the Project, the Architect shall be compensated for services performed prior to notice of such suspension. When If the Owner's suspension was for convenience rather than for cause, when the Project is resumed, the Architect shall be compensated for expenses incurred in the interruption and resumption of the Architect's services. The Architect's fees for the remaining services and the time schedules shall be equitably adjusted.

§ 9.3 If the Owner suspends the Project for more than 90 cumulative days for reasons other than the fault of the Architect, the Architect may terminate this Agreement by giving not less than seven days' written notice.

§ 9.4 Either party may terminate this Agreement upon not less than seven days' written notice should the other party fail substantially to perform in accordance with the terms of this Agreement through no fault of the party initiating the termination.

§ 9.5 The Owner may terminate this Agreement upon not less than seven days' written notice to the Architect for the Owner's convenience and without cause.

AlA Document B101<sup>TM</sup> – 2007 (formerly B151<sup>TM</sup> – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AlA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AlA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AlA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for resale.

§ 9.6 In the event of termination not the fault of the Architect, the Architect shall be compensated for services performed prior to termination, together with Reimbursable Expenses then due and all Termination Expenses as defined in Section 9.7. due.

**§ 9.7** Termination Expenses are in addition to compensation for the Architect's services and include expenses directly -attributable to termination for which the Architect is not otherwise compensated, plus an amount for the Architect's anticipated profit on the value of the services not performed by the Architect.

**§ 9.8** The Owner's rights to use the Architect's Instruments of Service in the event of a termination of this Agreement - are set forth in Article 7 and Section 11.9.

#### ARTICLE 10 MISCELLANEOUS PROVISIONS

§ 10.1 This Agreement shall be governed by the law of the place where the Project is located, except that if the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section -8.3. State of Illinois. Exclusive Venue for resolution of all disputes shall lie in Cook County, Illinois.

§ 10.2 Terms in this Agreement shall have the same meaning as those in AIA Document A201–2007, General Conditions of the Contract for Construction. Construction, as amended and included in the Construction Documents.

§ 10.3 The Owner and Architect, respectively, bind themselves, their agents, successors, assigns and legal representatives to this Agreement. Neither the Owner nor the Architect shall assign this Agreement without the written consent of the other, except that the Owner may assign this Agreement to a lender providing financing for the Project if the lender agrees to assume the Owner's rights and obligations under this Agreement.

§ 10.4 If the Owner requests the Architect to execute certificates, the proposed language of such certificates shall be submitted to the Architect for review at least 14 days prior to the requested dates of execution. If the Owner requests the Architect to execute consents reasonably required to facilitate assignment to a lender, the Architect shall execute all such consents that are consistent with this Agreement, provided the proposed consent is submitted to the Architect for review at least 14 days prior to execution. The Architect shall not be required to execute certificates or consents that would require knowledge, services or responsibilities beyond the scope of this Agreement.

§ 10.5 Nothing contained in this Agreement shall create a contractual relationship with or a cause of action in favor of a third party against either the Owner or Architect. There are no third party beneficiaries to this Agreement.

§ 10.6 Unless otherwise required in this Agreement, the Architect shall have no responsibility for the discovery, presence, handling, removal or disposal of, or exposure of persons to, hazardous materials or toxic substances in any form at the Project site.

§ 10.7 The Architect shall have the right to include photographic or artistic representations of the design of the Project among the Architect's promotional and professional materials. The Architect shall be given reasonable access to the completed Project to make such representations. However, the Architect's materials shall not include the Owner's confidential or proprietary information if the Owner has previously advised the Architect in writing of the specific information considered by the Owner to be confidential or proprietary. The Owner shall provide professional credit for the Architect in the Owner's promotional materials for the Project. proprietary.

§ 10.8 If the Architect or Owner receives information specifically designated by the other party as "confidential" or "business proprietary," the receiving party shall keep such information strictly confidential and shall not disclose it to any other person except to (1) its employees, (2) those who need to know the content of such information in order to perform services or construction solely and exclusively for the Project, or (3) its consultants and contractors whose contracts include similar restrictions on the use of confidential information.

§ 10.9 The Architect shall submit valid certificates of insurance and, if requested, policies, in form and substance satisfactory to Owner evidencing the effectiveness of the insurance policies along with original copies of the amendatory riders to any such policies to Owner for Owner's approval before Architect commences the performance of any services hereunder.

AIA Document B101<sup>TM</sup> – 2007 (formerly B151<sup>TM</sup> – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for regard

§ 10.10 The Architect hereby agrees to maintain the insurance described in Article 2 herein during the duration the performance of all Services. If the Architect fails to furnish and maintain the specified insurance, the Owner may terminate the Agreement.

§ 10.11 Architect shall include the Owner as additional insured by causing amendatory riders or endorsements to be attached to the Commercial General Liability insurance policies described in this Agreement. The insurance coverage afforded under the Commercial General Liability policies shall be primary to any insurance carried independently by the Indemnitees (as identified in Section 10.12 below). Said amendatory riders or endorsements shall indicate that as respects the Indemnitees, there shall be severability of interests under said insurance policies for coverages provided under said insurance policies.

§ 10.12 To the fullest extent permitted by law, the Architect hereby agrees to indemnify and hold the Owner, its directors, members, officers, agents, and employees (collectively the "Indemnitees") harmless from all losses, claims, liabilities, injuries, damages and expenses, including reasonable attorneys' fees, that the Indemnitees may incur to the extent arising out of the negligent performance or breach of performance by the Architect of its duties and obligations under or pursuant to this Agreement.

§ 10.13 Time is of the essence of this Agreement.

§ 10.14 Architect understands and acknowledges that its work, in whole or in part, will be performed on public school property where there may be direct, daily contact with school students. The Architect further understands and acknowledges that the State of Illinois requires that all employees of vendors, licensees, contractors or others having direct, daily contact with students are subject to a criminal background check and may not be listed on the State Sex Offender Registry. Prior to allowing any of its employees who will be performing the scope of work access to school property, the Architect agrees to provide the District with the following:

- Evidence that each employee, agent, contractor or other person performing work on school (1).property under this Agreement was subjected to a criminal background check in conformity with 105 ILCS 5/10-21.9; that said persons are not listed on said Registry; and said persons have no criminal convictions for the offenses listed under 105 ILCS 5/10-21.9;
- The Architect will provide the District, upon request, a copy of the criminal background (2).check conducted on each such person.

In the event the Architect plans to subcontract with or use the services of another person or firm that may have direct, daily contact with students on school property, in order to fulfill its obligations under its Agreement with the District then in that event Architect will require all such persons or firms to comply with the provisions of this paragraph and 105 ILCS 5/10-21.9.

In the event the Architect fails to comply with the provisions of this paragraph and 105 ILCS 5/10-21.9, and as a result a suit or claim is instituted by a student for harm caused by an employee of the Architect, or caused by an employee of a subcontractor to the Architect, then in that event the Architect agrees to fully defend and indemnify, including reimbursement of attorney's fees and costs, the District against any such claims.

#### ARTICLE 11 COMPENSATION

§ 11.1 For the Architect's Basic Services described under Article 3, the Owner shall compensate the Architect as follows:

(Insert amount of, or basis for, compensation.)

% of actual construction cost for construction related services as defined in Section 6.1 based on the Fees Schedule attached as Exhibit E which is incorporated into this Agreement. The Architect Fee shall be converted to a lump sum fee upon approval of the Construction Manager's Final Cost Estimate. The Architect Fees for Alternates shall be as set forth in Exhibit E.

Init.

<sup>§ 11.2</sup> For Additional Services designated in Section 4.1, the Owner shall compensate the Architect as follows:

AIA Document B101<sup>TM</sup> – 2007 (formerly B151<sup>TM</sup> – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved, WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under 19 the law. This document was produced by AIA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for resale. User Notes:

(Insert amount of, or basis for, compensation. If necessary, list specific services to which particular methods of compensation apply.)

§ 11.3 For <u>all</u> Additional Services that may arise during the course of the Project, including those under Section 4.3, the Owner shall compensate the Architect as follows: *(Insert amount of, or basis for, compensation.)* 

Compensation shall be established by mutual agreement in writing between the Owner and Architect prior to beginning any additional Services.

**§ 11.4** Compensation for Additional Services of the Architect's consultants when not included in Section 11.2 or 11.3, shall be the amount invoiced to the Architect plus percent (-%), or as otherwise stated below:

§ 11.5 Where compensation for Basic Services is based on a stipulated sum or percentage of the Cost of the Work, the compensation for each phase of services shall be as follows:

Schematic Design Phase Design Development Phase Construction Documents	fifteen twenty forty	percent ( percent ( percent (	$\frac{15}{20}$ $\frac{40}{40}$	%) %) %)
Phase Bidding or Negotiation Phase Construction Phase	<u>five</u> twenty	percent ( percent (	<u>5</u> <u>20</u>	%) %)
Total Basic Compensation	one hundred	percent (	100	%)

§ 11.6 When compensation is based on a percentage of the Cost of the Work and any portions of the Project are deleted or otherwise not constructed, compensation for those portions of the Project shall be payable to the extent services are performed on those portions, in accordance with the schedule set forth in Section 11.5 based on (1) the lowest bona fide bid or negotiated proposal, or (2) if no such bid or proposal is received, the most recent estimate of the Cost of the Work for such portions of the Project. The Architect shall be entitled to compensation in accordance with this Agreement for all services performed whether or not the Construction Phase is commenced.

§ 11.7 The hourly billing rates for services of the Architect and the Architect's consultants, if any, are set forth below. The rates shall be adjusted in accordance with the Architect's and Architect's consultants' normal review practices. *(If applicable, attach an exhibit of hourly billing rates or insert them below.)* 

Hourly billing rates for Architectural Services for the duration of this Agreement shall be in accordance with Architects' billing rates indicated on EXHIBIT D.

#### **Employee or Category**

Rate

#### § 11.8 COMPENSATION FOR REIMBURSABLE EXPENSES

§ 11.8.1 Reimbursable Expenses are in addition to compensation for Basic and Additional Services and include expenses incurred by the Architect and the Architect's consultants directly related to the Project, as follows:

- .1 Transportation and authorized out-of-town travel and subsistence;
  - .2 Long distance services, dedicated data and communication services, teleconferences, Project Web sites, and extranets;
  - .3 Fees paid for securing approval of authorities having jurisdiction over the Project;
  - .4 Printing, reproductions, plots, standard form documents;
  - .5 Postage, handling and delivery;
  - .6 Expense of overtime work requiring higher than regular rates, if authorized in advance by the Owner;

- .7 Renderings, models, mock-ups, professional photography, and presentation materials requested by the Owner;
- .8 Architect's Consultant's expense of professional liability insurance dedicated exclusively to this Project, or the expense of additional insurance coverage or limits if the Owner requests such insurance in excess of that normally carried by the <u>Architect and</u> Architect's consultants;
- .9 All taxes levied on professional services and on reimbursable expenses;
- .10 Site office expenses; and
- .11 Other similar Project-related expenditures.expenditures authorized in advance by the Owner.

§ 11.8.2 For Reimbursable Expenses the compensation shall be the expenses incurred by the Architect and the Architect's consultants plus percent ( %) of the expenses incurred.zero ( 0.0 ) of the expenses incurred. The Architect shall be reimbursed at a rate of 1.15% of the actual expenses incurred by the Architect only for specialty consultants which the Owner approves in writing.

#### § 11.9 COMPENSATION FOR USE OF ARCHITECT'S INSTRUMENTS OF SERVICE

If the Owner terminates the Architect for its convenience under Section 9.5, or the Architect terminates this Agreement under Section 9.3, the Owner shall pay a licensing fee as compensation for the Owner's continued use of the Architect's Instruments of Service solely for purposes of completing, using and maintaining the Project as follows:

#### § 11.10 PAYMENTS TO THE ARCHITECT

§ 11.10.1 An initial payment of  $\underline{\text{zero}}$  (\$ 0) shall be made upon execution of this Agreement and is the minimum payment under this Agreement. It shall be credited to the Owner's account in the final invoice.

§ 11.10.2 Unless otherwise agreed, payments for services shall be made monthly in proportion to services performed. Payments are due and payable upon presentation of the Architect's invoice. Amounts unpaid () days after the invoice date shall bear interest at the rate entered below, or in the absence thereof at the legal rate prevailing from time to time at the principal place of business of the Architect.

(Insert rate of monthly or annual interest agreed upon.)

The Architect's invoice shall be submitted in a format acceptable to the Owner, and shall show sufficient information to allow the Owner to determine the propriety thereof. At a minimum, each invoice shall state services completed during the billing period, amount due for Services performed during the billing period, amount previously paid, and agreed contract balance remaining. The Architect shall be paid in the time periods set forth in the Local Government Prompt Payment Act.

%

**§** 11.10.3 The Owner shall not withhold amounts from the Architect's compensation to impose a penalty or liquidated damages on the Architect, or to offset sums requested by or paid to contractors for the cost of changes in the Work unless the Architect agrees or has been found liable for the amounts in a binding dispute resolution proceeding.

§ 11.10.4 Records of Reimbursable Expenses, expenses pertaining to Additional Services, and services performed on the basis of hourly rates shall be available to the Owner at mutually convenient times.

§ 11.10.5 Owner reserves the right to recover from the Architect all or a portion of the costs associated with change orders issued to correct errors or work omitted in the construction documents prepared by the Architect. Consequential damages including any delay damages suffered by Owner, or which Owner may be liable, due to errors and omissions may be included in the recovery. In the case of omissions, Architect shall not be responsible for the direct cost of the construction contractor's Work to implement omitted Work to the extent that Architect demonstrates that, but for the omission, the cost of the omitted Work would have been fully included in the construction contractor's bid. Nothing in this paragraph (11.10.5) is intended to provide Owner or Architect with causes of action, defenses or measures of damages that it would not otherwise posses under applicable law.

Init.

AlA Document B101<sup>TM</sup> – 2007 (formerly B151<sup>TM</sup> – 1997). Copyright © 1974, 1978, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for

#### ARTICLE 12 SPECIAL TERMS AND CONDITIONS

Special terms and conditions that modify this Agreement are as follows:

None.

#### ARTICLE 13 SCOPE OF THE AGREEMENT

§ 13.1 This Agreement represents the entire and integrated agreement between the Owner and the Architect and supersedes all prior negotiations, representations or agreements, either written or oral. This Agreement may be amended only by written instrument signed by both Owner and Architect.

§ 13.2 This Agreement is comprised of the following documents listed below:

AIA Document B101<sup>™</sup>-2007, Standard Form Agreement Between Owner and Architect .1

AIA Document E201<sup>™</sup> 2007, Digital Data Protocol Exhibit, if completed, or the following: 2

Other documents: (List other documents, if any, including Exhibit A, Initial Information, and additional scopes of service, if any, forming part of the Agreement.)

This Agreement entered into as of the day and year first written above.

#### OWNER

#### ARCHITECT

(Signature)

(Printed name and title)

(Signature)

(Printed name and title)

#### Certification of Document's Authenticity

AIA<sup>®</sup> Document D401<sup>™</sup> – 2003

I, Kenneth M. Florey, hereby certify, to the best of my knowledge, information and belief, that I created the attached final document simultaneously with this certification at 16:31:20 on 02/11/2011 under Order No. 3032139990\_1 from AIA Contract Documents software and that in preparing the attached final document I made no changes to the original text of AIA<sup>®</sup> Document B101<sup>™</sup> - 2007, Standard Form of Agreement Between Owner and Architect, as published by the AIA in its software, other than changes shown in the attached final document by underscoring added text and striking over deleted text.

(Signed)

(Title)

(Dated)

AIA Document D401<sup>TM</sup> – 2003. Copyright © 1992 and 2003 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AIA software at 16:31:20 on 02/11/2011 under Order No.3032139990\_1 which expires on 08/16/2011, and is not for resale. **User Notes:** 

### 

#### Amendment to the Professional Services Agreement

Exhibit "A"

1

TO:

(Owner or Owner's Representative)

In accordance with the Agreement dated: <[ Services Agreement.

<DATE> described a

described as AIA Document B101-2007, Blanket

3

**BETWEEN** the Owner: (Name and address)

and the Architect: (Name and address) Legat Architect's, Inc. 2015 Spring Road, Suite 175 Oak Brook, IL 60523

for the Project: (Name and address) T.B.D. Summer 2011 Renovation

> Authorization is requested to proceed with Additional Services. to incur additional Reimbursable Expenses.

As Follows:

The following adjustments shall be made to compensation and time. (Insert provisions in accordance with the Agreement, or as otherwise agreed by the parties.)

Compensation: 8,5% of Costal Construction

Time:

Start of Project 6-8-11 Substantial Completion 8-13-11 Final Completion 9-13-11

AlA Document  $G802^{7\mu} - 2007$  (formerly  $G606^{7\mu} - 2000$ ). Copyright © 2000 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AlA<sup>6</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AlA<sup>6</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This document was produced by AlA software at 16:05:14 on 01/13/2009 under Order No.1000343289\_ which expires on 2/15/2009, and is not for resale. User Notes:

#### SUBMITTED BY:

(Signatur			1
Dr.	Dietra	Δ.	Millard
(Printed n	ame and title)	Pres	ident Board
(Date)		of	Education

AGREED TO:

(Signature)

Wayne Machnich, Lesat (Printed name and title)

(Date)

Altest: Secretary

Pate:

AlA Document G802<sup>TM</sup> – 2007 (formerly G606<sup>TM</sup> – 2000), Copyright © 2000 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and international Treatics. Unauthorized reproduction or distribution of this AIA<sup>®</sup> Document, or any portion of il, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the faw. This document was produced by AIA software at 16:05:14 on 01/13/2009 under Order No.1000343289\_ which expires on 2/15/2009, and is not for resale. User Notes: (2133476998)

#### SUBMITTED BY:

(Signatur			1
Dr.	Dietra	Δ.	Millard
(Printed n	ame and title)	Pres	ident Board
(Date)		of	Education

AGREED TO:

(Signature)

Wayne Machnich, Lesat (Printed name and title)

(Date)

A Hest: Secretary

Pate:

AlA Document G802<sup>TM</sup> – 2007 (formerly G606<sup>TM</sup> – 2000), Copyright © 2000 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and international Treatics. Unauthorized reproduction or distribution of this AIA<sup>®</sup> Document, or any portion of il, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the faw. This document was produced by AIA software at 16:05:14 on 01/13/2009 under Order No.1000343289\_ which expires on 2/15/2009, and is not for resale. User Notes: (2133476998)

# Oak Park and River Forest High School Construction Projects Long Range Plan

Construction Projects Long Range Plan Revised 12/14/10 rz (start of construction phase 1 - 6/8/11) ( Substantial completion -8/13/11)

		Comments and Clarifications	Comments and Clarifications Pricing reflects replacement in kind for press box. No allowance included for ticket booths or fencine. No allowance	Comments and Clarifications Pricing reflects replacement in kind for press box. No allowance included for ticket booths or fencing. No elevator included.	Comments and Clarifications Pricing reflects replacement in kind for press box. No allowance included for ticket booths or fencing. No elevator included. Pricing based on quote received by OPRF. Does not include rooms 173/174 which were completed in 2010. Assumes 25% floor fill; 9,000 sf tile. 184 and 212 were eliminated	Comments and Clarifications Pricing reflects replacement in kind for press box. No allowance included for ticket booths or fencing. No elevator included. Pricing based on quote received by OPRF. Pricing based on quote received by OPRF.	and Clarifications hind for press box. No allowance encing. No elevator included. ed by OPRF. Zd which were completed in 2010. if file. 184 and 212 were eliminated to future plans ns: recessed indirect lights in d cost for additional rooms.Murals stimate.	Comments and Clarifications Pricing reflects replacement in kind for press box. No allowance included for ticket booths or fencing. No elevator included. Pricing based on quote received by OPRF. Des not include rooms 173/174 which were completed in 2010. Des not include rooms; recessed indirect lights in correlite pendants in classrooms; recessed indirect lights in on walls in 475.	Comments and Clarifications Pricing reflects replacement in kind for press box. No allowance included for ticket booths or fencing. No elevator included. Pricing based on quote received by OPRF. Pricing based on quote received by OPRF. Pricing Sased on quote received by Antich were completed in 2010. Assumes 25% floor fill; 9,000 st file. 184 and 212 were eliminated from this versi life safety due to future plans from this versi life safety due to future plans Corelite pendants in classrooms; recessed indirect lights in corridors; new diffusers. Added cost for additional rooms. Murals on walls in 475 Need scope description and estimate. Corelite pendants in classrooms; recessed indirect lights in corridors; new diffusers. Added cost for additional rooms. Save existing cabinets in classrooms; recessed indirect lights in corridors; new diffusers. Added cost for additional rooms. Save existing cabinets in classrooms; recessed indirect lights in corridors; new diffusers.	Comments and Clarifications Pricing reflects replacement in kind for press box. No allowance included for ticket booths or fencing. No elevator included. Pricing based on quote received by OPRF. Does not include rooms 173/174 which were completed in 2010. Assumes 25% floor fill; 9,000 st file. 184 and 212 were eliminated from this vers life safety due to future plans from this vers life safety due to future plans corridors; new diffusers. Added cost for additional rooms. Murals on walls in 475 Need scope description and estimate. Correlite pendants in classrooms; recessed indirect lights in corridors; new diffusers. Added cost for additional rooms. Save existing cabinets in classrooms; recessed indirect lights in corridors; new diffusers. Added cost for additional rooms. Save existing cabinets in classrooms; recessed indirect lights in corridors; new diffusers. Need scope description and estimate.	and Clarifications hind for press box. No allowance encing. No elevator included. .74 which were completed in 2010. if tile. 184 and 212 were eliminated futue. 184 and 212 were eliminated file. 184 and 212 were eliminated file. 184 and 212 were alminated ans. recessed indirect lights in d cost for additional rooms. Murals stimate. It that were on life safety to a cost for additional rooms. Save 11 that were on life safety to so for strust in budget parameters to only based on actual field camera	and Clarifications and Clarifications encing. No elevator included. ed by OPRF. 7.4 which were completed in 2010. 7.4 which were completed in 2010. 7.4 which were completed in 2010. 7.5 which were completed in 2010. 7.6 for additional rooms. Murals as: recessed indirect lights in d cost for additional rooms. Murals stimate. 11 that were on life safety to a d cost for additional rooms. Save 11 that were on life safety to a d cost for additional rooms. Save 11 that were on life safety to a cost for additional rooms. Save 11 that were on life safety to based on actual field camera based on actual field camera	and Clarifications and Clarifications encing. No elevator included. .74 which were completed in 2010. .74 which were eliminated to future pland streassed indirect lights in d cost for additional rooms. Murals stimate. . recessed indirect lights in d cost for additional rooms. Save 11 that were on life safety to a cost for additional rooms. Save 11 that were on life safety to cost for additional rooms. Save 11 that were on life safety to a cost for additional rooms. Save 11 that were on life safety to so only. based on actual field camera	and Clarifications and Clarifications encing. No elevator included. encing. No elevator included. 7.34 which were completed in 2010. fifle. 184 and 212 were eliminated to future plans fittile. 184 and 212 were eliminated to future plans. ars: recessed indirect lights in d cost for additional rooms. Murals stimate. ars: recessed indirect lights in ars: recessed indirect lights in ars: recessed indirect lights in ars: recessed indirect lights in based on actual field camera based on actual field camera based on actual field camera	and Clarifications and Clarifications encing. No elevator included. ad by OPRF. 	and Clarifications and Clarifications encing. No elevator included. et by OPRF. 74 which were completed in 2010. 74 which were completed in 2010. 75 which were completed in 2010. 75 which were completed in 2010. 75 recessed indirect lights in as recessed indirect lights in as recessed indirect lights in a cost for additional rooms. Murals stimate. 11 that were on life safety to a cost for additional rooms. Save 11 that were on life safety to a cost for additional rooms. Save 11 that were on life safety to a cost for additional rooms. Save 11 that were on life safety to based on actual field camera based on actual field camera	and Clarifications and Clarifications encing. No elevator included. encing. No elevator included. 7.34 which were completed in 2010. 1.34 which were completed in 2010. In the second plans to future plans to future plans to future plans at recessed indirect lights in d cost for additional rooms. Murals stimate. Streessed indirect lights in a cost for additional rooms. Save at recessed indirect lights in a cost for additional rooms. Murals stimate. a cost for additional rooms. Save d cost for additional rooms. Sa	and Clarifications and Clarifications encing. No elevator included. A which were completed in 2010. Ta which were completed in 2010. In the set and 212 were eliminated to future file. 138 and 212 were eliminated to future file. 138 and 212 were eliminated the future file. 138 and 212 were eliminated the future file. 138 and 212 were eliminated the set of for additional rooms. Murals stimate. It that were on life safety to a cost for additional rooms. Save 11 that were on life safety to 20 were seed indirect lights in a cost for additional rooms. Save 11 that were on life safety to 3 were a seed indirect lights in a cost for additional fold camera based on actual field camera based on actual field camera	and Clarifications and Clarifications encing. No elevator included. Ed by OPRF. TA which were completed in 2010. TA which were completed in 2010. This	and Clarifications and Clarifications bencing. No elevator included. encing. No elevator included. .74 which were completed in 2010. .74 which were completed in 2010. .75 recessed indirect lights in d cost for additional rooms. Murals stimate. .75 recessed indirect lights in ms; recessed indirect lights in cost for additional rooms. Murals timate. .75 recessed indirect lights in ms; recessed indirect lights in ms; recessed indirect lights in so only .0.	and Clarifications and Clarifications encing. No elevator included. Ta which were completed in 2010. Ta which were completed in 2010. This 136 and 212 were eliminated to future plans the future plans the cost for additional rooms. Murals stimate. That were on life safety to a cost for additional rooms. Save 11 that were on life safety to 20 cost for additional rooms. Save 11 that were on life safety to 20 cost for additional rooms. Save 21 that were on life safety to 20 cost for additional rooms. Save 21 that were on life safety to 20 cost for additional rooms. Save 21 that were on life safety to 20 cost for additional rooms. Save 21 that were on life safety to 20 cost for additional rooms. Save 21 that were on life safety to 20 cost for additional rooms. Save 21 that were on life safety to 20 cost for additional rooms. Save 21 that were on life safety to 20 cost for additional rooms. Save 21 that were on life safety to 21 that were safety to 21 that were safety to 21 that were safety to 21 that were safety to 21 that were safety to 21 that were safety to 21 that
	Comments a					Need scope description and estimate.				Drains at loading dock; pricing based on actual field camera results.		Pricing based on bids from 2010.	651,000.00 Includes 41,000 SF								
	FY2011/2012	\$ 60,000.00	\$ 84,000.00		\$ 444,629.00	\$ 177,000.00	\$ 225,000.00	\$ 72,000.00	\$ 216,000.00	\$ 48,000.00	\$ 188,000.00	\$ 847,000.00	\$ 651,000.00	\$ 3,152,629.00	\$ 315,262.90	\$ 3,467,891.90	\$ 1,976,291.90	\$ 1,491,600.00		lotal	I otal
The second second second	Fund o & M. Life Safety, Tort, InHiathers	O&M	O&M	LS SI	SI	O&M	SI	LS .	LS LS	LS I	O&M	O&M	S								No. of Lot of Lot of Lot of Lot
and the second second second	Type of work maintenance, instructional, refurbly, replacement. Une Safety, Up grade, Project mainterment	Maintenance	Maintenance	Life Safety	Life Safety	Maintenance	Life Safety	Life Safety	Life Safety	Life Safety		Maintenance	Life Safety			No.					and the second second second
	Description	Press Box renovation, <del>Ticket Booths and Fencing</del>	Code upgrades Elevator new building - from 2010	LS item A-18 <del>373,174,484,212</del> ,273, 274, 340, <del>396</del> ,421 asbestos tile worn needs replaced. Replace with Fritztile.	LS ITEM A-10: All 4th floor rooms that need ceiling and light replacement. Locations: Rooms 407, 408, 409, 410, 411, 413, 415, 417, 421,425, 425, 425, 425, 425, 427, 428, 429, 430, 431, 433, 435, 437, 439, 430, 472, 378, 481, 482	New HVAC to rooms: 419A, 420, 421, 423, 424, 425, 425A,	rooms that need ceiling and light replacement. 311, 313, 315, 320, 321, 326, 329, 330, 331, 333, 335,	LS ITEM A-12: All 2nd floor rooms that need ceiling and light replacement. Locations: Rooms-229, 230, 231, 233, 235, 239, <del>200, 201</del> .	LS ITEM S P.5.1, P.5.2, P.5.3, P.5.4 (phase 2): 1907-1962 building - replace domestic hot water piping - (mains that support 2012 bathroom work)	LS iTEM P-6: { <del>phase 2} replace sanitary drains @ 1907 - 1924 {phase 2} replace 3rd</del> <del>and 4th floor</del> } Replace 8° waste on DP2.1 and P2.1 excluded from 2010 work	Znd floor corridor Asbestos abatement to Fritztile. Location: Znd floor - East- West corridor in 1907 Building; North-South corridor from Room 229 - 239; North!Maintenance South corridor from Room 215 - Stairs.	Air handlers 4: #2,3; 4 and S3	LS item A-19 2nd floor hallways, 67 addition Ceilings and lights	Sub-total	Contingency (use 10%)	Total	Total LS	Total 0 & M			
	Year	48 2011-2012	2011-2012	52 2011-2012	53 2011-2012	2011-2012	54 2011-2012	55 2011-2012	2011-2012	2011-2012	58 2011-2012	61 2011-2012	63 2011-2012			2011-2012					
	Line #	48	2	52	ES	S3A	54	55	56-2011	57	58	61	63								

		IFICATE OF LIA		and the second se	A DESCRIPTION OF THE OWNER OF THE	01	MM/DD/YYYY) /07/2009
W. 17	RODUCER Phone: (847) 623-0456 Fax VEST'S INSURANCE AGENCY 733 W WASHINGTON STREET VAUKEGAN IL 60085	(847) 023-5600	ONLY HOLD	AND CONFERS	ISSUED AS A MATTER OF NO RIGHTS UPON THE CI FICATE DOES NOT AMEND E AFFORDED BY THE POL	ERTIFICAT	
			INSURERS AF	FORDING COV	ERAGE		NAIC #
				Incinnati Insura	ince Company		10677
24	EGAT ARCHITECTS, INC.		INSURER B:				
W	AUKEGAN IL 60085		INSURER D:				
L	01/201020	- A - 199 MAR AN - 199 - 199 - 199 - 199 - 199 - 199 - 199 - 199 - 199 - 199 - 199 - 199 - 199 - 199 - 199 - 19	INSURER E:			,	
THE ANY MAY	OVERAGES E POLICIES OF INSURANCE LISTED BI Y REQUIREMENT, TERM DR CONDITIO Y PERTAIN, THE INSURANCE AFFORD LICIES, AGGREGATE LIMITS SHOWN M	ED BY THE POLICIES DESCRIBED H	IEREIN IS SUBJECT TO				
INSR	ADD1 TYPE OF INSURANCE	POLICY NUMBER	POLICY EFFECTIVE	POLICY EXPIRATIO		S	
	GENERAL LIABILITY	EBP0019851	05/14/0B	05/14/09	EACH OCCURRENCE	\$	1,000,000
	CLAIMS MADE X 0	BILITY			DAMAGE TO RENTED PREMISES (Es occurance)	5	1,000,000
A		CCOR			MED, EXP (Any one person)	\$	10,000
					PERSONAL & ADV INJURY GENERAL AGGREGATE	\$	1,000,000
	GENTL AGGREGATE LIMIT APPLIE				PRODUCTS-COMP/OP AGG.	s	2,000,000
	AUTOMOBILE LIABILITY	EBP0019851	05/14/08	05/14/09	COMBINED SINGLE LIMIT (Es accident)	s	1,000,000
	ALL OWNED AUTOS SCHEDULED AUTOS				BODILY INJURY (Per person)	s	.,,
A	X HIRED AUTOS X NON-OWNED AUTOS				BODILY INJURY (Per accident)	5	
	GARAGE LIABILITY	- EXHIBI	T "B" - TWO PA	GES	PROPERTY DAMAGE (Par accidant)	5	
	ANY AUTO				AUTO ONLY - EA ACCIDENT OTHER THAN AUTO ONLY: AGG		
Τ	EXCESS / UMBRELLA LIABILITY	EBP0019851	05/14/08	05/14/09	EACH DCCURRENCE	s	5,000,000
		ADE			AGGREGATE	s	5,000,000
						\$	
	DEDUCTIBLE RETENTION \$					\$	
W	VORKERS COMPENSATION AND				WC STATU	5	
	EMPLOYERS' LIABILITY				E.L. EACH ACCIDENT	s	
01	OFFICER/WEMBER EXCLUDED?				E.L. DISEASE-EA EMPLOYEE	s	
SP	yss, describe under PECIAL PROVISIONS below				E.L. DISEASE-POLICY LIMIT	\$	
0	OTHER:						
ESC	CRIPTION OF OPERATIONS/LC	CATIONS/VEHICLES/EXCLU	SIONS ADDED BY B	NDORSEMENT	SPECIAL PROVISIONS		
	ect Number: 021005.WV				I LUNE PROVISIONS		
ERI	TIFICATE HOLDER		CANCELLA	the second se			
			WRITTEN NOTICI TO DO SO SHALL	E TO THE CERTIFI	CRIBED POLICIES BE CANCELL ISSUING INSURER WILL ENDEA CATE HOLDER NAMED TO THE MON OR LIABILITY OF ANY KIND	VOR TO M	AIL 30 DAYS
	ntion:		AUTHORIZED REP	RESENTATIVE	Gestphalt	W	A

© ACORD CORPORATION 1988

					01	E (MM/DD/YYY)				
PRODUCER (217)528-5679 I Lee/O'Keefe Insurance Agency	FAX (217)528-2121			SUED AS A MATTER OF RIGHTS UPON THE CE						
2501 Chatham Rd., Ste 100	y, Luc.	HOLDER	<b>2. THIS CERTIFIC</b>	ATE DOES NOT AMEND	, EXTI	END OR				
P.O. Box 13080		ALIER	HE COVERAGE	AFFORDED BY THE POI		BELOW.				
Springfield, IL 62791-3080		INSURER	AFFORDING CO	VERAGE	N	LAIC #				
MSURED Legat Architects, Inc.		INSURER A:	Accident Fun	d Insurance Co						
24 N. Chapel Street		INSURER B:								
Waukegan, IL 60085		INSURER C:	_/41.5_51.00" ***********************************							
		INSURER D:								
		INSURER E:								
COVERAGES										
THE POLICIES OF INSURANCE LISTED BEL ANY REQUIREMENT, TERM OR CONDITION MAY PERTAIN, THE INSURANCE AFFORDEL POLICIES, AGGREGATE LIMITS SHOWN MA	OF ANY CONTRACT OR OTHER D BY THE POLICIES DESCRIBED	DOCUMENT WITH HEREIN IS SUBJE	I RESPECT TO WHI	CH THIS CERTIFICATE MAY	BEISS	SUED OR				
TT INSUL TYPE OF INSURANCE	POLICY NUMBER	POLICY EFFECTIV	E POLICY EXPIRATIO		TS					
GENERAL LIABILITY				EACH OCCURRENCE	5					
COMMERCIAL GENERAL LIABILITY				DAMAGE TO RENTED	5					
CLAIMS MADE OCCUR			1	MED EXP (Any one person)	5					
				PERSONAL & ADV INJURY	\$					
				GENERAL AGGREGATE	5					
GENL AGGREGATE LIMIT APPLIES PER:				FRODUCTS - COMPIOP AGG	5					
POLICY PRO- JECT LOC										
AUTOMOBILE LIABILITY				COMBINED SINGLE LIMIT (Ex accident)	5					
ALL OWNED AUTOS SCHEDULED AUTOS				EODILY INJURY (Per person)	5					
HIRED AUTOS				BODILY INJURY (Per accident)	s					
				PROPERTY DAMAGE (Per accidani)	5					
GARAGE LIABILITY				AUTO ONLY - EA ACCIDENT	5					
OTUA YMA				OTHER THAN EA ACC AUTO ONLY: AGG	5					
EXCESS/UMBRELLA LIABILITY				EACH OCCURRENCE	s					
OCCUR CLAIMS MADE				AGGREGATE	5					
					5					
DEDUCTIBLE					5					
RETENTION S					5					
WORKERS COMPENSATION AND	WCV60446230	05/14/2008	05/14/2009	X WC STATU- OTH- TORY LIMITS FR						
EMPLOYERS' LIABILITY				EL EACH ACCIDENT	5	500,0				
ANY PROPRIETOR/PARTILER/EXECUTIVE OFFICER/MEMBER EXCLUDED?				E.L. DISEASE - EA EMPLOYEE	5	500,0				
If yes, describe under SPECIAL PROVISIONS below				E.L. DISEASE - POLICY LIMIT	\$	500,0				
OTHER										
CRIPTION OF OPERATIONS / LOCATIONS / VEHICLE	ES / EXCLUSIONS ADDED BY ENDORSE	MENT I SPECIAL PRO	I VISIONE	L						
	agar server and the solution of the									
RTIFICATE HOLDER		CANCELLATION SHOULD ANY OF THE ABOVE DESCRIBED POLICIES BE CANCELLED BEFORE THE								
		and the second se		SUNG INSURER WILL ENDEAV						
			10 DAYS WRITTEN NOTICE TO THE CERTIFICATE HOLDER NAMED TO THE LEFT, BUT FAILURE TO MAIL SUCH NOTICE SHALL IMPOSE NO OBLIGATION OR LIABILITY							
				TS AGENTS OR REPRESENTATI	VES.					
		AUTHORIZED REF								
		Glenda C.	Richards							

ACORD 25 (2001/08)

©ACORD CORPORATION 1988

# Schedule of Billable Rates

EXHIBIT "D" 1 of 1

ORGANIZATION Oak Park & River Forest High School District 200

-	
CLASSIFICATION	BASE RATE
Principal	\$ 200.00
Senior Interior Designer	\$ 200.00
LEED Consultant	\$ 200.00
Staff Architect	\$ 180.00
LEED Document Coordinator	\$ 135.00
Associate Architect	\$115.00
Intern Architect	\$ 75.00
Clerical	\$ 55.00

## **LEGAT**ARCHITECTS

January 5, 2011 Revised: 1/25/11; 2/11/11

EXHIBIT "E"

Ms. Cheryl Witham Chief Financial Officer Oak Park & River Forest High School District 200 201 North Scoville, Room 270a Oak Park, Illinois 60302

Re: Life Safety Projects and Capital Improvements Projects Revised Proposal to Provide Professional Architectural Services

#### Dear Ms. Witham:

Thank you again for giving Legal Architects the opportunity to become Architect of Record for Oak Park & River Forest High School District 200. Pursuant to your request, Legat Architects ("architect") is pleased to submit this proposal to provide professional architectural services to Oak Park & River Forest High School District 200 ("client"). The purpose of this letter is to summarize the professional services to be provided and the related fees.

#### 1.0 Projects:

- 1.1 Annual Life Safety and Capital Improvements Projects where the scope of work has yet to be determined.
- 1.2 Oak Park & River Forest High School, 201 North Scoville, Oak Park, Illinois 60302.

### 2.0 Project Parameters:

- 2.1 The budget, scope of work, affected building area(s), estimates of probable cost, and schedules will be determined annually by the Client in conjunction with the Architect and the Construction Manager.
- 2.2 Project Activities Summary:
  - 2.2.1 Work with the Client to prioritize the scope of work line items to match available funds; and
  - 2.2.2 Provide a visual assessment of the existing conditions associated with the scope of work line items identified and review the Construction Manager's estimate of probable cost for the scope of work line items identified; and
  - 2.2.3 Complete drawings and specifications suitable for bidding; and
  - 2.2.4 Assist the Construction Manager in project bidding and the selection of the most qualified bidder; and
  - 2.2.5 Perform construction observation; and
  - 2.2.6 Assist the Construction Manager to facilitate project closeout.

## **LEGAT**ARCHITECTS

January 5, 2011 Revised: 1/25/11; 2/11/11

EXHIBIT "E"

Ms. Cheryl Witham Chief Financial Officer Oak Park & River Forest High School District 200 201 North Scoville, Room 270a Oak Park, Illinois 60302

Re: Life Safety Projects and Capital Improvements Projects Revised Proposal to Provide Professional Architectural Services

#### Dear Ms. Witham:

Thank you again for giving Legal Architects the opportunity to become Architect of Record for Oak Park & River Forest High School District 200. Pursuant to your request, Legat Architects ("architect") is pleased to submit this proposal to provide professional architectural services to Oak Park & River Forest High School District 200 ("client"). The purpose of this letter is to summarize the professional services to be provided and the related fees.

#### 1.0 Projects:

- 1.1 Annual Life Safety and Capital Improvements Projects where the scope of work has yet to be determined.
- 1.2 Oak Park & River Forest High School, 201 North Scoville, Oak Park, Illinois 60302.

### 2.0 Project Parameters:

- 2.1 The budget, scope of work, affected building area(s), estimates of probable cost, and schedules will be determined annually by the Client in conjunction with the Architect and the Construction Manager.
- 2.2 Project Activities Summary:
  - 2.2.1 Work with the Client to prioritize the scope of work line items to match available funds; and
  - 2.2.2 Provide a visual assessment of the existing conditions associated with the scope of work line items identified and review the Construction Manager's estimate of probable cost for the scope of work line items identified; and
  - 2.2.3 Complete drawings and specifications suitable for bidding; and
  - 2.2.4 Assist the Construction Manager in project bidding and the selection of the most qualified bidder; and
  - 2.2.5 Perform construction observation; and
  - 2.2.6 Assist the Construction Manager to facilitate project closeout.

## LEGATARCHITECTS

sustainability performance design

EXHIBIT "E"

Ms. Cheryl Witham Oak Park & River Forest High School District 200 **Revised Proposal to Provide Professional Architectural Services** January 5, 2011 *Revised: 1/25/11; 2/11/11* Page 2 of 3

#### 3.0 Architect's Scope of Service:

3.1 Legat Architects proposes to complete the scope of professional architectural services in accordance with the basic services outlined in AIA Document B101, Standard Form of Agreement Between Owner and Architect, 2007 edition. In the event of any inconsistencies between this Exhibit E and the AIA Document B101-2007, the AIA Document B101-2007 shall control.

#### 4.0 Compensation:

4.1 The Client will ccompensate the Architect based on a percentage of the construction cost and will be based on the sliding scale identified below. The Architect's compensation will be converted to a lump sum at the approval of the Construction Manager's final cost estimate, or guaranteed maximum price, inclusive of allowances and construction contingencies exclusive of the Owner's project contingency.

CONSTRUC	CIT	NCOST	PERCENTAGE FEE
\$0			Time & Material
\$ 300,000	to	\$ 1,000,000	
\$ 2,000,000	to	\$ 4,000,000	

- 4.1.1 Compensation for Alternates included in the Contract(s) for Construction shall be billed at 100% of the total cost of the Alternates as identified in the Construction Manager's final approved cost estimate
- 4.1.2 Compensation for Alternates not included in the Contract(s) for Construction shall be billed at 80% of the total cost of the Alternates as identified in the Construction Manager's final approved cost estimate.
- 4.1.3 The Architect will not be compensated for work related to Change Orders except when the change significantly increases the scope of work at no fault of the Architect and requires design/documentation.
- 4.2 Reimbursable Expenses will be in addition to the Architect's compensation and shall be invoiced using the multipliers indicated below times the expenses incurred by Legat Architects.
  - 4.2.1 Reproduction costs for drawings, specifications, addenda, reports, etc. required to be submitted at the end of each contractual phase and for bidding purposes shall be invoiced at 1.10 times.
  - 4.2.2 Postage and delivery charges for bid documents and materials requested by the Client and Construction Manager or required by authorities having jurisdiction shall be invoiced at 1.10 times.
  - 4.2.3 Necessary consultants, including professional civil engineering services, as approved by the Client will be invoiced at 1.25 times.
  - 4.2.4 Specialty consultants as approved by the Client will be invoiced at 1.25 times.

## **LEGAT**ARCHITECTS

EXHIBIT "E"

Ms. Cheryl Witham Oak Park & River Forest High School District 200 Revised Proposal to Provide Professional Architectural Services January 5, 2011 *Revised: 1/25/11; 2/11/11* Page 3 of 3

### 5.0 Additional Services:

- 5.1 <u>Additional On-Site Observation</u>: The Architect will perform two additional weekly job-site observations while construction operations are in progress. The Client will compensate the Architect based on 0.5% of the construction cost as identified in the Contract(s) for Construction.
- 5.2 <u>Issued for Construction Documents</u>: The Architect will provide Issued for Construction Documents in .pdf format which combine Issued for Bidding Documents and addenda items. The Client will compensate the Architect on a time-and-material basis.
- 5.3 <u>Record (As-Built) Drawings:</u> The Architect will provide Record Drawings in .pdf format which combine Issued for Bidding Documents, addenda items, and field changes documented by the Trade Contractors. A Record (As-Built) Floor Plan will be provided to the Client in AutoCAD format. The Client will compensate the Architect on a time-and-material basis.

#### 6.0 Miscellaneous Provisions:

6.1 Unless otherwise provided in this Agreement, the Architect and Architect's consultants will have no responsibility for the discovery, presence, handling, removal or disposal of or exposure of persons to hazardous materials in any form at the Project site, including but not limited to asbestos, asbestos products, polychlorinated biphenyl (PCB) or other toxic substances.

The AIA Document B101, Standard Form of Agreement Between Owner and Architect, 2007 edition, is currently being prepared by Robbins Schwartz Nicholas Lifton & Taylor, Ltd., attorneys for Oak Park & River Forest High School District 200. It is our understanding, the terms have been agreed upon. However, if you have any additional questions regarding the agreement or this exhibit, please contact me at your earliest convenience.

Thank you.

Sincerely, Legal Architects, Inc.

Robert W. Wroble, AIA, LEED AP

ATTACHMENTS None.

cc None.

EC Patrick Brosnan, Legal Architects Berardo DeSimone, Legat Architects

## Oak Park and River Forest High School District 200 201 North Scoville Avenue • Oak Park, IL 60302-2296

TO:	Board of Education	
FROM:	Lauren M. Smith, Director of Human Resources	
DATE:	February 24, 2011	
RE:	Policies	ACTION

Following the January 27, 2010 Board of Education meeting, the following policy was sent to the Alumni Association, APPLE, Boosters, Citizens' Council, the Concert Tour Association, PTO, Student Council, Faculty Senate, and the Instructional Council for their review and comment. One comment was received. It was the consensus of the Policy, Evaluation and Goals Committee (PEG) members to recommend it for second reading and action.

Policy 4122, Substitute Teachers

On February 14, 2011, the Policy Evaluation and Goals (PEG) Committee members recommended that the Board of Education approve the following policies for first reading at its regular February Board of Education meeting.

Policy 3310, Contracts/Purchasing Policy 3820, Energy Management Policy 4105, Equal Opportunity and Minority Recruitment

## RECOMMENDATION

Motion: Move to:

- A. Approve Policy 3310, Contracts/Purchasing, for First Reading, as Action presented
- B. Approval of Policy 3820, Energy Management, for First Reading, Action as presented
- C. Approval of Policy 4105, Equal Opportunity and Minority Action Recruitment for First Reading, as presented
- D. Amend Policy 4122, Substitute Teachers, as presented Action

Roll Call Vote

Agenda Item No. IX. A-D.

# FIRST

READING

## POLICY 3310, CONTRACTS/PURCHASING

The Board of Education is responsible for meeting the purchasing needs of the District, including those relating to materials, supplies, equipment, and services, of the quality and quantity required to operate Oak Park and River Forest High School. The Board finds that in order to maximize the interests of the District's residents and taxpayers, as well as suppliers and contractors, and to best protect those interests, a consistently applied policy is required.

All District funds shall be spent prudently and all expenditures of funds shall be made in compliance with the requirements of the School Code of Illinois ("School Code") and other relevant state laws. To this end, the Board of Education directs the Superintendent or a designee and the Chief Financial Officer to establish procedures necessary to achieve fiscal controls and price advantages through the implementation of the following policy of the Board of Education.

- A. <u>Application of School Code.</u> All purchasing, including leasing, shall comply with applicable provisions of the School Code. The Board authorizes the Superintendent or a designee to supervise the purchasing or leasing of all materials, goods, supplies and services for the District in accordance with budget allocations, state laws and sound purchasing practices.
- B. <u>Approval by Board</u>. In accordance with the procedures set forth in Section 10-20.21 of the School Code, the Board of Education will approve all contracts and purchases for supplies, materials or work involving an expenditure in excess of \$25,000, unless specifically exempted by the School Code, or unless specifically authorized to be approved in another manner in accordance with this policy.
- C. <u>Revenue-Generating Contracts</u>. All contracts and purchases for goods or services and that are intended to generate revenue or other remuneration for the District in excess of \$1,000, including without limitation, contracts for vending machines, sports and other attire, class rings, and photographic services, shall be approved by the School Board. The Superintendent or a designee shall ensure that, in accordance with Section 10-20.21(b-5) of the School Code, an attachment is included to the District's annual budget, in the form determined by the Illinois State Board of Education, indicating the names of vendors, the services or products provided, and the actual net revenue and non-monetary remuneration from each of the contracts and agreements identified by this paragraph. In addition, the report will indicate how the revenue was used, and to whom the non-monetary remuneration was distributed.

## D. <u>Quotations</u>

1. For purchases subject to dollar limitations to be awarded through quotations, the Superintendent or a designee shall be authorized to purchase, including by lease, any goods, work or service specifically budgeted which has a sale price within the parameters of the budget.

- 2. For purchases from \$2500 to \$4999, the Superintendent or a designee shall seek a minimum of two (2) competitive quotations.
- 3. For purchases from \$5,000 to \$25,000, the Superintendent or a designee shall seek a minimum of three (3) competitive quotations.
- 4. The Superintendent or a designee may accept or reject any or all quotations obtained through the procedures above.

## E. <u>Competitive Bidding</u>

- 1. For purchases in excess of \$25,000 the Superintendent or a designee shall advertise for sealed bids. Bids shall be awarded by the Board of Education in accordance with the requirements of Section 10-20.21 of the School Code, as well as the Prevailing Wage Act, best business practices as outlined in the Supplemental Regulations to this Policy, and all other applicable law or regulations, as amended from time to time.
- 2. Contractors, subcontractors, and vendors furnishing goods and services to the District shall be in compliance with all local, state, and federal laws and regulations applicable to persons and entities doing business with a School District. The Superintendent or a designee shall develop administrative rules setting forth these requirements.
- 3. In every solicitation for bids, the school district will state that firms owned by minorities and women would be encouraged to bid.
- F. <u>Approval of Lease</u>. Any lease of equipment or machinery shall not exceed five (5) years and shall be approved by affirmative vote of two-thirds (2/3) of the members of the Board, in accordance with Section 10-23.4a of the School Code.
- G. <u>Cooperative Purchasing</u>. The District may participate in cooperative purchasing with other school districts and/or other units of government to take advantage of lower prices for bulk purchasing and to reduce the administrative costs involved in purchasing.
- H. <u>Approval by Chief Financial Officer</u>. All purchases of goods and services with District funds shall be made on a purchase order or contract duly executed by the Chief Financial Officer.
- I. <u>Conflict of Interest</u>. In accordance with the School Code, the Gift Ban Act, 5 ILCS 430/10-10 and the Public Officer Prohibited Practice Act, 50 ILCS 105/0.01 et seq., no Board of Education member or District employee shall be directly or indirectly involved or own an interest in any contract, work, or business of the District, or in the purchase or sale of any real or personal property by or to the District.

J. Construction Project Change Orders. Notwithstanding any Board policy or other language to the contrary, the Board of Education authorizes the Chief Financial Officer to approve any individual construction contract change order valued less than \$75,000 except for any single construction contract change order that is greater than 10% of the total value of the construction contract subject to the change order in which case the Board of Education must approve the change order. The Chief Financial Officer shall provide the Board of Education with a list of these approved change orders on a monthly basis.

Amended Date(s):	August 27, 2009; October 26, 2006; March 24, 2005; August 25, 1994; May 28, 1992; October 17, 1985; December 19, 1984; November 18, 1982
Adopted Date:	
Review Date:	
Law References:	105 ILCS 5/10-20.21
Related Policies:	
Related Instructions	
And Guidelines:	
Cross Ref.:	

## **SUPPLEMENTAL REGULATIONS FOR POLICY 3310**

## I. EQUAL EMPLOYMENT OPPORTUNITY REQUIREMENTS

Contractors, subcontractors and vendors furnishing goods and services to the District shall comply with the Illinois Human Rights Commission's Equal Opportunity Clause, 44 Ill.Adm.Code 750.10. The following shall be incorporated by reference in every individual contract or in contract specification:

EQUAL EMPLOYMENT OPPORTUNITY - In the event of the contractors noncompliance with the provisions of this Equal Employment Opportunity Clause, the Illinois Human Rights Act or the Rules and Regulations of the Illinois Department of Human Rights ("Department"), the contractor may be declared ineligible for future contracts or subcontracts with the State of Illinois or any of its political subdivisions or municipal corporations, and the contract may be canceled or voided in whole or in part, and such other sanctions or penalties may be imposed or remedies invoked as provided by statute or regulation. During the performance of this contract the contractor agrees as follows:

- A. That it will not discriminate against any employee or applicant for employment because of race, color, religion, sex, marital status, sexual orientation, national origin or ancestry, age, physical or mental disability unrelated to ability, an unfavorable discharge from military service, or citizenship status; and further that it will examine all job classifications to determine if minority persons or women are underutilized and will take appropriate affirmative action to rectify any such underutilization.
- B. That, if it hires additional employees in order to perform this contract or any portion thereof, it will determine the availability (in accordance with the Department's Rules and Regulations) of minorities and women in the area(s) from which it may reasonably recruit and it will hire for each job classification for which employees are hired in such a way that minorities and women are not underutilized.
- C. That, in all solicitations or advertisements for employees placed by it on its behalf it will state that all applicants will be afforded equal opportunity without discrimination because of race, color, religion, sex, marital status, national origin or ancestry age, physical or mental disability unrelated to ability, or an unfavorable discharge form military service, or citizenship laws.
- D. That it will send to each labor organization or representative of workers with which it has or is bound by a collective bargaining or other agreement or understanding, a notice advising such labor organization or representative of the contractor's obligations under the Illinois Human Rights Act and the Department's Rules and Regulations. If any such labor organization or representative fails or

refuses to cooperate with the contractor in its efforts to comply with such Act and Rules and Regulations, the contractor will promptly so notify the Department and the contracting agency and will recruit employees from other sources when necessary to fulfill its obligations thereunder.

- E. That it will submit reports as required by the Department's Rules and Regulations, furnish all relevant information as may from time to time be requested by the Department or the contracting agency, and in all respects comply with the Illinois Human Rights Act and the Department's Rules and Regulations.
- F. That it will permit access to all relevant books, records, accounts and work sites by personnel or the contracting agency and the department for purposes of investigation to ascertain compliance with the Illinois Human Rights Act and the Department's Rules and Regulations.
- G. That it will include verbatim or by reference the provisions of this clause in every subcontract it awards under which any portion of the contract obligations are undertaken or assumed, so that such provisions will be binding upon such subcontractor. In the same manner as with other provisions of this contract, the contractor will be liable for compliance with applicable provisions of this clause by such subcontractors; and further it will promptly notify the contracting agency and the department in the event any subcontractor fails or refuses to comply therewith. In addition, the Contractor will not utilize any subcontractor declared by the Illinois Human Rights Commission to be ineligible for contracts or subcontracts with the State of Illinois or any of its political subdivisions or municipal corporations.

## II. WAGES OF EMPLOYEES ON PUBLIC WORKS

Each bidder shall be required to comply with all applicable provisions of the "Wages of Employees on Public Works Act" (Prevailing Wage Act, 820 ILCS 130/0.01 et seq.) The following stipulations required under law are deemed inserted by reference in all contracts with the school district:

- Not less than the prevailing rate of wages as found by the school district or Department of Labor or determined by the court on review shall be paid to all laborers, workers, and mechanics performing work under the contract.
- If, during the course of work under this contract, the Department of Labor revises the prevailing rate of hourly wages to be paid under this contract for any trade or occupation, owner will notify contractor and each subcontractor of the change in the prevailing rate of hourly wages. Contractor shall have the sole responsibility and duty to ensure that the revised prevailing rate of hourly wages is paid by contractor and all subcontractors to each worker to whom a revised rate is applicable. Revisions of the prevailing wage as set forth above shall not result in an increase in the contract sum.

## III. EMPLOYMENT OF ILLINOIS WORKERS ON PUBLIC WORK PROJECTS

Each bidder shall be required to comply with all applicable provisions of the -Employment of Illinois Workers on Public Works Projects." 30 ILCS 570/0.01 et seq.

The following provision shall be included in all contracts whenever there is a period of excessive unemployment in Illinois, as defined by statute:

The contractor shall employ only Illinois laborers on the public works project or improvement or for the cleaning up and on-site disposal of hazardous waste for school district as required by 30 ILCS 570/0.01 et seq.

## IV. SEXUAL HARASSMENT POLICY

Every eligible bidder and every party to a contract shall haven written sexual harassment policy that includes the following:

- the illegality of sexual harassment;
- the definition of sexual harassment under state law;
- a description of sexual harassment, utilizing examples;
- contractors internal complaint process including penalties:
- the legal recourse, investigative, and complaint process available through the Illinois Department of Human Rights and the Human Rights Commission; directions on how to contact the Department and Commission; and protection against retaliation as provided by Section 6-101 of the Human Rights Act.

## V. CHARTER BUS SERVICE CONTRACTS

Pursuant to Section 10-20.21(a) of *The School Code of Illinois*, all contracts for providing charter bus services for the sole purpose of transporting students regularly enrolled in grade 12 or below to or from interscholastic athletic or interscholastic or school-sponsored activities must contain clause (A) as set forth below, except that a contract with an out-of-state company may contain clause (B), as set forth below or clause (A). The clause must be set forth in the body of the contract in typeface of at least 12 points and all upper case letters:

(A) ALL OF THE CHARTER BUS DRIVERS WHO WILL BE PROVIDING SERVICES UNDER THIS CONTRACT HAVE OR WILL HAVE BEFORE ANY SERVICES ARE PROVIDED:

- (1) SUBMITTED THEIR FINGERPRINTS TO THE DEPARTMENT OF STATE POLICE IN THE FORM AND MANNER PRESCRIBED BY THE DEPARTMENT OF STATE POLICE. THESE FINGERPRINTS SHALL BE CHECKED AGAINST THE FINGERPRINT RECORDS NOW AND HEREAFTER FILED IN THE DEPARTMENT OF STATE POLICE AND FEDERAL BUREAU OF INVESTIGATION CRIMINAL HISTORY RECORDS DATABASES. THE FINGERPRINT CHECK HAS RESULTED IN A DETERMINATION THAT THEY HAVE NOT BEEN CONVICTED OF COMMITTING ANY OF THE OFFENSES SET FORTH IN SUBDIVISION (C-1)(4) OF SECTION 6-508 OF THE ILLINOIS VEHICLE CODE; AND
- (2) DEMONSTRATED PHYSICAL FITNESS TO OPERATE SCHOOL BUSES BY SUBMITTING THE RESULTS OF A MEDICAL EXAMINATON, INCLUDING TESTS FOR DRUG USE, TO A STATE REGULATORY AGENCY."

Amended Date(s):	October 26, 2006; August 25, 1994
Adopted Date:	May 28, 1992
Review Date:	
Law Reference:	
Related Policies:	
Related Instructions	
And Guidelines:	
Cross Ref.:	

## POLICY 3820, ENERGY MANAGEMENT

The District shall participate in energy conservation programs approved by the School Board. Energy management is necessary in order for the District to: minimize the impact energy cost increases will have on the budget, maintain a reliable supply of energy to meet the functional needs of the District, and ensure that energy is used efficiently.

A strong commitment on the part of the Board and the Administration is important to an effective energy management program. It shall be the responsibility of each District employee and student to actively participate in conservation efforts in order to reduce consumption to levels prescribed by state, federal and local rules.

The Director of Buildings and Grounds/designee shall implement, direct, monitor, evaluate and report District energy conservation efforts to the Board.

Amended Date(s):	
Adopted Date:	
<b>Review Date:</b>	
Law Reference:	IBC 101.4.7
<b>Related Policies:</b>	2009 International Energy Conservation Code.
<b>Related Instruction</b>	15.
And Guidelines:	Energy Policy Act.
Cross Ref.:	The School Code of Illinois
Revised:	

## ENERGY MANAGEMENT PROCEDURES

All operations of District facilities shall be governed by the following and participation is mandatory for all staff and students of Oak Park and River Forest High School District 200.

- A. Lighting
  - 1. All lights will be turned off in any area which will be unoccupied for a period in excess of fifteen (15) minutes except in corridors, stairwells and at exits as required by code.
  - 2. During design and relamping projects, consideration should be given to provide lighting within the following range:
    - a. Classrooms and offices 45-50 foot-candles (fc) but not less than 30 fc
    - b. Corridors 20 fc but not less than 10 fc
      - Storage Not less than 10 fc
    - d. Gyms 55-95 fc but not less than 30 fc
  - 3. Natural lighting shall be used where possible to attain lighting levels within the above ranges. For cleaning during off hours and in the morning when the building is being opened, the custodial staff shall only turn on lighting where needed. Building should be fully illuminated no more than 45 minutes before the normally scheduled arrival time for teaching and administrative staff. Old building lights will be turned off at 4:30pm Mon-Fri except on days when School is not in session or deemed necessary by the Director of Buildings and Grounds. Student Center and South Hall lights will remain on until 8:00pm Mon-Fri except for prior from the Director of Buildings and Grounds.
- **B.** Temperature Control

c.

- For the heating season, which generally runs from October 15 through May 15, temperatures in classrooms and offices will be maintained at a 68° set point in occupied mode and at 55° in unoccupied mode.
- 2. In those facilities that are air conditioned, a 76° occupied set point with an 82° unoccupied set point will be maintained during the cooling season, which generally runs from May 15<sup>th</sup> to October 15<sup>th</sup>.
- 3. The cooling systems in auditoriums shall maintain a 74° occupied set point.
- 4. Locker rooms shall maintain a 72° occupied set point during the heating season.
- 5. Non-classroom warehouse and garage facilities, when unoccupied by personnel, will be maintained at 55° during the heating season.
- 6. Special consideration will be given to certain daycare and special education classrooms where possible.
- 7. Personnel will not obstruct ventilation ducts or return grilles with books, charts, furniture or plants.

- 8. All windows and doors must be kept closed during the heating season or when air conditioning units are in operation.
- 9. Entrances and exits to all buildings shall be limited in their use when possible to minimize heat loss.
- 10. Broken windows, doors, non-functioning door closers, missing or damaged weather stripping, etc., shall be reported to the Building and Grounds office in a timely manner.
- 11. Unauthorized personnel or students found tampering (e.g., placing ice or wet towels on thermostats) with temperature regulating devices, such as thermostats or valves, will be provided guidance regarding compliance.
- 12. Portable space heaters or air conditioners of any kind are banned from use within District facilities, except where provided by Buildings and Grounds.
- 13. Employees and students are encouraged to wear sweaters, sweatshirts or similar clothing during the heating season.

## C. Scheduling

- 1. Small group activities will not be scheduled in large areas such as auditoriums and gymnasiums. Use of such areas will be coordinated with the maintenance staff to enable reduced lighting and heating during periods of non-use.
- 2. At the end of the school or office day, all windows shall be closed, the blinds or shades drawn to approximately <sup>3</sup>/<sub>4</sub> the distance from the top of the window to the windowsill, and the <u>lights turned off</u>. Cleaning staff will turn lights on only for the period when a specific area is being cleaned. On windows with a western exposure, the blinds should be adjusted to allow the sun to warm building during heating season or to block out the sun during cooling season where appropriate.
- 3. The District shall encourage coordinating facility usage with available heating and air conditioning units that serve the area to be used in order to reduce energy usage.

## D. Other

- 1. The domestic hot water temperature set point will be 120°. Food Services operations requiring higher temperature levels by code shall use booster units or dedicated water heaters when possible.
- 2. Pools shall be kept at a temperature of no less than 80°, but no warmer than 82°, consistent with the recommendation of the National Federation of State High School Association for school pools.
- 3. Office Equipment shut off copiers that do not have power saving options, laminators, etc., at the end of the day. Computers should utilize energy savings options within the operating system.

- 4. The use of personal appliances in classrooms, such as free standing lamps, electric coffee makers, microwaves, refrigerators, toaster ovens, pizza makers and /or other cooking or refrigeration appliances will not be allowed. The use of small fans, radios and desk lamps is allowed, but should be turned off when not in use.
- 5. Request for exemptions and hot and cold complaints must be addressed in work order form to the Director of Buildings and Grounds, at which time he/she will investigate the complaint or request for exemption. If the issue cannot be resolved while adhering to the energy policy, the Director of Buildings and Grounds shall make the determination as to what action, if any, will be taken. The Buildings and Grounds Department reserves the right to adjust set points up or down in a given area to provide the best overall performance of the HVAC system.

Vending Machines – Only energy efficient vending machines will be allowed within the District.

The District will continue to explore new energy savings technology in heating, air conditioning and lighting controls.

6. 7.

## Policy 4105, Equal Employment Opportunity and Minority Recruitment

## **General Personnel**

The School District shall provide equal employment opportunities to all persons regardless of their race, color, creed, religion, national origin, sex, sexual orientation, age, ancestry, marital status, arrest record, military status, order of protection status, unfavorable military discharge, citizenship status provided the individual is authorized to work in the United States, use of lawful products while not at work, being a victim of domestic or sexual violence, genetic information, physical or mental handicap or disability, if otherwise able to perform the essential functions of the job with reasonable accommodation, credit history, unless a satisfactory credit history is an established bona fide occupational requirement of a particular position, or other legally protected categories.

Persons who believe they have not received equal employment opportunities should report their claims to the Nondiscrimination Coordinator and/or a Complaint Manager for the Uniform Grievance Procedure. These individuals are listed below. No employee or applicant will be discriminated or retaliated against because he or she initiated a complaint, was a witness, supplied information, or otherwise participated in an investigation or proceeding involving an alleged violation of this policy or State or federal laws, rules or regulations, provided the employee or applicant did not make a knowingly false accusation nor provide knowingly false information.

Administrative Implementation

The Superintendent shall appoint a Nondiscrimination Coordinator for personnel who shall be responsible for coordinating the District's nondiscrimination efforts. The Nondiscrimination Coordinator may be the Superintendent or a Complaint Manager for the Uniform Grievance Procedure. The Superintendent shall insert into this policy the names, addresses, and telephone numbers of the District's current Nondiscrimination Coordinator and Complaint Managers.

Nondiscrimination Coordinator:

Name

Address

Telephone

## **Complaint Managers:**

Name	Name	
Address	Address	
Telephone	Telephone	

The Superintendent shall also use reasonable measures to inform staff members and applicants that the District is an equal opportunity employer, such as, by posting required notices and including this policy in the appropriate handbooks.

## **Minority Recruitment**

The District will attempt to recruit and hire minority employees. The implementation of this policy may include advertising openings in minority publications, participating in minority job fairs, and recruiting at colleges and universities with significant minority enrollments. This policy, however, does not require or permit the District to give preferential treatment or special rights based on a protected status without evidence of past discrimination.

Amended Date(s):	
Adopted Date:	
<b>Review Date:</b>	
Law Reference: Related Policies: Related Instructions	Age Discrimination in Employment Act, 29 U.S.C. §621; <u>et seq.</u> , Americans With Disabilities Act, Title I, 42 U.S.C. §12111 <u>et</u> <u>seq.</u> Civil Rights Act of 1991, 29 U.S.C. §§621 <u>et seq.</u> , 42 U.S.C. §1981 <u>et seq.</u> , §2000e <u>et seq.</u> , and §12101 <u>et seq.</u> , Equal Employment Opportunities Act (Title VII of the Civil Rights Act of 1964), 42 U.S.C. §2000e <u>et seq.</u> , 29 C.F.R. Part 1601, Equal Pay Act, 29 U.S.C. §206(d),Employee Credit Privacy Act, 820 ILCS 70/.; Genetic Information Nondiscrimination Act, 42 U.S.C. §2000ff <u>et seq.</u> ; Immigration Reform and Control Act, 8 U.S.C. §1324a <u>et seq.</u> Rehabilitation Act of 1973, 29 U.S.C. §791 <u>et seq.</u> ; Title VI of the Civil Rights Act of 1964, 42 U.S.C. §2000d <u>et seq.</u> ; Pregnancy Discrimination Act, 42. U.S.C. §2000e(k).Title IX of the Education Amendments, 20 U.S.C. §1681 <u>et seq.</u> , 34 C.F.R. Part 106; Uniformed Services Employment and Reemployment Rights Act (1994), 38 U.S.C. §§4301 <u>et seq.</u> ; <u>III. Constitution</u> , Art. I, §§17, 18, and 19; 105 ILCS 5/10-20.7, 5/10-20.7a, 5/10-21.1, 5/10-22.4, 5/10-23.5, 5/22-19, 5/24-4, 5/24-4.1, and 5/24-7; Genetic Information Protection Act, 410 ILCS 513/25; III. Whistleblower Act, 740 ILCS 174/.; III. Human Rights Act, 775 ILCS 35/5; III. Equal Pay Act of 2003, 820 ILCS 180/30; 23 III.Admin.Code §1.230.
And Guidelines:	

**Cross Ref.:** 

# SECOND

READING

## **Policy 4122, SUBSTITUTE TEACHERS**

A list of qualified substitute teachers will be maintained in the Personnel Office. Such teachers will be paid on a daily rate schedule adopted by the Board of Education.

The Superintendent may employ substitute teachers as necessary to replace teachers who are temporarily absent.

A substitute teacher must hold a valid teaching or substitute certificate and present a certificate of authorization from the Regional Superintendent showing that he or she is approved to substitute teach. Substitute teachers with a substitute certificate may teach only when an appropriate, fully-certificated teacher is unavailable.

A substitute teacher may teach only for a period not to exceed 90 paid school days or 450 paid school hours in any one school district in any one school term. However, a teacher holding an early childhood, elementary, high school, or special certificate may substitute teach for a period not to exceed 120 paid school days or 600 paid school hours in any one school district in any one school term, unless the subject area is one where the Regional Superintendent has certified that a personnel shortage exists.

The School Board establishes a daily rate of pay for substitute teachers. Substitute teachers receive only monetary compensation for time worked and no other benefits.

If members of the professional staff are assigned as substitutes during one of their unscheduled periods, they will be paid according to a the agreed upon internal substitute rate schedule adopted as reflected in the Faculty Senate and the <u>by</u> the Board of Education's Collective Bargaining Agreement.

Amended Date(s):	May 15, 1975; July 18, 1974;
Adopted Date:	September 23, 1968
Review Date:	
Law Reference:	23 Ill.Admin.Code §1.790;105 ILCS 5/21-9; 24-5.
Related Policies:	
<b>Related Instructions</b>	
And Guidelines:	
Cross Ref.:	

# Oak Park and River Forest High School District 200

201 North Scoville Avenue • Oak Park, IL 60302-2296

TO:Board of EducationFROM:Cheryl L. WithamDATE:February 24, 2011RE:Acceptance of Donations and Gifts

## BACKGROUND

All gifts and donations are presented to the Board of Education as received.

## SUMMARY OF FINDINGS

Mr. James Sarno, brother of employee Nickki Paplaczyk, donated 20 - 2011 Chicago Auto Show tickets for the students in the EAC II Program.

## **RECOMMENDATIONS (OR FUTURE DIRECTIONS)**

MOTION: To accept with gratitude the gifts as presented.

## ROLL CALLVOTE

AGENDA ITEM X. A.

## Oak Park and River Forest High School District 200

201 North Scoville Avenue • Oak Park, IL 60302-2296

TO: Board of Education

FROM: Cheryl L. Witham

DATE: February 24, 2011

RE: Construction Management Contract

## BACKGROUND

Provided is a draft copy of the proposed Construction Management Contract.

## SUMMARY OF FINDINGS

Attorney Ken Florey presented the contract and answered questions at the February 15<sup>th</sup> Finance Committee meeting.

## **RECOMMENDATIONS (OR FUTURE DIRECTIONS)**

MOTION: To accept the Construction Management Contract as presented.

ROLL CALL VOTE

AGENDA ITEM X. C.

# AIA Document A134" - 2009

Standard Form of Agreement Between Owner and Construction Manager as Constructor where the basis of payment is the Cost of the Work Plus a Fee without a Guaranteed Maximum Price

AGREEMENT made as of the « » day of « » in the year « » (In words, indicate day, month and year.)

BETWEEN the Owner. (Name, legal status and address)

« »« ». Oak Park and River Forest High School District 200 « »

and the Construction Manager (Name, legal status and address)

« »K » « »

for the following Project. (Name and address or location)

«Project Name» « »2011 Renovation

The Architect (Name, legal status and address)

« »« » Legat Architects 2015 Spring Road, Suite 175 Oak Brook, Illinois 60523 (C 2)

The Owner's Designated Representative (Name, address and other information)

« » ≪ ≫ «» « » «»

(C ))

The Construction Manager's Designated Representative (Name, address and other information)

a » 8 \$ K D « » « » (C 2)

ALA Document A124" - 2009 (formerly A131"CMc - 2003). Copyright c 1994, 2003 and 2009 by The American Institute of Architects. All rights reserved. WARNING: This ALA Document is protected by U.S. Copyright Law and International Treatise. Unauthorized reproduction or distribution of this ALA Document, or any portion of it, may result in severe cavil and criminal ponsitios, and will be prosented to the maximum extent possible under the law. This dreft was produced by AIA moftware at 11:33 37 on 12/17/2009 under Order No.6946126610 1 which User Note: (1446517284)

ADDITIONS AND DELETIONS: The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

AIA Document A201=-2007, General Conditions of the Contract for Construction, is adopted in this document by reference. Do not use with other general conditions unless this document is modified.

ELECTRONIC COPYING of any portion of this AIA Document to another electronic file is prohibited and constitutes a violation of copyright laws as set forth in the focter of this document.

The Architect's Designated Representative (Name, address and other information)

<b>(</b> <	5}					
	≫					
((	>>					
ĸ	3);					
-{{	>>					
	2>					

The Owner and Construction Manager agree as follows

AIA Document Alle" - 2009 (formerly AlliTick - 2003) Copyright C 1984, 2003 and 2009 by The American Institute of Architects All rights reserved. WARNING: This AIA' Document is protected by U.S. Copyright Law and International Treaties. Unsutherized rep-Judition or distribution of this AIA' Document, or any portion of it, may result in severe civil and eriminal poralies, and will be prosecuted to the maximum extent possible under the law This draft was produced by AJA software at 11 31:37 on 12/17/2008 under Order No.6946126610\_1 which expires on 08/16/2010, and is not for resals
User Motes. (1446517284)

#### **TABLE OF ARTICLES**

- 1 GENERAL PROVISIONS
- 2 CONSTRUCTION MANAGER'S RESPONSIBILITIES
- 3 OWNER'S RESPONSIBILITIES
- 4 COMPENSATION AND PAYMENTS FOR PRECONSTRUCTION PHASE SERVICES
- 5 COMPENSATION FOR CONSTRUCTION PHASE SERVICES
- 6 COST OF THE WORK FOR CONSTRUCTION PHASE
- 7 PAYMENTS FOR CONSTRUCTION PHASE SERVICES
- B INSURANCE AND BONDS
- 9 DISPUTE RESOLUTION
- 10 TERMINATION OR SUSPENSION
- 11 MISCELLANEOUS PROVISIONS
- 12 SCOPE OF THE AGREEMENT

#### ARTICLE 1 GENERAL PROVISIONS

§ 1.1 The Contract Documents

The Contract Documents consist of this Agreement, Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications, Addenda issued prior to the execution of this Agreement, other documents listed in this Agreement, and Modifications issued after execution of this Agreement, all of which form the Contract and are as fully a part of the Contract as if attached to this Agreement or repeated herein. Upon the Owner's approval of the Control Estimate, the Contract Documents will also include the documents described in Section 2.2.4 and revisions prepared by the Architect and furnished by the Owner as described in Section 2.2.5. The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations or agreements, either written or oral. If anything in the other Contract Documents, other than a Modification, is inconsistent with this Agreement, this Agreement shall govern.

#### § 1.2 Relationship of the Parties

The Construction Manager accepts the relationship of trust and confidence established by this Agreement and covenants with the Owner to cooperate with the Architect and exercise the Construction Manager's skill and judgment in furthering the interests of the Owner, to furnish efficient construction administration, management services and supervision, to furnish at all times an adequate supply of workers and materials, and to perform the Work in an expeditious and economical manner consistent with the Owner's interests. The Owner agrees to furnish or approve, in a timely manner, information required by the Construction Manager and to make payments to the Construction Manager in accordance with the requirements of the Contract Documents.

§ 1.2.1 The Construction Manager has been engaged to provide professional construction management services because of its character, experience, education, depth of experienced personnel, and qualifications in dealing with projects of similar scope, complexity, and magnitude.

#### § 1.3 General Conditions

For the Preconstruction Phase, AIA Document A201<sup>™</sup>-2007, General Conditions of the Contract for Construction, shall apply only as specifically provided in this Agreement. For the Construction Phase, the general conditions of the contract shall be as set forth in an amended A201-2007, which document is <u>attached hereto and incorporated as</u> <u>amended</u> herein by reference. The term "Contractor" as used in A201-2007 shall mean the Construction Manager

All Document All - 2008 (formerly All Document as protected by U.S. Copyright C 1994, 2003 and 2009 by The American Institute of Architects All rights reserved WANNING. This AIA Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA Document, or any pottion of it. may result in severe cival and criminal ponalities, and will be protected to the maximum extent possible under the law. This draft was produced by AIA software at 11-31:37 on 12/17/2009 under Order No.6946126610.1 which expises on 08/16/2010, and is not for resale

#### § 1.4 Contract Sum, Contract Time and Changes in the Work

The Contract Sum is the actual Cost of the Work as defined in Section 6.1.1 plus the Construction Manager's Fee as defined in Section 5.1. The Contract Time is the period of time, including authorized adjustments, allotted in the Contract Documents for Substantial Completion of the Work as certified by the Architect in accordance with Section 9.8 of AIA Document A201-2007. The Contract Time shall be measured from the date of commencement of the Construction Phase as established pursuant to Section 2.3.1.2 of this Agreement. Changes in the Work shall be governed by Section 5.2 of this Agreement and not by Article 7 of A201-2007. If, however, the Contract Time has been established in accordance with Section 2.2.4.5, Article 7 of A201-2007 shall control adjustments to the Contract Time-

#### ARTICLE 2 CONSTRUCTION MANAGER'S RESPONSIBILITIES

**<u>52.0.1**</u> The Construction Manager shall perform the services described in this Agreement.</u> The Construction Manager's Preconstruction Phase responsibilities are set forth in Sections 2.1 and 2.2. The Construction Manager's Construction Phase responsibilities are set forth in Section 2.3. The Owner and Construction Manager may agree, in consultation with the Architect, for the Construction Phase to commence prior to completion of the Preconstruction Phase, in which case, both phases will proceed concurrently. The Construction Manager shall identify a representative authorized to act on behalf of the Construction Manager with respect to the Project.

§2.0.2 The Construction Manager shall designate an authorized representative who shall have express authority granted by the Construction Manager to bind the Construction Manager with respect to all matters requiring the Construction Manager's approval or authorization. Construction Manager's representative shall have the authority to make representations on behalf of the Construction Manager concerning estimates and schedules, construction budgets, changes in the Work, and all other matters related to this Agreement.

§2.0.3 In addition to the other responsibilities set out below for pre-construction and bidding phases, the Construction Manager shall also perform the following tasks relative to Minority Owned Business Enterprises (MBE) and Female Owned Business Enterprises (FBE):

- a) The Construction manager will advise the Owner as to potential alternatives for increasingparticipation on the Project by MBE and FBE firms.
- b) If any of the alternatives are selected by the Owner, the Construction Manager will coordinate the planned use of increased MBE/FBE and labor goals throughout design, bidding, and construction phases.
- c) The Construction Manager will coordinate with the Project team (in the design phase) to recommend and review split bid packages, the divisions of work and how MBE/FBE firms and the labor force can meet the needs of those bid packages. Constructability of the project must not be compromised in the attainment of these goals.
- d) Construction Manager will ensure that the Instructions to Bidders and Bid Form shall require bidders to identify all MBE/FBE firms which the bidder contacted in order to solicit sub-bids.

#### § 2.1 Preconstruction Phase

§ 2.1.1 The Construction Manager shall provide a preliminary evaluation of the Owner's program (if any), schedule and construction budget requirements, each in terms of the other.

#### § 2.1.2 Consultation

The Construction Manager shall schedule and conduct meetings with the Architect and Owner to discuss such matters as procedures, progress, coordination, and scheduling of the Work. The Construction Manager shall advise the Owner and the Architect on proposed site use and improvements, selection of materials, and building systems and equipment. The Construction Manager shall also provide recommendations consistent with the Project requirements to the Owner and Architect on constructability, availability of materials and labor, time requirements for procurement, installation and construction, and factors related to construction cost including, but not limited to, costs of alternative designs or materials, preliminary budgets, life-cycle data, and possible cost reductions

§ 2.1.3 When Project requirements in Section 3.1.1 have been sufficiently identified, the Construction Manager shall prepare and periodically update a Project schedule for the Architect's review and the Owner's acceptance. The Construction Manager shall obtain the Architect's approval for the portion of the Project schedule relating to the performance of the Architect's services. The Project schedule shall coordinate and integrate the Construction Manager's services, the Architect's services, other Owner consultants' services, and the Owner's responsibilities

ATA Document A134 - 2009 (formerly A133 the - 2003). Copyright - 1994, 2003 and 2009 by the American Institute of Architects All rights reserved. Herere this ATA Research is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or destinations of this ATA Research as any portion of it, may result in severe civil and original penalties, and will be prosecuted to the espires on 08/16/2010, and is not for resale. (1446037284)

Formatted: Numbered + Level: 1 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75"

Formatted: Font color: Auto

and identify items that could affect the Project's timely completion. The updated Project schedule shall include the following. submission of the Control Estimate; the components of the Work, times of commencement and completion required of each Subcontractor, ordering and delivery of products, including those that must be ordered well in advance of construction, and the occupancy requirements of the Owner <u>All project schedules (including updates) shall be a Critical Path Method (CPM) schedules.</u>

§2.1.2.1 The Construction Manager shall assist the Owner and Architect in preparing Construction Contracts and advise the Owner on the acceptability of sub-tier subcontractors and material suppliers proposed by Subcontractors. The Construction Manager shall submit for review and approval to Owner, or its attorneys if designated by Owner, all agreements and conditions prepared for submission to bidders prior to release of bidding information.

#### § 2.1.4 Phased Construction

The Construction Manager shall provide recommendations with regard to accelerated or fast-track scheduling, procurement, or phased construction. The Construction Manager shall take into consideration cost reductions, cost information, constructability, provisions for temporary facilities and procurement and construction scheduling issues.

#### § 2.1.5 Preliminary Cost Estimates

§ 2.1.5.1 Based on the preliminary design and other design criteria prepared by the Architect, the Construction Manager shall prepare preliminary estimates of the Cost of the Work or the cost of program requirements using area, volume or similar conceptual estimating techniques for the Architect's review and Owner's approval 1f the Architect or Construction Manager suggest alternative materials and systems, the Construction Manager shall provide cost evaluations of those alternative materials and systems.

§ 2.1.5.2 As the Architect progresses with the preparation of the Schematic Design, Design Development and Construction Documents, the Construction Manager shall prepare and update, at appropriate intervals agreed to by the Owner, Construction Manager and Architect, estimates of the Cost of the Work of increasing detail and refinement and allowing for the further development of the design until such time as the Construction Manager submits a Control Estimate for the Work, pursuant to Section 2.2. Such estimates shall be provided for the Architect's review and the Owner's approval. The Construction Manager shall inform the Owner and Architect when estimates of the Cost of the Work exceed the latest approved Project budget and make recommendations for corrective action.

#### § 2.1.6 Subcontractors and Suppliers

The Construction Manager shall develop bidders' interest in the Project.

§ 2.1.6.1 Owner is a public entity bound by 110 ILCS 805/3-27.1 and is required to publicly bid all construction contracts and all construction contracts shall be awarded to the lowest responsible responsive bidder. Once the lowest responsible responsive multiple prime trade bidders. (hereinafter referred to as "Subcontractors") are identified. Owner shall assign the trade contracts to Construction Manager.

The trade contracts shall incorporate substantially the following provisions:

Though this Contract has been awarded by the Owner after advertisement for bids, upon award of the Contract by the Owner, the Owner shall be deemed to have assigned and set over portions of its rights and interest in this Agreement to [Construction Manager], which is identified herein as the Contractor.

By submitting its bid, the Contractor shall be deemed to have consented to the aforesaid assignment, and to have agreed to become an assigned subcontractor to the Contractor.

Upon assignment, the Subcontractor shall become a subcontractor of the Contractor pursuant to this Agreement, and, except as identified within this Agreement and as provided by law, will no longer have any rights under the contract against the Owner, but all of its rights under this Agreement shall be against the Contractor.

§ 2.1.6.2 Construction Manager shall prepare all bid invitations, instructions to bidders, and general and supplementary conditions. Drawings and technical specifications shall be provided by the Architect. During

AIA Document AING - 7008 (formarly AINITCM: - 2003). Copyright C 1994, 2003 and 2009 by The American Institute of Architects. All rights reserved. WANNING: This AIA Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosocuted to the maximum effect possible under the law. This deaft was produced by AIA software at 11:31:37 on 12/17/2009 under Order No 6945126410 (which septres on 06/15/2010, and is not for resule. User Notes

preparation of the Construction Documents by the Architect, the Construction Manager shall review said Construction Documents to ensure consistency with the bid documents prepared by the Construction Manager

§ 2.1.6.2.1 Construction Manager will ensure that the Instructions to Bidders require the bidders to contact at least three MBE/FBE firms in order to solicit bids, or in the alternative, to explain on the Bid Form why three firms could not be contacted. Additionally, Construction Manager will ensure that the Bid Form shall provide space for the bidder to identify the firms contacted, or explanation why firms were not contacted.

§2.1.6.3 Prior to advertisement for bids, Construction Manager shall provide all contractor or subcontract agreements to the Owner's attorneys for review and comment

§2.1.6.4 Construction Manager shall conduct pre-bid meetings with interested bidders in accordance with publicity announced scheduled meetings.

§2.1.6,5 The Owner shall receive bids, prepare bid analyses and make recommendations to the Owner for the Owner's award of contracts or rejection of bids.

§ 2.1.6.6 The Subcontract documents prepared by Construction Manager shall require full compliance with all state and local laws.

§ 2.1.7 The Construction Manager shall prepare, for the Architect's review and the Owner's acceptance, a procurement schedule for items that must be ordered well in advance of construction. The Construction Manager shall expedite and coordinate the ordering and delivery of materials that must be ordered well in advance of construction. The Construction Manager shall expedite and coordinate the ordering and delivery of materials that must be ordered well in advance of construction. The Construction Manager shall expedite and coordinate the ordering and delivery of materials that must be ordered well in advance of construction. The Construction for the Owner agrees to procure any items prior to the Owner's approval of the Construction sacceptable to the Construction Manager. Upon the Owner's approval of the Control Estimate, the Owner shall assign all contracts for these items to the Construction Manager and the Construction Manager shall thereafter accept responsibility for them.

#### § 2.1.8 Extent of Responsibility

The Construction Manager shall exercise reasonable care in preparing schedules and estimates The Construction Manager, however, does not warrant or guarantee estimates and schedules, including the Control Estimate and the estimated date of Substantial Completion, except as provided in Section 2.2.4.5. The Construction Manager is not required to ascertain that the Drawings and Specifications are in accordance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, but the Construction Manager shall promptly report to the Architect and Owner any nonconformity discovered by or made known to the Construction Manager as a request for information in such form as the Architect may require

#### § 2.1.9 Notices and Compliance with Laws

The Construction Manager shall comply with applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities applicable to its performance under this Contract, and with equal employment opportunity programs, and other programs as may be required by governmental and quasi governmental authorities for inclusion in the Contract Documents

#### § 2.2.11 PERMITS

The Construction Manager shall assist the Owner in obtaining building permits and special permits for permanent improvements, except for permits required to be obtained directly by the various Subcontractors. The Construction Manager shall verify that the Owner has paid applicable fees and assessments. The Construction Manager shall assist the Owner and Architect in connection with the Owner's responsibility for filing documents required for the approvals of governmental authorities having jurisdiction over the Project.

#### § 2.2 Control Estimate

§ 2.2.1 At the completion of the Construction Documents by the Architect, and before advertisement for bids, a time to be mutually agreed upon by the Owner and the Construction Manager and in consultation with the Architect, the Construction Manager shall prepare a Control Estimate for the Owner's review and acceptance. The Construction Estimate shall be the sum of the Construction Manager's estimate of the Cost of the Work and the Construction

ATA Document Al34" - 3009 (formerly Al31"CMc - 2003). Copyright © 1894, 2003 and 2008 by The American Institute of Architects. All sights reserved. utility = the site of the

Manager's Fee and shall include those items set forth in Section 2.2.4 below. When the Control Estimate is acceptable to the Owner, the Owner shall, upon request by the Construction Manager, acknowledge it in writing

§ 2.2.2 The Construction Manager shall develop and implement a detailed system of cost control that will provide the Owner and Architect with timely information as to the anticipated total Cost of the Work. The cost control system shall compare the Control Estimate with the actual cost for activities in progress and estimates for uncompleted tasks and proposed changes. This information shall be reported to the Owner, in writing, no later than the Construction Manager's first Application for Payment and shall be revised and submitted with each Application for Payment

§ 2.2.3 To the extent that the Drawings and Specifications are anticipated to require further development by the Architect, the Construction Manager shall provide in the Control Estimate for such further development consistent with the Contract Documents and reasonably inferable therefrom. Such further development does not include changes in scope, systems, kinds and quality of materials, finishes or equipment, all of which, if required, shall be incorporated in a revised Control Estimate by mutual agreement of the parties.

§ 2.2.4 The Control Estimate shall include

- .1 a list of the Drawings and Specifications, including all Addenda thereto, and the Conditions of the Contract;
- .2 a list of the clarifications and assumptions made by the Construction Manager in the preparation of the Control Estimate, including assumptions under Section 2.2.3, to supplement the information provided by the Owner and contained in the Drawings and Specifications.
- .3 a statement of the estimated Cost of the Work organized by trade categories or systems, allowances, and the Construction Manager's Fee;
- .4 the anticipated date of Substantial Completion upon which the Control Estimate is based, and a schedule for the issuance dates of the Construction Documents upon which the anticipated Substantial Completion date relies; and
- .5 a statement as to whether or not the duration from the stated date of commencement of the Construction Phase to the estimated date of Substantial Completion shall become the Contract Time and be subject to the provisions of Article 8 of A201-2007

§ 2.2.5 The Owner shall authorize the Architect to incorporate the agreed-upon assumptions and clarifications contained in the Control Estimate. The Owner shall promptly furnish those revised Drawings and Specifications to the Construction Manager as they are revised. The Construction Manager shall notify the Owner and Architect of any inconsistencies between the Control Estimate and the revised Drawings and Specifications.

§2.2.6 In the event that any portion of the Control Estimate materially differs (i.e., by more than 10% of the bid or applicable portion thereof) from the lowest bona fide bid for the respective portion of the Work covered by that portion of the Control Estimate, the Construction Manager shall;

- a) If the Owner chooses to accept the bid, compensate the Owner for additional design costs to modify the scope of Work to bring the cost materially within the cost set out in the Control Estimate; and
- b) Negotiate respective change orders with the respective Subcontractors to incorporate the aforesaid design changes, at no additional charge to the Owner, or
- c) If the Owner chooses to reject all bids and re-bid, compensate the Owner for additional design costs to modify the scope of work for re-bidding; and
- d) Administrate re-bid of the affected portions of Work at no additional charge to the Owner.

#### § 2.3 Construction Phase

§ 2.3.1 General

§ 2.3.1.1 For purposes of Section 8 1 2 of A201–2007, the date of commencement of the Work shall mean the date of commencement of the Construction Phase.

§ 2.3.1.2 The Construction Phase shall commence upon the Owner's approval of the Control Estimate or the Owner's issuance of a Notice to Proceed, whichever occurs carlier.

§2.3.1.3 Prior to commencement of the Construction Phase, the Construction Manager shall not incur any cost to be reimbursed as part of the Cost of the Work.

ATA Document A134" - 2008 (formerly A131"CMC - 2003) Copyright C 1994, 2003 and 2009 by The American Institute of Architects All rights reserved. HAULTS: The AIA Document is protected by U.S. Copyright Law and International Treations Unapthonized Reproduction distributed of this AIA Document, we any portion of it, may result in severe civil and oriminal penalties, and will be presented to the select stand period. The law This draft was produced by AIA software at 11:31:37 on 12/17/2009 under Order No. 6940226810.1 which separat on 08/16/2010, and is not for resule User Wolkes (1446517224)

Formatted: Numbered + Level: 1 + Numbering Style: a, b, c, ... + Start at: 1 + Aignment: Left + Aligned at: 0.5" + Indent at: 0.75"

#### § 2.3.2 Administration

§ 2.3.2.1 All trade Work shall be performed by Subcontractors, as assigned as set forth elsewhere in this Agreement If the Construction Manager has reasonable objection to any Subcontractor who is considered the lowest responsive bidder, it shall be the burden of Construction Manager to present demonstrative evidence that the bidder is not responsible The Construction Manager shall not "self perform" any trade Work Those portions of the Work that the Construction Manager does not oustomarily perform with the Construction Manager's own personnel shall be performed under subcontracts or by other appropriate agreements with the Construction Manager. The Owner may designate specific persons from whom, or entities from which, the Construction Manager shall obtain bids. The Construction Manager shall obtain bids from Subcontractors and from suppliers of materials or equipment fabricated especially for the Work and shall deliver such bids to the Architest. The Owner-shall then determine, with the advice of the Construction Manager and the Architect, which bids will be accepted. The Construction Manager shall not be required to contract with anyone to whom the Construction Manager has reasonable objection.

§ 2.3.2.2 Subcontracts or other agreements shall conform to the applicable payment provisions of this Agreement, and shall not be awarded on the basis of cost plus a fee without the prior consent of the Owner If the Subcontract is awarded on a cost-plus a fee basis, the Construction Manager shall provide in the Subcontract for the Owner to receive the same audit rights with regard to the Subcontractor as the Owner receives with regard to the Construction Manager in Section 6.11 below-

§2.3.2.3 Subcontracts or other agreements shall conform to the applicable payment provisions of this Agreement, and shall not be awarded on the basis of cost plus a fee.

§ 2.3.2.3 If the Construction Manager recommends a specific bidder that may be considered a "related party" according to Section 6.10, then the Construction Manager shall promptly notify the Owner in writing of such relationship and notify the Owner of the specific nature of the contemplated transaction, according to Section 6.10.2

§ 2.3.2.4 If the Construction Manager recommends a specific bidder that may be considered a "related party" according to Section 6.10, then the Construction Manager shall promptly notify the Owner in writing of such relationship and notify the Owner of the specific nature of the contemplated transaction, according to Section 6.10.2.

§ 2.3.2.5 The Construction Manager shall prepare and regularly update no less than once per month a Construction Schedule utilizing a Critical Path Method (CPM), or other Owner pre-approved scheduling method, which incorporates the activities of the Subcontractors on the Project, including activity sequences and durations, allocation of labor and materials, processing of shop drawings, Product Data and Samples and delivery of products requiring long lead time and procurement. Additionally, the schedule shall be in accordance with applicable Sections of A2017M-2007, as amended, including the Owner's occupancy requirements.

§ 2.3.2.6 The Construction Manager shall develop a system of cost control for the Work, including regular monitoring of actual costs for activities in progress and estimates for uncompleted tasks and proposed changes. The Construction Manager shall identify variances between actual and estimated costs and report the variances to the Owner and Architect and shall provide this information in its monthly reports to the Owner and Architect, in accordance with Section 2.3.2.7 above.

§ 2.3.2.7 In collaboration with the Architect, the Construction Manager shall establish and implement procedures for expediting the processing and approval of Shop Drawings, Product Data, Samples and other submittals. The Construction Manager shall review all Shop Drawings. Product Data. Samples and other submittals from the Subcontractors for apparent compliance with subcontract requirements. The Construction Manager shall transmit to the Architect those submittals which have been reviewed by the Construction Manager. The Construction Manager's actions shall be taken with such reasonable promptness as to cause no delay in the Work or in the activities of the Owner or Contractors. The Construction Manager shall receive and process all other submittals required by the Construction Documents, including certified payrolls, work plans, safety plans, periodic schedule updates, and the like.

§ 2.3.2.7.1 The Construction Manager shall create and maintain a log of all submittals showing, at a minimum, the dates of submittal, review, and return, disposition, and description.

§ 2.3.2.84 The Construction Manager shall schedule and conduct meetings to discuss such matters as procedures. progress, coordination, scheduling, and status of the Work The Construction Manager shall prepare and promptly distribute minutes to the Owner and Architect

AIA Document All<sup>4</sup> - 2009 (formerly All<sup>4</sup>CMc - 2003). Copyright C 1994, 2003 and 2009 by The American Institute of Architects. All rights reserved. WARNING This AIA Document is protocted by U.S Copyright Law and International Treates. Unsubfording temperatures of a state of the institute and the reserved is a state of the institute of the law. This draft was produced by AIA software at 11:31:37 on 12/17/2009 under Order No.8946126610\_1 which User Notes:

§ 2.3.2.5 Upon the Owner's approval of the Control Estimate, the Construction Manager shall prepare and submit to the Owner and Architect a construction schedule for the Work and submittal schedule in accordance with Section 3-10 of A201–2007.

§ 2.3.2.96 The Construction Manager shall record the progress of the Project. On a monthly basis, or otherwise as agreed to by the Owner, the Construction Manager shall submit written progress reports to the Owner and Architect, showing percentages of completion and other information required by the Owner. The Construction Manager shall also keep, and make available to the Owner and Architect, a daily log containing a record for each day of weather, portions of the Work in progress, number of workers on site, identification of equipment on site, problems that might affect progress of the Work, accidents, injuries, and other information required by the Owner.

#### § 2.4 Professional Services

Section 3.12.10 of A201-2007 shall apply to both the Preconstruction and Construction Phases

#### § 2.5 Hazardous Materials

Section 10.3 of A201-2007 shall apply to both the Preconstruction and Construction Phases. Construction Manager shall provide immediate written notice to the Owner and Architect if the Construction Manager discovers the existence, of any and all hazardous material, including, but not limited to any lead or lead based material and asbestos, asbestos-related products including the extent and location of same. Construction Manager shall not be responsible for the removal, encapsulation, transportation or disposal of any hazardous material.

Section 10.3 of A201-2007 shall apply to both the Preconstruction and Construction Phases.

**§2.6** CONSTRUCTION MANAGER STAFFING Construction Manager shall staff the Project. Construction Manager shall maintain one competent full-time superintendent at the Project site to coordinate and provide supervision of the Work and progress of the Subcontractors on the Project who shall devote full-time to the Project and who, so long as they remain in the employ of Construction Manager or any subsidiary or affiliate thereof, shall be removed from the Project except with the prior written consent of the Owner. If any such individual is no longer employed on the Project pursuant to this paragraph. Construction Manager shall propose to Owner the names of other individuals as substitutes. No person shall be employed on the Project as to whom Owner has reasonable objection. If Owner notifies Construction Manager that it has reasonable objection to any such individual, Construction Manager shall remove such person from the Project and propose substitutes to Owner for Owner's approval.

§2.7 CONTRACTOR INSURANCE AND SURETY DOCUMENTS Construction Manager shall secure Certificates of Insurance, as well as Performance and Payment Bonds, for each of the Subcontractors immediately upon award of the Contract and verify conformance of same with the Contract Documents prior to allowing the Subcontractors onto the Site. Insurance of Subcontractors shall name Owner, Construction Manager, and Architect as "Additional Insureds". Copies of same shall be transmitted to the Owner and Architect before Subcontractors may commence work on the project.

#### § 2.8 SCHEDULE

§2.8.1 Time is of the essence of this contract. Construction Manager, with the Owner's approval shall establish the Construction Schedule. It is the duty of the Construction Manager to schedule, sequence and coordinate the performance of the Subcontractors on the Project so that completion shall be in accordance with the Construction Schedule.

§2.8.2 The Construction Schedule shall commence with the notice to proceed, and shall schedule the Subcontractors' Work using a Critical Path Method or other scheduling methodology approved by Owner, Construction Manager shall update the Project Construction Schedule on a monthly basis to show current and forecasted progress and completion. A copy of each schedule shall be transmitted to the Owner, Architect and each Subcontractor. Additionally, Construction Manager shall develop a look-ahead schedule for discussion at each weekly construction coordination meeting. The weekly look-ahead shall, at a minimum, show the schedule for the past week and upcoming two weeks.

AIA Document Al34" - 3009 (formerly Al31"CMc - 2003). Copyright C 1984, 2003 and 2009 by The American Institute of Architetts. All rights reserved. WARNING: This AIA' Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA' Document, or any portion of it, may result in severe cavil and criminal penalties. and will be presecuted to the maximum extent possible under the law. This dfaft was produced by AIA software at 11:31:37 on 12/17/2009 under Order No.5945125610 (14465)72241 (14465)72241

**§2.8.3** If a monthly Project Schedule update indicates that the previously approved Project construction schedule may not be met, the Construction Manager shall recommend corrective action to the Owner.

§2.9 COST RECORDS Construction Manager shall verify and maintain in accordance with generally accepted accounting principles detailed cost accounting records on authorized work performed on the basis of unit costs, actual costs for labor and materials and other bases showing all costs, and shall maintain and transmit to Owner all receipts, invoices, purchase orders, canceled checks, bills of lading and other documents and evidence of payment or indebtedness to support such records as Owner may request. Construction Manager shall audit, verify and recommend approval of all Change Orders submitted by Subcontractors. Construction Manager shall afford the Owner access to these records and preserve them for a period of three (3) years after final payment.

#### §2.10 CONTRACT CLARIFICATIONS

§2.10.1 The Construction Manager shall transmit to the Architect requests for interpretations, information, or clarification of the meaning and intent of the Drawings and Specifications, and assist in the resolution of questions that may arise.

§ 2.10.2 The Construction Manager shall create and maintain a log of all requests showing, at a minimum, a description of the request, date of submission, response, and date of response.

#### § 2.11 SUBCONTRTACTOR CHANGE ORDERS

§2.11.1 Without invalidating this Agreement or any Subcontract, and without notice to any respective surety. Owner may make changes to the Subcontractor contracts. The Construction Manager shall review requests for changes, assist in negotiating Subcontractors' proposals, submit recommendations to the Architect and Owner, and, if they are accepted, prepare Change Orders and Construction Change Directives which incorporate the modifications to the Documents. For each change which is anticipated to be paid by lump sum, Construction Manager shall prepare a detailed estimate for the respective Change Order or Construction Change Directive initiated. Construction Manager shall compare the detailed value estimate with the submittal of the Subcontractor. Construction Manager shall require that each Subcontractor submit such invoices, purchase orders, labor and material records, and schedule analyses as is necessary to verify the accuracy of the Change Order or Construction Change Directive price and/or schedule adjustment. Construction Manager shall submit a detailed recommendation for the Owner's review and approval for each Change Order or Construction Change Directive. Construction Manager shall systematically maintain all documentation supporting each change and upon completion of the project turn over such documentation to the Owner. For the purposes of this paragraph, it is anticipated that only changes in which schedule considerations allow the change order price to be agreed prior to the performance of the change order work will be paid lump sum. It is further anticipated that changes in which the change order price must be agreed after the change order work is completed will be paid time and material reimbursable.

**§2.11.2** The Construction Manager shall create and maintain a log of all requests for changes showing, at a minimum, date of submission, description of request, requested change in contract price, requested change in contract time, and disposition

§ 2.11.3 The Construction Manager shall create and maintain a log of all approved change orders showing, at a minimum, dates of approval, description, reason for change, agreed price of change, and agreed change of contract time.

#### § 2.12 CLAIMS

§ 2.12.1 The Construction Manager shall assist the Owner and Architect in the review, evaluation and documentation of Claims. The Owner shall provide a written recommendation to the Owner suggesting the proper disposition of each claim. Construction Manager shall require that each Subcontractor submit such invoices, purchase orders, labor and material records, and schedule analyses as is necessary to verify the accuracy of the Claim cost and schedule impact. Construction Manager shall submit a detailed recommendation for the Owner's review and approval for each Claim.

AIA Document Al34" - 2009 (formerly Al31"CMC - 2003). Copyright C 7594, 2003 and 2009 by The American Institute of Architects. All rights reserved. WARNING This AIA' Document is protocted by U.S Copyright C 1494 and International Treaties Unauthorized reproduction or distribution of this AIA' Document, or any portion of it, may result in severe civil and criminal possibles, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 11:31:37 on 12/17/2009 under Order No 5945125610\_1 which expires on 08/16/2010, and is not for resele. § 2.12.2 The Construction Manager shall create and maintain a log of all Claims showing, at a minimum, date of submission, description of claim, demanded change in contract price, demanded change in contract time, and disposition.

§2.13 EOUITABLE EXTENSION OF TIME AND NO DAMAGE FOR DELAY To the extent that Construction Manager's Services are delayed by the Owner or causes not under control of Construction Manager. the Contract Time shall be equitably extended. Such extension of the Contract Time shall be Construction Manager's sole and exclusive remedy.

#### § 2.14 SITE SAFETY

§ 2.14.1 Construction Manager shall hold regular safety meetings at the site at least weekly with all Subcontractors Construction Manager shall maintain all required safety records and logs. The Construction Manager shall review. approve and coordinate the safety programs developed by each of the Subcontractors; Neither the Owner, Construction Manager, nor the Architect is responsible for Site Safety. The individual subcontracts shall provide that the subcontractors are solely responsible for the means and methods of prosecuting their respective work. The foregoing shall not relieve Subcontractors of their responsibility for site safety for their work and the work performed by their personnel.

#### § 2.15 OUALITY CONTROL

§2.15.1 Construction Manager shall inspect the Work of Subcontractors for defects and deficiencies in the Work in order to assure compliance with the requirements of the Contract Documents. As appropriate, the Construction Manager shall have authority, upon written authorization from the Owner, to require additional inspection or testing of the Work in accordance with the provisions of the Contract Documents, whether or not such Work is fabricated, installed or completed. The Construction Manager, in consultation with the Architect and Owner, may reject Work which does not conform to the requirements of the Contract Documents

§2.15.2 The Construction Manager shall assist the Owner in obtaining from the responsible Subcontractor correction of all portions of the project which do not comply with the Contract Documents or do not conform with good workmanlike performance.

§ 2.16 AS BUILT DOCUMENTS Construction Manager shall obtain data from Subcontractors and maintain a current set of accurate As Built Drawings and Specifications clearly marked to show reported actual changes during construction. Construction Manager shall transmit a copy of all As Built drawings and specifications to Architect for the preparation of final As Built Documents. Construction Manager shall periodically (not less than monthly) review Subcontractors' data to verify that they are recording information sufficient to allow preparation of the As Built documents

§ 2.17 SUBMISSION OF PROJECT DOCUMENTS Prior to final payment at the completion of the Project, the Construction Manager shall inventory and turn over to the Owner one copy of the following: Record/As Built Drawings and Specifications prepared by the Architect, the daily log, progress reports, project manual, field orders, change orders, requests for changes, requests for information, claims, Subcontractor pay applications, communications, submittals, and all other Construction Documents,

§ 2.18 OPERATIONS AND MAINTENANCE TRAINING AND MANUALS Construction Manager shall arrange and schedule operating and maintenance training on mechanical, electrical and other systems for Owner's personnel by factory authorized representatives commencing upon start-up and the commissioning of each system prior to substantial completion. Construction Manager shall receive all operating and maintenance manuals from Subcontractors, and, along with the Architect, shall review for completeness, clarity, and conformance with the respective contract documents. After all operations and maintenance manuals are received and approved, but no later than turnover of operations to Owner personnel. Construction Manager shall transmit all approved operations and maintenance manuals to the Owner.

52.19 PERFORMANCE AND PAYMENT BONDS Construction Manager shall obtain payment and performance bonds in accordance with Article 11 of the General Conditions.

AIR Document Alls" - 2005 (formarly Alls"Cake - 2003). Copyright < 1954, 2003 and 2009 by The American Institute of Architects All rights reserved. WARNING: This AIA Document is protected by U.S. Copyright Law and International T. attes. Unauthorized reproduction or All bottomit wide - avoid advantage and the - avoid conjunct, and a son avoid and avoid and avoid and the son and a son avoid a son and a son maximum extent possible under the law. This d espires ch 08/16/2010, and is not for resels. (1446537284)

Formatted: Font: Bold Formatted: Font: Bold Formatted: Font: (Default) Times New Roman, Not Bold Formatted: Normal

# ARTICLE 3 OWNER'S RESPONSIBILITIES

§ 3.1 Information and Services Required of the Owner

§ 3.1.1 The Owner shall provide information with reasonable promptness, regarding requirements for and limitations on the Project, including a written program which shall set forth the Owner's objectives, constraints, and criteria. including schedule, space requirements and relationships, flexibility and expandability, special equipment, systems sustainability and site requirements

§ 3.1.2 Prior to the Owner's approval of the Control Estimate, or within seven days of receiving the Owner's written acknowledgment required by Section 2.2.1, the Construction Manager may request in writing that the Owner provide reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the Contract. Thereafter, the Construction Manager may only request such evidence if (1) the Owner fails to make payments to the Construction Manager as the Contract Documents require, (2) a change in the Work materially changes the Contract Sum, or (3) the Construction Manager identifies in writing a reasonable concern regarding the Owner's ability to make payment when due. The Owner shall furnish such evidence as a condition recedent to commencement or continuation of the Work or the portion of the Work affected by a material change After the Owner furnishes the evidence, the Owner shall not materially vary such financial arrangements without prior notice to the Construction Manager and Architect-

§ 3.1.3 The Owner shall establish and periodically update the Owner's budget for the Project, including (1) the budget for the Cost of the Work as defined in Section 6.1.1, (2) the Owner's other costs, and (3) reasonable contingencies related to all of these costs. If the Owner significantly increases or decreases the Owner's budget for the Cost of the Work, the Owner shall notify the Construction Manager and Architect. The Owner and the Architect, in consultation with the Construction Manager, shall thereafter agree to a corresponding change in the budget for the Cost of the Work or in the Project's scope and quality

§ 3.1.4 Structural and Environmental Tests, Surveys and Reports. During the Preconstruction Phase, when requested by the Construction Manager, the Owner shall furnish the following information or services with reasonable promptness The Owner shall also furnish any other information or services under the Owner's control and relevant to the Construction Manager's performance of the Work with reasonable promptness after receiving the Construction Manager's written request for such information or services. The Construction Manager shall be entitled to rely on the accuracy of information and services furnished by the Owner but shall exercise proper precautions relating to the safe performance of the Work.

§ 3.1.4.1 The Owner shall furnish tests, inspections and reports required by law and as otherwise agreed to by the parties, such as structural, mechanical, and chemical tests, tests for air and water pollution, and tests for hazardous materials

§ 3.1.4.2 The Owner shall furnish surveys describing physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site The surveys and legal information shall include, as applicable, grades and lines of streets, alleys, pavements and adjoining property and structures; designated wetlands; adjacent drainage, rights-of-way, restrictions, easements, encroachments, zoning, deed restrictions, boundaries and contours of the site, locations, dimensions and necessary data with respect to existing buildings, other improvements and trees, and information concerning available utility services and lines, both public and private, above and below grade, including inverts and depths. All the information on the survey shall be referenced to a Project benchmark

§ 3.1.4.3 The Owner, when such services are requested, shall furnish services of geotechnical engineers, which may include but are not limited to test borings, test pits, determinations of soil bearing values, percolation tests, evaluations of hazardous materials, seismic evaluation, ground corrosion tests and resistivity tests, including necessary operations for anticipating subsoil conditions, with written reports and appropriate recommendations.

§ 3.1.4.4 During the Construction Phase, the Owner shall furnish information or services required of the Owner by the Contract Documents with reasonable promptness. The Owner shall also furnish any other information or services under the Owner's control and relevant to the Construction Manager's performance of the Work with reasonable promptness after receiving the Construction Manager's written request for such information or services

#### § 3.2 Owner's Designated Representative

The Owner shall identify a representative authorized to act on behalf of the Owner with respect to the Project The Owner's representative shall render decisions promptly and furnish information expeditiously, so as to avoid

ATA Document Al36" - 2008 [fermerly Al31"CMC - 2003]. Copyright > 1994, 2003 and 2009 by The American Institute of Architects. All rights reserved WANNING: This ATA Document is protected by US Copyright iaw and International Tratics. Unatherized reproduction of distribution of this ATA Document, or any portion of it, may result in severe civil and ariminal ponalties, and will be presecuted to the 12 maximum extant possible under the law. This diract was produced by ATA software at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market at 11.31:37 on 12/17/2009 under Order No.6966126610\_1 which Var Market a (1446537284)

unreasonable delay in the services or Work of the Construction Manager Except as otherwise provided in Section 4.2.1 of A201-2007, the Architect does not have such authority. The term "Owner" means the Owner or the Owner's authorized representative.

§ 3.2.1 Legal Requirements. The Owner shall furnish all legal, insurance and accounting services, including auditing services, that may be reasonably necessary at any time for the Project to meet the Owner's needs and interests

#### § 3.3 Architect

The Owner shall retain an Architect to provide services, duties and responsibilities as described in ALA Document B103™ 2007, Standard Form of Agreement Between Owner and Architect, including any additional services requested by the Construction Manager that are necessary for the Preconstruction and Construction Phase services under this Agreement. The Owner shall provide the Construction Manager a copy of the executed agreement between the Owner and the Architect, and any further modifications to the agreement.

ARTICLE 4 COMPENSATION AND PAYMENTS FOR PRECONSTRUCTION PHASE SERVICES § 4.1 Compensation

§ 4.1.1 For the Construction Manager's Preconstruction Phase services, tThe Owner shall compensate the Construction Manager as follows:

The Construction Manager will be compensated for all items depicted on the Cost Assignment (attached hereto as Attachment \_\_\_) as being included in "CM Fee." "CM Reimbursable." and "Direct Cost of the Work." The portions of Construction Manager's compensation shall be calculated and paid as follows:

- Construction Manager's Fee shall be \_\_\_\_\_\_percent (\_\_\_\_%) of the final cost of construction as its Construction Services Fee Compensates for overhead and profit. The Fee shall be paid in equal a monthly lump sum amount of \_\_\_\_\_\_\_ and 00/100 Dollars (\$\_\_\_\_\_\_).
- Reimbursable Expenses: Reimbursable Expenses are expenses directly incurred by Construction Manager in connection with management of on-site activities. Such expenses shall be invoiced monthly. No markup will be allowed for Reimbursable Expenses
- 3. Direct Cost of the Work as enumerated in Article 6 herein: Direct Cost of Work includes work performed by trade contractors. Construction Manager shall be compensated for the Direct Cost of Work on a monthly basis. No mark-up will be allowed on the Direct Cost of Work.

§4.1.2 For the Construction Manager's Preconstruction Phase services described in Sections 2.1 and 2.2: (Insert amount of, or basis for, compensation and include a list of reimbursable cost items, as applicable.)

§4.1.3 If the Preconstruction Phase services covered by this Agreement have not been completed within «-» ( «-» ) months of the date of this Agreement, through no fault of the Construction Manager, the Construction Manager's compensation for Preconstruction Phase services shall be equitably adjusted.

§ 4.1.4 Compensation based on Direct Personnel Expense includes the direct salaries of the Construction Manager's personnel providing Preconstruction Phase services and the Construction Manager's costs for the mandatory and customary contributions and benefits related thereto, such as employment taxes and other statutory employee benefits, insurance, sick leave, holidays, vacations, employee retirement plans and similar contributions.

## § 4.2 Payments

§ 4.2.1 Unless otherwise agreed, payments for services shall be made monthly in proportion to services performed

§ 4.2.2 Payments are due and payable upon thirty days following presentation of the Construction Manager's invoice, unless Owner properly objects to the invoice, or portion thereof within said thirty days. All interest on improperly paid invoices shall be calculated in accordance with the Illinois Local Government Prompt Payment Act, Amounts unpaid (a) (in (a) days after the invoice date shall be ar interest at the rate entered below, or in the

ALA Document Al34" - 2005 (formerly Al31"CMs - 2003). Copyright C 1594, 2003 and 2009 by The American Institute of Architects. All rights reserved. WARNING. This ALA' Document is protected by U.S. Copyright Law and International Treaties. Unauthorized teproduction or distribution of this ALA' Document, or any portion of it, may result in sovere civil and criminal ponalties, and will be presecuted to the maximum oxight possible under the law. This draft was produced by ALA software at 11:31:37 on 12/17/2009 under Order No 6946126201 which espires on 00/16/2010, and is not for resale. (1466337284)

Formatted: Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.25" + Tab after: 0.5" + Indent at: 0.5"

Formatted: Tab stops: Not at 0.5"

Manager (Insert rate of monthly or annual inte	rest agreed upon.)
<del>«~» % « »</del>	
N - 7 7 0 N 2	
ARTICLE 5 COMPENSATION FOR CO	
5.1 For the Construction Manager's Construction Manager the Contract S	performance of the Work as described in Section 2.3, the Owner shall-pay the tum in current funds for the Construction Manager's performance of the
Contract. The Contract Sum is the Co	on an enterne target of the construction Manager's performance of the ost of the Work as defined in Section 61.1 plus the Construction Manager's
Fee	
5.1.1-The Construction Manager's F	lan.
	t of the Work or other provision for determining the Construction Manager's
Fee.)	2 · · · · · · · · · · · · · · · · · · ·
<del></del>	
5.1.2 The method of adjustment of the	he Construction Manager's Fee for changes in the Work
<del>5.1.3 Limitations, if any, on a Subco Vork-</del>	<del>ntractor's overhead and profit for increases in the cost of its portion of th</del> e
landard rate paid at the place of the P 5.1.5 Unit prices, if any.	
landard rate paid at the place of the P 5.1.5 Unit prices, if any.	
landard rate paid at the place of the P 5.1.5 Unit prices, if any.	roject.
tandard rate paid at the place of the P 5.1.6 Unit prices, if any. Identify and state the unit price; state	roject. The quantity limitations, if any, to which the unit price will be applicable.)
<del>landard rate paid at the place of the P.</del> <del>5.1.5 Unit prices, if any.</del> I <del>dentify and state the unit price: state</del> <del>Itom</del>	roject. The quantity limitations, if any, to which the unit price will be applicable.)
tendard rate paid at the place of the P 5.1.6 Unit prices, if any. Identify and state the unit price; state tom 5.3 Changes in the Work 5.3.1 The Owner may, without invalue	the quantity limitations, if any, to which the unit price will be applicable.) Units and Limitations Price per Unit (\$0.00) dating the Contract, order changes in the Work within the general scope of the
tendard rate paid at the place of the P 5.1.5 Unit prices, if any. Identify and state the unit price; state tem 5.3 Changes in the Work 5.3.1 The Owner may, without invalie ontract consisting of additions, deleti	the quantity limitations, if any, to which the unit price will be applicable.) Units and Limitations Price per Unit (\$0.00) dating the Contract, order changes in the Work within the general scope of the
tendard rate paid at the place of the P 5.1.5 Unit prices, if any. Identify and state the unit price; state Item 5.3 Changes in the Work 5.3.1 The Owner may, without invalia ontract consisting of additions, deleti rehitect may make minor changes in	the quantity limitations, if any, to which the unit price will be applicable.) Units and Limitations Price per Unit (\$0.00) dating the Contract, order changes in the Work within the general scope of the
tendard rate paid at the place of the P 5.1.5 Unit prices, if any. Identify and state the unit price; state Item 5.3 Changes in the Work 5.3.1 The Owner may, without invalia ontract consisting of additions, deleti rehitect may make minor changes in	Angest. The quantity limitations, if any, to which the unit price will be applicable.) Units and Limitations Price per Unit (\$0.00) dating the Contract, order changes in the Work within the general scope of the ions or other revisions. The Owner shall issue such changes in writing. The the Work as provided in Section 7.4 of AIA Document A201–2007. General ction. The Construction Manager shall be entitled to an equitable adjustment
tendard rate paid at the place of the P 5.1.5 Unit prices, if any. Identify and state the unit price; state tom 5.3 Changes in the Work 5.3.1 The Owner may, without invalid ontract consisting of additions, deleti rehitect may make minor changes in onditions of the Contract for Constru- the Contract Time as a result of char	And the second s
tandard rate paid at the place of the P 5.1.5 Unit prices, if any. Identify and state the unit price; state tom 5.3 Changes in the Work 5.3.1 The Owner may, without invalid ontract consisting of additions, deleti rehitect may make minor changes in onditions of the Contract for Constru- the Contract Time as a result of char 5.3.2 Adjustments to the Construction	Angest. The quantity limitations, if any, to which the unit price will be applicable.) Units and Limitations Price per Unit (\$0.00) dating the Contract, order changes in the Work within the general scope of the ions or other revisions. The Owner shall issue such changes in writing. The the Work as provided in Section 7.4 of AIA Document A201–2007. General ction. The Construction Manager shall be entitled to an equitable adjustment
tandard rate paid at the place of the P 5.1.5 Unit prices, if any. Identify and state the unit price; state tom 5.3 Changes in the Work 5.3.1 The Owner may, without invalid ontract consisting of additions, deleti rehitect may make minor changes in onditions of the Contract for Constru- the Contract Time as a result of char 5.3.2 Adjustments to the Construction	Angest. The quantity limitations, if any, to which the unit price will be applicable.) Units and Limitations Price por Unit (\$0.00) dating the Contract, order changes in the Work within the general scope of the ions or other revisions. The Owner shall issue such changes in writing. The the Work as provided in Section 7.4 of AIA Document A201–2007. General ction. The Construction Manager shall be entitled to an equitable adjustment nges in the Work. 1 Manager's compensation on account of changes in the Work may be
tendard rate paid at the place of the P 5.1.6 Unit prices, if any. Identify and state the unit price; state itom 5.3 Changes in the Work 5.3.1 The Owner may, without invalid ontract consisting of additions, deleti- rehitect may make minor changes in onditions of the Contract for Constru- the Contract Time as a result of char 5.3.2 Adjustments to the Construction termined by any of the methods listed ontract for Construction.	Aroject. The quantity limitations, if any, to which the unit price will be applicable.) Units and Limitations Price per Unit (\$0.00) dating the Contract, order changes in the Work within the general scope of the ions or other revisions. The Owner shall issue such changes in writing. The the Work as provided in Section 7.4 of AIA Document A201–2007. General ction. The Construction Manager shall be entitled to an equitable adjustment nges in the Work. I Manager's compensation on account of changes in the Work may be ad in Section 7.3.3 of AIA Document A201–2007. General Conditions of the
tendard rate paid at the place of the P 5.1.5 Unit prices, if any. Identify and state the unit price; state tom 5.3 Changes in the Work 5.3.1 The Owner may, without invalid ontract consisting of additions, deleti- rehitect may make minor changes in onditions of the Contract for Constru- the Contract Time as a result of char 5.3.2 Adjustments to the Construction termined by any of the methods listed ontract for Construction. 5.3.5 If no specific provision is made changes in the Work, or if the extent	And the section 5.1.2 for adjustment of the Construction Manager's Fee in the case in Section 5.1.2 for adjustment of the Construction Manager's Fee in the case in Section 5.1.2 for adjustment of the Construction Manager's Fee in the case in Section 5.1.2 for adjustment of the Construction Manager's Fee in the case in Section 5.1.2 for adjustment of the Construction Manager's Fee in the case in Section 5.1.2 for adjustment of the Construction Manager's Fee in the case in Section 5.1.2 for adjustment of the Construction Manager's Fee in the case in Section 5.1.2 for adjustment of the Construction Manager's Fee in the case
tendard rate paid at the place of the P     5.1.5 Unit prices, if any.     Identify and state the unit price; state     Item     S.3 Changes in the Work     5.3.1 The Owner may, without invalid     ontract consisting of additions, deleti     rehitect may make minor changes in     onditions of the Contract for Constru-     the Contract Time as a result of char     5.3.2 Adjustments to the Construction     termined by any of the methods liste     ontract for Construction.     5.3.5 If no specific provision is made     (changes in the Work, or if the extent     ovisions of Section 5, 1, 2 will cause s	the quantity limitations, if any, to which the unit price will be applicable.) Units and Limitations Price per Unit (\$0.00)  dating the Contract, order changes in the Work within the general scope of the ions or other revisions. The Owner shall issue such changes in writing. The the Work as provided in Section 7.4 of AIA Document A201–2007, General ction. The Construction Manager shall be entitled to an equitable adjustment nges in the Work.  Manager's compensation on account of changes in the Work may be ad in Section 7.3.3 of AIA Document A201–2007, General Conditions of the in Section 5.1.2 for adjustment of the Construction Manager's Fee in the case is of such changes is such, in the aggregate, that application of the adjustment ubstantial inequity to the Owner or Construction Manager, the Construction
<b>5.1.5</b> Unit prices, if any. <b>State of the Price of the Pri</b>	Adding the Contract, order changes in the Work within the general scope of the Units and Limitations Price per Unit (\$0.00) dating the Contract, order changes in the Work within the general scope of the ions or other revisions. The Owner shall issue such changes in writing. The the Work as provided in Section 7.4 of AIA Document A201–2007, General ction. The Construction Manager shall be entitled to an equitable adjustment nges in the Work. 1 Manager's compensation on account of changes in the Work may be ad in Section 7.3.3 of AIA Document A201–2007, General Conditions of the in Section 5.1.2 for adjustment of the Construction Manager's Fee in the case is such, in the aggregate, that application of the adjustment ubstantial inequity to the Owner or Construction Manager. the Construction ted on the same basis that was used to establish the Fee for the original Work.
tendard rate paid at the place of the P     5.1.5 Unit prices, if any.     Identify and state the unit price; state     Item     S.3 Changes in the Work     5.3.1 The Owner may, without invalid     ontract consisting of additions, deleti     rehitect may make minor changes in     onditions of the Contract for Constru-     the Contract Time as a result of char     5.3.2 Adjustments to the Construction     termined by any of the methods liste     ontract for Construction.     5.3.5 If no specific provision is made     (changes in the Work, or if the extent     ovisions of Section 5, 1, 2 will cause s	Adding the Contract, order changes in the Work within the general scope of the Units and Limitations Price per Unit (\$0.00) dating the Contract, order changes in the Work within the general scope of the ions or other revisions. The Owner shall issue such changes in writing. The the Work as provided in Section 7.4 of AIA Document A201–2007, General ction. The Construction Manager shall be entitled to an equitable adjustment nges in the Work. 1 Manager's compensation on account of changes in the Work may be ad in Section 7.3.3 of AIA Document A201–2007, General Conditions of the in Section 5.1.2 for adjustment of the Construction Manager's Fee in the case is such, in the aggregate, that application of the adjustment ubstantial inequity to the Owner or Construction Manager. the Construction ted on the same basis that was used to establish the Fee for the original Work.
tendard rate paid at the place of the P 5.1.5 Unit prices, if any. Identify and state the unit price; state term 5.3 Changes in the Work 5.3.1 The Owner may, without invalue ontract consisting of additions, deleti rehitect may make minor changes in onditions of the Contract for Constru- the Contract Time as a result of char 5.3.2 Adjustments to the Construction termined by any of the methods liste ontract for Construction. 5.3.5 If no specific provision is made changes in the Work, or if the extent ovisions of Section 5.1.2 will cause s anager's Fee shall be equitably adjust d the Guaranteed Maximum Price sha 5.2.1 The Owner may, without invalid 5.3.1 The Owner may, without invalid 5.3.2 Changes in the Work 5.3.1 The Owner may, without invalid 5.3.1 The Owner may, without invalid 5.3.2 Changes in the Work 5.3.1 The Owner may, without invalid 5.3.1 The Owner may, without invalid 5.3.2 Changes in the Work 5.3.1 The Owner may, without invalid 5.3.1 The Owner may, without invalid 5.3.2 Changes in the Work 5.3.1 The Owner may, without invalid 5.3.1 The Owner may, without invalid 5.3.2 Changes in the Work 5.3.1 The Owner may, without invalid 5.3.1 The Owner may, without invalid 5.3.2 Changes in the Work 5.3.2 Changes in the Work 5.3.3 Changes in the Work 5.3.4 The Owner may, without invalid 5.3.3 Changes in the Work 5.3.4 The Owner may, without invalid 5.3.3 Changes in the Work 5.3.4 The Owner may, without invalid 5.3.3 Changes in the Work 5.3.4 The Owner may, without invalid 5.3.4 The Owner may, without invalid 5.3.4 The Owner may, without invalid 5.3.5 Changes in the Work 5.3.5 Ch	Integration       Image: State S
tendard rate paid at the place of the P 5.1.5 Unit prices, if any. Identify and state the unit price; state term 5.3 Changes in the Work 5.3.1 The Owner may, without invalue ontract consisting of additions, deleti rehitect may make minor changes in onditions of the Contract for Constru- the Contract Time as a result of char 5.3.2 Adjustments to the Construction termined by any of the methods liste ontract for Construction. 5.3.5 If no specific provision is made changes in the Work, or if the extent ovisions of Section 5.1.2 will cause s anager's Fee shall be equitably adjust d the Guaranteed Maximum Price sha 5.2.1 The Owner may, without invalid mtract consisting of additions, deletion 5.3.5 If no specific provision is made changes in the Work, or if the extent ovisions of Section 5.1.2 will cause s anager's Fee shall be equitably adjust d the Guaranteed Maximum Price sha 5.2.1 The Owner may, without invalid mtract consisting of additions, deletion	Integration of the same basis that was used to establish the Fee for the original Work, all be applied by the same basis that was used to establish the Fee for the original Work, all be applied by the same basis that was used to establish the Fee for the original Work, all be applied by the work of the same basis that was used to establish the Fee for the original Work, all be applied to the original work, all be applied to the original work, all be applied to the original the original the original the original the original the work and the original the
Sandard rate paid at the place of the Pl	Integration       Image: State S

accestvad. MARNING: This ALA Document is protected by U.S. Copyright Law and International Treaties; Unartherised repredaction or distribution of this ALA Document; or any portion of it. may result in severe civil and criminal possible under the law This draft was produced by AIA software at 11:31:37 on 12/17/2009 under Order No 694612602.0 which empires on 00/16/2010, and is not for resals User Notes: (1446537204)

in the Contract Time as a result of changes in the Work. The Construction Manager shall incorporate all changes in the Work and Contract Time as separate entries in the Control Estimate.

§ 5.2.2 Increased costs for the items set forth in Sections 6.1 through 6.7 that result from changes in the Work shall become part of the Cost of the Work, and the Construction Manager's Fee shall be adjusted as provided in Section 5.1.2

§ 5.2.3 If the Construction Manager receives any Drawings, Specifications, interpretations or instructions from the Owner or Architect which are inconsistent with the Contract Documents, or encounters manticipated conditions, any of which will result in a significant change in the Cost of the Work or estimated date of Substantial Completion m comparison with the Control Estimate, the Construction Manager shall promptly notify the Owner and Architect in writing and shall not proceed with the affected Work until the Construction Manager receives further written instructions from the Owner and Architect.

§5.2.4 If no specific provision is made in Section 5.1.2 for adjustment of the Construction Manager's Fee in the case of changes in the Work, or if the extent of such changes is such, in the aggregate, that application of the adjustment provisions of Section 5.1.2 will cause substantial inequity to the Owner or Construction Manager, the Construction Manager's Fee shall be equitably adjusted on the same basis that was used to establish the Fee for the original Work.

#### ARTICLE 6 DIRECT COST OF THE WORK FOR CONSTRUCTION PHASE § 6.1 Costs to Be Reimbursed

§ 6.1.1 The terms "Direct Cost of Work" or "Cost of the Work" shall mean the Construction Manager's Compensation for the proper performance of trade Subcontracts assigned to Construction Manager after public bid. The term Cost of the Work shall mean costs necessarily incurred by the Construction Manager in the proper performance of the Work. Such costs shall be at rates not higher than the standard paid at the place of the Project except with prior consent of the Owner. The Cost of the Work shall include only the items set forth in Sections 6.1

§ 6.1.2 Where any cost is subject to the Owner's prior approval, the Construction Manager shall obtain this approval prior to incurring the cost

## § 6.2 Labor Costs

through 6.7.

5 6.2.1 Wages of construction workers directly employed by the Construction Manager to perform the construction of the Work at the site or, with the Owner's prior approval, at off-site workshops

§ 6.2.2 Wages or salaries of the Construction Manager's supervisory and administrative personnel when stationed at the site with the Owner's prior approval.

(If it is intended that the wages or salaries of certain personnel stationed at the Construction Manager's principal or other offices shall be included in the Cost of the Work, identify in Section 11.5, the personnal to be included, whether for all or only part of their time, and the rates at which their time will be charged to the Work.)

§ 6.2.3 Wages and salaries of the Construction Manager's supervisory or administrative personnel engoged at factories, workshops or on the road, in expediting the production or transportation of materials or equipment required for the Work, but only for that portion of their time required for the Work.

§ 6.2.4 Costs paid or incurred by the Construction Manager for taxes, insurance, contributions, assessments and benefits required by law or collective bargaining agreements and, for personnel not covered by such agreements, customary benefits such as sick leave, medical and health benefits, holidays, vacations and pensions, provided such costs are based on wages and salaries included in the Cost of the Work under Sections 6.2.1 through 6.2.3.

§ 6.2.5 Bonuses, profit sharing, incentive compensation and any other discretionary payments paid to anyone hired by the Construction Manager or paid to any Subcontractor or vendor, with the Owner's prior approval.

#### § 6.3 Subcontract Costs

Payments made by the Construction Manager to Subcontractors and suppliers in accordance with the requirements of the subcontracts shall be reimbursed. In turn, reimbursements from Subcontractors shall be credited to the Owner Payments made by the Construction Manager to Subcontractors in accordance with the requirements of the subcontracts

AIR Docement Al34" - 2005 (formerly Al31"CMC - 2003). Copyright C 1994, 2003 and 2009 by The American Institute of Architects All sights reserved. WARKING: This AIR Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIR Document, or any portion of it, may result in severe civil and criminal penalties, and will be presecuted to the paramemerkent possible under the law. This draft was produced by AIR software at 11:31:37 on 12/17/2009 under Order No.65426501 which sepires on 00/16/2010, and is not for resale (1446537204)

#### § 6.4 Costs of Materials and Equipment Incorporated in the Completed Construction §5.4.1 Costs, including transportation and storage, of materials and equipment incorporated or to be incorporated in the completed construction

§ 5.4.2 Costs of materials described in the preceding Section 6.4.1 in excess of those actually installed to allow for reasonable waste and spoilage. Unused excess materials, if any, shall become the Owner's property at the completion of the Work or, at the Owner's option, shall be sold by the Construction Manager. Any amounts realized from such sales shall be credited to the Owner as a doduction from the Cost of the Work-

# § 6.5 Costs of Other Materials and Equipment, Temporary Facilities and Related Kems

§ 6.5.1 Costs of transportation, storage, installation, maintenance, dismantling and removal of materials, supplies, temporary facilities; machinery, equipment and hand tools not customarily owned by construction workers that are provided by the Construction Manager at the site and fully consumed in the performance of the Work. Costs of materials, supplies, temporary facilities, machinery, equipment and tools that are not fully consumed shall be based on the cost or value of the item at the time it is first used on the Project site less the value of the item when it is no longer used at the Project site. Costs for items not fully consumed by the Construction Manager shall mean fair market value.

§ 6.5.2 Rental charges for temporary facilities, machinery, equipment and hand tools not customarily owned by construction workers that are provided by the Construction Manager at the site and costs of transportation. installation, minor repairs, dismantling and removal. The total rental cost of any Construction Manager owned stem may not exceed the purchase price of any comparable item. Rates of Construction Manager owned equipment and quantities of equipment shall be subject to the Owner's prior approval-

§ 5.5.3 Costs of removal of debris from the site of the Work and its proper and legal disposal.

§ 6.6.4 Costs of document reproductions, facsimile transmissions and long distance telephone calls, postage and parcel delivery charges, telephone service at the site and reasonable petty eash expenses of the site office.

§ 6.5.5 That portion of the reasonable expenses of the Construction Manager's supervisory or administrative personnel incurred while traveling in discharge of duties connected with the Work-

§ 6.5.6 Costs of materials and equipment suitably stored off the site at a mutually acceptable location, subject to the **Owner's prior approval.** 

#### § 6.6 Miscellaneous Reimbursable Costs

§ 6.6.1 Premiums for that portion of insurance and bonds required by the Contract Documents that can be directly attributed to this Contract. Self-insurance for either full or partial amounts of the coverages required by the Contract Documents, with the Owner's prior approval.

§ 6.6.2 Sales, use or similar taxes imposed by a governmental authority that are related to the Work and for which the Construction Manager is liable.

§ 6.6.3 Fees and assessments for the building permit and for other permits, licenses and inspections for which the Construction Manager is required by the Contract Documents to pay

§ 6.6.4 Fees of laboratories for tests required by the Contract Documents, except those related to defective or nonconforming Work for which reimbursement is excluded by Section 13.5 3 of AIA Document A201-2007 or by other provisions of the Contract Documents, and which do not fall within the scope of Section 6.7.3.

§ 6.6.5 Royalties and license fees paid for the use of a particular design, process or product required by the Contract Decuments; the cost of defending suits or claims for infringement of patent rights arising from such requirement of the Contract Documents; and payments made in accordance with legal judgments against the Construction Manager resulting from such suits or claims and payments of settlements made with the Owner's consent. However, such costs of legal defenses, judgments and sattlements shall not be included in the calculation of the Construction Manager's Fee. If such royalties, fees and costs are excluded by the last sentence of Section 3.17 of AIA Document A201 2007 or other provisions of the Contract Documents, then they shall not be included in the Cost of the Work

remarked. HANNING: This ALX. Document is protected by U.S. Copyright I 1994, 2003 and 2008 by The American Institute of Architects. All rights distribution of this ALX. Document, or any potics of it, may result in severe oval and criminal ponalizes, and will be presecuted to the marking ortent possible under the law. This draft was produced by ALX software at 11.31:37 on 12/17/2009 under Order No.6946126610\_1 which espines on 00/16/20100, and is not for resale.

§ 6.6.6 Costs for electronic equipment and software, directly related to the Work with the Owner's prior approval-

\$6.6.7 Deposits lost for causes other than the Construction Manager's negligence or failure to fulfill a specific responsibility in the Contract Documents.

§ 6.6.8 Legal, mediation and arbitration costs, including attorneys' fees, other than those arising from disputes between the Owner and Construction Manager, or arising due to the fault of Construction Manager, reasonably incurred by the Construction Manager after the execution of this Agreement in the performance of the Work and with the Owner's prior approval, which shall not be unreasonably withheld. Construction Manager shall obtain approval for expenditures for legal services prior to incurring expenses. At Owner's option, the Construction Manager shall assign its rights and liabilities in any legal action relating to this Agreement to the Owner, Legal mediation and arbitration costs; including attorneys' fees, other than those arising from disputes between the Owner and Construction Manager, reasonably incurred by the Construction Manager after the execution of this Agreement in the performance of the Work and with the Owner's prior approval, which shall not be unreasonably withheld.

§ 6.6.9 Subject to the Owner's prior approval, expenses incurred in accordance with the Construction Manager's standard written personnel policy for relocation and temporary living allowances of the Construction Manager's personnol-required for the Work.

## § 6.7 Other Costs and Emergencies

§ 6.7.1 Other costs incurred in the performance of the Work if, and to the extent, approved by Change Order with the Owner Other costs incurred in the performance of the Work if, and to the extent, approved in advance in writing by the Owner.

§ 6.7.2 Costs incurred in taking action to prevent threatened damage, injury or loss in case of an emergency affecting, the safety of persons and property, as provided in Section 10.4 of AIA Document A201-2007.

§ 6.7.3 Costs of repairing or correcting damaged or nonconforming Work executed by the Construction Manager, Subcontractors or suppliers, provided that such damaged or nonconforming Work was not caused by negligence or failure to fulfill a specific responsibility of the Construction Manager and only to the extent that the cost of repair or correction is not recovered by the Construction Manager from insurance, sureties, Subcontractors, suppliers, or others

§ 6.7.4 The costs described in Sections 6.1 through 6.7 shall be included in the Cost of the Work notwithstanding any provision of AIA Document A201-2007 or other Conditions of the Contract which may require the Construction Manager to pay such costs, unless such costs are excluded by the provisions of Section 6.8

#### 5.8 Costs Not to Be Reimbursed

§ 6.8.1 The Cost of the Work shall not include the items listed below:

- Salaries and other compensation of the Construction Manager's personnel stationed at the Construction Manager's principal office or offices other than the site office, except as specifically provided in Section 6.2, or as may be provided in Article 11;
- Expenses of the Construction Manager's principal office and offices other than the site office;
- Overhead and general expenses, except as may be expressly included in Sections 6.1 through 6.7.
- The Construction Manager's capital expenses, including interest on the Construction Manager's capital employed for the Work;
- Except as provided in Section 6.7.3 of this Agreement, costs due to the negligence or failure of the Construction Manager, Subcontractors and suppliers or anyone directly or indirectly employed by any of them or for whose acts any of them may be liable to fulfill a specific responsibility of the Contract,
- Any cost not specifically and expressly described in Sections 6.1 through 6.7; and
- Costs for services incurred during the Preconstruction Phase.

#### § 6.9 Discounts, Rebates and Refunds

 $\frac{1}{5}$  6.9.1 Cash discounts obtained on payments made by the Construction Manager shall accrue to the Owner if (1) before making the payment, the Construction Manager included them in an Application for Payment and received payment from the Owner, or (2) the Owner has deposited funds with the Construction Manager with which to make payments; otherwise, cash discounts shall accrue to the Construction Manager Trade discounts, rebates, refunds and

ATA Document Alle" - 2005 (formarly Alle"CHC - 2003). Copyright C 1994, 2003 and 2009 by The American Institute of Architects All rights ATA Document Also " dues insumity mass one - aver, copragnet due and International Treaties. Unauthorized reproduction or reserved HARNING: This AIA Document, or any portion of it, may result in severe civil and criminal penaltime, and will be presented to the 17 maximum extent possible under the law. This draft was produced by AIA software at 12:31:37 on 12/17/2009 under Order No.6946126610 1 which expires on 08/16/2010, and is not for resale. User Notes (1446517284)

amounts received from sales of surplus materials and equipment shall accrue to the Owner, and the Construction Manager shall make provisions so that they can be obtained

§ 6.9.2 Amounts that accrue to the Owner in accordance with the provisions of Section 6 9.1 shall be credited to the Owner as a deduction from the Cost of the Work.

#### § 6.10 Related Party Transactions

§ 6.10.1 For purposes of Section 6.10, the term "related party" shall mean a parent, subsidiary, affiliate or other entity having common ownership or management with the Construction Manager, any entity in which any stoekholder in, or management employce of, the Construction Manager owns any interest in excess of ten pero the aggregate, or any person or entity which has the right-to control the business or affairs of the Construction Manager. The term "related party" includes any member of the immediate family of any person identified above.

§ 6.10.2 If any of the costs to be reimbursed arise from a transaction between the Construction Manager and a related party, the Construction Manager shall notify the Owner of the specific nature of the contemplated transaction, including the identity of the related party and the anticipated cost to be incurred, before any such transaction is consummated or cost incurred. If the Owner, after such notification, authorizes the proposed transaction, then the cost incurred shall be included as a cost to be reimbursed, and the Construction Manager shall procure the Workequipment, goods or service from the related party, as a Subcontractor, according to the terms of Sections 2.3.2.1, 23.2.2 and 2.3.2.3. If the Owner fails to authorize the transaction, the Construction Manager shall produte the Work, equipment, goods or service from some person or entity other than a related party according to the terms of Sections 2.3.2.1, 2.3.2.2 and 2.3.2.3.

## § 6.11 Accounting Records

The Construction Manager shall keep full and detailed records and accounts related to the cost of the Work and exercise such controls as may be necessary for proper financial management under this Contract and to substantiate all costs incurred. The accounting and control systems shall be satisfactory to the Owner. The Owner and the Owner's auditors shall, during regular business hours and upon reasonable notice, be afforded access to, and shall be permitted to audit and copy, the Construction Manager's records and accounts, including complete documentation supporting accounting entries, books, correspondence, instructions, drawings, receipts, subcontracts, Subcontractor's proposals, purchase orders, vouchers, memoranda and other data relating to this Contract. The Construction Manager shall preserve these records for a period of three years after final payment, or for such longer period as may be required by law.

ARTICLE 7 PAYMENTS FOR CONSTRUCTION PHASE SERVICES

# § 7.1 Progress Payments

§ 7.1.1 Based upon Applications for Payment submitted to the Architect by the Construction Manager and Certificates for Payment issued by the Architect, the Owner shall make progress payments on account of the Contract Sum to the Construction Manager as provided below and elsewhere in the Contract Documents.

§7.1.1.1 The Owner shall withhold from each payment, except Final Payment, ten percent of Construction Manager's Fee as retention. All of retention shall be paid as part of Final Payment

§7.1.2 The period covered by each Application for Payment shall be one calendar month ending on the last day of the month, or as follows

## **« »**

§ 7.1.3 Upon receipt of a properly submitted Application for payment, payment shall be made as provided in the Illinois Local Government Prompt Payment Act, Provided that an Application for Payment is received by the Architect not later than the « -- day of a month, the Owner shall make payment of the certified amount to the Construction Manager not later than the «-- and a soft the -- month. If an Application for Payment is received by the Architect after the application date fixed above, payment shall be made by the Owner not later than «-» ( « -» ) days after the Architect receives the Application for Payment.

(Federal, state or local laws may require payment within a certain period of time.)

§7.1.4 With each Application for Payment, the Construction Manager shall submit payrolls, petty each accounts, receipted invoices or invoices with cheek vouchers attached, and any other evidence required by the Owner or

AIA Document Alls" - 2009 (fermarly Alls"nee - 2003). Copyright 6 1994, 2003 and 2009 by the American Institute of Architects. All rights resarved. WANNING: This AIA: Document is protected by U.S. Copyright Law and International Treaties: Unarthorized repredention or distribution of this AIA: Document, or any portion of it. may robult an severe oivil and criminal ponalties, and will be presented to the maximum extent possible under the law. This draft was produced by AIA software at 11:31:37 on 12/17/2009 under Order No.696612680\_1 which expires On 03/16/2010, and is not for resale. User Notes: (1446537284)

Architect to demonstrate that each disbursements already made by the Construction Manager on account of the Cost of the Work equal or exceed progress payments already received by the Construction Manager, less that portion of those payments attributable to the Construction Manager's Fee, plus payrolls for the period covered by the present Application for Payment-

§ 7.1.5 Each Application for Payment shall be based on the most recent schedule of values submitted by the Construction Manager in accordance with the Contract Documents. The schedule of values shall allocate the entire Cost of the Work, except that the Construction Manager's Fee and items to be reimbursed shall be shown as single separate items. The schedule of values shall be prepared in such form and supported by such data to substantiate its accuracy as the Architect may require. This schedule, unless objected to by the Architect, shall be used as a basis for reviewing the Construction Manager's Applications for Payment Applications for Payment shall show the Cost of the Work actually incurred by the Construction Manager through the end of the period covered by the Application for Payment and for which the Construction Manager has made or intends to make actual payment prior to the next Application for Payment

§ 7.1.6 Applications for Payment shall show the percentage of completion of each portion of the Work as of the end of the period covered by the Application for Payment. The percentage of completion shall be the percentage of that portion of the Work which has actually been completed, Subject to other provisions of the Contract Documents, the amount of each progress payment shall be computed as follows:

Take the Cost of the Work as described in Section 6.1.1;

Add the Construction Manager's Fee, less retainage of - + percent ( + + %). The Construction Manager's 2 Fee shall be computed upon the Cost of the Work described in the preceding Section 7.1.6.1 at the rate stated in Section 5.1 1: or if the Construction Manager's Fee is stated as a fixed sum in that Section, an amount that bears the same ratio to that fixed sum Fee as the Cost of the Work bears to a reasonable estimate of the probable Cost of the Work-upon its completion;

3 Subtract retainage of a spercent ( \* \* %) from that portion of the Work that the Construction Manager self-performs-

4 Subtract the aggregate of previous payments made by the Owner;

5 Subtract the shortfall, if any, indicated by the Construction Manager in the documentation required by Section 7.1.4 or resulting from errors subsequently discovered by the Owner's auditors in such documentation, and Subtract amounts, if any, for which the Architect has withheld or withdrawn a Certificate for Payment as provided in the Contract Documents

§7.1.7 The Owner and Construction Manager shall agree upon (1) a mutually acceptable procedure for review and approval of payments to Subcontractors and (2) the percentage of retainage held on Subcontracts, and the Construction Manager shall execute subcontracts in accordance with those agreements.

§ 7.1.8 The Owner and Construction Manager shall agree upon (1) a mutually acceptable procedure for review and approval of payments to Subcontractors and (2) the percentage of retainage held on Construction Manager's Fee and Subcontracts, and the Construction Manager shall execute subcontracts in accordance with those agreements. The Construction Documents shall depict that retention of ten percent shall be held from Construction Manager's Fee and from each Subcontractor Application for Payment, and such retention shall be paid as part of Final Payment,

§ 7.1.9 Except with the Owner's prior approval, the Construction Manager shall not make advance payments to suppliers for materials or equipment which have not been delivered and stored at the site.

§7.1.10 In taking action on the Construction Manager's Applications for Payment, the Architect shall be entitled to rely on the accuracy and completeness of the information furnished by the Construction Manager and shall not be deemed to represent that the Architect has made a detailed examination, audit or arithmetic verification of the documentation submitted in accordance with Section 7,1,4 or other supporting data; that the Architect has made exhaustive or continuous on-site inspections; or that the Architect has made examinations to ascertain how or for what purposes the Construction Manager has used amounts previously paid on account of the Contract. Such examinations, audits and verifications, if required by the Owner, will be performed by the Owner's auditors acting in the sole interest of the Owner.

#### §7.1.11 CONSTRUCTION MANAGER'S PAYMENT APPLICATIONS

§ 7.1.11.1 Construction Manager shall submit monthly Applications for Payment in a format acceptable to the Owner. The statements shall be fully detailed so as to allow the Owner to determine the propriety of the statement.

All Document Al3(" - 2009 (formally Al31"Cmc - 2003). Copyright ( 1994, 2003 and 2009 by The American Institute of Architects. All rights reserved. ADMINO: This AIX' Document is protected by U.S. Copyright Law and International Treaties. Unambhorized reproduction or distribution of this AIX' Document, or any portion of it, may result in severe civil and criminal ponalties, and will be preserved to the maximum outent possible under the law. This dealt was produced by AIX softwars at 11:31:37 on 12/17/2009 under Joiner No.546126810\_1 which maximum extent possible under the law. This c septres on 08/16/2010, and is not for resale User Motes. (14465)7284)

Field Code Changed

At a minimum, monthly statements shall detail the amount due for the current period, subcontractor costs, agreed change orders or modifications, previous amounts billed, reimbursable expenses, and balance of contract outstanding.

#### § 7.2 Final Payment

§7.2.1 Final payment, constituting the entire unpaid balance of the Contract Sum, shall be made by the Owner to the Construction Manager when

- .1 the Construction Manager has fully performed the Contract except for the Construction Manager's responsibility to correct Work as provided in Section 12.2.2 of AIA Document A201-2007, and to satisfy other requirements, if any, which extend beyond final payment;
- the Construction Manager has submitted a final accounting for the Cost of the Work and a final .2 Application for Payment; and
- a final Certificate for Payment has been issued by the Architect.

The Owner's final payment to the Construction Manager shall be made no later than 30 days after the issuance of the Architect's final Certificate for Payment, or as follows,

#### **(( )**

§7.1.8 Except with the Owner's prior approval, the Construction Manager shall not make advance payments to suppliers for materials or equipment which have not been delivered and stored at the site

§7.1.9 In taking action on the Construction Manager's Applications for Payment, the Architect shall be entitled to rely on the accuracy and completeness of the information furnished by the Construction Manager and shall not be deemed to represent that the Architect has made a detailed examination, audit or arithmetic verification of the documentation submitted in accordance with Section 7.1.4 or other supporting data; that the Architect has made exhaustive or continuous on site inspections; or that the Architect has made examinations to ascertain how or for what-purposes the Construction Manager has used amounts previously paid on account of the Contract. Such examinations, audits and verifications, if required by the Owner, will be performed by the Owner's auditors acting in the sole interest of the Owner-

## § 7.2 Final Payment

§7.2.1 Final payment, constituting the entire unpaid balance of the Contract Sum, shall be made by the Owner to the Construction Manager when

- .1 the Construction Manager has fully performed the Contract except for the Construction Manager's responsibility to correct Work as provided in Section 12.2.2 of AIA Document-A201-2007, and to satisfy other requirements, if any, which extend beyond final payment;
- the Construction Manager has submitted a final accounting for the Cost of the Work and a final Application for Payment; and
- a final Certificate for Payment has been issued by the Architegt.

The Owner's final payment to the Construction Manager shall be made no later than 30 days after the issuance of the Architect's final Certificate for Payment, or as follows:

#### 

§7.2.2 The Owner's auditors will review and report in writing on the Construction Manager's final accounting within 30 days after delivery of the final accounting to the Architect by the Construction Manager. Based upon such Cost of the Work as the Owner's auditors report to be substantiated by the Construction Manager's final accounting, and provided the other conditions of Section 7.2.1 have been met, the Architect will, within seven days after receipt of the written report of the Owner's auditors, either issue to the Owner a final Certificate for Payment with a copy to the Construction Manager, or notify the Construction Manager and Owner in writing of the Architect's reasons for withholding a certificate as provided in Section 9.5.1 of the AIA Document A201-2007. The time periods stated in this Section supersede those stated in Section 9.4.1 of the AIA Document A201-2007. The Architect is not responsible for verifying the accuracy of the Construction Manager's final accounting.

§7.2.3 If the Owner's auditors report the Cost of the Work as substantiated by the Construction Manager's final accounting to be less than claimed by the Construction Manager, the Construction Manager shall be entitled to

All Dormmant Alls" - 2008 (formmarky Alls"Cdc - 2003). Copyright C 1994, 2003 and 2009 by The American Institute of Architects. All rights reserved. MANNING: This AIA Dorument is protected by U.S. Copyright Law and International Treates. Unauthorised eproduction or distribution of this AIA Dorument. or any portion of it, may result in sovere civil and criminal ponalise, and will be presented to the maximum action test possible under the law. This draft was produced by AIA software at 11:31:37 on 12/17/2009 under Order No.65461266101 which expires on 00/16/2010, and is not for resals (1446537284)

request mediation of the disputed amount without seeking an initial decision pursuant to Section 15.2 of A201– 2007. A request for mediation shall be made by the Construction Manager within 30 days after the Construction Manager's receipt of a copy of the Architect's final Certificate for Payment. Failure to request mediation within this 30 day period shall result in the substantiated amount reported by the Owner's auditors becoming binding on the Construction Manager. Pending a final resolution of the disputed amount, the Owner shall pay the Construction Manager the amount certified in the Architect's final Certificate for Payment.

§7.2.4 If, subsequent to final payment and at the Owner's request, the Construction Manager-mours costs described in Section 6.1.1 and not excluded by Section 6.8 to correct defective or nonconforming Work, the Owner shall reimburse the Construction Manager such costs and the Construction Manager's Fee applicable thereto on the same basis as if such costs had been incurred prior to final payment.

#### ARTICLE 8 INSURANCE AND BONDS

For all phases of the Project, the Construction Manager and the Owner shall purchase and maintain insurance, and the Construction Manager shall provide bonds as set forth in Article 11 of AIA Document A201-2007 (State bonding requirements, if any, and limits of liability for insurance required in Article 11 of AIA Document A201-2007.)

Type of Insurance or Bond

Limit of Liability or Bond Amount (\$9.00)

## ARTICLE 9 DISPUTE RESOLUTION

§ 9.1 Any Claim between the Owner and Construction Manager shall be resolved in accordance with the provisions set forth in this Article 9 and Article 15 of A201–2007. However, for Claims arising from or relating to the Construction Manager's Preconstruction Phase services, no decision by the Initial Decision Maker shall be required as a condition precedent to mediation or binding dispute resolution, and Section 9.3 of this Agreement shall not apply.

§ 9.2 For any Claim subject to, but not resolved by mediation pursuant to Section 15.3 of AIA Document A201-2007, the method of binding dispute resolution shall be as follows:

(Check the appropriate box. If the Owner and Construction Manager do not select a method of binding dispute resolution below, or do not subsequently agree in writing to a binding dispute resolution method other than httgation, Claims will be resolved by litigation in a court of competent jurisdiction.)

- [ C ] Arbitration pursuant to Section 15.4 of AIA Document A201-2007
- [\* ) Litigation in a court of competent jurisdiction
- [ 赵 🛛 ] Other: (Specify)

#### «As stated in General Conditions »

#### § 9.3 Initial Decision Maker

The Architect will serve as the Initial Decision Maker pursuant to Section 15.2 of AIA Document A201 2007 for Claims arising from or relating to the Construction Manager's Construction Phase services, unless the parties appoint below another individual, not a party to the Agreement, to serve as the Initial Decision Maker. (If the parties mutually agree, insert the name, address and other contact information of the Initial Decision Maker, if other than the Architect.)

\*\*--->> \*\*--->>

ARTICLE 10 TERMINATION OR SUSPENSION

§ 10.1 Termination Prior to Owner's Approval of the Control Estimate

§ 10.1.1 Prior to the Owner's approval of the Control Estimate, the The Owner may terminate this Agreement upon not less than seven days' written notice to the Construction Manager for the Owner's convenience and without cause

AIA Document Al3(\* - 2009 [formerly Al3]\*CMG - 2003). Copyright C 1994, 2003 and 2008 by The American Institute of Architects. All rights seserved. WANHING: This AIA' Document, or any portion of it, may result in severe civil and criminal peralties, unauthorized reproduction or distribution of this AIA' Document, or any portion of it, may result in severe civil and criminal peralties, and will be preserved to the maximum attent pessible under the law This dfaft was produced by AIA software at 11.31:37 on 12/17/2009 under Order No 6946126801 which espires on 00/16/2010, and is not for resule. User Botes: (1446337281) and the Construction Manager may terminate this Agreement, upon not less than seven days' written notice to the Owner, for the reasons set forth in Section 14.1.1 of A201-2007

§ 10.1.2 In the event of termination of this Agreement pursuant to Section 10 1.1, the Construction Manager shall be equitably compensated for Preconstruction Phase services performed prior to receipt of a notice of termination. In no event shall the Construction Manager's compensation under this Section exceed the compensation set forth in Section 4.1

§ 10.1.3 In the event of termination of this Agreement pursuant to Section 10.1.1, after the commencement of the Construction Phase but prior to the Owner's approval of the Control Estimate, the Owner shall pay to the Construction Manager under Section 1012 an amount calculated as follows, which amount shall be in addition to any compensation paid to the Construction Manager under Section 10.1.2:

- Take the Cost of the Work incurred by the Construction Manager to the date of termination. .1
- 2 Add the Construction Manager's Fee computed upon the Cost of the Work to the date of termination at the rate stated in Section 5.1 or, if the Construction Manager's Fee is stated as a fixed sum in that Section, an amount that bears the same ratio to that fixed-sum Fee as the Cost of the Work at the time of termination bears to a reasonable estimate of the probable Cost of the Work upon its completion, and
- .3 Subtract the aggregate of previous payments made by the Owner for Construction Phase services.

The Owner shall also pay the Construction Manager fair compensation, either by purchase or rental at the election of the Owner, for any equipment owned by the Construction Manager which the Owner elects to retain and which is not otherwise included in the Cost of the Work under Section 10 1 3 1. To the extent that the Owner elects to take legal assignment of subcontracts and purchase orders (including rental agreements), the Construction Manager shall, as a condition of receiving the payments referred to in this Article 10, execute and deliver all such papers and take all such steps, including the legal assignment of such subcontracts and other contractual rights of the Construction Manager, as the Owner may require for the purpose of fully vesting in the Owner the rights and benefits of the Construction Manager under such subcontracts or purchase orders. All Subcontracts, purchase orders and rental agreements entered into by the Construction Manager will contain provisions allowing for assignment to the Owner as described above

If the Owner accepts assignment of subcontracts, purchase orders or rental agreements as described above, the Owner will reimburse or indemnify the Construction Manager for all costs arising under the subcontract, purchase order or rental agreement if those costs would have been reimbursable as Cost of the Work if the contract had not been terminated. If the Owner chooses not to accept assignment of any subcontract, purchase order or rental agreement that would have constituted a Cost of the Work had this agreement not been terminated, the Construction Manager will terminate the subcontract, purchase order or rental agreement and the Owner will pay the Construction Manager the costs necessarily incurred by the Construction Manager because of such termination

§ 10.2 Termination Subsequent to the Owner's Approval of the Control Estimate For Cause

§ 10.2.1 The Owner may terminate this Agreement for Cause for the reasons, and in the manner set out in Section 14.2 of the General Conditions, Subsequent to the Owner's approval of the Control Estimate, the Contract may be terminated as provided in Sections 14.1.1, 14.1.2 and 14.2.1 of A201 2007. The provisions of Article 14 of A201-2007 do not otherwise apply to this Section 10.2.

§ 10:22 In the event of such termination by the Owner, the amount payable to the Construction Manager shall not exceed the amount the Construction Manager would have been entitled to receive pursuant to Sections 10.1.2 and 10.1.3 of this Agreement, less any compensation that may be awarded to the Owner pursuant to Article 9.

§ 10.2.3 In the event of such termination by the Construction Manager, the amount payable to the Construction Manager shall be in accordance with Sections 10.1.2 and 10.1.3 of this Agreement, except that the Construction Manager's Fee shall be calculated as if the Work had been fully completed by the Construction Manager, including a reasonable estimate of the Cost of the Work for Work not actually completed.

§ 10.24 In addition to the Owner's right to terminate this Agreement for cause as provided in Section 14.2.1 of A201-2007, the Owner may terminate this Agreement for convenience as provided in Section 14.4; however, the Owner shall then only pay the Construction Manager an amount calculated as follows:

ATA Document Al36" - 2005 (dummarly Al31"CMc - 2003). Copyright C 1994, 2003 and 2004 by The American Institute of Architects. All rights The second warming the second se ited to the 22 supires on 00/16/2010, and is not for resals (1446537284)

.1 — Take the Cost of the Work incurred by the Construction Manager to the date of termination.

- .2 Add the Construction Manager's Fee computed upon the Cost of the Work to the date of termination at the rate stated in Section 5.1.1 or, if the Construction Manager's Fee is stated as a fixed sum in that Section, an amount that bears the same ratio to that fixed sum Fee as the Cost of the Work at the time of termination bears to a reasonable estimate of the probable Cost of the Work upon its completion; and
- 3 Subtract the aggregate of previous payments made by the Owner-

## § 10.3 Suspension

The Work may be suspended for a period of thirty (30) days by the Owner as provided in A2017M-2007; without an equitable adjustment. Upon lifting of a suspension in excess of thirty (30) days, the fee for Construction Management services and General Conditions shall be equitably adjusted.

The Work may be suspended by the Owner as provided in Article 14 of AIA Document A201-2007. In such ease, the Control Estimate and Contract Time shall be increased as provided in Section 14.3.2 of AIA Document A201-2007, except that the term "profit" shall be understood to mean the Construction Manager's File as described in Sections 5.1 and 5.2.4 of this Agreement.

#### ARTICLE 11 MISCELLANEOUS PROVISIONS

§ 11.1 Terms in this Agreement shall have the same meaning as those in A201-2007.

§11.2 Nothing contained in this Agreement shall create a contractual relationship with or a cause of action in favor of a third party against either the Owner or Construction Manager. Owner and Construction Manager acknowledge and agree that the obligations of the Construction Manager are solely for the benefit of the Owner and not intended in any respect to benefit the Architect, Subcontractors, or any other third parties.

§ 11.3 In case of conflict between this Agreement and any other document incorporated or referenced herein, this Agreement shall prevail, followed by the General Conditions, supplementary conditions, Owner's request for qualifications or proposal from Construction Manager, the drawings and specifications, and Construction Manager's proposal.

§11.4 The Construction Manager shall endeavor to keep the project free from mechanic's liens. In the event that the Owner is required to withhold money pursuant to a notice of lien or lien against public funds due to fault of the Construction Manager, the Construction Manager shall defend and hold harmless the Owner for all costs incurred, including attorneys' fees. The Construction Manager shall ensure that a similar provision is incorporated into the Subcontracts. In the event that the funds for this Project due to Construction Manager's fault, the Construction Manager shall ensure for all costs incurred due to the lien, including attorneys' fees. Construction Manager for all resulting costs incurred due to the lien, including attorneys' fees. Construction Manager shall assure that similar language protecting the Owner and Construction Manager is included in the subcontract documents.

§11.5 Construction Manager acknowledges that this is a public works project governed by the Illinois Prevailing Wage Act. Construction Manager shall pay its laborers if any and assure the Owner that Subcontractors shall pay its laborers not less than the established prevailing rate of wages. 820 ILCS 130/1 et seq. Construction Manager shall comply with all reporting requirements of the Illinois Prevailing Wage Act. Similarly, the Construction Manager shall assure that all Subcontractors and sub-tier subcontractors comply with the reporting requirements of the Illinois Prevailing Wage Act.

§ 11.6 Construction Manager represents that it has in place a Sexual Harassment Policy in accordance with the Illinois Human Rights Act and shall assure the Owner that Trade Contractors shall have in place a Sexual Harassment Policy prior to commencement of work on the Project. 775 ILCS 5/1-105.

§ 11.7 Construction Manager represents that it does not discriminate in its hiring practices based upon race, color, religion, sex, marital status, national origin or ancestry, age, physical or mental handicap unrelated to ability, or an unfavorable discharge from military service. Construction Manager shall assure the Owner that Trade Contractors shall not discriminate as set forth in this paragraph, 775 ILCS 5/2-1053; 44 III, Admin, Code Section 750 et seq.

§ 11.8 Construction Manager represents that it is in conformance with the Drug Free Workplace Act \_30 ILCS 580/1 et seq.

AIA Document AI34" - 3009 (fermacly AI31"CMc - 2003). Copyright C 1994, 2003 and 2009 by The American Institute of Architects. All rights reserved. WANNING: This AIA' Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction of distribution of this AIA' Document, or any portion of it, may result in sovero civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 11:31:37 on 12/17/2009 under Order No 6946126610\_1 which explays on 09/16/2010, and is not for resale User Netws: (14465372204)

## §11.9 Construction Manager certifies it is not barred from contracting as a result of bid rigging or bid rotation 720 ILCS 5/33 E-11.

# § 11.102 Ownership and Use of Documents

Section 1 5 of A201-2007 shall apply to both the Preconstruction and Construction Phases

#### § 11.113 Governing Law

This Agreement shall be governed by the law of the State of Illinois Section 13.1 of A201-2007 shall apply to both the Preconstruction and Construction Phases.

## § 11.124 Assignment

The Owner and Construction Manager, respectively, bind themselves, their agents, successors, assigns and legal representatives to this Agreement. Except for the assignment of the Subcontracts to Construction Manager, neither the Owner nor the Construction Manager shall assign this Agreement without the written consent of the other. Except as provided in Section 13.2.2 of A201-2007, neither party to the Contract shall assign the Contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract. The Owner and Construction Manager, respectively, bind thomselves, their agents, successors, assigns and legal representatives to this Agreement. Neither the Owner nor the Construction Manager, respectively, bind thomselves, their agents, successors, assigns and legal representatives to this Agreement. Neither the Owner nor the Construction Manager, respectively, bind thomselves, their agents, successors, assigns and legal representatives to this Agreement. Neither the Owner nor the Construction Manager shall assign this Agreement without the written consent of the other, except that the Owner as rights and obligations under this Agreement. Except as provided in Section 13.2.2 of A201-2007, neither party to the Contract shall assign the Contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract as a whole without party shall nevertheless remain legally responsible for all obligations under the Contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract as a whole without the party shall nevertheless remain legally responsible for al

§ 11.135 Other provisions

« None »

## ARTICLE 12 SCOPE OF THE AGREEMENT

§ 12.1 This Agreement represents the entire and integrated agreement between the Owner and the Construction Manager and supersedes all prior negotiations, representations or agreements, either written or oral. This Agreement may be amended only by written instrument signed by both Owner and Construction Manager

§ 12.2 The following documents comprise the Agreement

- .1 AIA Document A134-2009, Standard Form of Agreement Between Owner and Construction Manager as Constructor where the basis of payment is the Cost of the Work Plus a Fee with a Guaranteed Maximum Price
- .2 AIA Document A201-2007, General Conditions of the Contract for Construction, as amended and attached hereto.
- .1 ALA Document A134 2009, Standard Form of Agreement Between Owner and Construction Manager as Constructor where the basis of payment is the Cost of the Work Plus a Fee without a Guaranteed Maximum Price
- .2 AIA Document A201-2007, General Conditions of the Contract for Construction
- .3 AIA Document E2017 2007, Digital Data Protocol Exhibit, if completed, or the following:
- .4 AIA Document E202™ 2008, Building Information Modeling Protocol Exhibit, if completed, or the following.

	77	

-

.5 - Other documents.

(List other documents, if any, forming part of the Agreement.)

AIA Document AI34" - 2009 (formatly AI31"CMc - 2003). Copyright & 1994, 2003 and 2009 by The American Institute of Architects Ail sights tessreed, WANNING: ns AIA Document is protocted by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA Document, or any portion of it, may result in severe civil and criminal ponalities, and will be proceeded to the maximum extent persible under the law. This draft was produced by AIA software at 11:31:37 on 12/17/2009 under Order No.6946316650 1 which septres on 00/12/2010 and is not for ressis (1446537284) This Agreement is entered into as of the day and year first written above

OWNER (Signature)

CONSTRUCTION MANAGER (Signature)

« »« » (Printed name and ntle) « »« » (Printed name and title)

ATA Document Aller - 2009 [Gournely AllInces - 2003]. Copyright - 1994, 2003 and 2009 by The American Institute of Architects All rights reserved WARHING: This AIX Document is protocled by U.S. Copyright Law and International Treates Unauthorized reproduction of the presented to the maximum extent possible under the law. This draft was produced by AIA software at 11:31:37 on 12/17/2009 under Order Wo.654612610\_1 which explicas on 08/16/2010, and is not for resale (1446537284).

AIA Document A201" - 2007

General Conditions of the Contract for Construction

for the following PROJECT: (Name and location or address) 2011 Renovation

## THE OWNER:

(Name and address) Oak Park and River Forest High School District 200 201 North Scoville Oak Park, Illinois 60302

## THE ARCHITECT:

(Name and address) Legat Architects 2015 Spring Road, Suite 175 Oak Brook, Illinois 60523

# TABLE OF ARTICLES

- 1 GENERAL PROVISIONS
- 2 OWNER
- 3 CONTRACTOR
- 4 ARCHITECT
- 5 SUBCONTRACTORS
- 6 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS
- 7 CHANGES IN THE WORK
- 8 TIME
- 9 PAYMENTS AND COMPLETION
- 10 PROTECTION OF PERSONS AND PROPERTY
- 11 INSURANCE AND BONDS
- 12 UNCOVERING AND CORRECTION OF WORK
- 13 MISCELLANEOUS PROVISIONS

ELECTRONIC COPYING of any portion of this AIA Document to another electronic file is prohibited and constitutes a violation of copyright laws as set forth in the footer of this document.

1

AIA Document A201<sup>M</sup> - \$007. Copyright # 1911, 1915, 1918, 1925, 1937, 1851, 1958, 1961, 1963, 1966, 1870, 1876, 1967, 1987 and 2007 by The American Institute of Arinityts. All rights reserved. WORNDO: This AIA Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA Document, or any portion of it, may result in source oivil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA softwase at 14 15 10 on C/15/2026 under Order No.1000314794 2 which expires on 8/17/2008, and is not for result. User Notes: (3921870803)

ADDITIONS AND DELETIONS: The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed.

This document has important legal consequences. Consultation with an attorney is ancouraged with respect to its completion or modification.

- 14 TERMINATION OR SUSPENSION OF THE CONTRACT
- 15 CLAIMS AND DISPUTES

AIA Document A201" - 2007. Copyright © 1911, 1915, 1918, 1923, 1937, 1931, 1958, 1961, 1965, 1970, 1974, 1987, 1987 and 3007 by Tre American Institute of Suchined's All sights reserved WARNING: This AIA Document's protected by U.S. Copyright Law and Televratismal Treaties. Unathonized reproduction of distribution of this AIA" Document, or amy pottion of it, may result is serves served and eximised penalties, and will be prosocuted to the maximum extent possible under the law. This draft was produced by AIA software set 14:15:10 on 01/19/2000 under Order No 1000314794\_1 which expires on 0/37/2000, and is not for resule. User Notes: (3521870003)

2

(Numbers and Topics in Bold are Section Headings) Acceptance of Nonconforming Work 9.66.993, 12.3 Acceptance of Work 966, 9.82, 993, 9101, 9103, 123 Access to Work 3.16, 6 2.1, 12 j Accident Prevention 10 Acts and Omissions 3 2, 3 3 2, 3 12 8, 3 18, 4 2 3, 8 3 1, 9 5 1, 10 2 5, 10 2 8, 13 4 2, 13 7 1, 14 1, 15 2 Addenda 11.1, 311,1 Additional Costs, Claims for 3 7 4, 3 7 5, 6 1.1, 7.3 7 5, 10 3, 15 1 4 **Additional Inspections and Testing** 942, 9.83, 12.21, 13.5 Additional Insured 1114 Additional Time, Claims for 3 2.4, 3 7 4, 3 7 5, 3 10 2, 8 3.2, 15.1.5 Administration of the Contract 313, 4.2, 94, 95 Advertisement or Invitation to Bid 111 Aesthetic Effect 4213 Allowances 3.8, 7.3 8 All-risk Insurance 1131, 11311 Applications for Payment 4.2.5, 7 3 9, 9 2, 9.3, 9 4, 9 5 1, 9 6 3, 9 7 1, 9 10, 11.1.3 Approvals 2 1 1, 2.2.2, 2 4, 3 1.3, 3.10 2, 3.12 8, 3 12 9, 3 12.10, 42.7,932,1351 Arbitration 8.3.1, 11 3 10, 13.1 1, 15 3 2, 15.4 ARCHITECT Architect, Definition of 4.1.1 Architect, Extent of Authority 2 4 1, 3.12 7, 4 1, 4.2, 5.2, 6.3.1, 7.1 2, 7.3.7, 7 4, 9 2 1, 9 3 1, 9 4, 9 5, 9 6 3, 9 8, 9 10 1, 9 10 3, 12 1, 12.2.1, 13.5 1, 13.5 2, 14 2 2, 14.2 4, 15 1 3, 15 2 1 Architect, Limitations of Authority and Responsibility 2 1 1, 3.12.4, 3 12.8, 3 12.10, 4.1 2, 4 2 1, 4 2.2, 4.2 3, 4 2 6, 4 2.7, 4.2.10, 4 2 12, 4 2.13, 5 2 1, 7.4 1, 942,953,964,1513,152 Architect's Additional Services and Expenses

INDEX

241,11311,1221,13.52,1353,1424 Architect's Administration of the Contract 3.1.3, 4 2, 3 7.4, 15 2, 9 4.1, 9 5 Architect's Approvals 241,313,351,3102,427 Architect's Authority to Reject Work 351, 426, 12.12, 122.1 Architect's Copyright 117,1.5 Architect's Decisions 3.7 4, 4 2 6, 4 2 7, 4 2 11, 4.2 12, 4 2 13, 4 2 14, 631, 737, 739, 81.3, 8.3.1, 9.21, 94.1, 9.5, 984, 991, 1352, 152, 153 Architect's Inspections 374, 4.22, 4.29, 9.4.2, 983, 992, 9101, 135 Architect's Instructions 324, 3.3.1, 426, 427, 13.52 Architect's Interpretations 4211,4212 Architect's Project Representative 4210 Architect's Relationship with Contractor 1 1 2, 1 5, 3.1 3, 3 2.2, 3 2 3, 3.2 4, 3 3.1, 3 4 2, 3 5 1, 3 7 4, 3 7 5, 3.9 2, 3 9.3, 3 10, 3 11, 3 12, 3 16, 3 18, 4 1 2, 4 1.3, 4 2, 5.2, 6.2.2, 7, 8 3 1, 9 2, 9 3, 9 4, 9.5, 97, 98, 99, 10.26, 103, 1137, 12, 1342, 135, 152 Architect's Relationship with Subcontractors 1 1 2, 4 2 3, 4 2 4, 4 2 6, 9 6 3, 9 6 4, 11 3 7 Architect's Representations 942,951,9.101 Architect's Site Visits 3 7.4, 4.2 2, 4 2.9, 9.4 2, 9 5 1, 9 9 2, 9 10 1, 13 5 Asbestos 1031 Attomeys' Fees 3 18 1, 9 10 2, 10.3 3 Award of Separate Contracts 611.612 Award of Subcontracts and Other Contracts for Portions of the Work 5.2 **Basic Definitions** 1.1 **Bidding Requirements** 1.1.1, 5.2.1, 11 4 1 Binding Dispute Resolution 971, 1139, 11310, 1311, 1525, 15261, 1531, 15 3.2, 15.4.1 **Boiler and Machinery Insurance** 11.3.2 Bonds, Lien 7374,910.2,9.10.3 Bonds, Performance, and Payment 7 3.7.4, 9.6 7, 9 10 3, 11 3 9, 11.4 **Building Permit** 3.7.1

з

AIA Document B201\* - 2007. Copyright © 1911, 1913, 1918, 1923, 1937, 1931, 1955, 1961, 1963, 1966, 1976, 1976, 1977, 1987 and 2007 by The American Institute of Architects. All sights reserved. MARNING: This AIA Document is protected by U.S. Copyright Law and International Tradical Unauthorized reproduction of distribution of this AIA Document, or any portion of it, may result is severe civil and erisimal penalitis. Unauthorized reproduction of distribution of this AIA Document, or any portion of it is, may result is severe civil and erisimal Demalities, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 14:15:10 on 01/16/2008 under Order No.1000314794\_1 which expires on 8/17/2008, and is not for resale. User Notes:

Capitalization 1.3 Certificate of Substantial Completion 98.3, 984, 985 **Certificates for Payment** 421, 425, 429, 933, 94, 95, 961, 966, 971, 9.10.1, 9.10.3, 14.1.1.3, 14.2.4, 15.1.3 Certificates of Inspection, Testing or Approval 1354 Certificates of Insurance 9 10.2, 11 1 3 **Change Orders** 1.1 1, 2 4 1, 3 4 2, 3 7.4, 3.8 2 3, 3 11 1, 3 12 8, 4 2.8, 5 2 3, 7 1 2, 7 1 3, 7.2, 7 3 2, 7 3 6, 7 3 9, 7 3 10, 8 3 1, 9.3 1 1, 9.10 3, 10 3 2, 11 3 1 2, 11 3 4, 11 3 9, 12 1 2, 15.1 3 Change Orders, Definition of 7.2.1 CHANGES IN THE WORK 221, 311, 428, 7, 721, 731, 74, 741, 831, 9.3.1 1, 11 3.9 Claims, Definition of 15.1.1 CLAIMS AND DISPUTES 3 2 4, 6 1 1, 6.3 1, 7 3 9, 9 3 3, 9 10 4, 10 3 3, 15, 154 Claims and Timely Assertion of Claims 1541 **Claims for Additional Cost** 3 2 4, 3 7 4, 6 1 1, 7 3 9, 10 3 2, 15.1.4 **Claims for Additional Time** 3 2.4, 3.7.46.1 1, 8.3 2, 10 3 2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3 2 4, 3.18, 6 1.1, 8 3 3, 9 5 1, 9 6 7, 10 3 3, 11 1 1, 11 3 5, 11 3 7, 14 1 3, 14 2 4, 15 1 6 Claims Subject to Arbitration 1531, 154.1 **Cleaning Up** 3.15,63 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10 1, 3.12 6, 5.2 1, 5.2 3, 6 2 2, 8.1.2, 8 2 2, 8.3 1, 11 1, 11 3 1, 11 3 6, 11 4.1, 1514 Commencement of the Work, Definition of 8.1.2 **Communications Facilitating Contract** Administration 391, 4.2.4 Completion, Conditions Relating to 3 4 1, 3 11, 3 15, 4 2.2, 4 2 9, 8 2, 9 4.2, 9 8, 9 9.1, 9 10, 12 2, 13.7, 14 1.2 COMPLETION, PAYMENTS AND Q Completion, Substantial 4 2 9, 8 1 1, 8 1 3, 8.2.3, 9 4.2, 9 8, 9 9 1, 9 10 3, 12.2, 13.7

12 2, 13.7
Contractor, Definition of
ATA Document A201<sup>m</sup> - 2007. Copyright © 1911, 1915, 1915, 1925, 1937, 1931, 1939, 1951, 1961, 1964, 1970, 1976, 1987, 1987 and 2007 by The
ATA Document a protocted by U.S. Copyright Law and International
Treaties. Unauthorized reproduction or distribution of this ATA Document, or any portion of it, may result an severe civil and distribution
Treaties. Unauthorized reproduction by distribution of this ATA Document, or any portion of it, may result an severe civil and distribution
publics, and will be prosented to the maximum action possible under the law. This diaft was produced by ATA software at 14:15:10 on
01/18/2000 under Order No 1000314794\_1 which espires on 8/17/2000 and is not for resale
User Notes:
(1521870803)

Compliance with Laws 161, 323, 36, 37, 31210, 313, 411, 964, 10.2 2, 11.1, 11 3, 13 1, 13.4, 13 5.1, 13 5 2, 13 6, 14.1.1, 14 2 1 3, 15.2 8, 15 4 2, 15 4.3 Concealed or Unknown Conditions 37.4, 4.28, 83.1, 103 Conditions of the Contract 1.11, 6.1.1, 6.14 Consent, Written 3 4 2, 3.7.4, 3 12 8, 3 14 2, 4 1.2, 9.3.2, 9.8.5, 9.9 1, 9.10.2, 9.10.3, 11.3.1, 13.2, 13.4.2, 15.4.4.2 **Consolidation or Joinder** 15.4.4 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS 114,6 Construction Change Directive, Definition of 7.3.1 **Construction** Change Directives 1.1 1, 3 4 2, 3 12 8, 4 2.8, 7 1.1, 7.1 2, 7 1 3, 7.3, 9.3.1.1 Construction Schedules, Contractor's 3.10, 3.12, 1, 3, 12, 2, 6, 1, 3, 15, 1, 5, 2 **Contingent Assignment of Subcontracts** 5.4, 14.2.2.2 **Continuing Contract Performance** 15.1.3 Contract, Definition of 1.1.2 CONTRACT, TERMINATION OR SUSPENSION OF THE 5 4 1 1, 11 3.9, 14 Contract Administration 3.1.3, 4, 94, 95 Contract Award and Execution, Conditions Relating to 3 7 1, 3 10, 5.2, 6 1, 11 1 3, 11.3 6, 11 4 1 Contract Documents, The 1.1.1 Contract Documents, Copies Furnished and Use of 1 5.2, 2.2 5, 5.3 Contract Documents, Definition of 1.1.1 Contract Sum 374, 38, 5.2.3, 7.2, 73, 7.4, 9.1, 942, 9.5.1.4, 967, 9.7, 1032, 1131, 1424, 1432, 1514, 1525 Contract Sum, Definition of 9.1 Contract Time 3 7 4, 3 7 5, 3 10.2, 5.2.3, 7,2.1.3, 7 3 1, 7 3 5, 7.4, 8 1 1, 8 2 1, 8 3 1, 9 5 1, 9.7 1, 10 3 2, 12 1.1, 14 3 2, 15151.152.5 Contract Time, Definition of 8.1.1 CONTRACTOR

4

3.1.6.1.2 **Contractor's Construction Schedules** 3.10, 3.12 1, 3.12 2, 6.1 3, 15 1.5 2 Contractor's Employees 3.3.2, 3.4.3, 3 8.1, 3 9, 3 18 2, 4 2.3, 4.2 6, 10 2, 10 3, 11 1 1, 11.3 7, 14.4, 14 2.1 1, **Contractor's Liability Insurance** 11.1 Contractor's Relationship with Separate Contractors and Owner's Forces 3 12.5, 3 14 2, 4.2 4, 6, 11 3.7, 12 1 2, 12.2 4 Contractor's Relationship with Subcontractors 1.2.2, 3 3.2, 3 18 1, 3 18.2, 5, 9 6.2, 9 6 7, 9 10 2, 1131.2, 1137, 1138 Contractor's Relationship with the Architect 1 1 2, 1.5, 3 1 3, 3 2.2, 3.2 3, 3 2.4, 3.3.1, 3.4.2, 3.5 1, 3 7 4, 3.10, 3 11, 3 12, 3 16, 3 18, 4.1.3, 4.2, 5 2, 6 2 2, 7, 8 3 1, 9 2, 9.3, 9.4, 9 5, 9 7, 9 8, 9 9, 10 2.6, 10 3, 11.3 7, 12, 13 5, 15 1 2, 15 2 1 Contractor's Representations 3 2 1, 3 2 2, 3 5.1, 3 12 6, 6.2 2, 8 2.1, 9.3 3, 9 8 2 Contractor's Responsibility for Those Performing the Work 3 3.2, 3.18, 5.3.1, 6 1 3, 6 2, 9.5 1, 10.2 8 Contractor's Review of Contract Documents 3.2 Contractor's Right to Stop the Work 97 Contractor's Right to Terminate the Contract 14.1, 15.16 Contractor's Submittals 3 10, 3 11, 3 12 4, 4 2 7, 5.2.1, 5 2 3, 9 2, 9 3, 9 8 2, 983,991,910.2,9103,11.1.3,1142 Contractor's Superintendent 39, 1026 Contractor's Supervision and Construction Procedures 1 2.2, 3.3, 3 4, 3 12 10, 4 2 2, 4 2 7, 6 1 3, 6 2 4, 7 1.3, 7 3 5, 7 3 7, 8.2, 10, 12, 14, 15 1.3 Contractual Liability Insurance 11118,112 Coordination and Correlation 1 2, 3 2 1, 3 3 1, 3 10, 3 12 6, 6 1 3, 6 2 1 Copies Furnished of Drawings and Specifications 1 5, 2.2 5, 3 11 Copyrights 1.5, 3.17 Correction of Work 2 3, 2 4, 3 7.3, 9.4 2, 9.8 2, 9 8.3, 9 9 1, 12 1 2, 12.2 **Correlation and Intent of the Contract Documents** 1.2 Cost, Definition of 7.3.7 Costs 2 4 1, 3.2.4, 3 7 3, 3.8 2, 3.15 2, 5.4.2, 6 1 1, 6 2 3, 7.3.3 3, 7.3.7, 7 3.8, 7.3 9, 9 10 2, 10 3 2, 10 3 6, 11.3, 12 1.2, 12 2 1, 12.2 4, 13 5, 14 **Cutting and Patching** 

3.14.62.5 Damage to Construction of Owner or Separate Contractors 3 14.2, 6.2.4, 10.2.1.2, 10 2 5, 10 4, 11 1 1, 11 3, 12.2.4 Damage to the Work 3.14.2, 9.9.1, 10.2.1.2, 10.2.5, 10.4.1, 11.3.1, 12.2.4 Damages, Claims for 3 2 4, 3 18, 6 1.1, 8 3 3, 9 5.1, 9 6 7, 10 3 3, 11.1.1, 11 3 5, 11 3 7, 14 1 3, 14 2 4, 15 1 6 Damages for Delay 6.1.1, 8 3 3, 9.5 1.6, 9.7, 10.3.2 Date of Commencement of the Work, Definition of 8.1.2 Date of Substantial Completion, Definition of 8.1.3 Day, Definition of 8.1.4 Decisions of the Architect 3 7.4, 4.2 6, 4.2 7, 4 2 11, 4.2 12, 4 2 13, 15 2, 6.3, 7.3 7, 7.3.9, 8 1.3, 8 3 1, 9 2.1, 9 4, 9.5 1, 9.8 4, 9 9 1, 13 5.2, 14 2 2, 14 2 4, 15 1, 15.2 **Decisions to Withhold Certification** 9.4 1, 9.5, 9.7, 14 1 1.3 Defective or Nonconforming Work, Acceptance, Rejection and Correction of 2 3.1, 2.4.1, 3 5 1, 4.2 6, 6.2 5, 9.5 1, 9.5.2, 9.6 6, 9.8 2, 9.9.3, 9.10 4, 12.2 1 Defective Work, Definition of 3.5.1 Definitions 1 1, 2.1.1, 3 1 1, 3.5 1, 3.12 1, 3 12.2, 3.12 3, 4.1 1, 15.1 1, 5.1, 6.1.2, 7.2.1, 7 3 1, 8.1, 9 1, 9 8.1 Delays and Extensions of Time 3 2 , 3 7 4, 5.2.3, 7 2 1, 7.3 1, 7.4.1, 8.3, 9 5 1, 9 7 1, 10.3.2, 10 4 1, 14 3.2, 15 1.5, 15 2 5 Disputes 631, 7.3.9, 151, 152 Documents and Samples at the Site 3.11 Drawings, Definition of 1.1.5 Drawings and Specifications, Use and Ownership of 3.11 Effective Date of Insurance 8.2.2, 11.1.2 Emergencies 10.4, 14 1.1.2, 15 1 4 Employees, Contractor's 3 3 2, 3 4 3, 3.8.1, 3.9, 3.18 2, 4 2.3, 4 2 6, 10 2, 10.3 3, 11.1 1, 11 3 7, 14 1, 14 2.1 1 Equipment Labor, Materials or 1.1.3, 1.1.6, 3.4, 3.5, 1, 3.8, 2, 3.8.3, 3, 12, 3, 13, 1, 3 15 1, 4 2 6, 4 2.7, 5 2.1, 6.2.1, 7 3 7, 9 3 2, 9 3 3, 9513, 9102, 1021, 1024, 14211, 14212 Execution and Progress of the Work

5

AIA Documant &201" - 3007. Copyright D 1911, 1915, 1918, 1828, 1937, 1951, 1858, 1861, 1963, 1966, 1970, 1876, 1887, 1997 and 2007 by The American Institute of Architects. All rights reserved. RANNING. This AIA Document is protocted by U.S. Copyright Law and International Fronties Unauthorized reproduction on distribution of this AIA Document, or any portion of it, may result in severe only and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 14-15:10 on 0/16/2005 under Order No.1000314754 1 which expires on 8/17/2008, and is not for resale. User Motes:

1.1 3, 1 2 1, 1 2.2, 2 2 3, 2 2 5, 3.1, 3 3 1, 3 4 1, 3 5 1, 3 7 1, 3.10.1, 3 12, 3 14, 4.2, 6 2.2, 7 1 3, 7 3 5, 8 2, 9 5 1, 9.9 1, 10.2, 10 3, 12 2, 14 2, 14 3 1, 15 1.3 Extensions of Time 3 2.4, 3 7 4, 5 2 3, 7.2 1, 7 3, 7 4 1, 9 5 1, 9 7 1, 10 3 2, 10 4 1, 14 3, 15 1 5, 15 2 5 Failure of Payment 9 5 1 3, 9.7, 9 10.2, 13 6, 14 1.1 3, 14 2 1.2 Faulty Work (See Defective or Nonconforming Work) **Final Completion and Final Payment** 4.2 1, 4 2.9, 9 8.2, 9.10, 11 1 2, 11.1 3, 11 3 1, 11 3.5, 12 3.1, 14 2.4, 14.4 3 Financial Arrangements, Owner's 2.2 1, 13 2 2, 14.1.1 4 Fire and Extended Coverage Insurance 11311 **GENERAL PROVISIONS** 1 Governing Law 13.1 Guarantees (See Warranty) **Hazardous Materials** 10 2 4, 10.3 Identification of Subcontractors and Suppliers 521 Indemnification 3 17 1, 3.18, 9.10 2, 10 3 3, 10 3 5, 10 3 6, 11 3.1 2, 1137 Information and Services Required of the Owner 2 1.2, 2.2, 3.2.2, 3 12 4, 3 12 10, 6 1.3, 6 1 4, 6.2.5, 961,964,992,9103,1033,1121,114,1351, 13 5 2, 14 1 1 4, 14 1 4, 15 1 3 Initial Decision 15.2 Initial Decision Maker, Definition of 11.8 Initial Decision Maker, Decisions 14.2.2, 14.2.4, 15.2.1, 15.2.2, 15.2.3, 15.2.4, 15.2.5 Initial Decision Maker, Extent of Authority 14 2 2, 14 2 4, 15 1 3, 15 2 1, 15 2 2, 15 2 3, 15 2 4, 15.2.5 Injury or Damage to Person or Property 10.2.8, 10.4.1 Inspections 3.1 3, 3.3 3, 3 7.1, 4 2 2, 4 2.6, 4.2.9, 9 4 2, 9 8.3, 992,9101,1221,135 Instructions to Bidders 111 Instructions to the Contractor 3 2 4, 3 3 1, 3 8 1, 5 2 1, 7, 8 2.2, 12, 13 5 2 Instruments of Service, Definition of 1.1.7 Insurance 3 18 1, 6.1.1, 7.3 7, 9 3.2, 9 8.4, 9.9 1, 9 10 2, 11 **Insurance, Boiler and Machinery** 11.3.2 Insurance, Contractor's Liability

11.1 Insurance, Effective Date of 8 2 2, 11.1.2 Insurance, Loss of Use 11.3.3 Insurance, Owner's Liability 11.2 **Insurance**, Property 1025,11.3 Insurance, Stored Materials 932, 114.14 INSURANCE AND BONDS 11 Insurance Companies, Consent to Partial Occupancy 9.9.1, 11 4.1.5 Insurance Companies, Settlement with 11410 Intent of the Contract Documents 121,42.7,42.12,42.13,74 Interest 13.6 Interpretation 1.2.3, 1.4, 4 1 1, 5 1, 6.1.2, 15 1 1 Interpretations, Written 4 2.11, 4 2 12, 15.1 4 Judgment on Final Award 154.2 Labor and Materials, Equipment 1.1.3, 1.1.6, 3.4, 3 5 1, 3 8 2, 3.8.3, 3 12, 3 13, 3 15 1, 4 2.6, 4 2 7, 5.2 1, 6.2 1, 7 3 7, 9.3.2, 9 3 3, 9 5.1 3, 9 10.2, 10 2 J, 10 2 4, 14 2 I 1, 14 2 1.2 Labor Disputes 8.3.1 Laws and Regulations 1 5, 3 2.3, 3 6, 3 7, 3 12.10, 3 13.1, 4 1 1, 9.6 4, 9 9 1, 10 2 2, 11 1.1, 11 3, 13 1 1, 13 4, 13 5 1, 13 5.2, 1361, 14, 1528, 154 Liens 2.1.2, 9 3 3, 9 10 2, 9.10 4, 15.2.8 Limitations, Statutes of 12 2.5, 13.7, 15.4 1 1 Limitations of Liability 231, 322, 351, 31210, 3171, 3181, 4.2.6, 4.2 7, 4.2.12, 6 2 2, 9 4 2, 9.6 4, 9 6 7, 10.2 5, 10 3 3, 11 1 2, 11 2.1, 11 3 7, 12 2 5, 13 4.2 Limitations of Time 2.1.2, 2.2, 2.4, 3.2.2, 3.10, 3.11, 3.12.5, 3.15.1, 4.2.7, 5.2, 5 3.1, 5.4 1, 6 2 4, 7.3, 7 4, 8.2, 9.2 1, 9.3 1, 933, 941, 95, 96, 97.1, 98, 99, 910, 1113, 11 3 1 5, 11 3 6, 11.3.10, 12.2, 13 5, 13 7, 14, 15 Loss of Use Insurance 11.3.3 Material Suppliers 1 5, 3 12 1, 4 2 4, 4 2 6, 5 2 1, 9 3, 9 4 2, 9 6, 9 10 5 Materials, Hazardous 10 2 4, 10.3 Materials, Labor, Equipment and

6

ATA Document A201<sup>m</sup> - 2007. Copyright © 1011, 1915, 1018, 1023, 1037, 1051, 1958, 1061, 1963, 1966, 1870, 1076, 1887, 1987 and 2007 by The Averscan Institute of Architects All sights reserved. WANNING: This AIX Document is protocted by U.S. Copyright Eaw and International Treaties. Document of a production or discribution of this AIX Document of the may portion of it, may result in securit in a deciminal panalties, and will be presented to the maximum extent possible under the law. This draft was produced by AIX software at 14:15:10 on C/104/200 under Order No.1000314794\_1 which expires on 8/17/2708, and is not for resele User Notes:

1 1 3, 1.1 6, 1 5.1, 3 4 1, 3 5 1, 3.8 2, 3 8.3, 3 12, 3 13 1, 3 15 1, 4.2 6, 4 2.7, 5 2 1, 6.2 1, 7 3.7, 9.3 2, 9 3.3, 9.5 1.3, 9.10 2, 10 2.1.2, 10.2.4, 14 2.1 1, 14 2.1.2 Means, Methods, Techniques, Sequences and Procedures of Construction 3 3 1, 3 12 10, 4.2 2, 4.2 7, 9.4 2 Mechanic's Lien 21.2, 15.2.8 Mediation 8 3 1, 10 3 5, 10 3 6, 15 2 1, 15 2 5, 15 2 6, 15.3, 154.1 Minor Changes in the Work 1.1 1, 3 12 8, 4 2 8, 7 1, 7.4 MISCELLANEOUS PROVISIONS 13 Modifications, Definition of 1.1.1 Modifications to the Contract 1.1.1, 1.1.2, 3.11, 4.1.2, 4.2.1, 5.2.3, 7, 8.3.1, 9.7.1, 10.3.2, 11.3.1 **Mutual Responsibility** 6.2 Nonconforming Work, Acceptance of 9.6.6, 9 9.3, 12.3 Nonconforming Work, Rejection and Correction of 2 3 1, 2.4 1, 3 5 1, 4 2 6, 6 2 4, 9 5.1, 9 8.2, 9.9 3, 9 10 4, 12.2 1 Notice 2 2 1, 2 3 1, 2 4 1, 3 2 4, 3 3 1, 3 7 2, 3 12 9, 5 2 1, 971, 9.10, 1022, 1113, 1146, 122.2.1, 13.3, 13 5 1, 13 5 2, 14 1, 14.2, 15 2 8, 15 4 1 Notice, Written 2 3 1, 2.4 1, 3.3 1, 3.9.2, 3.12.9, 3.12 10, 5 2 1, 9 7 1, 9 10, 10 2 2, 10 3, 11 1 3, 11 3 6, 12 2 2 1, 13 3, 14, 1528, 15.41 Notice of Claims 3 7 4, 4.5, 10.2 8, 15.1.2, 15.4 Notice of Testing and Inspections 13 5.1, 13 5 2 Observations, Contractor's 3.2, 374 Occupancy 2.2 2, 9 6 6, 9 8, 11 3 1 5 Orders, Written 1.1 1, 2.3, 3 9.2, 7, 8 2 2, 11 3 9, 12 1, 12 2 2 1, 13 5 2, 14 3 1 OWNER 7 Owner, Definition of 2.1.1 Owner, Information and Services Required of the 2 1 2, 2, 2, 3, 2, 2, 3 12, 10, 6, 1, 3, 6, 1, 4, 6, 2, 5, 9, 3, 2, 961, 964, 99.2, 9103, 1033, 11.21, 113, 1351, 13 5 2, 14 1.1.4, 14 1 4, 15 1 3 **Owner's Authority** 1 5, 2 1 1, 2.3.1, 2 4 1, 3 4 2, 3 8 1, 3 12 10, 3 14.2, 4.1 2, 4 1 3, 4 2 4, 4 2.9, 5.2 1, 5 2 4, 5 4 1, 6 1, 6 3.1,

721,731,822,831,931,932,951,964 9 9.1, 9.10 2, 10 3 2, 11 1.3, 11 3 3, 11 3 10, 12.2 2, 12.3 1, 13 2 2, 14 3, 14.4, 15 2 7 **Owner's Financial Capability** 2 2 1, 13.2 2, 14.1.1 4 **Owner's Liability Insurance** 11.2 **Owner's Loss of Use Insurance** 11.3.3 Owner's Relationship with Subcontractors 1 1 2, 5 2, 5.3, 5 4, 9 6 4, 9 10 2, 14 2 2 **Owner's Right to Carry Out the Work** 2.4, 14.2.2 **Owner's Right to Clean Up** 6.3 **Owner's Right to Perform Construction and to Award Separate Contracts** 6.1 **Owner's Right to Stop the Work** 2.3 Owner's Right to Suspend the Work 14.3 Owner's Right to Terminate the Contract 14.2 **Ownership and Use of Drawings, Specifications** and Other Instruments of Service 1 1, 1 1 6, 1.1.7, 1.5, 2 2 5, 3.2 2, 3.11 1, 3.17 1, 4 2 12, 5 3 1 **Partial Occupancy or Use** 96.6, 9.9, 11.3.1.5 Patching, Cutting and 3.14,625 Patents 317 Payment, Applications for 4 2 5, 7 3 9, 9.2 1, 9.3, 9.4, 9 5, 9.6 3, 9.7 1, 9 8 5, 9 10 1, 14.2.3, 14.2.4, 14 4.3 Payment, Certificates for 4 2.5, 4 2.9, 9.3.3, 9.4, 9.5, 9 6.1, 9.6 6, 9.7 1, 9 10 1, 9.103, 137, 141.1.3, 1424 **Payment**, Failure of 9513, 9.7, 9.10.2, 13.6, 14113, 142.12 Payment, Final 4 2 1, 4 2 9, 9 8.2, 9.10, 11 1 2, 11.1.3, 11 4 1, 11.4 5, 12 3.1, 13 7, 14 2 4, 14 4 3 Payment Bond, Performance Bond and 7 3.7.4, 9 6 7, 9.10.3, 11 4 9, 11.4 Payments, Progress 9 3, 9.6, 9 8 5, 9 10 3, 13 6, 14.2.3, 15 1 3 **PAYMENTS AND COMPLETION** Payments to Subcontractors 5 4 2, 9 5.1.3, 9 6.2, 9 6 3, 9 6 4, 9.6 7, 11 4.8, 14.2.1.2 PCB 10.3.1 Performance Bond and Payment Bond 73.74,967,9.103,1149,114

AIA Document A201 - 2007. Copyright 6 1911, 1915, 1918, 1935, 1937, 1951, 1958, 1951, 1963, 1963, 1976, 1976, 1977, 1987, 1987 and 2007 by The American Institute of Austricets. All rights reserved WANNING This AIA Document is protected by U.S. Copyright Law and International Transies. Unauthorized reproduction of distribution of this AIA Document, or any portion of it, may result in averse eight and principal panalties, and will be preserved to the maximum extent possible under the law. This draft was produced by AIA software at 14:15:10 on 01/12/2008 under Order No.100C314794\_1 which expires on 8/17/2008, and is hot for resule. (3921870803)

7

Permits, Fees, Notices and Compliance with Laws 2.2 2, 3.7, 3.13, 7 3.7 4, 10 2 2 PERSONS AND PROPERTY, PROTECTION OF 10 Polychlorinated Biphenyl 1031 Product Data, Definition of 3.12.2 Product Data and Samples, Shop Drawings 3 II, 3.12, 4 2 7 Progress and Completion 4.2.2, 8.2, 98, 991, 14 I 4, 15 1 3 Progress Payments 9.3, 9.6, 985, 9103, 136, 1423, 1513 Project, Definition of the 1.1.4 Project Representatives 4 2 10 **Property Insurance** 10 2 5. 11.3 PROTECTION OF PERSONS AND PROPERTY 10 Regulations and Laws 1 5, 3.2 3, 3 6, 3 7, 3 12.10, 3 13, 4 1 1, 9 6 4, 9 9 1, 10.2 2, 11.1, 11.4, 13 1, 13 4, 13 5.1, 13 5 2, 13 6, 14, 15 2.8, 15.4 Rejection of Work 351,426,1221 Releases and Waivers of Liens 9.10.2 Representations 3 2 1, 3 5 1, 3 12 6, 6 2 2, 8 2 1, 9 3 3, 9 4 2, 9 5 1, 982,9101 Representatives 2 1 1, 3 1 1, 3 9, 4 1 1, 4 2 1, 4 2 2, 4 2 10, 5 1.1, 512, 13.21 Responsibility for Those Performing the Work 3 3.2, 3 18, 4 2.3, 5 3 1, 6 1 3, 6 2, 6.3, 9 5 1, 10 Retainage 931,962,985,991,910.2,9103 **Review of Contract Documents and Field Conditions by Contractor** 3.2, 3 12 7, 6 1.3 Review of Contractor's Submittals by Owner and Architect 3 10.1, 3 10.2, 3 11, 3 12, 4 2, 5 2, 6 1.3, 9 2, 9 8 2 Review of Shop Drawings, Product Data and Samples by Contractor 3 12 **Rights and Remedies** 1 1 2, 2 3, 2 4, 3.5 1, 3 7 4, 3.15.2, 4.2 6, 4 5, 5 3, 5 4, 6 1, 6 3, 7 3 1, 8 3, 9.5 1, 9 7, 10 2 5, 10 3, 12.2 2, 12.2 4, 13.4, 14, 15 4 **Royalties, Patents and Copyrights** 3.17 Rules and Notices for Arbitration 1541

Safety of Persons and Property 10.2, 10.4 Safety Precautions and Programs 3 3 1, 4 2 2, 4 2 7, 5 3 1, 10.1, 10.2, 10 4 Samples, Definition of 3.12.3 Samples, Shop Drawings, Product Data and 311, 3.12, 427 Samples at the Site, Documents and 3.11 Schedule of Values 9.2, 931 Schedules, Construction 1 4 1 2, 3 10, 3 12 1, 3 12.2, 6.1 3, 15.1 5.2 Separate Contracts and Contractors 1 1 4, 3 12 5, 3 14 2, 4 2 4, 4 2 7, 6, 8 3 1, 11 4 7, 12.1.2 Shop Drawings, Definition of 3.12.1 Shop Drawings, Product Data and Samples 3.11, 3.12, 4 2 7 Site, Use of 3.13, 6 1 1, 6.2 1 Site Inspections 3.2.2, 3 3 3, 3 7.1, 3 7 4, 4 2, 9.4.2, 9 10 1, 13 5 Site Visits, Architect's 3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.5.1, 9.9.2, 9.10.1, 13.5 Special Inspections and Testing 4 2 6, 12 2.1, 13.5 Specifications, Definition of the 1.1.6 Specifications, The 1.1, 1, 1.1.6, 1.2.2, 1.5, 3.11, 3.12, 10, 3.17, 4.2.14 Statute of Limitations 137, 15.4.1.1 Stopping the Work 2.3, 9.7, 10.3, 14 1 Stored Materials 6.2.1, 9.3.2, 10.2.1.2, 10.2.4, 11.4.1.4 Subcontractor, Definition of 5.1.1 SUBCONTRACTORS Subcontractors, Work by 1.2 2, 3 3.2, 3 12 1, 4 2.3, 5.2 3, 5 3, 5.4, 9 3 1 2, 9.67 Subcontractual Relations 5.3, 5 4, 9.3 1 2, 9 6, 9 10, 10 2 1, 11 4 7, 11 4 8, 141,14.21 Submittals 3 10, 3 11, 3.12, 4.2 7, 5.2.1, 5 2 3, 7.3 7, 9 2, 9 3, 9.8, 9.9.1, 9.10.2, 9.10.3, 11.1.3 Submittal Schedule 3 10 2, 3 12 5, 4.2 7 Subrogation, Waivers of 61.1, 11.45, 11.3.7 Substantial Completion

ATA Document A201<sup>20</sup> - 2007. Copyright © 1911, 1915, 1919, 1927, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1867, 1897 and 2007 by The Arerican Listitute of Architects. All rights reserved founded the ATA' Document is protocoided by U.S. Copyright Law and International Treaties Unauthorized reproduction or distribution of this ATA' Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum action possible under the law. This draft use produced by ATA software at 14:15 10 on 01/14/2008 under Order No.1000314794\_1 which expires on 0/17/2008, and is not for result

8

4.2 9, 8.1.1, 8 1 3, 8 2 3, 9 4.2, 9.8, 9 9 1, 9 10 3, 12 2, 13.7 Substantial Completion, Definition of 9.8.1 Substitution of Subcontractors 523,524 Substitution of Architect 413 Substitutions of Materials 3.42,351,738 Sub-subcontractor, Definition of 5.1.2 Subsurface Conditions 374 Successors and Assigns 13.2 Superintendent 3.9, 10.2 6 Supervision and Construction Procedures 122, 3.3, 3.4, 3 12 10, 4.22, 427, 61.3, 624, 7 1 3, 7 3 7, 8 2, 8 3 1, 9 4.2, 10, 12, 14, 15.1 3 Surety 5 4 1 2, 9 8 5, 9 10 2, 9 10 3, 14 2 2, 15 2 7 Surety, Consent of 9 10 2, 9.10 3 Surveys 223 Suspension by the Owner for Convenience 14.3 Suspension of the Work 542,14.3 Suspension or Termination of the Contract 5411,1149,14 Taxes 36, 3.8.21, 7374 Termination by the Contractor 14.1, 15.1.6 Termination by the Owner for Cause 5411, 14.2, 15.16 Termination by the Owner for Convenience 14.4 Termination of the Architect 41.3 Termination of the Contractor 1422 TERMINATION OR SUSPENSION OF THE CONTRACT 14 **Tests and Inspections** 3.1.3, 3 3 3, 4 2 2, 4 2 6, 4 2.9, 9 4 2, 9.8 3, 9.9 2, 9 10 1, 10 3.2, 11 4 1 1, 12 2.1, **13.5** TIME 8 Time, Delays and Extensions of 3 2 4, 3 7 4, 5.2.3, 7 2 1, 7 3.1, 7 4 1, 8.3, 9.5 1, 9 7 1, 10 3 2, 10.4 1, 14.3 2, 15.1 5, 15 2 5 Time Limits

2 1 2, 2.2, 2 4, 3 2 2, 3.10, 3 11, 3.12 5, 3 15.1, 4.2, 4.4, 4 5, 5 2, 5 3, 5 4, 6.2.4, 7.3, 7 4, 8 2, 9 2, 9 3 1, 933,941,95,96,97,98,99,910,1113, 11 4 1 5, 11 4 6, 11.4 10, 12 2, 13 5, 13 7, 14, 15 1 2, 154 **Time Limits on Claims** 3.7 4, 10 2.8, 13.7, 15.1 2 Title to Work 932,933 Transmission of Data in Digital Form 1.6 UNCOVERING AND CORRECTION OF WORK 12 Uncovering of Work 12.1 Unforeseen Conditions, Concealed or Unknown 374,8.31,103 Unit Prices 7.3.3.2, 7.3.4 Use of Documents 1.1.1, 1.5, 2.2.5, 3 12.6, 5.3 Use of Site 3.13, 61.1, 6.2.1 Values, Schedule of 9.2, 9.3 1 Waiver of Claims by the Architect 13.4.2 Waiver of Claims by the Contractor 9 10 5, 11 4.7, 13.4 2, 15.1.6 Waiver of Claims by the Owner 9.9 3, 9 10.3, 9 10 4, 11 4 3, 11 4.5, 11 4 7, 12.2 2 1, 13 4 2, 14.2 4, 15 1 6 Waiver of Consequential Damages 14.2.4, 15.1.6 Waiver of Liens 9 10.2, 9 10 4 Waivers of Subrogation 61.1, 11.45, 11.3.7 Warranty 35, 429, 9.3.3, 984, 991, 9104, 122.2, 1371 Weather Delays 15.1.5.2 Work, Definition of 1.1.3 Written Consent 1.5.2, 3.42, 3.74, 3 128, 3 14.2, 4 1.2, 9 3 2, 9 8 5. 991, 9.102, 910.3, 11.41, 132, 1342, 15442 Written Interpretations 4 2.11, 4.2 12 Written Notice 2.3, 2 4, 3 3 1, 3 9, 3 12 9, 3 12.10, 5 2 1, 8 2 2, 9.7, 9 10, 10 2 2, 10 3, 11 1 3, 11 4.6, 12 2 2, 12 2 4, 13.3, 14, 15 4 1 Written Orders 1 1 1, 2.3, 3 9, 7, 8 2 2, 11 4 9, 12.1, 12 2, 13 5 2, 14 3.1, 15.1.2

9

AIA Document A101<sup>m</sup> - 2007. Copyright 6 1911, 1915, 1918, 1925, 1937, 1851, 1958, 1951, 1963, 1965, 1976, 1976, 1887, 1887 and 2007 by The Averitan Institute of Atchitects. All sights reserved. MORNING: This AIA' Document is protected by U.S. Copyright Law and International Troaties. Unsubcrited reproduction or distribution of this AIA' Document, or any portion of it, may result in severe civil and orininal penalities, and will be proceduced to the maximum extent possible under the law Birdduced by AIA software at 14:15:10 on 01/18/2008 under Order No.1000314794\_1 which expires on 0/17/2008, and is not for resale. User Motes

#### ARTICLE 1 GENERAL PROVISIONS

§ 1.1 BASIC DEFINITIONS

§ 1.1.1 THE CONTRACT DOCUMENTS

The Contract Documents are enumerated in the Agreement between the Owner and Contractor (hereinafter the Agreement) and consist of the Agreement, Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications, Addenda issued prior to execution of the Contract, <u>other documents included</u> in the Project Manual, other documents listed in the Agreement and Modifications issued after execution of the Contract. A Modification is (1) a written amendment to the Contract signed by both parties, <u>or</u>(2) a Change Order, (3) a Construction Change Directive or (1) a written order-for a minor-change-in-the Work issued by the Architect Unless specifically enumerated in the Agreement, the Contract Documents do not include the advertisement or invitation to bid, Instructions to Bidders, sample forms, other information furnished by the Owner in anticipation of receiving bids or proposals, the Contractor's bid or proposal, or portions of Addenda relations

#### § 1.1.2 THE CONTRACT

The Contract Documents form the Contract for Construction The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations or agreements, either written or oral. The Contract may be amended or modified only by a Modification. The Contract Documents shall not be construed to create a contractual relationship of any kind (1) between the Contractor and the Architect or the Architect's consultants, (2) between the Owner and a Subcontractor or a Sub-subcontractor, (3) between the Owner and the Architect or the Architect's consultants or (4) between any persons or entities other than the Owner and the Contractor. The Architect shall, however, be entitled to performance and enforcement of obligations under the Contract intended to facilitate performance of the Architect's duties.

## § 1.1.3 THE WORK

The term "Work" means all of the Contractor's obligations under the Contract Documents, including the construction and services required by the Contract Documents, whether completed or partially completed, and

includes all other labor, materials, equipment and services provided or to be provided by the Contractor to fulfill the Contractor's obligations. The Work may constitute the whole or a part of the Project

## § 1.1.4 THE PROJECT

The Project is the total construction of which the Work performed under the Contract Documents may be the whole or a part and which may include construction by the Owner and by separate contractors.

# § 1.1.5 THE DRAWINGS

The Drawings are the graphic and pictorial portions of the Contract Documents showing the design, location and dimensions of the Work, generally including plans, elevations, sections, details, schedules and diagrams

# § 1.1.6 THE SPECIFICATIONS

The Specifications are that portion of the Contract Documents consisting of the written requirements for materials, equipment, systems, standards and workmanship for the Work, and performance of related services

#### § 1.1.7 INSTRUMENTS OF SERVICE

Instruments of Service are representations, in any medium of expression now known or later developed, of the tangible and intangible creative work performed by the Architect and the Architect's consultants under their respective professional services agreements. Instruments of Service may include, without limitation, sludies, surveys, models, sketches, drawings, specifications, and other similar materials

## § 1.1.8 INITIAL DECISION MAKER

The Initial Decision Maker, if any is the person identified in the Agreement to render initial decisions on Claims in accordance with Section 15.2, and certify termination of the Agreement under Section 14.2.2.

## § 1.2 CORRELATION AND INTENT OF THE CONTRACT DOCUMENTS

§ 1.2.1 The intent of the Contract Documents is to include all items necessary for the proper execution and completion of the Work by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all, performance by the Contractor shall be required only to the extent consistent with the Contract Documents and reasonably inferable from them as being necessary to produce the indicated results.

AIA Decumant #201 - 2007. Copyright + 1911, 1915, 1918, 1925, 1937, 1951, 1951, 1951, 1955, 1955, 1970, 1976, 1997, 1997 and 2007 by The American Instatute of Auchitects - All rights reserved. HARNING: This AIA: Document is protected by U.S. Copyright Law and Toternational Treaties Unauthorized reproduction or distribution of this AIA: Document, or any portion of it, may result in severe civil and criminal penaltics, and will be prosecuted to the maximum extent possible under the law. This diff was produced by AIA software at 14:15:10 on 0/18/2008 under Crief No. 1000314794\_1 which empires on #/17/2008, and im not for resals. (1921870803) § 1.2.2 Organization of the Specifications into divisions, sections and articles, and arrangement of Drawings shall not control the Contractor in dividing the Work among Subcontractors or in establishing the extent of Work to be performed by any trade

§ 1.2.3 Unless otherwise stated in the Contract Documents, words that have well-known technical or construction industry meanings are used in the Contract Documents in accordance with such recognized meanings

85.2.3. Line of part and there is an energy to do Contro (Populated or any), a doctor is the track of the doctor of the second track, or for non-difference doctor of the process of the second track of th

## § 1.3 CAPITALIZATION

Terms capitalized in these General Conditions include those that are (1) specifically defined, (2) the titles of numbered articles or (3) the titles of other documents published by the American Institute of Architects.

## § 1.4 INTERPRETATION

In the interest of brevity the Contract Documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement

§ 1.5 OWNERSHIP AND USE OF DRAWINGS, SPECIFICATIONS AND OTHER INSTRUMENTS OF SERVICE § 1.5.1 The Architect and the Architect's consultants shall be deemed the authors and owners of their respective Instruments of Service, including the Drawings and Specifications, and will retain all common law, statutory and other reserved rights, including copyrights. The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers shall not own or claim a copyright in the Instruments of Service. Submittal or distribution to meet official regulatory requirements or for other purposes in connection with this Project is not to be construed as publication in derogation of the Owner's the Architect's or Architect's consultants' reserved rights

§ 1.5.2 The Contractor, Subcontractors, Sub-subcontractors and material or equipment suppliers are authorized to use and reproduce the Instruments of Service provided to them solely and exclusively for execution of the Work All copies made under this authorization shall bear the copyright notice, if any, shown on the Instruments of Service. The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers may not use the Instruments of Service on other projects or for additions to this Project outside the scope of the Work without the specific written consent of the Owner, Architect and the Architect's consultants

## § 1.6 TRANSMISSION OF DATA IN DIGITAL FORM

If the parties intend to transmit Instruments of Service or any other information or documentation in digital form, they shall endeavor to establish necessary protocols governing such transmissions, unless otherwise already provided in the Agreement or the Contract Documents.

#### ARTICLE 2 OWNER § 2.1 GENERAL

§ 2.1.1 The Owner is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Unless provided in the Agreement, the Owner shall designate in writing a representative who shall have express authority to bind the Owner with respect to all matters requiring the Owner's approval or authorization. Except as otherwise provided in Section 4.2.1, the Architect does not have such authority. The term "Owner" means the Owner or the Owner's authorized representative

§21.2 The Owner shall furnish to the Contractor within fifteen days after receipt of a written request, information necessary and relevant for the Contractor to evaluate, give notice of or enforce mechanic's lien rights. Such information shall include a correct statement of the record legal title to the property on which the Project is located, usually referred to as the site, and the Owner's interest therein.

## § 2.2 INFORMATION AND SERVICES REQUIRED OF THE OWNER

§ 22.1 Prior to commencement of the Work, the Contractor may request in writing that the Owner provide reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the

Al& Document A201<sup>m</sup> - 2007. Copyright © 1911, 1915, 1915, 1925, 1937, 1931, 1954, 1951, 1951, 1951, 1955, 1976, 1977, 1987 and 2007 by The American Institute of Acchitects. All rights reserved. KARHING: This AIA Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA" Document, or any portion of it, may result in severe civil and craminal publics, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 14:15:10 on 03/18/2005 under Order No.1000314764\_1 which sepires on 8/17/2008, and is not for resale. User Notes:

Formatted: Font: Bold

Contract. Thereafter, the Contractor may only request such evidence if (1) the Owner fails to make payments to the Contractor as the Contract Documents require; (2) a change in the Work materially changes the Contract Sum; or (3) the Contractor identifies in writing a reasonable concern regarding the Owner's ability to make payment when due. The Owner shall furnish such evidence as a condition precedent to commencement or continuation of the Work or the portion of the Work affected by a material change. After the Owner furnishes the evidence, the Owner shall not materially vary such financial arrangements without prior notice to the Contractor.

§ 2.2.2 Except for permits and fees that are the responsibility of the Contractor under the Contract Documents, including those required under Section 3.7.1, the Owner shall secure and pay for necessary approvals, easements, assessments and charges required for construction, use or occupancy of permanent structures or for permanent changes in existing facilities <u>the Contractor in the Interventer Contractor</u> or the contractor of the contrac

§ 2.2.3 The Owner shall furnish surveys describing physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site. The Contractor shall be entitled to rely on the accuracy of information furnished by the Owner but shall exercise proper precautions relating to the safe performance of the Work

§ 2.2.4 The Owner shall furnish information or services required of the Owner by the Contract Documents with reasonable promptness. The Owner shall also furnish any other information or services under the Owner's control and relevant to the Contractor's performance of the Work with reasonable promptness after receiving the Contractor's written request for such information or services

§ 2.2.5 Unless otherwise provided in the Contract Documents, the Owner shall furnish to the Contractor one copy of the Contract Documents for purposes of making reproductions pursuant to Section 1.5.2

## § 2.3 OWNER'S RIGHT TO STOP THE WORK

If the Contractor fails to correct Work that is not in accordance with the requirements of the Contract Documents as required by Section 12.2 or repeatedly fails to carry out Work in accordance with the Contract Documents, the Owner may issue a written order to the Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, the right of the Owner to stop the Work shall not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Contractor or any other person or entity, except to the extent required by Section 6.1.3 <u>Line Owner or the benefit of the Contractor or any other person or entity, except to the extent required by Section 6.1.3 Line Owner or the benefit of the Contractor or any other person or entity, except to the extent required by Section 6.1.3 Line Owner or the benefit of the Contractor or any other person or entity, except to the extent required by Section 6.1.3 Line Owner or the orthogeneric to require the rest of the Contractor or any other person or entity.</u>

#### § 2.4 OWNER'S RIGHT TO CARRY OUT THE WORK

If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents and the standard of the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents and the Contractor of such default or neglect with diligence and promptness, the Owner to commence and continue correction of such default or neglect with diligence and promptness, the Owner may, without prejudice to other remedies the Owner may have, correct such deficiencies. In such case an appropriate Change Order shall be issued deducting from payments then or thereafter due the Contractor the reasonable cost of correcting such deficiencies, including Owner's expenses (including attorneys' fees and consequential or incidental expenses) and compensation for the Architect's additional services made necessary by such default, neglect or failure Such action by the Owner and amounts charged to the Contractor are both subject to prior approval of the Architect-If payments then or thereafter due the Contractor such amounts, the Contractor shall pay the difference to the Owner within thirty (30) days after request.

## ALL ON STREEKINGHE FOR A DIS

All Document Al01 - 2007. Copyright 0 1011, 1015, 1018, 1027, 1037, 1031, 1056, 1051, 1055, 1055, 1057, 1076, 1077, 1007 and 2007 by The Aresican Institute of Architectry. All sights reserved HARNINO: This Alf Document is protected by U.S. Copyright Law and International Treaties, and will be projected to the maximum extent possible under the law. This disft was produced by AIR software at 14:15:10 on 01/16/2008 under Order No.1000314794\_1 which expires on 6/17/2008, and is not for resale. (3921070803)

#### Formatted: Font: Bold

Formatted: Tab stops: Not at 0.5"

Formatted: Justified, Tab stops: Not at 0.5"

#### ARTICLE 3 CONTRACTOR **§ 3.1 GENERAL**

§ 3.1.1 The Contractor is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number The Contractor shall be lawfully licensed, if required in the jurisdiction where the Project is located The Contractor shall designate in writing a representative who shall have express authority to bind the Contractor with respect to all matters under this Contract. The term "Contractor" means not be deemed an agent of the Owner for any reason,

§ 3.1.2 The Contractor shall perform the Work in strict accordance with the Contract Documents

§ 3.1.3 The Contractor shall not be relieved of obligations to perform the Work in strict accordance with the Contract Documents either by activities or duties of the Architect in the Architect's administration of the Contract, or by tests, inspections or approvals required or performed by persons or entities other than the Contractor

## § 3.2 REVIEW OF CONTRACT DOCUMENTS AND FIELD CONDITIONS BY CONTRACTOR

§ 3.2.13 section of the Contraction the Contractor is a representation that the Contractor has seened the site become generally tumblar with local conclutions under which the Work is to be performed with concluded personal observations with requirements of the Contract Documents I vecution of the Contract by the Contractor is a representation that the Contractor has visited the site, become generally familiar with local conditions paller which the Work is to be performed, correlated personal observations with requirements of the Contract Docimpents, and has satisfied uself as to the nature and location of the work, the general and local conditions, it cluding those bearing upon access (including partial or total restriction in access), transportation, disposal, staging, handling and storage of materials, availability of labor, water, electric power, roads and uncertainties of weather ground water table or similar physical conditions of the exornal, the character, maility and quantity of existing conditions to be encountered, the character of equipment and facilities needed prior to and during the prosecution of the work and all other matters which can many way effect the work of the cost thereof under this Contract. Any fadure by the Contractor to acquaint useff with all the available information concerning these conditions will not refrese the Contractor from any obligations with respect to his Contract.

§ 3.2.2 Because the Contract Documents are complementary, the Contractor shall, before starting each portion of the Work, carefully study and compare the various Contract Documents relative to that portion of the Work, as well as the information furnished by the Owner pursuant to Section 2.2.3, shall take field measurements of any existing conditions related to that portion of the Work, and shall observe any conditions at the site affecting it. These obligations are for the purpose of facilitating coordination and construction by the Contractor and are not for the purpose of discovering latent errors, omissions, or inconsistencies in the Contract Documents; however, the Contractor shall promptly report to the Architect any patent errors, inconsistencies or omissions discov made known to the Contractor as a request for information in such form as the Architect may require. It is recognized that the Contractor's review is made in the Contractor's capacity as a contractor and not as a licensed design professional, unless otherwise specifically provided in the Contract Documents

§ 3.2.3 The Contractor is not required to ascertain that the Contract Documents are in accordance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, but the Contractor shall promptly report to the Architect any nonconformity discovered by or made known to the Contractor as a request for information in such form as the Architect may require.

§ 3.2.4 If the Contractor believes that additional cost or time is involved because of clarifications or instructions the Architect issues in response to the Contractor's notices or requests for information pursuant to Sections 3.2.2 or 3 2 3, the Contractor shall make Claims as provided in Article 15 If the Contractor fails to perform the obligations of Sections 3.2.2 or 3.2.3, the Contractor shall pay such costs and damages to the Owner as would have been avoided if the Contractor had performed such obligations. If the Contractor performs those obligations, the Contractor shall not be liable to the Owner or Architect for damages resulting from errors, inconsistencies or omissions in the Contract Documents, for differences between field measurements or conditions and the Contract Documents, or for nonconformities of the Contract Documents to applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities

1. Show you introve the second secon second sec 2. All in the procession from a positive pringers. These Area applied by do Wain pod data area procession.

AIA Document AID1W - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1965, 1970, 1976, 1987, 1987, 1987 and 2007 by The American Institute of Jubblects. All sights reserved . UNING: This AIA Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA Document, or any portion of it, may result in severe civil and criminal ponaltics, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 24:15:10 on 01/12/2008 under Order No 1000314794\_i which expires on 8/17/2008, and is not for resals. 13 Veer Notes

concare a finite concentration to concentration of the second concentration of the sec

# § 3.3 SUPERVISION AND CONSTRUCTION PROCEDURES

§ 3.3.1 The Contractor shall supervise and direct the Work, using the Contractor's best skill and attention. The Contractor shall be solely responsible for, and have control over, construction means, methods, techniques, sequences and procedures and for coordinating all portions of the Work under the Contract, unless the Contract Documents give other specific instructions concerning these matters. If the Contract Documents give specific instruction means, methods, techniques, sequences or procedures, the Contractor shall evaluate the jobsite safety thereof and, except as stated below, shall be fully and solely responsible for the jobsite safety of such means, methods, techniques, sequences or procedures. If the Contractor determines that such means, methods, techniques, sequences or procedures lift the Contractor determines that such means, methods, techniques, sequences or procedures lift be Contractor shall give timely written notice to the Owner and Architect and shall not proceed with that portion of the Work without further written instructions from the Architect. If the Contractor is then instructed to proceed with the required means, methods, techniques, sequences or procedures sequences or procedures sequences, methods, techniques, sequences or procedures without acceptance of changes proposed by the Contractor, the Owner shall be solely responsible for any loss or damage arising solely from those Owner-required means, methods, techniques, sequences or procedures without acceptance of changes proposed by the Contractor, the Contractor, sequences or procedures without secuences or procedures means, methods, techniques, sequences or procedures without secuences or procedures are provided with the required means, methods, techniques, sequences or procedures without acceptance of changes proposed by the Contractor, the Owner shall be solely responsible for any loss or damage arising solely from those Owner-required means, methods, techniques, sequences or procedures without acceptance of changes proposed by

§ 3.3.2 The Contractor shall be responsible to the Owner for acts and omissions of the Contractor's employees, Subcontractors and their agents and employees, and other persons or entities performing portions of the Work for, or on behalf of, the Contractor or any of its Subcontractors

§ 3.3.3 The Contractor shall be responsible for inspection of portions of Work already performed to determine that such portions are in proper condition to receive subsequent Work

e <u>Seel</u> the Condectory had <u>evende</u> of the transferration of second and <u>oth</u>er become on which presserves <u>a stan</u>t

§ 3.3.5 You die volgen per binnend wet in de volgendied en dat Chinke op Auchine (Theat Robert Chine) auf dat Franzöhnte eksternen son en im

## § 3.4 LABOR AND MATERIALS

§ 3.4.1 Unless otherwise provided in the Contract Documents, the Contractor shall provide and pay for labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services necessary for proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work

§ 3.4.2 Except in the case of minor changes in the Work authorized by the Architect in accordance with Sections 3.12.8 or 7.4, the Contractor may make substitutions only with the consent of the Owner, after evaluation by the Architect and in accordance with a Change Order or Construction Change Directive

§ 3.4.3 The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the Work. The Contractor shall not permit employment of unfit persons or persons not properly skilled in tasks assigned to them.

and a state of the second s I have a verification of the second second

AIA Document A301 - - 2007. Copyright 0 1911, 1915, 1918, 1919, 1931, 1951, 1951, 1951, 1963, 1963, 1966, 1970, 1976, 1987, 1897 and 2007 by The American Institute of ArtUnice(: All sights sessared WARNING This AIA' Document is protocted by 0.5 Copyright Law and International Treaties: Unauthorized seproduction or distribution of this AIA' Document, or any portion of it, may result in severe divil and criminal ponaltics; and will be prosecuted to the maximum extern possible under the law This Graft was produced by AIA software at 14'13:10 on 01/16/2008 under Order No.1000314794\_1 which expires on 8/17/2008. and is not for resule (35218'0803)

the second s the state of the second s

#### § 3.5 WARRANTY

The Contractor warrants to the Owner and Architect that materials and equipment furnished under the Contract will be of good quality and new unless the Contract Documents require or permit otherwise. The Contractor further warrants that the Work will be of each group and that senate conform to the requirements of the Contract Documents and will be free from defects , except for those inherent in the quality of the Work the Contract Documents require or permit. Work, materials, or equipment not conforming to these requirements may be considered defective The Contractor's warranty excludes remedy for damage or defect caused by abuse, alterations to the Work not executed by the Contractor, improper or insufficient maintenance, improper operation, or normal wear and tear and normal usage. If required by the Architect, the Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment.

#### **§ 3.6 TAXES**

The Contractor shall pay sales, consumer, use and similar taxes for the Work provided by the Contractor that are legally enacted when bids are received or negotiations concluded, whether or not yet effective or merely scheduled to go into effect

#### § 3.7 PERMITS, FEES, NOTICES, AND COMPLIANCE WITH LAWS

§ 3.7.1 Unless otherwise provided in the Contract Documents, the Contractor shall secure and pay for the building permit as well as for other permits, fees, licenses, and inspections by government agencies necessary for proper execution and completion of the Work that are customarily secured after execution of the Contract and legally required at the time bids are received or negotiations concluded

§ 3.7.2 The Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities applicable to performance of the Work.

§ 3.7.3 If the Contractor performs Work !-rules and regulations, or lawful orders of public authorities, the Contractor shall assume appropriate responsibility for such Work and shall bear the costs attributable to correction.

§ 3.7.4 Concealed or Unknown Conditions. If the Contractor encounters conditions at the site that are (1) subsurface or otherwise concealed physical conditions that differ materially from those indicated in the Contract Documents or (2) unknown physical conditions of an unusual nature, that differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Contract Documents, the Contractor shall promptly provide notice to the Owner and the Architect before conditions are disturbed and in no event later than 21 days after first observance of the conditions. The Architect will promptly investigate such conditions and, if the Architect determines that they differ materially and cause an increase or decrease in the Contractor's cost of, or time required for, performance of any part of the Work, will recommend an equitable adjustment in the Contract Sum or Contract Time, or both 1f the Architect determines that the conditions at the site are not materially different from those indicated in the Contract Documents and that no change in the terms of the Contract is justified, the Architect shall promptly notify the Owner and Contractor in writing, stating the reasons If either party disputes the Architect's determination or recommendation, that party may proceed as provided in Article 15.

§ 3.7.5 If, in the course of the Work, the Contractor encounters human remains or recognizes the existence of burial markers, archaeological sites or wetlands not indicated in the Contract Documents, the Contractor shall immediately suspend any operations that would affect them and shall notify the Owner and Architect. Upon receipt of such notice, the Owner shall promptly take any action necessary to obtain governmental authorization required to resume the operations The Contractor shall continue to suspend such operations until otherwise instructed by the Owner but shall continue with all other operations that do not affect those remains or features. Requests for adjustments in the Contract Sum and Contract Time arising from the existence of such remains or features may be made as provided in Article 15

#### § 3.8 ALLOWANCES

§ 3.8.1 The Contractor shall include in the Contract Sum all allowances stated in the Contract Documents. Items covered by allowances shall be supplied for such amounts and by such persons or entities as the Owner may direct,

AIR Document A201 - 2007. Copyright © 1911, 1915, 1916, 1925, 1937, 1851, 1856, 1961, 1961, 1964, 1966, 1976, 1976, 1987, 1987 and 2007 by The Areatoch Inditute of Acchitects: All sights seesawed HORNING: This AIA Document is protected by U.S. Copyright Law and International Treaties Unathorized teproduction or distribution of this AIA Document, or any potion of it, may result in severe civil and original ponalties, and will be projected to the maximum extend possible under the law field and produced by AIA softwars at 14/13/10 en 07/18/7008 under Order No.1000314784\_1 which empires on 8/17/2038, and is not for resale User Notes: 15

but the Contractor shall not be required to employ persons or entities to whom the Contractor has reasonable objection-

§ 3.8.2 Unless otherwise provided in the Contract Documents,

- .1 allowances shall cover the cost to the Contractor of materials and equipment delivered at the site and all required taxes, less applicable trade discounts;
- .2 Contractor's costs for unloading and handling at the site, labor, installation costs, overhead, profit and other expenses contemplated for stated allowance amounts shall be included in the Contract Sum but not in the allowances; and
- •3 whenever costs are more than or less than allowances, the Contract Sum shall be adjusted accordingly by Change Order. The amount of the Change Order shall reflect (1) the difference between actual costs and the allowances under Section 3.8.2.1 and (2) changes in Contractor's costs under Section 3.8.2.2

§ 3.8.3 Materials and equipment under an allowance shall be selected by the Owner with reasonable promptness.

#### § 3.9 SUPERINTENDENT

§ 3.9.1 The Contractor shall employ a competent superintendent and necessary assistants who shall be in attendance at the Project site during performance of the Work on site. The superintendent shall represent the Contractor, and communications given to the superintendent shall be as binding as if given to the Contractor. It is group as the to the during the during the during the during the during during the transformation of the transformation of the transformation  $M_{1} = \frac{1}{2} \frac{1}{$ 

§ 3.9.2 The Contractor, as soon as practicable after award of the Contract, shall furnish in writing to the Owner through the Architect the name and qualifications of a proposed superintendent. The Architect may reply within 14 days to the Contractor in writing stating (1) whether the Owner or the Architect has reasonable objection to the proposed superintendent or (2) that the Architect requires additional time to review. Failure of the Architect to reply within the 14 day period shall constitute notice of no reasonable objection.

§ 3.9.3 The Contractor shall not employ a proposed superintendent to whom the Owner or Architect has made reasonable and timely objection. The Contractor shall not change the superintendent without the Owner's consent, which shall not unreasonably be withheld or delayed.

# § 3.10 CONTRACTOR'S CONSTRUCTION SCHEDULES

§ 3.10.1 The Contractor, promptly after being awarded the Contract, shall prepare and submit for the Owner's and Architect's information a Contractor's construction schedule for the Work. The schedule shall not exceed time limits current under the Contract Documents, shall be revised at appropriate intervals as required by the conditions of the Work and Project, shall be related to the entire Project to the extent required by the Contract Documents, and shall provide for expeditious and practicable execution of the Work.

§ 3.10.1.1 The Contractor's construction schedules shall be in a bar chart format, and shall depict of the visit of the set of the set of the set of the set from each flow weak of any set of the se

a A 18,6 kašti ir čeniti daita kies Perior data (kaj daita) a septi na terra ingli fasti kies kato man na serie Sea more da Rio andrite datali pergan Biglini.

§ 3.10.1.2 No less than once per month, the Contractor shall submit an updated construction schedule. The updated construction schedule shall depict actual start and completion dates for work commenced and, if appropriate, work completed. Additionally, the updated construction schedules shall depict updated estimates of anticipated commencement and completion dates for all upcoming work.

§ 3.10.1.3 Submission of the initial construction schedule and monthly schedule updates shall be absolute prerequisites of certification of the Contractor's application for payment.

§ 3.10.2 The Contractor shall prepare a submittal schedule, promptly after being awarded the Contract and thereafter as necessary to maintain a current submittal schedule, and shall submit the schedule(s) for the Architect's approval

ATA Document A201<sup>M</sup> - 2007. Copyright 0 1911, 1915, 1918, 1925, 1937, 1951, 1951, 1951, 1951, 1956, 1970, 1976, 1997, 1997 and 2007 by The American Institute of Arceitectus All sights reserved MARHINO This AIA Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA Document, or any portion of it, may result in server six11 and original positize, and will be presented to the surface possible undor the law. This draft way produced by AIA software st 13:10 on 0/18/2008 under Order No.1000314754 1 which supress on 2/17/2008, and is not for resale. User Notes: (3921470603)

#### Formatted: Font: (Default) Times New Roman

Formatted: Font: (Default) Times New Roman

The Architect's approval shall not unreasonably be delayed or withheld. The submittal schedule shall (1) be coordinated with the Contractor's construction schedule, and (2) allow the Architect reasonable time to review submittals. If the Contractor fails to submit a submittal schedule, the Contractor shall not be entitled to any increase in Contract Sum or extension of Contract Time based on the time required for review of submittals

§ 3.10.3 The Contractor shall perform the Work in general accordance with the most recent schedules submitted to the Owner and Architect heade up contracting all backets to the design of the rest of the is a manufactor first ministry with our main separative and an exception of pathy our manufactor and I have a first a second off the meeting of the contract of the track of the Owner shall be entitled to rely on Contractor's schedules for coordination of its and activities, as well as the activities of other contractors working at the Project site or on the Project

## § 3.11 DOCUMENTS AND SAMPLES AT THE SITE

The Contractor shall maintain at the site for the Owner one copy of the Drawings, Specifications, Addenda, Change Orders and other Modifications, in good order and marked currently to indicate field changes and selections made during construction, and one copy of approved Shop Drawings, Product Data, Samples and similar required submittals (all collectively referred to herein as "Record Documents") These shall be available to the Architect and shall be delivered to the Architect for submittal to the Owner upon completion of the Work as a record of the Work as constructed

§ 3.11.1 The Contractor will make the Record Documents available for inspection by the Architect upon reasonable notice. Adequate maintenance of the Record Documents shall be a prerequisite to certification of the Contractor's application for payment.

#### § 3.12 SHOP DRAWINGS, PRODUCT DATA AND SAMPLES

§ 3.12.1 Shop Drawings are drawings, diagrams, schedules and other data specially prepared for the Work by the Contractor or a Subcontractor, Sub-subcontractor, manufacturer, supplier or distributor to illustrate some portion of the Work

§ 3.12.2 Product Data are illustrations, standard schedules, performance charts, instructions, brochures, diagrams and other information furnished by the Contractor to illustrate materials or equipment for some portion of the Work

§ 3.12.3 Samples are physical examples that illustrate materials, equipment or workmanship and establish standards by which the Work will be judged

§ 3.12.4 Shop Drawings, Product Data, Samples and similar submittals are not Contract Documents. Their purpose is to demonstrate the way by which the Contractor proposes to conform to the information given and the design concept expressed in the Contract Documents for those portions of the Work for which the Contract Documents require submittals The Contractor shall submit Product Data for all equipment and materials incorporated into the finished work. Review by the Architect is subject to the limitations of Section 4.2.7 Informational submittals upon which the Architect is not expected to take responsive action may be so identified in the Contract Documents. Submittals that are not required by the Contract Documents may be returned by the Architect without action.

§ 3.12.5 The Contractor shall review for compliance with the Contract Documents, approve and submit to the Architect Shop Drawings, Product Data, Samples and similar submittals required by the Contract Documents in accordance with the submittal schedule approved by the Architect or, in the absence of an approved submittal schedule, with reasonable promptness and in such sequence as to cause no delay in the Work or in the activities of the Owner or of separate contractors

§ 3.12.6 By submitting Shop Drawings, Product Data, Samples and similar submittals, the Contractor represents to the Owner and Architect that the Contractor has (1) reviewed and approved them, (2) determined and verified materials, field measurements and field construction criteria related thereto, or will do so and (3) checked and coordinated the information contained within such submittals with the requirements of the Work and of the Contract Documents

§ 3.12.7 The Contractor shall perform no portion of the Work for which the Contract Documents require submittal and review of Shop Drawings, Product Data, Samples or similar submittals until the respective submittal has been approved by the Architect

AIA Document A201<sup>22</sup> - 2007 Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1951, 1951, 1963, 1966, 1876, 1976, 1967, 1987 and 2007 by The American InStitute of Architect: All rights reserved. WARNING: This AIA Document is protocted by U.S. Copyright Law and International Treatices Unauthorized reproduction or distribution of this AIA" Document; or any portion of it, may result in revere civil and criminal penaltics, and will be prosecuted to the maximum extent possible under the law. This deaf was produced by AIA software at 16:15:10 on 01/16/2008 under Order No.1000314794\_1 which expires on 0/17/2009, and is not for resele 17 139218708031

§ 3.12.8 The Work shall be in accordance with approved submittals except that the Contractor shall not be relieved of responsibility for deviations from requirements of the Contract Documents by the Architect's approval of Shop Drawings, Product Data, Samples or similar submittals unless the Contractor has specifically informed the Architect in writing of such deviation at the time of submittal and (1) the Architect has given written approval to the specific deviation as a minor change in the Work, or (2) a Change Order or Construction Change Directive has been issued authorizing the deviation. The Contractor shall not be relieved of responsibility for errors or omissions in Shop Drawings, Product Data, Samples or similar submittals by the Architect's approval thereof.

§ 3.12.9 The Contractor shall direct specific attention, in writing or on resubmitted Shop Drawings, Product Data, Samples or similar submittals, to revisions other than those requested by the Architect on previous submittals. In the absence of such written notice, the Architect's approval of a resubmission shall not apply to such revisions

§ 3.12.10 The Contractor shall not be required to provide professional services that constitute the practice of architecture or engineering unless such services are specifically required by the Contract Documents for a portion of the Work or unless the Contractor needs to provide such services in order to carry out the Contractor's responsibilities for construction means, methods, techniques, sequences and procedures. The Contractor shall not be required to provide professional services in violation of applicable law. If professional design services or certifications by a design professional related to systems, materials or equipment are specifically required of the Contractor by the Contract Documents, the Owner and the Architect will specify all performance and design criteria that such services must satisfy. The Contractor shall cause such services or certifications to be provided by a properly licensed design professional, whose signature and seal shall appear on all drawings, calculations, specifications, certifications, Shop Drawings and other submittals prepared by such professional Shop Drawings and other submittals related to the Work designed or certified by such professional, if prepared by others, shall bear such professional's written approval when submitted to the Architect. The Owner and the Architect shall be entitled to rely upon the adequacy, accuracy and completeness of the services, certifications and approvals performed or provided by such design professionals, provided the Owner and Architect have specified to the Contractor all performance and design criteria that such services must satisfy Pursuant to this Section 3 12 10, the Architect will review, approve or take other appropriate action on submittals only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Contractor shall not be responsible for the adequacy of the performance and design criteria specified in the Contract Documents

#### § 3.13 USE OF SITE

The Contractor shall confine operations at the site to areas permitted by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities and the Contract Documents and shall not unreasonably encumber the site with materials or equipment

#### § 3.14 CUTTING AND PATCHING

§ 3.14.1 The Contractor shall be responsible for cutting, fitting or patching required to complete the Work or to make its parts fit together properly. All areas requiring cutting, fitting and patching shall be restored to the condition existing prior to the cutting, fitting and patching, unless otherwise required by the Contract Documents

§ 3.14.2 The Contractor shall not damage or endanger a portion of the Work or fully or partially completed construction of the Owner or separate contractors by cutting, patching or otherwise altering such construction, or by excavation. The Contractor shall not cut or otherwise alter such construction by the Owner or a separate contractor except with written consent of the Owner and of such separate contractor, such consent shall not be unreasonably withhold from the Owner or a separate contractor the Contractor's consent to cutting or otherwise altering the Work.

#### § 3.15 CLEANING UP

§ 3.15.1 The Contractor shall keep the premises and surrounding area free from accumulation of waste materials or rubbish caused by operations under the Contract. At completion of the Work, the Contractor shall remove waste materials, rubbish, the Contractor's tools, construction equipment, machinery and surplus materials from and about the Project. The tractor's tools, construction equipment, machinery and surplus materials from and about the Project. The tractor of the Work at the tractor of the t

AIA Documant A201" - 2007 Copyright © 1911, 1916, 1918, 1925, 1917, 1981, 1988, 1981, 1983, 1988, 1970, 1976, 1987, 1987, 1987 and 2007 by The American Institute of Acchinetry All sights reserved. USUBNO Classific District Spreaded by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this RAX Document, or any portion of it, may result in severe civil and criminal ponalizes, and will be presecuted to the maximum extent possible under the law. This diaft was produced by AIA software at 14:15:10 on 0/1/8/2008 under Order No.1000314794 1 which empires on 0/17/2008, and is not for resale (3521870803) § 3.15.2 If the Contractor fails to clean up as provided in the Contract Documents, the Owner may do so and Owner shall be entitled to reimbursement from the Contractor

#### § 3.16 ACCESS TO WORK

The Contractor shall provide the Owner and Architect access to the Work in preparation and progress wherever located

## § 3.17 ROYALTIES, PATENTS AND COPYRIGHTS

The Contractor shall pay all royalties and license fees The Contractor shall defend suits or claims for infringement of copyrights and patent rights and shall hold the Owner and Architect harmless from loss on account thereof, but shall not be responsible for such defense or loss when a particular design, process or product of a particular manufacturer or manufacturers is required by the Contract Documents, or where the copyright violations are contained in Drawings, Specifications or other documents prepared by the Owner or Architect. However, if the Contractor has reason to believe that the required design, process or product is an infringement of a copyright or a patent, the Contractor shall be responsible for such loss unless such information is promptly furnished to the Architect

#### § 3.18 INDEMNIFICATION

§ 3.18.1 To the fullest extent permitted by law the Contractor shall indemnify and hold harmless the Owner, Architect, Architect's consultants, and agents and employees of any of them from and against claims, damages, losses and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work, provided that such claim, damage, loss or expense is attributable to bodily injury, siekness, disease or death, or to injury to or destruction of tangible property (other than the Work itself), but only to the extent caused by the market and acts or omissions of the Contractor, a Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts they may be liable, regardless of whether or not such claim, damage, loss or expense is caused in part by a party indemnified hereunder. Such obligation shall not be construed to negate, abridge, or reduce other rights or obligations of indemnity which would otherwise exist as to a party or person described in this Section 3 18

§ 3.18.2 In claims against any person or entity indemnified under this Section 3 18 by an employee of the Contractor, a Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts they may be liable, the indemnification obligation under Section 3 18 1 shall not be limited by a limitation on amount or type of damages, compensation or benefits payable by or for the Contractor or a Subcontractor under workers' compensation acts, disability benefit acts or other employee benefit acts

## ARTICLE 4 ARCHITECT

## § 4.1 GENERAL

§ 4.1.1 The Owner shall retain an architect lawfully licensed to practice architecture or an entity lawfully practicing architecture in the jurisdiction where the Project is located. That person or entity is identified as the Architect in the Agreement and is referred to throughout the Contract Documents as if singular in number.

§ 4.1.2 Duties, responsibilities and limitations of authority of the Architect as set forth in the Contract Documents shall not be restricted, modified or extended without written consent of the Owner, Contractor and Architect. Consent shall not be unreasonably withheld.

§4.1.3 If the employment of the Architect is terminated, the Owner shall employ a successor architect as to whom the Contractor has no reasonable objection and whose status under the Contract Documents shall be that of the Architect.

## § 4.2 ADMINISTRATION OF THE CONTRACT

§ 4.2.1 The Architect will provide administration of the Contract as described in the Contract Documents and will be an Owner's representative during construction until the date the Architect issues the final Certificate For Payment. The Architect will have authority to act on behalf of the Owner only to the extent provided in the Contract Documents

§ 4.2.2 The Architect will visit the site at intervals appropriate to the stage of construction, or as otherwise agreed with the Owner, to become generally familiar with the progress and quality of the portion of the Work completed,

AIL Document AIO1" - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1959, 1961, 1965, 1970, 1976, 1987, 1987 and 2007 by The American Institute of Archiveria All Eights reserved. KAURING: Thia Alf. Document's protocted by U.S. Copyright Law and International Ponalties, and will be presecuted to the maximum extent possible under the law. This draft was produced by AIA softwars at 14-15:10 on U/19/2008 under Order Wo.1000314794 1 which expires on 8/17/2008, and is not for reseals. User Motes: (3921270803) and to determine in general if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect will not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. The Architect will not have control over, charge of, or responsibility for, the construction means, methods, techniques, sequences or procedures, or for the safety precautions and programs in connection with the Work, since these are solely the Contractor's rights and responsibilities under the Contract Documents, except as provided in Section 3.3.1.

§ 4.2.3 On the basis of the site visits, the Architect will keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and report to the Owner (1) known deviations from the Contract Documents and from the most recent construction schedule submitted by the Contractor, and (2) defects and deficiencies observed in the Work. The Architect will not be responsible for the Contractor's failure to perform the Work in accordance with the requirements of the Contract Documents. The Architect will not have control over or charge of and will not be responsible for acts or omissions of the Contractor, Subcontractors, or their agents or employees, or any other persons or entities performing portions of the Work

# § 4.2.4 COMMUNICATIONS FACILITATING CONTRACT ADMINISTRATION

Except as otherwise provided in the Contract Documents or when direct communications have been specially authorized, the Owner and Contractor shall endeavor to communicate with each other through the Architect about matters arising out of or relating to the Contract. Communications by and with the Architect's consultants shall be through the Architect Communications by and with Subcontractors and material suppliers shall be through the Contractor Communications by and with separate contractors shall be through the Owner.

§ 4.2.5 Based on the Architect's evaluations of the Contractor's Applications for Payment, the Architect will review and certify the amounts due the Contractor and will issue Certificates for Payment in such amounts

§ 4.2.6 The Architect has authority to reject Work that does not conform to the Contract Documents Whenever the Architect considers it necessary or advisable, the Architect will have authority to require inspection or testing of the Work in accordance with Sections 13.5.2 and 13.5.3, whether or not such Work is fabricated, installed or completed However, neither this authority of the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Contractor, Subcontractors, material and equipment suppliers, their agents or employees, or other persons or entities performing portions of the Work

§ 4.2.7 The Architect will review and approve, or take other appropriate action upon, the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Architect's action will be taken in accordance with the submittal schedule approved by the Architect or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time in the Architect's professional judgment to permit adequate review. Review of such submittals is not conducted for the purpose of determining the accuracy and completeness of other details such as dimensions and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of the Contractor as required by the Contract Documents. The Architect's review of the Contractor's submittal shall not relieve the Contractor of the obligations under Sections 3 3, 3.5 and 3 12. The Architect's review shall not constitute approval of safety precautions or, unless otherwise specifically stated by the Architect, of any construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.

§ 4.2.8 The Architect will prepare Change Orders and Construction Change Directives, and may authorize minor changes in the Work as provided in Section 7.4. The Architect will investigate and make determinations and recommendations regarding concealed and unknown conditions as provided in Section 3.7.4.

§ 4.2.9 The Architect will conduct inspections to determine the date or dates of Substantial Completion and the date of final completion; issue Certificates of Substantial Completion pursuant to Section 9 8, receive and forward to the Owner, for the Owner's review and records, written warranties and related documents required by the Contract and assembled by the Contractor pursuant to Section 9 10; and issue a final Certificate for Payment pursuant to Section 9 10

AIA Decument A201<sup>m</sup> - 2007. Copyright 0 1011, 1018, 1018, 1025, 1037, 1051, 1051, 1051, 1066, 1970, 1376, 1897, 1097 and 2007 by The Averican Institute of Scientiscies. All rights remerved. WARNING: This AIA Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA Document, or any portion of it, may result in severe civil and criminal posities, and will be producted to the maximum oxtont porsible under the law. This diaft was produced by AIA software it 14.15:10 on 0/12/2000 under Crdwr No.1000514794 2 which expires on 8/17/2008, and is not for resele. (3921870803) § 4.2.10 If the Owner and Architect agree, the Architect will provide one or more project representatives to assist in carrying out the Architect's responsibilities at the site. The duties, responsibilities and limitations of authority of such project representatives shall be as set forth in an exhibit to be incorporated in the Contract Documents.

§ 4.2.11 The Architect will interpret and decide matters concerning performance under, and requirements of, the Contract Documents on written request of either the Owner or Contractor. The Architect's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness.

§ 4.2.12 Interpretations and decisions of the Architect will be consistent with the intent of, and reasonably inferable from, the Contract Documents and will be in writing or in the form of drawings. When making such interpretations and decisions, the Architect will endeavor to secure faithful performance by both Owner and Contractor, will not show partiality to either and will not be liable for results of interpretations or decisions rendered in good faith

§ 4.2.13 The Architect's decisions on matters relating to aesthetic effect will be final if consistent with the intent expressed in the Contract Documents.

§ 4.2.14 The Architect will review and respond to requests for information about the Contract Documents. The Architect's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness. If appropriate, the Architect will prepare and issue supplemental Drawings and Specifications in response to the requests for information. If requested by the Architect, all requests for information shall be submitted to the Architect in a format acceptable to the Architect.

# ARTICLE 5 SUBCONTRACTORS

## § 5.1 DEFINITIONS

§ 5.1.1 A Subcontractor is a person or entity who has a direct contract with the Contractor to perform a portion of the Work at the site. The term "Subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Subcontractor or an authorized representative of the Subcontractor. The term "Subcontractor" does not include a separate contractor or subcontractors of a separate contractor

§ 5.1.2 A Sub-subcontractor is a person or entity who has a direct or indirect contract with a Subcontractor to perform a portion of the Work at the site. The term "Sub-subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Sub-subcontractor or an authorized representative of the Subsubcontractor

# § 5.2 AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OF THE WORK

§ 5.2.1 Unless otherwise stated in the Contract Documents or the bidding requirements, the Contractor, as soon as practicable after award of the Contract, shall furnish in writing to the Owner through the Architect the names of persons or entities (including those who are to furnish materials or equipment fabricated to a special design) proposed for each principal portion of the Work. The Architect may reply within 14 days to the Contractor in writing stating (1) whether the Owner or the Architect has reasonable objection to any such proposed person or entity or (2) that the Architect requires additional time for review. Failure of the Owner or Architect to reply within the 14 day period shall constitute notice of no reasonable objection.

§ 5.2.2 The Contractor shall not contract with a proposed person or entity to whom the Owner or Architect has made reasonable and timely objection. The Contractor shall not be required to contract with anyone to whom the Contractor has made reasonable objection.

§ 5.2.3 If the Owner or Architect has reasonable objection to a person or entity proposed by the Contractor, the Contractor shall propose another to whom the Owner or Architect has no reasonable objection. If the proposed but rejected Subcontractor was reasonably capable of performing the Work, the Contract Sum and Contract Time shall be increased or decreased by the difference, if any, occasioned by such change, and an appropriate Change Order shall be issued before commencement of the substitute Subcontractor's Work. However, no increase in the Contract Sum or Contract Time shall be allowed for such change unless the Contractor has acted promptly and responsively in submitting names as required

§ 5.2.4 The Contractor shall not substitute a Subcontractor, person or entity previously selected if the Owner or Architect makes reasonable objection to such substitution

ATA Decrement A201 - 2007. Copyright 6 1911, 1915, 1918, 1925, 1937, 1931, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1987, 1987 and 2007 by The American Institute of Archivects All rights reserved. WARNING: This All Document is protected by U.S. Copyright Law and international Trasties. Unauthorized reproduction or distribution of this ALX Document, or any portion of 16, may result is severe cavil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 14:15:10 on 0/1/16/2008 under Order No 1000314794 1 which expires on 8/17/2008, and is not for resule. User Notes

## § 5.3 SUBCONTRACTUAL RELATIONS

By appropriate agreement, written where legally required for validity, the Contractor shall require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities, including the responsibility for safety of the Subcontractor's Work, which the Contractor, by these Documents, assumes toward the Owner and Architect Each subcontract agreement shall preserve and protect the rights of the Owner and Architect Documents with respect to the Work to be performed by the Subcontractor so that subcontracting thereof will not prejudice such rights, and shall allow to the Subcontractor, unless specifically provided otherwise in the subcontract Documents, has against the Owner. Where appropriate, the Contractor shall require each Subcontractor to enter into similar agreements with Sub-subcontract agreement, copies of the Contract Documents to which the Subcontractor, and upon written request of the Subcontractor, identify to the Subcontractor terms and conditions of the proposed subcontract agreement that may be at variance with the Contract Documents Subcontractors will similarly make copies of applicable portions of such documents available to their respective proposed Sub-subcontractors.

#### § 5.4 CONTINGENT ASSIGNMENT OF SUBCONTRACTS

§ 5.4.1 Each subcontract agreement for a portion of the Work is assigned by the Contractor to the Owner, provided that

- .1 assignment is effective only after termination of the Contract by the Owner accepts by notifying the Subcontractor and Contractor in writing, and
- .2 assignment is subject to the prior rights of the surety, if any, obligated under bond relating to the Contract.

When the Owner accepts the assignment of a subcontract agreement, the Owner assumes the Contractor's rights and obligations under the subcontract

3.3.2 Communication of the state of the s

2.4 as 4 per set of the set

# ARTICLE 6 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS § 6.1 OWNER'S RIGHT TO PERFORM CONSTRUCTION AND TO AWARD SEPARATE CONTRACTS

§ 6.1.1 The Owner reserves the right to perform construction or operations related to the Project with the Owner's own forces, and to award separate contracts in connection with other portions of the Project or other construction or operations on the site under Conditions of the Contract identical or substantially similar to these including those portions related to insurance and waiver of subrogation. If the Contractor claims that delay or additional cost is involved because of such action by the Owner, the Contractor shall make such Claim as provided in Article 15

§ 6.1.2 When separate contracts are awarded for different portions of the Project or other construction or operations on the site, the term "Contractor" in the Contract Documents in each case shall mean the Contractor who executes each separate Owner-Contractor Agreement.

§ 6.1.3 The Owner shall provide for coordination of the activities of the Owner's own forces and of each separate contractor with the Work of the Contractor, who shall cooperate with them. The Contractor shall participate with other separate contractors and the Owner in reviewing their construction schedules. The Contractor shall make any revisions to the construction schedule deemed necessary after a joint review and mutual agreement. The construction schedules shall then constitute the schedules to be used by the Contractor, separate contractors and the Owner until subsequently revised

# 4.4.1. A start of the second start of the s

AIA Document A201<sup>m</sup> - 2007. Copyright © 1811, 1815, 1818, 1825, 1837, 1851, 1858, 1961, 1866, 1970, 1876, 1887, 1987 and 2007 by The Areactan InStitute of Archivuts All rights reserved. HARHINO This AIA' Document is protected by U.S. Copyright Law and International ponalizes, and will be protected to the maximum extent possible under the law. This draft was produced by AIA software at 14:15:10 on U/18/2008 and is not for resals Uart Metes: (3921870803) provide the strength of the st

### § 6.2 MUTUAL RESPONSIBILITY

§ 6.2.1 The Contractor shall afford the Owner and separate contractors reasonable opportunity for introduction and storage of their materials and equipment and performance of their activities, and shall connect and coordinate the Contractor's construction and operations with theirs as required by the Contract Documents

§ 6.2.2 If part of the Contractor's Work depends for proper execution or results upon construction or operations by the Owner or a separate contractor, the Contractor shall, prior to proceeding with that portion of the Work, promptly report to the Architect apparent discrepancies or defects in such other construction that would render it unsuitable

for such proper execution and results Failure of the Contractor so to report shall constitute an acknowledgment that the Owner's or separate contractor's completed or partially completed construction is fit and proper to receive the Contractor's Work, except as to defects not then reasonably discoverable.

§ 6.2.3 The Contractor shall reimburse the Owner for costs the Owner incurs that are payable to a separate contractor because of the Contractor's delays, improperly timed activities or defective construction. <u>Alternatively, the</u> <u>Contractor shall pay separate contractors directly for such costs</u>. The <u>separate contractors Owner shall be</u> responsible to the Contractor for costs the Contractor incurs because of a separate contractor's delays, improperly timed activities, damage to the Work or defective construction. For the purposes of facilitating this Section, the <u>Contractor and separate contractors shall be deemed third party beneficiaries to each other's respective contracts</u> with the Owner.

§ 6.2.4 The Contractor shall promptly remedy damage the Contractor wrongfully causes to completed or partially completed construction or to properly of the Owner, separate contractors as provided in Section 10.2.5.

§ 6.2.5 The Owner and each separate contractor shall have the same responsibilities for cutting and patching as are described for the Contractor in Section 3 14

### § 6.3 OWNER'S RIGHT TO CLEAN UP

If a dispute arises among the Contractor, separate contractors and the Owner as to the responsibility under their respective contracts for maintaining the premises and surrounding area free from waste materials and rubbish, the Owner may clean up and the Architect will allocate the cost among those responsible

ARTICLE 7 CHANGES IN THE WORK

§ 7.1 GENERAL

### § 7.1.1 The Owner may, without invalidating the Contract and without notice to the surety, direct changes in the

Work Changes in the Work may be accomplished after execution of the Contract, and without invalidating the Contract, by Change Order, Construction Change Directive or order for a minor change in the Work, subject to the limitations stated in this Article 7 and elsewhere in the Contract Documents

§ 7.1.2 A Change Order shall be based upon agreement among the Owner, Contractor and Architect, a Construction Change Directive requires agreement by the Owner and Architect and may or may not be agreed to by the Contractor; an order for a minor change in the Work may be issued by the Architect alone.

§ 7.1.3 Changes in the Work shall be performed under applicable provisions of the Contract Documents, and the Contractor shall proceed promptly, unless otherwise provided in the Change Order, Construction Change Directive or order for a minor change in the Work

§7.1.4 Notwithstanding any other provision to the contrary in the Contract Documents, upon receipt of a direction for Change, the Contractor shall immediately implement the Change into the Work

§7.1.5 No Change Order shall be approved unless preceded by written direction for Change is provided by the Owner or Architect. This requirement cannot be waived. There shall be no implicit or constrictive change order.

### § 7.2 CHANGE ORDERS

§ 7.2.1 A Change Order is a written instrument prepared by the Architect and signed by the Owner, Contractor and Architect stating their agreement upon all of the following:

ATA Bocument A101 - 2007. Copyright to 1911, 1915, 1818, 1825, 1837, 1951, 1858, 1951, 1863, 1866, 1876, 1876, 1887, 1887, 1887 and 2007 by The American Institute of Broblect's All rights reserved. WARNING: This AIA '...cument is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA Document, or any portion of it, may result in server civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 14:15:10 on 01/16/2010 under Order No.1000314794 1 which expires on 8/17/2008, and is not for resels (3921070603)

Formatted: Font: (Default) Times New Roman

Formatted: Font: (Default) Times New Roman, Not Bold

- .1 The change in the Work,
- .2 The amount of the adjustment, if any, in the Contract Sum, and
- .3 The extent of the adjustment, if any, in the Contract Time.

### § 7.2.2 No payment for change in the Work shall be made until such change has been memorialized in an executed Change Order and the Change has been implemented into the Work. Partial payments on partially implemented changes shall be paid similarly as partial payments on base contract work.

§ 7,2.3 Contractors shall be permitted the following markaps for charves orders. Additional markap for psurance or bonds will not be allowed. All change order requests must be submitted with the following backup information or they will not be reviewed by the Architect or Owner. Provide material and labor quantities material and costs. Labor rates and any other substantiating data to explain the change order amount.

Permitted markup for Change Orders

Additive Change Order 10% Deductive Change Order 5% -

### § 7.3 CONSTRUCTION CHANGE DIRECTIVES

§7.3.1 A Construction Change Directive is a written order prepared by the Architect and signed by the Owner and Architect, directing a change in the Work prior to agreement on adjustment, if any, in the Contract Sum or Contract Time, or both The Owner may by Construction Change Directive, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions, the Contract Sum and Contract Time being adjusted accordingly.

§7.3.2 A Construction Change Directive shall be used in the absence of total agreement on the terms of a Change Order for the purposes of defining the change and how payment shall be calculated, but not for the purpose of approving payment

§ 7.3.3 If the Construction Change Directive provides for an adjustment to the Contract Sum, the adjustment shall be based on one of the following methods

- .1 Mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to permit evaluation,
- .2 Unit prices stated in the Contract Documents or subsequently agreed upon,
- .3 Cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or percentage fee, or
- .4 As provided in Section 7.3.7

§ 7.3.4 If unit prices are stated in the Contract Documents or subsequently agreed upon, and if quantities originally contemplated are materially changed (i.e., by more than 25%) in a proposed Change Order or Construction Change Directive, so that application of such unit prices to quantities of Work proposed will cause substantial inequity to the Owner or Contractor, the applicable unit prices shallmay be equitably adjusted.

§7.3.5 Upon receipt of a Construction Change Directive, the Contractor shall promptly proceed with the change in the Work involved and advise the Architect of the Contractor's agreement or disagreement with the method, if any, provided in the Construction Change Directive for determining the proposed adjustment in the Contract Sum or Contract Time

§ 7.3.6 A Construction Change Directive signed by the Contractor indicates the Contractor's agreement therewith, including adjustment in Contract Sum and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Change Order.

§ 7.3.7 If the Contractor does not respond promptly or disagrees with the method for adjustment in the Contract Sum, the Architect shall determine the method and the adjustment on the basis of reasonable expenditures and savings of those performing the Work attributable to the change, including, in case of an increase in the Contract Sum, an amount for overhead and profit <u>restriction (Stated in S7.2.3) as set forth in the Agreement, or if no such amount is set forth in the Agreement, or if no such amount is set forth in the Agreement, in such case, and also under Section 7.3.3.3, the Contractor shall keep and present, in such form as the Architect may prescribe, an itemized accounting together with appropriate supporting</u>

ATA Document ASO1<sup>20</sup> - 2007. Copyright © 1913, 1915, 1915, 1928, 1937, 1953, 1953, 1954, 1963, 1966, 1970, 1976, 1987, 1987 and 2007 by The Averian Institute of Aschitects All Eights reserved. KANNING: This AIA Document is protected by U.S. Copyright Law and International prostice Unauthorized reproduction or distribution of this AIA" Document is protected by U.S. Copyright Law and International penalties, and will be prosocuted to the maximum extent possible under the Law. This draft was produced by AIA softwate at 14:15.10 on 0//8/2008 under Order No 1000314784 1 which expires on 8/27/2008, and is not for resale. User Notes:

### Formatted: AIA Body Text Indented

Formatted: AIA Body Text Indented, Indent: Hanging: 0.5" data Unless otherwise provided in the Contract Documents, costs for the purposes of this Section 7.3.7 shall be limited to the following

- .1 Costs of labor, including social security, old age and unemployment insurance, fringe benefits required by agreement or custom, and workers' compensation insurance;
- .2 Costs of materials, supplies and equipment, including cost of transportation, whether incorporated or consumed.
- .3 Rental costs of machinery and equipment, exclusive of hand tools, whether rented from the Contractor or others,
- Costs of premiums for all bonds and insurance, permit fees, and sales, use or similar taxes related to .4 the Work, and
- Additional costs of supervision and field office personnel directly attributable to the change .5

§ 7.3.8 The amount of credit to be allowed by the Contractor to the Owner for a deletion or change that results in a net decrease in the Contract Sum shall be actual net cost as confirmed by the Architect. When both additions and credits covering related Work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change

§7.3.9 Pending final determination of the total cost of a Construction Change Directive to the Owner, the Contractor may request payment for Work completed under the Construction Change Directive in Applications for Payment. The Architect will make an interim determination for purposes of monthly certification for payment for those costs and certify for payment the amount that the Architect determines, in the Architect's professional judgment, to be reasonably justified The Architect's interim determination of cost shall adjust the Contract Sum on the same basis as a Change Order, subject to the right of either party to disagree and assert a Claim in accordance with Article 15.

§ 7.3.10 When the Owner and Contractor agree with a determination made by the Architect concerning the adjustments in the Contract Sum and Contract Time, or otherwise reach agreement upon the adjustments, such agreement shall be effective immediately and the Architect will prepare a Change Order Change Orders may be issued for all or any part of a Construction Change Directive.

### § 7.4 MINOR CHANGES IN THE WORK

The Architect has authority to order minor changes in the Work not involving adjustment in the Contract Sum or extension of the Contract Time and not inconsistent with the intent of the Contract Documents. Such changes will be effected by written order signed by the Architect and shall be binding on the Owner and Contractor

### ARTICLE 8 TIME

### **§ 8.1 DEFINITIONS**

§ 8.1.1 Unless otherwise provided, Contract Time is the period of time, including authorized adjustments, allotted in the Contract Documents for Substantial Completion of the Work

§ 8.1.2 The date of commencement of the Work is the date established in the Agreement

§ 8.1.3 The date of Substantial Completion is the date certified by the Architect in accordance with Section 9.8

§ 8.1.4 The term "day" as used in the Contract Documents shall mean calendar day unless otherwise specifically defined

### § 8.2 PROGRESS AND COMPLETION

§ 8.2.1 Time limits stated in the Contract Documents are of the essence of the Contract. By executing the Agreement the Contractor confirms that the Contract Time is a reasonable period for performing the Work

§ 8.2.2 The Contractor shall not knowingly, except by agreement or instruction of the Owner in writing, prematurely commence operations on the site or elsewhere prior to the effective date of insurance required by Article 11 to be furnished by the Contractor and Owner The date of commencement of the Work shall not be changed by the effective date of such insurance

§ 8.2.3 The Contractor shall proceed expeditiously with adequate forces and shall achieve Substantial Completion and Final Completion within the Contract Time Unless provided elsewhere in the Contract Documents, Final Completion shall be completed within thirty (30) days following Substantial Completion

AIA Document A201" - 2007. Copyright & 1911, 1915, 1918, 1973, 1977, 1951, 1954, 1961, 1963, 1966, 1976, 1987, 1987, 1987 and 2007 by The Arrical Institute of Architects. All rights reserved. MANNNG. This AIA Document is protected by U.S. Copyright faw and International Treaties Unauthorized reproduction or distribution of this AIA Document, or any portion of it, may result in severe divid and criminal penaltics, and will be prosecuted to the maximum extent possible under the law This draft was produced by AIA software at 14:15:10 on 01/18/2008 under Order Mo.1000314794 1 which expires on 8/17/2008, and is not for resale. 25 139218708031

### § 8.3 DELAYS AND EXTENSIONS OF TIME

§ 8.3.1 If the Contractor is delayed at any time in the commencement or progress of the Work by an act or neglect of the Owner or Architect, or of an employee of either, or of a separate contractor employed by the Owner; or by changes ordered in the Work; or by labor disputes, fire, unusual delay in deliveries, unavoidable casualties or other causes beyond the Contractor's control, or by delay authorized by the Owner pending mediation and arbitration, or by other causes that the Architect determines may justify delay, then the Contract Time shall be extended by Change Order for such reasonable time as the Architect may determine

§ 8.3.2 Claims relating to time shall be made in accordance with applicable provisions of Article 15

§ 8.3.3 This Section 8.3 does not preclude recovery of damages for delay by either party under other provisions of the Contract Documents- Extension of Contract Time pursuant to this Article shall be the Contractor's sole and exclusive remedy for delay

§ 8.3.4 Extension of Contract Time resulting from Changes in the Work shall be negotiated into respective Change Orders. Whenever the Contractor seeks an adjustment in the Contract Time as part of a Claim or Change Order, the Contractor shall justify the request with proper reference to the approved construction schedules. All executed Change Orders shall be deemed to include adjustments in the Contract Time, if any, resulting from the underlying Change in the Work.

§8.3.5 to estimate of the broad add common additional services necessitated by Contractor's failure to achieve Substantial Completion within the time established in the Agreement and for more than one inspection for each Substantial Completion and Final Completion.

### ARTICLE 9 PAYMENTS AND COMPLETION

### § 9.1 CONTRACT SUM

The Contract Sum is stated in the Agreement and, including authorized adjustments, is the total amount payable by the Owner to the Contractor for performance of the Work under the Contract Documents.

### § 9.2 SCHEDULE OF VALUES

Where the Contract is based on a stipulated sum or Guaranteed Maximum Price, the Contractor shall submit to the Architect, before the first Application for Payment, a schedule of values allocating the entire Contract Sum to the various portions of the Work and prepared in such form and supported by such data to substantiate its accuracy as the Architect may require This schedule, unless objected to by the Architect, shall be used as a basis for reviewing the Contractor's Applications for Payment <u>As such, approval of the schedule of values (and revisions thereto) shall be an absolute prerequisite to certification of the applications for payment.</u>

### § 9.3 APPLICATIONS FOR PAYMENT

§ 9.3.1 At least ten days before the date established for each progress payment, the The Contractor shall submit to the Architect an itemized Application for Payment prepared in accordance with the schedule of values, if required under Section 9.2, for completed portions of the Work Such application shall be notarized, if required, and supported by such data substantiating the Contractor's right to payment as the Owner or Architect may require, such as copies of requisitions from Subcontractors and material suppliers, and shall reflect retainage if provided for in the Contract Documents. The payment is the tract of the t

§ 9.3.1.1 As provided in Section 7 3.9, such applications may include requests for payment on account of changes in the Work that have been properly authorized by Construction Change Directives, or by interim determinations of the Architect, but not yet included in Change Orders

§ 9.3.1.2 Applications for Payment shall not include requests for payment for portions of the Work for which the Contractor does not intend to pay a Subcontractor or material supplier, unless such Work has been performed by others whom the Contractor intends to pay

§ 9.3.2 Unless otherwise provided in the Contract Documents, payments shall be made on account of materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work. If approved in advance

EIA Document A201 - 2007. Copyright 0 1011, 1015, 1018, 1025, 1097, 1091, 1058, 1061, 1061, 1066, 1076, 1076, 1087, 1087 and 2007 by The Arerican Institute of Acceltences. All rights reserved. WARNING: This AIX Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIX Document, or any portion of it, may result in severe civil and criminal penaltics, and will be prosecuted to the maximum maximum pressible under the law This draft was produced by AIA software at 14:15:10 or 0/18/2008 under Order No.1000314784 1 which aspires on 8/17/2008, and is not for resels. User Noves.

by the Owner, payment may similarly be made for materials and equipment suitably stored off the site at a location agreed upon in writing Payment for materials and equipment stored on or off the site shall be conditioned upon compliance by the Contractor with procedures satisfactory to the Owner to establish the Owner's title to such materials and equipment or otherwise protect the Owner's interest, and shall include the costs of applicable insurance, storage and transportation to the site for such materials and equipment stored off the site.

§ 9.3.3 The Contractor warrants that title to all Work covered by an Application for Payment will pass to the Owner no later than the time of payment. The Contractor further warrants that upon submittal of an Application for Payment all Work for which Certificates for Payment have been previously issued and payments received from the Owner shall, to the best of the Contractor's knowledge, information and belief, be free and clear of liens, claims, security interests or encumbrances in favor of the Contractor, Subcontractors, material suppliers, or other persons or entities making a claim by reason of having provided labor, materials and equipment relating to the Work

§ 9.3.4 All Applications for Payment shall be accompanied by lien waivers from the Contractor and applicable Subcontractors The lien waivers then taken together, shall equal the sum due under the respective Application for Payment

§ 9.3.5 All Applications for Payment shall be accompanied by the Contractor's and Subcontractors' certified payrolls as required by the Illinois Prevailing Wage Act.

§ 9.3.6 Submission of properly executed lien waivers and the certified payrolls shall be absolute prerequisites to certification of the respective Application for Payment.

### § 9.4 CERTIFICATES FOR PAYMENT

§ 9.4.1 The Architect will, within seven days after receipt of the Contractor's Application for Payment, either issue to the Owner a Certificate for Payment, with a copy to the Contractor, for such amount as the Architect determines is properly due, or notify the Contractor and Owner in writing of the Architect's reasons for withholding certification in whole or in part as provided in Section 9.5.1

§ 9.4.2 The issuance of a Certificate for Payment will constitute a representation by the Architect to the Owner, based on the Architect's evaluation of the Work and the data comprising the Application for Payment, that, to the best of the Architect's knowledge, information and belief, the Work has progressed to the point indicated and that the quality of the Work is in accordance with the Contract Documents. The foregoing representations are subject to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to results of subsequent tests and inspections, to correction of minor deviations from the Contract Documents prior to completion and to specific qualifications expressed by the Architect. The issuance of a Certificate for Payment will further constitute a representation that the Contractor is entitled to payment in the amount certified. However, the issuance of a Certificate for Payment will not be a representation that the Architect has (1) made exhaustive or continuous onsite inspections to check the quality or quantity of the Work, (2) reviewed construction means, methods, techniques, sequences or procedures, (3) reviewed copies of requisitions received from Subcontractors and material suppliers and other data requested by the Owner to substantiate the Contractor's right to payment, or (4) made examination to ascertain how or for what purpose the Contractor has used money previously paid on account of the Contract Sum

### § 9.5 DECISIONS TO WITHHOLD CERTIFICATION

§ 9.5.1 The Architect may withhold a Certificate for Payment in whole or in part, for failure of the Contractor to satisfy prerequisites expressly set forth in the Contract Documents and/or to the extent reasonably necessary to protect the Owner, if in the Architect's opinion the representations to the Owner required by Section 9.4.2 cannot be made. If the Architect is unable to certify payment in the amount of the Application, the Architect will notify the Contractor and Owner as provided in Section 9.4.1. If the Contractor and Architect cannot agree on a revised amount, the Architect will promptly issue a Certificate for Payment for the amount for which the Architect is able to make such representations to the Owner The Architect may also withhold a Certificate for Payment or, because of subsequently discovered evidence, may nullify the whole or a part of a Certificate for Payment previously issued, to such extent as may be necessary in the Architect's opinion to protect the Owner from loss for which the Contractor is responsible, including loss resulting from acts and omissions described in Section 3.3.2, because of

- .1 defective Work not remedied;
- third party claims filed or reasonable evidence indicating probable filing of such claims unless .2 security acceptable to the Owner is provided by the Contractor,

AIA Document AJ01= - 2007. Copyright 6 1511, 1915, 1918, 1935, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WORNING: This AIA Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA' Document; or any portion of it, may result in severe divil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This dist was produced by AIA software at 14:15:10 on 01/18/2008 under Order No.1000314794 i which septres on 8/17/2008, and is not for resale 27

(3921070003)

- .3 failure of the Contractor to make payments properly to Subcontractors or for labor, materials or equipment,
- .4 reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum,
- .5 damage to the Owner or a separate contractor,
- .6 reasonable evidence that the Work will not be completed within the Contract Time, and that the unpaid balance would not be adequate to cover actual or liquidated damages for the anticipated delay, or
- .7 repeated failure to carry out the Work in accordance with the Contract Documents.

§ 9.5.2 When the above reasons for withholding certification are removed, certification will be made for amounts previously withheld

§ 9.5.3 If the Architect withholds certification for payment under Section 9 5.1.3, the Owner may, at its sole option, issue joint checks to the Contractor and to any Subcontractor or material or equipment suppliers to whom the Contractor failed to make payment for Work properly performed or material or equipment suitably delivered. If the Owner makes payments by joint check, the Owner shall notify the Architect and the Architect will reflect such payment on the next Certificate for Payment.

89.5.4 interaction of the stability states, server the neuron scatter for which it estils and the may have been as the state of the

### § 9.6 PROGRESS PAYMENTS

§ 9.6.1 After the Architect has issued a Certificate for Payment, the Owner shall make payment in the manner and within the time provided in the Contract Documents, and shall so notify the Architect

§ 9.6.2 The Contractor shall pay each Subcontractor no later than seven days after receipt of payment from the Owner the amount to which the Subcontractor is entitled, reflecting percentages actually retained from payments to the Contractor on account of the Subcontractor's portion of the Work. The Contractor shall, by appropriate agreement with each Subcontractor, require each Subcontractor to make payments to Sub-subcontractors in a similar manner.

§ 9.6.3 The Architect will, on request, furnish to a Subcontractor, if practicable, information regarding percentages of completion or amounts applied for by the Contractor and action taken thereon by the Architect and Owner on account of portions of the Work done by such Subcontractor

§ 9.6.4 The Owner has the right to request written evidence from the Contractor that the Contractor has properly paid Subcontractors and material and equipment suppliers amounts paid by the Owner to the Contractor for subcontracted Work. If the Contractor fails to furnish such evidence within seven days, the Owner shall have the right to contact Subcontractors to ascertain whether they have been properly paid. Neither the Owner nor Architect shall have an obligation to pay or to see to the payment of money to a Subcontractor, except as may otherwise be required by law

§ 9.6.5 Contractor payments to material and equipment suppliers shall be treated in a manner similar to that provided in Sections 9 6 2, 9 6.3 and 9.6 4

§ 9.6.6 A Certificate for Payment, a progress payment, or partial or entire use or occupancy of the Project by the Owner shall not constitute acceptance of Work not in accordance with the Contract Documents

§ 9.6.7 Unless the Contractor provides the Owner with a payment bond in the full penal sum of the Contract Sum, payments received by the Contractor for Work properly performed by Subcontractors and suppliers shall be held by the Contractor for those Subcontractors or suppliers who performed Work or furnished materials, or both, under contract with the Contractor for which payment was made by the Owner Nothing contained herein shall require money to be placed in a separate account and not commingled with money of the Contractor, shall create any fiduciary liability or tort liability on the part of the Contractor for breach of trust or shall entitle any person or entity to an award of puntive damages against the Contractor for breach of the requirements of this provision.

All Document A201" - 2007. Copyright © 1913, 1915, 1918, 1928, 1937, 1951, 1958, 1951, 1958, 1958, 1958, 1970, 1876, 1957, 1957 and 2007 by The American Institute of Architecta. All rights remarked. WARNING: This AIX Document, 'protected by U.S. 'Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIX Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extont possible under the law. This draft was produced by AIA software at 14:15:10 on 01/18/2008 under Order No.1000314784\_1 which empires on 8/17/2008, and is not for respice User Nates. (1921870803)

### Formatted: Justified, Tab stops: Not at 0.5"

provide the Original Haddingship in an exclusion of the high space of the providence of the payment of and the second second

### § 9.7 FAILURE OF PAYMENT

If the Architect does not issue a Certificate for Payment, through no fault of the Contractor, within seven days after receipt of the Contractor's Application for Payment, or if the Owner does not pay the Contractor within seven days after the date established in the Contract Documents the amount certified by the Architect or awarded by binding dispute resolution, then the Contractor may, upon seven additional days' written notice to the Owner and Architect, stop the Work until payment of the amount owing has been received. The Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable costs of shutdown, delay and start-up, plus interest as provided for in the Contract Documents

### § 9.8 SUBSTANTIAL COMPLETION

§ 9.8.1 Substantial Completion is the stage in the progress of the Work when the Work or designated portion thereof is sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work for its intended use the work all not to considered suite ble for substantial Completion is generalized with inter steer meltinge in the West as report and as designed and schedided of the signal due, eggined termine of the speed and of the source of the structure of the sector of the sector of the sector of the sector in our dingly preption of search is in a completed and downarties, and at find finisher without it for arms the only of the only remaining Work on II Is prepared in the OV is a could deter the in a number of the completent of the Sept. By the Control to yould not reasonally interface of interest the the second of th souther process the Coorty on the Long start of remained. Work will be completed within the sector on foll the the Day of Substants I Competition. The Competitor shall secure and deliver to the Owner write rethe literation of the last state of the second states and states and states bearing the date of Stations 1 and a set of the set of a set of the set of in a para a Doomagn . The Contractic example for the war and of elt Work, whether a ritorne is not a and a subjects at poy their. If an die is the Contracted does not an aplege of granning synch scalar, forty five (45) or Substantial completion Dear Oalt zi , the Contract Mathematics of the remaining Work to be in the second second second of the second The spiner notice the Owner post of angla press pay the content Well in a core of which it allow conversion to the Confractor of Post or more by Oel C. For in pull to offset light ( Contractory fight n Letter

§ 9.8.2 When the Contractor considers that the Work, or a portion thereof which the Owner agrees to accept separately, is substantially complete, the Contractor shall prepare and submit to the Architect a comprehensive list of items to be completed or corrected prior to final payment (the "Punchlist"). Failure to include an item on such list does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents

§ 9.8.3 Upon receipt of the Contractor's list, the Architect will make an inspection to determine whether the Work or designated portion thereof is substantially complete. If the Architect's inspection discloses any item, whether or not included on the Contractor's list, which is not sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work or designated portion thereof for its intended use, the Contractor shall, before issuance of the Certificate of Substantial Completion, complete or correct such item upon notification by the Architect. In such case, the Contractor shall then submit a request for another inspection by the Architect to determine Substantial Completion

§ 9.8.4 When the Work or designated portion thereof is substantially complete, the Architect will prepare a Certificate of Substantial Completion that shall establish the date of Substantial Completion, shall establish responsibilities of the Owner and Contractor for security, maintenance, heat, utilities, damage to the Work and insurance, and shall fix the time within which the Contractor shall finish all items on the list accompanying the Certificate Warranties required by the Contract Documents shall commence on the date of Substantial Completion of the Work or designated portion thereof unless otherwise provided in the Certificate of Substantial Completion.

§ 9.8.5 The Certificate of Substantial Completion shall be submitted to the Owner and Contractor for their written acceptance of responsibilities assigned to them in such Certificate Upon such acceptance and consent of surety, if

AIR Decement A201 - 2007. Copyright © 1811, 1915, 1818, 1925, 1937, 1851, 1958, 1841, 1843, 1946, 1876, 1876, 1877, 1877, 1877, American Institute of Architectic All Eights reserved. MANUMO: This AIX Document is protected by U.S. Copyright Law and Toternat onal Tratics. Dantborlied repredention or distribution of this AIX Document, or any portion of it, may result in severe sivil and crimical penalties, and will be presented to the maximum acteon to possible under the law. This deaft was produced by AIX software at 14:15:10 on U/14/2000 under Order No.1000314794\_1 which expires on 0/17/2008, and is not for resale. 29 (3921870803)

Formatted: Justified

any, the Owner shall make payment of retainage applying to such Work or designated portion thereof as set out in the Agreement. Such payment shall be adjusted for Work that is incomplete or not in accordance with the requirements of the Contract Documents

**§ 9.8.6** Upon Substantial Completion, via this Contract, the Contractor and applicable Subcontractors (via the subcontract flow-through provision(s)) assign all vendor and manufacturers' warranties to the Owner. The manufacturers' warranties shall be submitted to the Architect prior to submission of the final Application for Payment.

§ 9.8.7 Among other items identified elsewhere in the Contract Documents, submission of the following shall be a prerequisite to Substantial Completion:

- All Record Documents
- b. All Operations and Maintenance Manuals (3 copies in 3-ring binders)
- c. All Manufacturers' warranties

### § 9.9 PARTIAL OCCUPANCY OR USE

§ 9.9.1 The Owner may occupy or use any completed or partially completed portion of the Work at any stage when such portion is designated by separate agreement with the Contractor, provided such occupancy or use is consented to by the msurer as required under Section 11.3.1.5 and authorized by public authorities having jurisdiction over the Project. Such partial occupancy or use may commence whether or not the portion is substantially complete, provided the Owner and Contractor have accepted in writing the responsibilities assigned to each of them for payments, retainage, if any, security, maintenance, heat, utilities, damage to the Work and insurance, and have agreed in writing concerning the period for correction of the Work and commencement of warranties required by the Contract Documents. When the Contractor considers a portion substantially complete, the Contractor shall prepare and submit a list to the Architect as provided under Section 9.8.2. Consent of the Contract to partial occupancy or use shall not be unreasonably withheld. The stage of the progress of the Work shall be determined by written agreement between the Owner and Contractor or, if no agreement is reached, by decision of the Architect.

§ 9.9.2 Immediately prior to such partial occupancy or use, the Owner, Contractor and Architect shall jointly inspect the area to be occupied or portion of the Work to be used in order to determine and record the condition of the Work

§ 9.9.3 Unless otherwise agreed upon, partial occupancy or use of a portion or portions of the Work shall not constitute acceptance of Work not complying with the requirements of the Contract Documents

### § 9.10 FINAL COMPLETION AND FINAL PAYMENT

§ 9.10.1 All Work depicted on the Contractor's Punchlist and thereafter identified in the Architect's inspection shall be completed within thirty days of issuance of the Certificate of Substantial Completion Upon receipt of the Contractor's written notice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Architect will promptly make such inspection and, when the Architect finds the Work acceptable under the Contract Documents and the Contract fully performed, the Architect will promptly issue a final Certificate for Payment stating that to the best of the Architect's knowledge, information and belief, and on the basis of the Architect's on-site visits and inspections, the Work has been completed in accordance with terms and conditions of the Contract Documents and that the entire balance found to be due the Contractor and noted in the final Certificate is due and payable. The Architect's final Certificate for Payment will constitute a further representation that conditions listed in Section 9 10 2 as precedent to the Contractor's being entitled to final payment have been fulfilled

§ 9.10.2 Neither final payment nor any remaining retained percentage shall become due until the Contractor submits to the Architect (1) an affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the Owner or the Owner's property might be responsible or encumbered (less amounts withheld by Owner) have been paid or otherwise satisfied, (2) a certificate evidencing that insurance required by the Contract Documents to remain in force after final payment is currently in effect and will not be canceled or allowed to expire until at least 30 days' prior written notice has been given to the Owner, (3) a written statement that the Contract Documents, (4) consent of surety, if any, to final payment and (5), if required by the Owner, other data establishing payment or satisfaction of obligations, such as receipts, releases and waivers of liens, claims, security

ATA Decumant A201" - 2007. Copyright © 1911, 1915, 1916, 1925, 1937, 1951, 1958, 1951, 1956, 1970, 1976, 1987, 198

Formatted: Font: (Default) Times New Roman Formatted: Font: (Default) Times New Roman

### Formatted: Font: Not Bold

Formatted: Font: (Default) Times New Roman
Formatted: Numbered + Level: 1 +

Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.75" interests or encumbrances arising out of the Contract, to the extent and in such form as may be designated by the Owner If a Subcontractor refuses to furnish a release or waiver required by the Owner, the Contractor may furnish a bond satisfactory to the Owner to indemnify the Owner against such lien. If such lien remains unsatisfied after payments are made, the Contractor shall refund to the Owner all money that the Owner may be compelled to pay in discharging such lien, including all costs and reasonable attorneys' fees

§ 9.10.3 If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Contractor or by issuance of Change Orders affecting final completion, and the Architect so confirms, the Owner shall, upon application by the Contractor and certification by the Architect, and without terminating the

Contract, make payment of the balance due for that portion of the Work fully completed and accepted, less retention. If the remaining balance for Work not fully completed or corrected is less than retainage stipulated in the Contract Documents, and if bonds have been furnished, the written consent of surety to payment of the balance due for that portion of the Work fully completed and accepted shall be submitted by the Contractor to the Architect prior to certification of such payment Such payment shall be made under terms and conditions governing final payment, except that it shall not constitute a waiver of claims.

### § 8.10.4 The making of final payment shall constitute a waiver of Claims by the Owner except those arising from liens, Claims, security interests or encumbrances arising out of the Contract and unsettlede

- failure of the Work to comply with the requirements of the Contract Documents; or
- terms of special warranties required by the Contract Documents.

§ 9.10.5 Acceptance of final payment by the Contractor, a Subcontractor or material supplier shall constitute a waiver of claims by that payee except those previously made in writing and specifically identified by that payee as unsettled at the time of final Application for Payment

### ARTICLE 10 PROTECTION OF PERSONS AND PROPERTY

§ 10.1 SAFETY PRECAUTIONS AND PROGRAMS

The Contractor shall be responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the performance of the Contract, Netherary Course to a stall of then the or case dety presentional or programs in connecture with the Works

### § 10.2 SAFETY OF PERSONS AND PROPERTY

§ 10.2.1 The Contractor shall take reasonable precautions for safety of, and shall provide reasonable protection to prevent damage, injury or loss to

- employees on the Work and other persons who may be affected thereby, .1
- .2 the Work and materials and equipment to be incorporated therein, whether in storage on or off the site, under care, custody or control of the Contractor or the Contractor's Subcontractors or Subsubcontractors; and
- .3 other property at the site or adjacent thereto, such as trees, shrubs, lawns, walks, pavements, roadways, structures and utilities not designated for removal, relocation or replacement in the course of construction

§ 10.2.2 The Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities bearing on safety of persons or property or their protection from damage, injury or loss

§ 10.2.3 The Contractor shall erect and maintain, as required by existing conditions and performance of the Contract, reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards, promulgating safety regulations and notifying owners and users of adjacent sites and utilities

§ 10.2.4 When use or storage of explosives or other hazardous materials or equipment or unusual methods are necessary for execution of the Work, the Contractor shall exercise utmost care and carry on such activities under supervision of properly qualified personnel

§ 10.2.5 The Contractor shall promptly remedy damage and loss (other than damage or loss insured under property insurance required by the Contract Documents) to property referred to in Sections 10 2.1 2 and 10.2.1.3 caused in whole or in part by the Contractor, a Subcontractor, a Sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be hable and for which the Contractor is responsible under

AIA Document A201<sup>m</sup> - 3007. Copyright D 1911, 1915, 1918, 1928, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1987, 1987 and 2007 by The merican Institute of Alchitects. All rights remerved. WARNING: This AIA Document is protected by U.S. Copyright Law and International Treates Unauthorized reproduction or distribution of this AIA Document; or any portion of it, may reall in severe divid and eriminal ponalties, and will be prosecuted to the maximum extent possible under the law. This dist was produced by AIA software at 14:13:10 an 01/18/2008 under order No 1000314794\_1 which expires on 8/17/2008, and is not for resele. 31 User Notes: (3921070803)

Sections 10.2.1.2 and 10.2.1.3, except damage or loss attributable to acts or omissions of the Owner or Architect or anyone directly or indirectly employed by either of them, or by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to the Contractor's obligations under Section 3.18

§ 10.2.6 The Contractor shall designate a responsible member of the Contractor's organization at the site whose duty shall be the prevention of accidents. This person shall be the Contractor's superintendent unless otherwise designated by the Contractor in writing to the Owner and Architect.

§ 10.2.7 The Contractor shall not permit any part of the construction or site to be loaded so as to cause damage or create an unsafe condition

### § 10.2.8 INJURY OR DAMAGE TO PERSON OR PROPERTY

If either party suffers injury or damage to person or property because of an act or onission of the other party, or of others for whose acts such party is legally responsible, written notice of such injury or damage, whether or not insured, shall be given to the other party within a reasonable time not exceeding 21 days after discovery. The notice shall provide sufficient detail to enable the other party to investigate the matter.

### § 10.3 HAZARDOUS MATERIALS

§ 10.3.1 The Contractor is responsible for compliance with any requirements included in the Contract Documents regarding hazardous materials. If the Contractor encounters a hazardous material or substance not addressed in the Contract Documents and if reasonable precautions will be inadequate to prevent foreseeable bodily injury or death to persons resulting from a material or substance, including but not limited to asbestos or polychlorinated biphenyl (PCB), encountered on the site by the Contractor, the Contractor shall, upon recognizing the condition, immediately stop Work in the affected area and report the condition to the Owner and Architect in writing.

§ 10.3.2 Upon receipt of the Contractor's written notice, the Owner shall obtain the services of a licensed laboratory to verify the presence or absence of the material or substance reported by the Contractor and, in the event such material or substance is found to be present, to cause it to be rendered harmless. Unless otherwise required by the Contract Documents, the Owner shall furnish in writing to the Contractor and Architect the names and qualifications of persons or entities who are to perform tests verifying the presence or absence of such material or substance or who are to perform the task of removal or safe containment of such material or substance. The Contractor and the Architect will promptly reply to the Owner in writing stating whether or not either has reasonable objection to the persons or entities proposed by the Owner if either the Contractor or Architect has an objection to a person or entity proposed by the Owner shall propose another to whom the Contractor and the Architect have no reasonable objection. When the material or substance has been rendered harmless, Work in the affected area shall resume upon written agreement of the Owner and Contractor. By Change Order, the Contractor's reasonable additional costs of shut-down, delay and start-up

§ 10.3.3 To the fullest extent permitted by law, the Owner shall indemnify and hold harmless the Contractor, Subcontractors, Architect, Architect's consultants and agents and employees of any of them from and against claims, damages, losses and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work in the affected area if in fact the material or substance presents the risk of bodily injury or death as described in Section 10.3.1 and has not been rendered harmless, provided that such claim, damage, loss or expense is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the Work itself), except to the extent that such damage, loss or expense is due to the fault or negligence of the party seeking indemnity

§ 10.3.4 The Owner shall not be responsible under this Section 10.3 for materials or substances the Contractor brings to the site unless such materials or substances are required by the Contract Documents. The Owner shall be responsible for materials or substances required by the Contract Documents, except to the extent of the Contractor's fault or negligence in the use and handling of such materials or substances.

§ 10.3.5 The Contractor shall indemnify the Owner for the cost and expense the Owner incurs (1) for remediation of a material or substance the Contractor brings to the site and negligently handles, or (2) where the Contractor fails to perform its obligations under Section 10.3.1, except to the extent that the cost and expense are due to the Owner's fault or negligence

ATA Decrement A101<sup>m</sup> - 2007. Copyright © 1911, 1915, 1918, 1923, 1937, 1951, 1958, 1951, 1958, 1954, 1956, 1976, 1976, 1977, 1897 and 2007 by The Areficen Institute of Aichitotio All rights reserved WADNING: This AIA' Document is protected by U.S. Copyright Law and International Treatics. Unauthorized reproduction or distribution of this AIA' Document, or any portion of it may result in severe civil and craminal penalties, and will be prosocuted to the maximum extent possible under the law This draft was produced by ATA software at 14:15:10 on 01/18/2005 under Order No.1000314784 1 which expires on 8/17/2008, and is not for resels. (3921870803) § 10.3.6 If, without negligence on the part of the Contractor, the Contractor is held liable by a government agency for the cost of remediation of a hazardous material or substance solely by reason of performing Work as required by the Contract Documents, the Owner shall indemnify the Contractor for all cost and expense thereby incurred.

### § 10.4 EMERGENCIES

In an emergency affecting safety of persons or property, the Contractor shall act, at the Contractor's discretion, to prevent threatened damage, injury or loss Additional compensation or extension of time claimed by the Contractor on account of an emergency shall be determined as provided in Article 15 and Article 7

### ARTICLE 11 INSURANCE AND BONDS

### § 11.1 CONTRACTOR'S LIABILITY INSURANCE

§ 11.1.1 The Contractor shall purchase from and maintain in a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located such insurance as will protect the Contractor from claims set forth below which may arise out of or result from the Contractor's operations and completed operations under the Contract and for which the Contractor may be legally liable, whether such operations be by the Contractor or by a Subcontractor or by anyone directly or indirectly employed by any of them, or by anyone for whose acts any of them may be liable:

- .1 Claims under workers' compensation, disability benefit and other similar employee benefit acts that are applicable to the Work to be performed,
- .2 Claims for damages because of bodily injury, occupational sickness or disease, or death of the Contractor's employees,
- .3 Claims for damages because of bodily injury, sickness or disease, or death of any person other than the Contractor's employees,
- Claims for damages insured by usual personal injury liability coverage; .4
- .5 Claims for damages, other than to the Work itself, because of injury to or destruction of tangible property, including loss of use resulting therefrom;
- .6 Claims for damages because of bodily injury, death of a person or property damage arising out of ownership, maintenance or use of a motor vehicle,
- .7 Claims for bodily injury or property damage arising out of completed operations, and .8 Claims involving contractual liability insurance applicable to the Contractor's obligations under
- Section 3 18

§ 11.1.2 The insurance required by Section 11.1.1 shall be written for not less than limits of liability specified in the Contract Documents or required by law, whichever coverage is greater. Coverages, whether written on an occurrence or claims-made basis, shall be maintained without interruption from the date of commencement of the Work until the date of final payment and termination of any coverage required to be maintained after final payment, and, with respect to the Contractor's completed operations coverage, until the expiration of the period for correction of Work or for such other period for maintenance of completed operations coverage as specified in the Contract Documents

§ 11.1.3 Certificates of insurance acceptable to the Owner shall be filed with the Owner prior to commencement of the Work and thereafter upon renewal or replacement of each required policy of insurance. These certificates and the insurance policies required by this Section 11.1 shall contain a provision that coverages afforded under the policies will not be canceled or allowed to expire until at least 30 days' prior written notice has been given to the Owner An additional certificate evidencing continuation of liability coverage, including coverage for completed operations, shall be submitted with the final Application for Payment as required by Section 9 10 2 and thereafter upon renewal or replacement of such coverage until the expiration of the time required by Section 11.1.2. Information concerning reduction of coverage on account of revised limits or claims paid under the General Aggregate, or both, shall be furnished by the Contractor with reasonable promptness

§ 11.1.4 The Contractor shall cause the commercial liability coverage required by the Contract Documents to include (1) the Owner, the Architect and the Architect's Consultants as additional insureds for claims caused in whole or in part by the Contractor's negligent acts or omissions during the Contractor's operations, and (2) the Owner as an additional insured for claims caused in whole or in part by the Contractor's negligent acts or omissions during the Contractor's completed operations The applicable policies shall be endorsed to indicate that they are primary as respects the additional insureds, and not contributory with any other insurance available to the additional insureds,

AIA Document A201<sup>22</sup> - 2007 Copyright 0 1911, 1915, 1918, 1925, 1937, 1951, 1951, 1953, 1963, 1966, 1970, 1976, 1987, 1987 and 2007 by The American Institute of Architecist All sights reserved WARNING: This AIA Document is protocted by U.S. Copyright Law and International Trotices Unsuffering reproduction or distribution of this AIA Document, or any portion of it, may result in severe civil and eriminal penaltics, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 14:15:10 on 0/13/2008 under Order No 1000314794 1 which sepires on 8/17/2006, and is not for resals 33

(3921070803)

§11.1.5 Contractor shall maintain, at its own expense, the following insurance coverages on an occurrence basis insuring the Contractor, its employees and agents, and the Indemnitees as required in Section 3.18 herein which insurance shall be placed with insurance companies rated at least "A", "XIV" by Best's Key Rating Guide and shall incorporate a provision requiring the giving of written notice to Owner at least thirty (30) days prior to the cancellation or non-renewal of any such policies.

### §11,1.6.1 Contractor's Liability Insurance

The Contractor shall not commence Work under this Contract until all insurance required herein is obtained and approved by the Owner; nor shall the Contractor allow any Subcontractor to commence any portion of the Work.

§11.1.6.2 Commercial General Liability Insurance (including limited form contractual liability and completed operations, explosion, collapse and underground hazards), covering personal injury, bodily injury and property damages in the amount of One Million Dollars per occurrence and Two Million Dollars aggregate (\$1,000,000/\$2,000,000) covering personal injury, bodily injury and property damage.

§11.1.6.3 Automobile Liability Insurance, including hired and non-owed vehicles, if any, in the amount of One Million Dollars (\$1.000.000) covering personal iniury, bodily injury and property damage.

§11.1.6.4 Workmen's Compensation Insurance in the amount of the statutory maximum with an Employer's Liability coverage of at least Five Hundred Thousand Dollars (\$500.000).

§ 11.1.6.5 Failure of either the Architect or Owner to demand Certificates of Insurance and/or policies shall not constitute a waiver of the contractor's responsibility hereunder. Nor shall review and/or approval by either the Owner or Architect in any way relieve Contractor of its responsibility for furnishing sufficient amounts and coverages of insurance. Said endorsements or amendatory riders shall indicate that as respects said additional insureds, there shall be severability of interests under said insurance policies. The Certificates and amendatory riders or endorsements shall clearly indicate the specific coverage (including contractual liability for the Contractor's obligation under 3.18) and shall contain provision requiring the giving of written notice to Owner at least thirty (30) days prior to the cancellation, non-renewal or material modification of any such policies, as evidenced by return receipt of United States Certified Mail.

§11.1.6.6 Under no circumstances shall the Contractor be relieved of providing insurance as required by this Contract. If inspection of Certificates by Owner would reasonably reveal any deficiencies in coverage as required by this Contract. Contractor shall not be relieved of its obligation to prove insurance coverages as required herein and may not assert any defense of waiver, acquiescence, estoppel, or otherwise by the failure of Owner, or its agents to object to the form of the Certificate. Policies or other documents provided by the Contractor to certify that the Contractor complied with the provisions of this Contract regarding insurance coverage.

§11.1.7 Contractor shall also protect the Owner by specifically incorporating this Article 11 into every subcontract entered into and also requiring that every Subcontractor incorporate this paragraph into every sub-subcontract it enters into. Notwithstanding this requirement, this Article 11 is deemed incorporated into every subcontract and sub-subcontract via such document's flow-through provisions.

§ 11.3.1.8 Liability of Contractor or Subcontractor is not limited by purchase of insurance. Nothing contained in the insurance requirements of the Contract Documents is to be construed as limiting the liability of the Contractor, the liability of any Subcontractor of any tier, or the liability of the Architect, or either of their respective insurance carriers. Owner does not, in any way, represent that the coverages or limits of insurance specified is sufficient or adequate to protect the Owner, Contractor, Architect, or any Subcontractor's interest or liabilities, but are merely minimums. The obligation of the Contractor and every Subcontractor of any tier to purchase insurance shall not, in any way, limit their obligations to the Owner in the event that the Owner should suffer an injury or loss in excess of the amount recoverable through insurance, or any loss or portion of the loss which is not covered by either the Architect's, Contractor's or any Subcontractor's insurance

### § 11.2 OWNER'S LIABILITY INSURANCE

The Owner shall be responsible for purchasing and maintaining the Owner's usual liability insurance

AIR Document A201 - 2007. Copyright 0 1911, 1015, 1018, 1025, 1037, 1031, 1038, 1054, 1054, 1056, 1076, 1077, 1007, 1007 he American Institute of Aschritects. All rights reserved WARNING: This AIX Document is protected by U.S. Copyright Law and International Treaties Databorised represented to the surface of this AIX Document, or any portion of it, may result in severe civil and erminal penalties, and will be presented to the surface to seasure of the AIX Document is not for resale. User Notes. (1921870803).

Formatted: Justified

### § 11.3 PROPERTY INSURANCE

§ 11.3.1 Unless otherwise provided, the <u>Construction ManagerOwner</u> shall purchase and maintain, in a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located, property insurance written on a builder's risk "all-risk" or equivalent policy form in the amount of the initial Contract Sum, plus value of subsequent Contract Modifications and cost of materials supplied or installed by others, comprising total value for the entire Project at the site on a replacement cost basis without optional deductibles. Such property insurance shall be maintained, unless otherwise provided in the Contract Documents or otherwise agreed in writing by all persons and entities who are beneficiaries of such insurance, until final payment has been made as provided in Section 9.10 or until no person or entity other than the Owner has an insurable interest in the property required by this Section 11.3 to be covered, whichever is later. This insurance shall include interests of the Owner, the Contractor, Subcontractors and Sub-subcontractors in the Project.

§ 11.3.1.1 Property insurance shall be on an "all-risk" or equivalent policy form and shall include, without limitation, insurance against the perils of fire (with extended coverage) and physical loss or damage including, without duplication of coverage, theft, vandalism, malicous mischief, collapse, earthquake, flood, windstorm, falsework, testing and startup, temporary buildings and debris removal including demolition occasioned by enforcement of any applicable logal resurgements and shall locate rescapeble accurate the applicable.

applicable legal requirements, and shall cover reasonable compensation for Architect's. <u>Owner's</u>, and Contractor's services and expenses required as a result of such insured loss

§ 11.3.1.2 If the Owner does not intend to purchase such property insurance required by the Contract and with all of the coverages in the amount described above, the Owner shall so inform the Contractor in writing prior to commencement of the Work. The Contractor may then effect insurance that will protect the interests of the Contractor. Subcontractors and Sub-subcontractors in the Work, and by appropriate Change Order the cost thereof shall be charged to the Owner. If the Contractor is damaged by the failure or neglect of the Owner to purchase or monitoin insurance as described above, without so notifying the Contractor in writing, then the Owner shall been all reasonable costs properly attributable thereto.

§ 11.3.1.3 If the property insurance requires deductibles, the <u>Contractor or Construction Manager, as the case may</u> be, <del>Owner</del> shall pay costs not covered because of such deductibles.

§ 11.3.1.4 This property insurance shall cover portions of the Work stored off the site, and also portions of the Work in transit.

§ 11.3.1.5 Partial occupancy or use in accordance with Section 9.9 shall not commence until the insurance company or companies providing property insurance have consented to such partial occupancy or use by endorsement or otherwise. The Owner and the Contractor shall take reasonable steps to obtain consent of the insurance company or companies and shall, without mutual written consent, take no action with respect to partial occupancy or use that would cause cancellation, lapse or reduction of insurance

### § 11.3.2 BOILER AND MACHINERY INSURANCE

The Owner shall purchase and maintain boiler and machinery insurance required by the Contract Documents or by law, which shall specifically cover such insured objects during installation and until final acceptance by the Owner, this insurance shall include interests of the Owner, Contractor. Subcontractors and Sub-subcontractors in the Work, and the Owner and Contractor shall be named insureds.

### § 11.3.3 LOSS OF USE INSURANCE

The Owner, at the Owner's option, may purchase and maintain such insurance as will insure the Owner against loss of use of the Owner's property due to fire or other hazards, however caused The Owner waives all rights of action against the Contractor for loss of use of the Owner's property, including consequential losses due to fire or other hazards however caused.

§ 11.3.4 If the Contractor requests in writing that insurance for risks other than those described herein or other special causes of loss be included in the property insurance policy, the Owner shall, if possible, include such insurance, and the cost thereof shall be charged to the Contractor by appropriate Change Order.

§ 11.3.5 If during the Project construction period the Owner insures properties, real or personal or both, at or adjacent to the site by property insurance under policies separate from those insuring the Project, or if after final payment property insurance is to be provided on the completed Project through a policy or policies other than those insuring

AIR Decumant A301<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1938, 1951, 1958, 1958, 1970, 1974, 1987, 1987 and 2007 by The Averican Institute of Architects. All rights reserved. MARVING: This AIA Document is protected by U.S. Copyright Law and International Tractics. Unauthorized reproduction or distribution of this AIA Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 14-15:10 on 01/18/2008 under Cider No.1000314764\_1 which expires on 8/17/2008, and is not for resele. User Notes the Project during the construction period, the Owner shall waive all rights in accordance with the terms of Section 11.3.7 for damages caused by fire or other causes of loss covered by this separate property insurance. All separate policies shall provide this waiver of subrogation by endorsement or otherwise

§ 11.3.6 Before an exposure to loss may occur, the Owner shall file with the Contractor a copy of each policy that includes insurance coverages required by this Section 11.3 Each policy shall contain all generally applicable conditions, definitions, exclusions and endorsements related to this Project Each policy shall contain a provision that the policy will not be canceled or allowed to expire, and that its limits will not be reduced, until at least 30 days' prior written notice has been given to the Contractor.

### § 11.3.7 WAIVERS OF SUBROGATION

The Owner and Contractor waive all rights against (1) each other and any of their subcontractors, subsubcontractors, agents and employees, each of the other, and (2) the Architect, Architect's consultants, separate contractors described in Article 6, if any, and any of their subcontractors, sub-subcontractors, agents and employees, for damages caused by fire or other causes of loss to the extent covered by property insurance obtained pursuant to this Section 11.3 or other property insurance applicable to the Work, except such rights as they have to proceeds of such insurance held by the Owner as fiduciary. The Owner or Contractor, as appropriate, shall require of the Architect's consultants, separate contractors described in Article 6, if any, and the aubcontractors, subsubcontractors, agents and employees of any of them, by appropriate agreements, written where legally required for validity, similar waivers each in favor of other parties enumerated herein. The policies shall provide such waivers of subrogation by endorsement or otherwise. As waiver of subrogation shall be effective as to a person or entity even though that person or entity would otherwise have a duty of indemnification, contractual or otherwise, did not pay the insurance premium directly or indirectly, and whether or not the person or entity had an insurable interest in the property damaged. The Owner and Contractor shall not waive rights of subrogation.

§ 11.3.6 A loss insured under the Owner's property insurance shall be adjusted by the <u>Construction ManagerOwner</u> as fiduciary and made payable to the <u>Construction ManagerOwner</u> as fiduciary for the insureds, as their interests may appear, subject to requirements of any applicable mortgagee clause and of Section 11.3.10. The Contractor shall pay the <u>Owner and/or</u> Subcontractors their just shares of insurance proceeds received by the <u>Construction</u>, and by appropriate agreements, written where legally required for validity, shall require Subcontractors to make payments to their Sub-subcontractors in similar manner.

§ 11.3.9 If required in-writing by a party in interest, the Owner as fiduciary shall, upon occurrence of an insured loss, give bond for proper performance of the Owner's duties. The cast of required bonds shall be charged against proceeds received as fiduciary. The Owner shall deposit in a separate account proceeds so received, which the Owner shall distribute in accordance with such agreement as the parties in interest may reach, or as determined in accordance with the method of binding dispute resolution selected in the Agreement between the Owner and Contractor. If after such loss no other special agreement is made and unless the Owner terminates the Contract for convenience, replacement of damaged property shall be performed by the Contractor after notification of a Change in the Work-in accordance with Article 7.

§ 11.3.10 The <u>Construction Manager</u>Owner as fiduciary shall have power to adjust and settle a loss with insurers unless one of the parties in interest shall object in writing within five days after occurrence of loss to the <u>Construction Manager</u>'s exercise of this power; if such objection is made, the dispute shall be resolved in the manner selected by the Owner and Contractor as the method of binding dispute resolution, the <u>Construction Manager</u>Owner as fiduciary shall make settlement with insurers or, in the case of a dispute over distribution of insurance proceeds, in accordance with the directions of the arbitrators

### § 11.4 PERFORMANCE BOND AND PAYMENT BOND

§ 41.4.1 The Owner shall have the right to require the Contractor to furnish bonds covering faithful performance of the Contract and payment of obligations arising thereunder as stipulated in bidding requirements or specifically required in the Contract Documents on the date of execution of the Contract.

§ 11.4.12 Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the Contract, the Contractor shall promptly furnish a copy of the bonds or shall authorize a copy to be furnished

ATA Document A201 - 1007. Copyright + 1011, 1915, 1918, 1925, 1917, 1951, 1955, 1961, 1965, 1966, 1970, 1976, 1987, 1987 and 2007 by The American Institute of Architectur. All sights reserved. KARNING: This AIX Document is protected by U.S. Copyright Law and International Treaties. Numethorized reproduction or distribution of this AIX Document, or any postion of it, may result in severe civil and criminal ponalities, and will be prosecuted to the maximum extent possible under the law. This dreft was produced by A3A software at 14.15:10 on 01/18/2038 under Order No.2000314794 1 which empires on 0/17/2008, and is not for resele (3921070803) § 11.4.2 Contractor, within ten (10) days after receiving notice of the award, shall furnish a performance bond and a payment bond in the full amount of the Contract agreeing to perform the Work and fulfill all obligations in accordance with all of the provisions of the Contract Documents with a surety rated not less than A. VI by Best's Insurance Guide Key.

§ 11.4.3 If, at any time, the Owner shall become reasonably dissatisfied with any surety or sureties then upon the bonds, or for any other reason such bonds shall cease to be adequate security for the Owner. Contractor shall, within five (5) days after notice to do so, substitute an acceptable bond in such form and sum, and signed by such other sureties as may be satisfactory to the Owner. No further payment shall be deemed due nor shall be made until the new sureties shall have qualified.

§ 11.4.4 The Performance Bond and Payment Bond shall be executed in conformity with American Institute of Architects, Doc. A312 Said bond forms shall be deemed modified to the extent to be consistent with this Article 11.4. A certified copy of the Power of Attorney from the surety company stating that the person executing the bond is duly authorized by the surety to execute the bond, shall accompany the bond.

§ 11.4.5 Whenever the Contractor shall be and is declared by the Owner to be in default under the Contract, the surety of the Contractor shall be responsible to compensate the Owner for the following costs incurred by the Owner as they result from the default: 1) any and all extra work, 2) additional Architect costs, 3) accounting costs, 4) legal costs and reasonable attorneys' fees. 5) testing, consulting, and other engineering costs, 6) any other costs incurred tresulting from the default. Notwithstanding, the performance bond surety's payment obligation shall not exceed the penal sum of the bond.

§ 11.4.6 It shall be the duty of the Surety to give an unequivocal notice in writing to Owner within ten (10) days after receipt of a Declaration of Default and notice of termination of the Agreement of the Surety's election either to remedy the default or defaults promptly or to perform the remaining Work promptly or to pay to Owner costs as herein provided, time being of the essence. In said Notice of Election, the Surety shall indicate the date on which the remedy or performance will commence and it shall then be the duty of the Surety to give prompt notice in writing to Owner immediately upon completion of (a) the remedy and/or correction of each default. (b) the remedy and/or correction of each item of conderned Work, (c) the furnishing of each omitted item of Work, and (d) the performance of the Contract. The Surety shall not assert solvency of its principal or its principal's denial of default as justification for its failure to give notice of election or for its failure to promptly requires additional time to investigate the declaration of default, it shell within the aforesaid ten (10) days so notify the Owner. In such case, the Owner may, without prejudice to its rights under the performance bond, continue construction of the Work, and charge the costs of such Work to the Surety. Upon completion of the Work, and charge the costs of such Work to the Surety. Upon the performance bond, continue construction of the Work, and charge the costs of such Work to the Surety. Upon to perform and under the bond or herein, or in any event not in excess of a reasonable time, the Surety may exercise its rights otherwise held herein and under the bond.

§ 11.4.9 In the event the said Surety shall fail to act promptly as hereinbefore provided, then Owner may cause ten (10) days notice of such failure to be given, both to said principal and Surety, and that at the expiration of said ten (10) days, the obligee shall have the authority to cause said Work to be done, and when the same is completed and the cost thereof quantified, the said principal and Surety shall hereby agree to pay any excess in the cost of said Work above the agreed price to be paid under said Contract, but not exceeding the penal sum of the bond.

§ 11.4.9 As the Work is completed by a completing contactor in accordance with the terms of the Contract Documents, to the extent that any portion of the Contract Price remains owing after setoff. Owner shall pay completing contractor in accordance with the Schedule of Values as certified by the Architect, and upon the completion of the Work pursuant to the Agreement, if any funds remain due on said Contract, the same shall be paid to said Surety.

§ 11.4.10 In case of any conflict between any provision of the performance bond and the Contract Documents, the provisions of the Contract Documents shall prevail.

§ 11.4.12 Any provisions contained within the bonds creating a prerequisite or condition precedent for Owner not otherwise required herein these General Conditions, or abrogating Owner's rights or remedies otherwise available in contract, law, or equity, are void.

AIR Decumant A201<sup>20</sup> - 2007. copyright 0 1011, 1915, 1918, 1825, 1937, 1951, 1958, 1961, 1963, 1965, 1970, 1976, 1987, 1987 and 2007 by The Averigan Institute of Archivets. All sights reserved NORVINO: This AIR' Document is protocted by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIR' Document, or any portion of it, may result in servers mixtl and crisical possibles, and will be prosecuted to the maximum actent possible under the law. This draft was produced by AIR software 14.15/10 on 07/18/2008 under Order No 1000314794 1 which expires on 0/17/2008, and is not for resale. User Sotes: (3921870603) § 11.4.13 In the event the Surety shall make any assignment for the benefit of creditors or commit any act of bankruptcy, or if it shall be declared bankrupt, or if it shall file a voluntary petition in bankruptcy, or shall in the opinion of the Owner be insolvent, the Contractor agrees forthwith upon request of the Owner to furnish and maintain other corporate surety with respect to said bonds satisfactory to the Owner.

# ARTICLE 12 UNCOVERING AND CORRECTION OF WORK § 12.1 UNCOVERING OF WORK

 $\S$  12.1.1 If a portion of the Work is covered contrary to the <u>specification</u> Architect's request or to requirements specifically expressed in the Contract Documents, it must, if requested in writing by the Architect, be uncovered for the <u>specification</u> Architect's examination and be replaced at the Contractor's expense without change in the Contract Time

§ 12.1.2 If a portion of the Work has been covered that the Octave Architect has not specifically requested to examine prior to its being covered, the Octave Architect may request to see such Work and it shall be uncovered by the Contractor. If such Work is in accordance with the Contract Documents, costs of uncovering and replacement shall, by appropriate Change Order, be at the Owner's expense. If such Work is not in accordance with the Contract Documents, such costs and the cost of correction shall be at the Contractor's expense unless the condition was caused by the Owner or a separate contractor in which event the Owner shall be responsible for payment of such costs

### § 12.2 CORRECTION OF WORK

### § 12.2.1 BEFORE OR AFTER SUBSTANTIAL COMPLETION

The Contractor shall promptly correct Work rejected by the Architect or failing to conform to the requirements of the Contract Documents, whether discovered before or after Substantial Completion and whether or not fabricated, installed or completed. Costs of correcting such rejected Work, including additional testing and inspections, the cost of uncovering and replacement, and compensation for the Architect's services and expenses made necessary thereby, shall be at the Contractor's expense.

### § 12.2.2 AFTER SUBSTANTIAL COMPLETION

§ **12.22.1** In addition to the Contractor's obligations under Section 3.5, if, within one year after the date of Substantial Completion of the Work or designated portion thereof or after the date for commencement of warranties established under Section 9.9.1, or by terms of an applicable special warranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Contractor shall correct it promptly after receipt of written notice from the Owner to do so unless the Owner has previously given the Contractor a written specific acceptance of such condition. The Owner shall give such notice promptly

after discovery of the condition During the one-year period for correction of Work, if the Owner fails to notify the Contractor and give the Contractor an opportunity to make the correction, the Owner waives the rights to require correction by the Contractor and to make a claim for breach of warranty If the Contractor fails to correct noncontonning Work within a reasonable time during that period after receipt of notice from the Owner or Architect, the Owner may correct it in accordance with Section 2.4

§ 12.2.2.2 The one-year period for correction of Work shall be extended with respect to portions of Work first performed after Substantial Completion by the period of time between Substantial Completion and the actual completion of that portion of the Work

§ 12.2.2.3 The one-year period for correction of Work shall not be extended on the specific defective work by corrective Work performed by the Contractor pursuant to this Section 12.2 for a period of one year from the time of completion of the corrective work

§ 12.2.3 The Contractor shall remove from the site portions of the Work that are not in accordance with the requirements of the Contract Documents and are neither corrected by the Contractor nor accepted by the Owner

§ 12.2.4 The Contractor shall bear the cost of correcting destroyed or damaged construction, whether completed or partially completed, of the Owner or separate contractors caused by the Contractor's correction or removal of Work that is not in accordance with the requirements of the Contract Documents.

§ 12.2.5 Nothing contained in this Section 12.2 shall be construed to establish a period of limitation with respect to other obligations the Contractor has under the Contract Documents. Establishment of the one-year period for

At Document A201 - - 2007. Copyright @ 1911, 1918, 1928, 1923, 1937, 1953, 1953, 1953, 1955, 1979, 1976, 1997, 1997 and 2007 by The Anarican Institute of Rechrickets All sights reserved. WARNING: This AIA" Document is protected by U.S. Sopyright Law and International Transfies, and will be presecuted to the maximum extent possible under the law. This draft was produced by AIA software exvil and criminal penalties, and will be presecuted to the maximum extent possible under the law. This draft was produced by AIA software exvil and criminal User Notes: (3921870803)

Formatted: Justified

correction of Work as described in Section 12.2.2 relates only to the specific obligation of the Contractor to correct the Work, and has no relationship to the time within which the obligation to comply with the Contract Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Contractor's liability with respect to the Contractor's obligations other than specifically to correct the Work

b and the second s

### § 12.3 ACCEPTANCE OF NONCONFORMING WORK

If the Owner prefers to accept Work that is not in accordance with the requirements of the Contract Documents, the Owner may do so instead of requiring its removal and correction, in which case the Contract Sum will be reduced as appropriate and equitable. Such adjustment shall be effected whether or not final payment has been made

### ARTICLE 13 MISCELLANEOUS PROVISIONS

### § 13.1 GOVERNING LAW AND VENUE

The Contract shall be governed by the law of the <u>State of Illinois</u>. Exclusive venue for any <u>dispute resolution</u> proceeding shall be <u>Cook</u>. County, Illinois, place where the Project is located except that, if the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section 15.4.

### A FIR HAY RIGHTS ACT.

1. Set <u>2007</u> De la Contra d

- 1 If ut village the open of the second se
- 2. <u>A self deated</u>, <u>discussioned becomplete</u>, <u>i</u> <u>i</u> <u>non-net chili</u> e deated at the regularity organization opportunity with a the non-pitered self of replanet in a discussion of accessing a self-term operation in the <u>b</u> high programmer a basis by accessing to re-
- 1. In the state of the stat
- (4) The second secon

AIR Document A201<sup>m</sup> - 2007. Copyright © 1011, 1913, 1918, 1923, 1937, 1951, 1958, 1964, 1965, 1970, 1976, 1987, 1987 and 2007 by The American Institute of Architetts All rights reserved. WANTHO: This AIR Document is protected by U.S. Copyright Law and International Treaties. Unsutherized reproduction or distribution of this AIR Document, or any portion of it, may result in severe civil and criminal penaltiss, and will be presecuted to the maximum extent possible under the law. This diaft was produced by AIR software at 14:15:10 on 01/18/2006 under Order No.1000314784 1 which empires on 8/17/2008, and is not for resels. User Notes. a) A set where the set is not a marked three set on the dynamics of the set of set. Mathematic address on the second and the second se second sec pd have the formation of the real sector of the device of the figure of the real sector of the real sector of the and the proceeding of the

### of a left of a left of

(b) Section 1. (a) address of the section of the first section of the section

Formatted: AIA Body Text Indented, Don't keep with next, Don't keep lines together, Tab stops: 0.5", Left

AIA Document A301 - 2007 Copyright © 1911, 1915, 1916, 1925, 1937, 1951, 1958, 1961, 1963, 1964, 1976, 1976, 1987, 1987 and 2007 by The American Institute G. Alchnicate all rights reserved HOUNING: This AIX Document is protected by U.S. Copyright Law and International Treaties Unauthorized reproduction or distribution of this AIX Document or any portion of it may rosult in severe civil and enimal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIX software at 14:15:10 on 0//18/2008 under Ordet No.1000314794\_1 which express on 8/17/2008, and is not for resale. 40 User Notes (3921870803)

3. C. F. David G. Eldeff, S. Ora, Physical and Consistent substantiation of Surveying all for an analytic case. a state of a second second

TALKA STATE AT TIME ASSAULT A TRACE OF AN ADDRESS AND ADDRESS THAT A STATE ADDRESS ADDR the new new process of same be in a set to be in the second by the secon a townor on a proceeding the Construction of Ann boundary a contrast under and the observed on the off B C. 1915 I C. 18 Dwner, and other path of addition of the is an also is the start of the action Buy a second of a general hadron of the short of the grant as the out of

THE REPORT OF A DESCRIPTION OF A DOMESNEED DOMESNEED AND A DESCRIPTION OF A DOMESNEED AND A DOMESN A DOMESNEED AND A DOMESNEED na hanna a shina da yommar na mashina ha many Subcontacen, there a to jety all lybered, workness, nd na hanna isadama ka adala solar sina ba yayahing waya cin the tarahisi siyadeerati oriyas al'yadaman a where the submit again where and the general probability on the feel holionan pull providing by the controllings at the start Disconduction of Lobor and President to Blenou Jose and statings in orders og availab Const. 13.1 \*\*\* in Contegor and Scherwagenerst of comply out thingers haven't, How, Solat R. Start Bill you was a SALED of half and half and the Buls the surger for the constant research and use a speciation and the term the providing saves the torse, by and Sid matcher prefits in Wey, by the Project will be an on the conforming to the change and shall be confirmed as the shall be a terminal selfer of an de and the second of the former of the former of the second of the prevalet, so a solid of the second of the former o Challen in writing to filmore law

I below the second by her there as a molecule of the Act of has demants as mult if an it should be the best of the a non-Maries by the University of the version of the Constant and the state of the the second star and experimentation of the second start of the second start of the second start of the second start is not second start in the sec the set of the ledelity of damage in many of the or and wells, in the Chance, Construction day, is or refiners is to more more the more with a and employing with denoteds jugite by the Herrin or a the more shown a signed and a barrier of the little of the affect signed to the signed response to a the second state of the second state of the state of the second state of the and a star of the second star and the second star of the second star way and the second star and the second star

15154 here here an event to the test of the transferred to be payrolls as to Owner in complement d), the Illusia dire galine (Wane et al.)

### H LEVOIS DRUG-EREL WORKPLACE AS I

113.1.6 " prot resides a contraster shall regide a terms and an complete the applicable ( has law conductive and Owan to the Constantar

early the sum control and the second cost and to the anterestated by the Contractory shall compare up to the 31 Contraction Contraction of the Linger Street ondition and in set of design appointments and coupley sees map the of the market market to the test solution is set care [130 II CS 550 cross] )

### THE STREET AND OUS PROVISION OF EAW.

It is a start moving the table of a three of an and the product of a separate of start and in an an a start we take the case of the rate of has their prevationer as preversion was a start of the and the set of a map at any a direction monsisting monsisting the providence of the the class of control to be a set of an the treatment becaused, shall be deemed to be used of herein and the set of the presence of the set of outperson and the set of the set o of the second straight the rest of a trad concept incited in a rhout blacking of other or of the in the production of the state of the state

The state of the second of the second of the second s is device the other independently reasonable and in addition she was she as double the shirt of the stand protects wheat the Oak stretche all the set

ATA Decumant A201 - 2007. Copyright 0 1911, 1918, 1918, 1928, 1937, 1983, 1983, 1963, 1963, 1969, 1970, 1876, 1987, 1897, 1897 and 2007 by The American Institute of Architects: All Raghts systemed. WORNING: This ATA Document is protected by U.S. Copyright Law and International Treations: Unauthorized reproduction or distribution of this ATA Document, or any portion of it, may result in severe civil and criminal ponaltics; and will be presented to the maximum extent possible under the law. This draft was produced by ATA software at 14:15:10 on 0/18/2008 under Order No.1000314794\_1 which expires on 0/17/2008, and is not for resals. 41 (3921070003)

Formatted: Font: Not Bold

 $\frac{1}{2} = \frac{1}{2} + \frac{$ 

### § 13.2 SUCCESSORS AND ASSIGNS

§ 13.2.1 The Owner and Contractor respectively bind themselves, their partners, successors, assigns and legal representatives to covenants, agreements and obligations contained in the Contract Documents Except as provided in Section 13 2.2, neither party to the Contract shall assign any obligation, rights, or interest in this Contract the Contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract

§ 13.2.2 The Owner may, without consent of the Contractor, assign the Contract to a lender providing construction financing for the Project, if the lender assumes the Owner's rights and obligations under the Contract Documents. The Contractor shall exceute all consents reasonably required to facilitate such assignment.

### § 13.3 WRITTEN NOTICE

Written notice shall be deemed to have been duly served if delivered in person to the individual, to a member of the firm or entity, or to an officer of the corporation for which it was intended, or if delivered at, or sent by registered or certified mail or by courser service providing proof of delivery to, the last business address known to the party giving notice

### § 13.4 RIGHTS AND REMEDIES

§ 13.4.1 Duties and obligations imposed by the Contract Documents and rights and remedies available thereunder shall be in addition to and not a limitation of duties, obligations, rights and remedies otherwise imposed or available by law

§ 13.4.2 No action or failure to act by the Owner, Architect or Contractor shall constitute a waiver of a right or duty afforded them under the Contract, nor shall such action or failure to act constitute approval of or acquiescence in a breach there under, except as may be specifically agreed in writing

### § 13.5 TESTS AND INSPECTIONS

§ 13.5.1 Tests, inspections and approvals of portions of the Work shall be made as required by the Contract Documents and by applicable laws, statutes, ordinances, codes, rules and regulations or lawful orders of public authorities. Unless otherwise provided, the Contractor shall make arrangements for such tests, inspections and approvals with an independent testing laboratory or entity acceptable to the Owner, or with the appropriate public authority, and shall bear all related costs of tests, inspections and approvals. The Contractor shall give the Architect timely notice of when and where tests and inspections are to be made so that the Architect may be present for such procedures. The Owner shall bear costs of (1) tests, inspections or approvals that do not become requirements until after bids are received or negotiations concluded, and (2) tests, inspections or approvals where building codes or applicable laws or regulations prohibit the Owner from delegating their cost to the Contractor

§ 13.5.2 If the Architect, Owner or public authorities having jurisdiction determine that portions of the Work require additional testing, inspection or approval not included under Section 13.5.1, the Architect will, upon written authorization from the Owner, instruct the Contractor to make arrangements for such additional testing, inspection or approval by an entity acceptable to the Owner, and the Contractor shall give timely notice to the Architect of when and where tests and inspections are to be made so that the Architect may be present for such procedures. Such costs, except as provided in Section 13.5.3, shall be at the Owner's expense.

§ 13.5.3 If such procedures for testing, inspection or approval under Sections 13.5.1 and 13.5.2 reveal failure of the portions of the Work to comply with requirements established by the Contract Documents, all costs made necessary

All Document Al01" - 2007. Copyright © 1911, 1915, 1915, 1925, 1937, 1951, 1958, 1953, 1956, 1978, 1978, 1978, 1977, 1997 and 2007 by The American Institute of Suchitets: All rights reserved. WARNING: This All Document is protected by U.S. Copyright has and International Traties. Unauthorized copyrightation or distribution of this Al% Document, or any portion of it, may result in severe civil and crisical ponalizes, and will be prosecuted to the maximum extent possible under the law. This draft was produced by ALA software at 14:15:10 on 0/10/2000 under Order No 1000314754\_1 which expires on 8/17/2008, and is not for resale User Notes: (3921870803)

by such failure including those of repeated procedures and compensation for the Architect's services and expenses shall be at the Contractor's expense

§ 13.5.4 Required certificates of testing, inspection or approval shall, unless otherwise required by the Contract Documents, be secured by the Contractor and promptly delivered to the Architect

§ 13.5.5 If the Architect is to observe tests, inspections or approvals required by the Contract Documents, the Architect will do so promptly and, where practicable, at the normal place of testing

§ 13.5.6 Tests or inspections conducted pursuant to the Contract Documents shall be made promptly to avoid unreasonable delay in the Work

### § 13.6 INTEREST

Payments due and unpaid under the Contract Documents shall bear interest from the date payment is due at such rate as the parties may agree upon in writing or, in the absence thereof, at the legal rate prevailing from time to time at the place where the Project is located <u>No interest shall accrue on amounts held in recention</u>.

### § 13.7 TIME LIMITS ON CLAIMS

The Owner and Contractor shall commence all claims and causes of action, whether in contract, lort, breach of warranty or otherwise, against the other arising out of or related to the Contract in accordance with the requirements of the final dispute resolution method selected in the Agreement within the time period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Contractor waive all claims and causes of action not commenced in accordance with this Section 13.7.

### ARTICLE 14 TERMINATION OR SUSPENSION OF THE CONTRACT § 14.1 TERMINATION BY THE CONTRACTOR

§ 14.1.1 The Contractor may terminate the Contract if the Work is stopped for a period of 30 consecutive days through no act or fault of the Contractor or a Subcontractor, Sub-subcontractor or their agents or employees or any other persons or entities performing portions of the Work under direct or indirect contract with the Contractor, for

- any of the following reasons .1 Issuance of an order of a court or other public authority having jurisdiction that requires all Work to be stopped.
  - .2 An act of government, such as a declaration of national emergency that requires all Work to be stopped.
  - .3 Because the Architect has not issued a Certificate for Payment and has not notified the Contractor of the reason for withholding certification as provided in Section 9.4.1, or because the Owner has not made payment on a Certificate for Payment within the time stated in the Contract Documents, or
  - The contract Documents of the second process of the contract Documents of the contr

 mit
 tell
 tell

§ 14.1.3 If one of the reasons described in Section 14.1 I or 14.1.2 exists, the Contractor may, upon seven days' written notice to the Owner and Architect, terminate the Contract and recover from the Owner payment for Work

§ 14.1.4 If the Work is stopped for a period of the consecutive days through no act or fault of the Contractor or a Subcontractor or their agents or employees or any other persons performing portions of the Work under contract with the Contractor because the Owner has repeatedly failed to fulfill the Owner's obligations under the Contract Documents with respect to matters important to the progress of the Work, the Contractor may, upon seven additional days' written notice to the Owner and the Architect, terminate the Contract and recover from the Owner as provided in Section 14.1.3

AIA Document A201<sup>20</sup> - 2007. Copyright © 1911, 1915, 1918, 1823, 1837, 1951, 1951, 1951, 1861, 1861, 1966, 1970, 1376, 1867, 1867 and 2007 by The American Institute of Allowington of distribution of this AIA Document is protected by U.S. Copyright av and International Treaties. Unauthorized reproduction or distribution of this AIA Document, or any portion of it, may result in sovere civil and common ponities, and will be prosocuted to the maximum acteant possible under the law. This draft was produced by AIA softwars at 14:15.10 on 01/18/2008 under Cader No.1000314784\_1 which espires on 8/17/2008, and is not for resule. User Notes:

### § 14.2 TERMINATION BY THE OWNER FOR CAUSE and the second Formatted: AIA Agreement Body Text is to entry and the statement of the second statement and a second statement of the ---- I------ die Constantin mit die 1. I the second s 4 A second provide a second se second sec Child In the second second second second The the Contract of Million and Contract of Contract of Contract of Contract of Contract of Contract of Contract in all of the second second second management of the second s 3.15 Formatted: AIA Body Text Indented, Left 2. By Conversely a climation of the incommendation of the state of the under the heaters! By a many a second, he supported by the last or all or upon rade and fighted or the hole upper, in modeling the limit former fight out with genner reading a water to the the transferred and the second second and the Case of a deal ar writes, his mability to pay of dible scorally sither to me he he is done if going I seganeral er the last fit a bis greation of the control hand participation and the participation of the second an incompatible build an you must carey Formatted: AIA Body Text Indented, Justified 25 Sector graphics and the above of the sector of the s Formatted: AIA Body Text Indented, Justified 1 Source of the state of the Formatted: AIA Body Text Hanging and the start of the second seco Formatted: Font: Bold Formatted: AIA Body Text Hanging, Indent: 0.0 Left: 0.49", Hanging: 0.01" Formatted: AIA Body Text Hanging when the second of the later of the second for star and the second s in a near their or build as a sensible softwarty history of the son of the Work of the sector in Formatted: AIA Body Text Hanging, Justified, Indent: Left: 0.49", Hanging: 0.01" Freedomente Formatted: AIA Body Text Hanging, Justified \$14.22 Los Oxner - Changer and the second state of the provide state of the operation of the operation of the second state of Formatted: Justified, Indent: Left: 0", First stor Les or materials and a more often as the Committee or and the Streps restor the Performence (166) and materials line: 0" the rad May gl Poys on Bond discused in Longraph II there of the day's stiften notes, hereighter or o play as a office Contractor. It traces all by the Overet, the Contract is shed to pays or pathot all of the current and her only supple. I contraste of the Protect wallings is 10700, then the date of such residence the gradie come functions if any the outer of such the next forth and the control the Conternal's galary to dy south Over a differenced right to the hyperdy or estight you much mereor and stephes the Compactor's press. the most found that an an analysis the second section of the contraction of the second section of the weather that Weather the Weather that we are the second section of the secti (a) A set of the Octorie set of the set o Formatted: AIA Agreement Body Text § 14.2.2 When any of the above reasons exist, the Owner, upon certification by the Initial Decision Maker that sufficient cause exists to justify such action, may without prejudice to any other rights or remedies of the Owner and after giving the Contractor and the Contractor's surety, if any, seven days' written notice, terminate employment of the Contractor and may, subject to any prior rights of the surely A ----- Exclude the Contractor from the site and take possession of all materials, equipment, tools, and Formatted: AIA Agreement Body Text construction equipment and machinery thereon owned by the Contractor. 2 Accept assignment of subcontracts pursuant to Section 5.4, and - Finish the Work by whatever reasonable method the Owner may deem expedient. Upon written request of -6the Contractor, the Owner shall furnish to the Contractor a detailed accounting of the costs incurred by the Owner in finishing the Work-ATA Decument A201" - 2007. Copyright © 1911, 1915, 1918, 1925, 1917, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The Averian Institute of Architects. All sights reserved. WOUNTING: This AIA' Document is protected by U.S. Copyright Law and International Treaties. Unsutherized reproduction or distribution of this XIA' Document, or any portion of it, may result in sevure civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 14-15:10 on 0/178/2009 under Order Wo.1000314794\_i which expires on 0/17/2000, and is not for resule ()921070001 44

## § 14.2.3 When the Owner terminates the Contract for one of the reasons stated in Section 14.2.1, the Contractor shall not be entitled to receive further payment until the Work is finished

### § 14.3 SUSPENSION BY THE OWNER FOR CONVENIENCE

§ 14.3.1 The Owner may, without cause, order the Contractor in writing to suspend, delay or interrupt the Work in whole or in part for such period of time as the Owner may determine

§ 14.3.2 The Contract Sum and Contract Time shall be adjusted for increases in the cost and time caused by suspension, delay or interruption as described in Section 14.3.1. Adjustment of the Contract Sum shall include profit. No adjustment shall be made to the extent

- .1 that performance is, was or would have been so suspended, delayed or interrupted by another cause for which the Contractor is responsible, or
- .2 that an equitable adjustment is made or denied under another provision of the Contract

### § 14.4 TERMINATION BY THE OWNER FOR CONVENIENCE

§ 14.4.1 The Owner may, at any time, terminate the Contract for the Owner's convenience and without cause

§ 14.4.2 Upon receipt of written notice from the Owner of such termination for the Owner's convenience, the Contractor shall

- .1 cease operations as directed by the Owner in the notice,
- .2 take actions necessary, or that the Owner may direct, for the protection and preservation of the Work, and
- .3 except for Work directed to be performed prior to the effective date of termination stated in the notice, terminate all existing subcontracts and purchase orders and enter into no further subcontracts and purchase orders

§ 14.4.3 In case of such termination for the Owner's convenience, the Contractor shall be entitled to receive payment for Work \_\_\_\_\_\_ executed\_\_\_\_\_\_

### ARTICLE 15 CLAIMS AND DISPUTES § 15.1 CLAIMS § 15.1.1 DEFINITION

A Claim is a demand or assertion by one of the parties seeking, as a matter of right, payment of money, or other relief with respect to the terms of the Contract. The term "Claim" also includes other disputes and matters in question between the Owner and Contractor arising out of or relating to the Contract. The responsibility to substantiate Claims shall rest with the party making the Claim

### § 15.1.2 NOTICE OF CLAIMS

Claims by either the Owner or Contractor must be initiated by written notice to the Owner-other party and to the Initial Decision Maker. If any, with a copy sent to the Architect, if the Architect is not serving as the Initial Decision

ATA Document A201 - 5007. Copyright © 1911, 1918, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1987, 1987, 1987 And 2007 by The American Institute of Architects. All mights reserved. WARVING: This XIX Document is protected by U.S. Copyright Law and International Treation. Unsuffering of Architects. All mights reserved. WARVING: This XIX Document is protected by U.S. Copyright Law and International ponalise, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 14-15-10 on 0//3/2008 under Order No 2000314794 2 which expires on 8/17/2008, and is not for resele. (39214704001)

Formatted: Justified

Maker Claims by either party the Contractor must be initiated within 21 days after occurrence of the event giving rise to such Claim or within 21 days after the <u>Contractor elaimant</u> first recognizes the condition giving rise to the Claim, whichever is later.

### § 15.1.3 CONTINUING CONTRACT PERFORMANCE

Pending final resolution of a Claim, except as otherwise agreed in writing or as provided in Section 9.7 and Article 14, the Contractor shall proceed diligently with performance of the Contract and the Owner shall continue to make payments in accordance with the Contract Documents. The Architect will prepare Change Orders and issue Certificates for Payment in accordance with the decisions of the Initial Decision Maker.

### § 15.1.4 CLAIMS FOR ADDITIONAL COST

If the Contractor wishes to make a Claim for an increase in the Contract Sum, written notice as provided herein shall be given before proceeding to execute the Work. Prior notice is not required for Claims relating to an emergency endangering life or property arising under Section 10.4.

### § 15.1.5 CLAIMS FOR ADDITIONAL TIME

§ 15.1.5.1 If the Contractor wishes to make a Claim for an increase in the Contract Time, written notice as provided herein shall be given. The Contractor's Claim shall include an estimate of <del>cost and of</del> probable effect of delay on progress of the Work. In the case of a continuing delay, only one Claim is necessary <u>For Claims for Additional time</u> to the extent that an estimated extension of time is warranted, such extension shall be the Contractor's sole and exclusive remedy.

§ 15.1.5.2 If adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data substantiating that weather conditions were abnormal for the period of time, could not have been reasonably anticipated and had an adverse effect on the scheduled construction

51515.3 For all Claims for Additional Line, the Contractor shall support such Claims in the same manuer as supporting additional time for Change Orders.

### § 15.1.6 CLAIMS FOR CONSEQUENTIAL DAMAGES

The Contractor and Owner warve Claims against each other for consequential damages arising out of or relating to this Contract. This mutual waiver includes

- .1 damages meurred by the Owner for rental expenses, for losses of use, income, profit, financing, business and reputation, and for loss of management or employee productivity or of the services of such persons; and
- damages incurred by the Contractor for principal office expenses including the compensation of personnel stationed there, for losses of financing, business and reputation, and for loss of profit except anticipated profit arising directly from the Work.

This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination in accordance with Article 14. Nothing contained in this Section 15.1.6 shall be deemed to proclude an award-of liquidated damages, when applicable, in accordance with the requirements of the Contract Documents.

### § 15.2 INITIAL DECISION

§ 15.2.1 Claims by the Contractor, excluding those arising under Sections 10 3, 10 4, 11 3, 9, and 11 3, 10, shall be referred to the Initial Decision Maker for initial decision. The Architect will serve as the Initial Decision Maker, unless otherwise indicated in the Agreement Except for those Claims excluded by this Section 15.2.1, an initial decision shall be required as a condition precedent to mechationarbitration or litigation, as the case may be, of any <u>Contractor</u> Claim arising prior to the date final payment is due, unless 30 days have passed after the Claim has been referred to the Initial Decision Maker with no decision having been rendered. Unless the Initial Decision Maker and all affected parties agree, the Initial Decision Maker will not decide disputes between the Contractor and persons or entities other than the Owner

§ 15.2.2 The Initial Decision Maker will review Claims and within ten days of the receipt of a Claim take one or more of the following actions (1) request additional supporting data from the claimant or a response with supporting data from the other party, (2) reject the Claim in whole or in part, (3) approve the Claim, (4) suggest a compromise, or (5) advise the parties that the Initial Decision Maker is unable to resolve the Claim if the Initial Decision Maker lacks sufficient information to evaluate the merits of the Claim or if the Initial Decision Maker concludes that, in the

AIR Document A201<sup>20</sup> - 2007. Copyright © 1913, 1915, 1918, 1928, 1937, 1955, 1958, 1963, 1963, 1965, 1970, 1976, 1987 and 2007 by The American Institute of Acchivetus. All sights reserved WOUNTHO: This AIR' Document is protocted by U.S. Copyright Law and International Trotaties Unauthorized septeduction or distribution of this AIR' Document, or any portion of it, may result in server civil and cristical penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIR software at 1.05:00 on C1/18/2008 under Order No 1000314794 1 which expires on 8/17/2008, and is not for reale. User Notes: (3921870803)

Initial Decision Maker's sole discretion, it would be mappropriate for the Initial Decision Maker to resolve the Claim

§ 15.2.3 In evaluating Claims, the Initial Decision Maker may, but shall not be obligated to, consult with or seek information from either party or from persons with special knowledge or expertise who may assist the Initial Decision Maker in rendering a decision The Initial Decision Maker may request the Owner to authorize retention of such persons at the Owner's expense

§ 15.24 If the Initial Decision Maker requests a party to provide a response to a Claim or to furnish additional supporting data, such party shall respond, within ten days after receipt of such request, and shall either (i) provide a response on the requested supporting data, (2) advise the Initial Decision Maker when the response or supporting data will be furnished or (3) advise the Initial Decision Maker that no supporting data will be furnished Upon receipt of the response or supporting data, if any, the Initial Decision Maker will either reject or approve the Claim in whole or in part

§ 15.2.5 The Initial Decision Maker will render an initial decision approving or rejecting the Claim, or indicating that the Initial Decision Maker is unable to resolve the Claim. This initial decision shall (1) be in writing, (2) state the reasons therefor, and (3) notify the parties and the Architect, if the Architect is not serving as the Initial Decision Maker, of any change in the Contract Sum or Contract Time or both. The initial decision shall be final and binding on the parties but subject to mediation and, if the parties fail to resolve their dispute through mediation, to binding dispute resolution

§ 15.2.6 Either party may file for mediation of an initial decision at any time, subject to the terms of Section 15.2.6.1

§15.2.6.1 Either party may, within 30 days from the date of an initial decision, demand in writing that the other party file for mediation within 60 days of the initial decision. If such a demand is made and the party receiving the demand fails to file for mediation within the time required, then both parties waive their rights to mediate or pursue binding dispute resolution proceedings with respect to the initial decision.

§ 15.2.7 In the event of a Claim against the Contractor, the Owner may, but is not obligated to, notify the surety, if any, of the nature and amount of the Claim If the Claim relates to a possibility of a Contractor's default, the Owner may, but is not obligated to, notify the surety and request the surety's assistance in resolving the controversy

§ 15.28 If a Claim relates to or is the subject of a mechanic's lien, the party asserting such Claim may proceed in accordance with applicable law to comply with the lien notice or filing deadlines.

### § 15.3 MEDIATION

§ 15.3.1 Claims, disputes, or other matters in controversy arising out of or related to the Contract except those waived as provided for in Sections 9.10.4, 9.10.5, and 15.1.6 shall be subject to mediation as a condition procedent to binding dispute resolution

§ 15.3.2 The parties shall endeavor to resolve their Claims by mediation which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Mediation Procedures in effect on the date of the Agreement. A request for mediation shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the mediation. The request may be made concurrently with the filing of binding dispute resolution proceedings but, in such event, mediation shall proceed in advance of binding dispute resolution proceedings, which shall be stayed pending mediation for a period of 60 days from the date of filing, unless stayed for a longer period by agreement of the parties or court order. If an arbitration is stayed pursuant to this Section 15.3.2, the parties may nonetholess proceed to the selection of the arbitrator(s) and agree upon a schedule for later proceedings.

§ 15.3.3 The parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction thereof.

### § 15.4 ARBITRATION

§ 15.4.1 Stang out an include the standard of the Characher all character of the standard of t a las martis son contact a survey more a toring to be a ferre or and contact and

AIA Document A201<sup>m</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1863, 1966, 1970, 1976, 1987, 1887, 1887 and 2007 by The American Institute of Alcharacts. All sights reserved. MARNING This AIA Document is protected by U.S. Copyright Law and International Troaties Unauthorized reproduction or distribution of this AIA Document, or any portion of it, may result in sevare civil and criminal ponalies, and will be prosecuted to the maximum extent possible under the law. This desit was produced by AIA software at 14-13:20 on 03/18/2006 under Order No.1000314794\_1 which expires on 8/17/2008, and is not for resale 47 (3921070803)

1. Construction of the second seco

1. The second second

All statistics in a state where a new state of the new state of a state of the s

P. P. P. M. By, traje to the bits at the ball exable to gravity or control of the ball exable to gravity or control of the control of the

by the arbitrator

a 1 g a r d and a star of the prevailing party

If the parties have selected arbitration as the method for binding dispute resolution in the Agreement, any Claim subject to, but not resolved by, mediation shall be subject to arbitration which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Arbitration Rules in effect on the date of the Agreement. A demand for arbitration shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the arbitration. The party filing a notice of demand for arbitration must assert in the demand all Claims then known to that party on which arbitration is permitted to be demanded.

§15.4.1.1 A demand for arbitration shall be made no earlier than concurrently with the filing of a request for mediation, but in no event shall it be made after the date when the institution of legal or equitable proceedings based on the Claim would be barred by the applicable statute of limitations. For statute of limitations purposes, receipt of a written demand for arbitration by the person or entity administering the arbitration shall constitute the institution of legal or equitable proceedings based on the Claim.

§ 15.4.2 The award rendered by the arbitrator or arbitrators shall be final, and judgment may be entered upon it in accordance with applicable law in any court having jurisdiction thereof.

AIA Decomment AIO1\* - 2007. Copyright © 1811, 1918, 1918, 1923, 1837, 1853, 1853, 1863, 1863, 1866, 1870, 1876, 1867, 1867, 1867 and 2007 by The American Institute of Frencience. All sights reserved HARVING This AIA' Document is protocted by U.S. Copyright Law and Frentional Treaties. Unauthorized regraduation or distribution of this AIA' Document, or any portion of it may result to severe exvil and criminal penalties, and wall be presected to the maximum extool possible under the law. This draft was produced by AIA software at 14.15:10 on 01/13/2006 under Order No.1000314754 1 which expires on 8/17/2009. and is not for resale. (1921070805),

Formatted: Font: (Default) Times New Roman

Formatted: Font: (Default) Times New Roman

Formatted: Font: (Default) Times New Roman

§ 15.4.3 The foregoing agreement to arbitrate and other agreements to arbitrate with an additional person or entity duly consented to by parties to the Agreement shall be specifically enforceable under applicable law in any court having jurisdiction thereof.

### § 15.4.4 CONSOLIDATION OR JOINDER

§ 15.4.4.1 Either party, at its sole discretion, may consolidate an arbitration conducted under this Agreement with any other arbitration to which it is a party provided that (1) the arbitration agreement governing the other arbitration permits consolidation, (2) the arbitrations to be consolidated substantially involve common questions of law or fact, and (3) the arbitrations employ materially similar procedural rules and methods for selecting arbitrator(s).

§ 15.4.4.2 Either party, at its sole discretion, may include by joinder persons or entities substantially involved in a common question of law or fact whose presence is required if complete relief is to be accorded in arbitration, provided that the party sought to be joined consents in writing to such joinder. Consent to arbitration involving an additional person or entity shall not constitute consent to arbitration of any claim, dispute or other matter in question not described in the written consent.

§ 15.4.4.3 The Owner and Contractor grant to any person or entity made a party to an arbitration conducted under this Section 15.4, whether by joinder or consolidation, the same rights of joinder and consolidation as the Owner and Contractor under this Agreement.

AIA Document AIO1 - 2007. Copyright © 1911, 1915, 1915, 1925, 1937, 1951, 1951, 1954, 1966, 1976, 1976, 1976, 1987, 1987 and 2007 by The Aresican Institute of Architects. All rights reserved. WANNING: This AIA Document is protected by U.S. Copyright Law and International Froaties: Nonathorized reproduction or distribution of this AIA' Document, or any portion of it, may result in severe civil and criminal ponalties, and will be proseculed to the maximum extent possible under the law This draft was produced by AIA software at 14:13/10 on 0/18/2009 under order No 1000314764 1 which expires on 8/17/2008. and is not for ressle. User Notes

## Oak Park and River Forest High School District 200

201 North Scoville Avenue • Oak Park, IL 60302-2296

TO: Board of Education

FROM: Lauren Smith, Director of Human Resources

DATE: February 24, 2011

Cc: Stephen Isoye, Superintendent

RE: Personnel Recommendation/Actions

### BACKGROUND

The personnel report for February 24, 2011 includes a recommendation for new hires and leaves of absence.

### SUMMARY OF FINDINGS

New hires include:

- 0.55 New FTE (4 hours and 24 minutes) CPA for Business Office Support
- 1.0 New FTE for Safety & Support (Cafeteria monitor)
- 1.0 Replacement FTE for Safety & Support

You will also see that you have been provided with leave information and request(s). This includes what type of leave, the location, and the duration if known. Names for mandated leaves have not been included due to privacy rights under HIPAA.

### Next Steps

Motion: Move to approve the Personnel Recommendations as presented.

TEL: (708) 383-0700

Oak Park and River Forest High School District 200 – Board of Education February 24, 2011 Personnel Report

# NEW HIRES

# **Classified Personnel**

1943		
<b>Date</b>	7, 2011	
fective	February 7	
ц <u></u>		
	\$15.29/hour	
ß	\$15.2	
if Exp	s	
ars of	10 years	
ž		
jree		
ist Dec	Bachelors	-
Highe	Ba	
ae	lary Beth	Erickson
Sa	Mary	Eric
uo	Office	
Locati	siness	
	Bu	
	ffice	
New	ness Oi	hoddu
	Busil	0
1		

# Safety and Support Team

			1
Effective Date	February 7, 2011	•	
Salary	\$18.62/hour		
Years of Exp	0		-
<b>Highest Degree</b>	Bachelors		
Name	Andre Harris		
Location	Safety & Support		
New	Part-time Cafeteria	Monitor (3 hours)	

Safety & S
------------

# MANDATORY LEAVES

Effective Date	January 31, 2011 – February 25, 2011	February 3, 2011 – March 2, 2011	February 14, 2011 – February 25, 2011
Location	Math	B&G	Business Office
Type	FMLA	FMLA	FMLA

# LEAVES

Effective Date	February 11, 2011 - TBD
Reason	Medical
Location	Safety & Support
Name	Climent Edmond Jr.

## Oak Park and River Forest High School District 200

201 North Scoville Avenue 

Oak Park, IL 60302-2296

Office of Human Resources

(708) 434-3214 Fax (708) 434-3919

RE:	Sabbatical Leave Applications
DATE:	February 24, 2011
Cc:	Dr. Steven Isoye, Superintendent
FROM:	Lauren Smith, Director of Human Resources
TO:	Board of Education

### BACKGROUND

As is the custom of the District and in keeping with the Collective Bargaining Agreement with Faculty Senate, the Board of Education receives applications for Sabbatical Leaves in the month of February. The Sabbatical review and approval process involves three phases:

- 1) The District Sabbatical Review Committee
- 2) The Superintendent
- 3) The Board of Education

This year there is one application for the Board of Education to consider from Avi Lessing. Mr. Lessing has been an English teacher with the District since 2003.

### **SUMMARY OF FINDINGS**

The District's Sabbatical Review Committee and Dr. Isoye have approved Mr. Lessing's application. We believe that he has successfully communicated his motives for the leave, the activities in which he plans to participate, and how the sabbatical will benefit him, the students, and his colleagues. Mr. Lessing has also demonstrated, through his previous activities, his commitment to professional growth and student achievement.

### Next Steps

The Sabbatical Committee and the Superintendent recommends the approval of the Sabbatical Leave request of Avi Lessing for the 2011-2012 school year.

Roll Call Vote

Agenda Item No. XI. B

Date: 1/6/11

### OAK PARK AND RIVER FOREST HIGH SCHOOL APPLICATION FOR SABBATICAL LEAVE

Directions:	Agreement. Answer the Return the completed	e questions below applications by J	as completely as pos anuary 18, 2011 to	ssible, attaching ac room 215. Pleas	the Board/Faculty Senate Iditional pages as required. e type all portions of this
Name:	LESSENG Last	Aus	Ĵ Middle	ENGL	In
	Last	First	Middle	Divisio	n
Number of year	s employed at OPRFHS	(including current y	rear):	,	
Have you previo	ously been on sabbatical	leave while employ	ved at OPRFHS?	Yes	<u> </u>
	state the dates and the a gram of the school:	activities complete	d and the impact the	e activities had for	your teaching and for the
Dates:	From:		To:		
Activitie	es:				
When do you w	ish to be on sabbatical le	ave? From: _	To: _		
Please respond	to the following question	s and attach your a	answers to this applic	ation:	
1. What is your	primary motive for reques	sting a sabbatical le	eave?		
	activities in which you pla sent on each activity.)	an to participate ar	nd the goals of each a	activity? (Indicate a	pproximate amount of time
3. How will your	sabbatical leave be of be	enefit to you, your s	students, and your co	lleagues?	
4. How do you i	ntend to share the inform	ation gained from y	your sabbatical leave	with the school cor	nmunity?
	have you made in the la /our basic job responsibili		nance your profession	nal growth and/or t	o contribute to the district's
6. Will you rece amount.	ive any type of outside gr	ant or remuneratio	n as a result of your s	sabbatical activities	? If so, please indicate the

For Committee Use -Comments:

Recommendations:

Endorsements and Approval:

Division Head or Program Administrator

Signature

Ila/n Date

Director of Human Resources

Signature

Date

Superintendent

Signature

Date

Board of Education

Signature

Date

### 1. What is your primary motive for requesting a sabbatical leave?

I hope to take a sabbatical leave from August of 2011 to June of 2012 to write my dissertation for a PhD in Curriculum Studies at UIC. I want my work to better ground me in pedagogical theory, create new curricula for my courses (and potentially the school), and allow me to reflect deeply on my teaching practice. Under the mentorship of William Ayers, Distinguished Professor of Education and Senior University Scholar (now retired), I have recently finished my coursework at the University. Dr. Bill Schubert, a renowned scholar in the field and author and editor of over 20 books, is my doctoral advisor on my PhD research project.

My work will focus on the field of Social Emotional Learning (SEL) and how it can impact student learning. This long-held interest, has already led to the emergence of a new course, Experiments in Reading Literature and the World ("Experiments"), which I began teaching in 2005. The course strives to reach students on intellectual, emotional, intuitive, and kinesthetic levels, and has positively impacted hundreds of seniors already. Students often say it was the place where they understood how their own lives were bound with the lives of their classmates. Moreover, the SEL techniques I have learned have enhanced my effectiveness in all of my courses. SEL addresses and removes obstacles that so often stymie students. More and more research shows the interconnectedness of academic learning and SEL, and many studies show enormous benefit to high school students when teachers directly incorporate SEL in academic classes (CASEL). I imagine further research in the field will foster a stronger SEL framework (for both my classes and the school) that will expand students' capacity for learning.

# 2. What are the activities in which you plan to participate and the goals of each activity? (Indicate approximate amount of time to be spent on each activity.)

Since all of the necessary coursework as well as the preliminary exam, proposal, and oral preliminary exam will have been completed by the fall of 2011, the primary activity of the sabbatical will be to write my dissertation. The dissertation will be a qualitative, case study of my classroom experiences. I want to examine how teachers and students encounter each other from often shifting ways of knowing and then examine the scenarios and situations that make them change. This research will hopefully offer educators some long-term strategies and moment-to-moment tactics to address issues such as self-awareness, social-awareness, and decision-making.

First quarter I will write a literature review, an in-depth study of what others have said in the fields of SEL, phenomenology, and Curriculum Studies. In the winter, I will develop my theoretical framework: my plan is to critique the way progressive education glosses over some of the difficulties of SEL and in effect discredits its aims. Also, in the winter, I plan to systematically interview former students to discern how they made meaning in my classroom and whether their sense of those experiences has shifted through the years. I am particularly curious about the way the institution of school impacted their perceptions. Those interviews will explore the moments when their own understanding of their learning, their classmates, or their teacher changed and how they articulated that shift. These interviews (conducted both in groups and one-on-one) will serve as the data set of my case study. The bulk of the dissertation will be written in the winter and early spring. The summer will be spent taking what I have learned and implementing it into the curriculum of the courses I will be teaching. In addition, I will have begun to conceive of ways to integrate SEL school-wide (I can outline some of those ideas to the committee if desired.).

### 3. How will your sabbatical leave be of benefit to you, your students, and your colleagues?

I have no doubt that the sabbatical would make me a substantially better teacher at OPRF. After teaching for more than a decade, I have become aware of my weaknesses and am intent upon addressing them. A sabbatical devoted to academic research and writing would enable me to implement curricular work that would increase my impact on student achievement and sharpen my technique in the classroom. When I return with promising ideas and methods drawn from my research and writing, I will have gained even greater command of how SEL can fit into the scope and sequence of curriculum, evaluation, and learning theory.

The sabbatical will also directly affect my teaching and students in a more immediate way. The act of writing such a sustained piece will improve the writing instruction I provide in my English classes. I agree with Peter Elbow, former department chair of the University of Massachusetts' writing program, who often lamented that teachers could not offer effective writing instruction unless they themselves wrote. I know all of the members of my learning team from last year, Teachers Who Write, benefited from the act of analyzing our own writing in conjunction with understanding how students construct written pieces. Teachers who engage in the very activities they ask of their students become attuned to the requirements of strong writing and intimately familiar with obstacles that students must negotiate to become skilled, confident writers. In writing a book length dissertation, I will continue to practice the skills that I want to impart to my students: storytelling, argumentation, research, conferencing, revising, and editing.

The sabbatical will also offer benefits for my colleagues including the two teams I teach in, departmental learning teams, and the inter-divisional connections I have enjoyed with teachers in mathematics, theater, and history. Grants that I have received from the school have proven helpful to my colleagues, since whatever knowledge I gained was immediately shared with co-teachers or learning team members. Just as I can say without qualification that James Bell has made me a better crafter of lesson plans and that Mike Soffer has helped me implement backward design in all my classes, I know that they would similarly express how I have helped them reach more students, be more innovative, and reflect more deeply about their teaching practice. A sabbatical would make those benefits less incalculable and more deliberate. Those teams could serve as exemplars of how to inculcate SEL within an existing curriculum.

Just as we need teachers to do further study in their subject area, we need teachers to do in-depth training in Curriculum Studies, particularly SEL. It is crucial to reflect systematically on the significant curriculum questions. What knowledge is worth having? How do we teach what is worth knowing? How do we know when students have internalized that knowledge? What obstacles do students face in the pursuit of that knowledge? How do we develop student capacity for resilience? How as teachers do we understand and respond to students' resistance? My study will offer meaningful insight to those questions and can be a valuable resource to my colleagues. Research shows that "schools that help students and staff develop greater social and emotional competence, in turn help ensure positive short- and long-term academic and personal outcomes for students, and higher levels of teaching and work satisfaction for staff" (CASEL). After my sabbatical, I will be in a strong position to help the school fulfill SEL state standards. We need to systematically implement SEL across the curriculum and this work could be a springboard towards such an effort.

4. How do you intend to share the information gained from your sabbatical leave with the school community?

Teaching, at its best, is as an endless, open, and honest process of collaboration, reflection, productive work, and feedback. I anticipate sharing my work both formally and informally in a number of ways:

- In my team teaching
- > With the SEL team Jeremiah Wiencek has assembled
- > With Amy Hill and her work around our school's climate and mission
- > With Dan Cohen around training the department in SEL
- With Cindy Milojevic planning class assemblies (this has already begun; see below)
- > With Peter Kahn and his student mentoring program
- > With Judith Lopez assembling SEL resources for students in the library, and
- With students engendering greater transparency in curricular, methodological, and pedagogical choices.

My hope, in addition, is to convert the dissertation into a published book that can be useful to colleagues in discussion, planning, and teaching. I am confident that my training and record of publication have put me in a position to succeed. My college honors thesis, a 108-page essay on how my experiences in college with death<sup>1</sup> mixed reflection on personal experience with academic research. In recognition of its quality the supervising committee conferred upon me the honor of magna cum laude. My most recent paper, titled "A Soft Approach to Hard Teaching," was published last year in the *Handbook of Social Justice in Education*, edited by William Ayers, Therese Quinn, and David Stovall. In addition, Dr. Ayers and Rick Ayers, a professor at University of San Francisco, wrote a book entitled *Teaching the Taboo* in which one chapter considers some of the innovations I have made in the classroom.

My hope is that a book of essays I write will make its own contribution to the field of Curriculum Studies and will clarify some of the techniques available to teachers attempting to do SEL work in the classroom. I hope this book, like the work of so many of my colleagues, will bring local, state-wide, and national prominence to Oak Park River Forest High School (OPRF) as one of the few large, public high schools with a diverse population doing cutting edge work. It could make us a "beacon" school in yet another way.

5. What efforts have you made in the last five years to enhance your professional growth and/or to contribute to the district's efforts beyond your basic job responsibilities?

<sup>&</sup>lt;sup>1</sup> Within six months of my junior year in college my freshmen year roommate died of a brain tumor, my father got prostate cancer, my two best friends were hit by a car and one was killed, and I developed a tumor in my neck.

As a teacher I believe I should do everything I can to meet the expectations of my students and this community. For that reason, I have always taken advantage of the opportunities to engage in extended professional learning through OPRF's teacher grant program. I have benefited from leadership training opportunities in Oregon (Inquiry Based Learning), Colorado (Matrix Leadership), Philadelphia (Westtown Seminar) and Illinois (Internal Family System). In addition, I have attended several other leadership conferences and met with people from Collaborative for Academic, Social and Emotional Learning (CASEL) which later led to a learning team I proposed that included reading in SEL.<sup>2</sup> In theater, I have benefited from training at Second City, Improv Olympic, and courses in directing through Chicago Dramatists. All of this is in addition to the professional development I have gained through the PhD program at UIC.

In terms of what I have offered the district, over the last five years I have directed a play each year and twice during the summer. My work has included challenging plays such as Ionesco's *Rhinoceros* and Pirandello's *Five Characters in Search of an Author*, as well as less conventional work including an original Improv show. These different kinds of plays in both style and content – experimental, traditional, intellectual, and original – have consistently been rewarding experiences for the students involved. In much of my extra-curricular work with youth, the students become part of ensembles that are often life changing. Students continually comment to me that they remain a support group to one another well after the production has passed. Here is one direct way SEL impacts both student experience and outcome. Their experience with one another and me impacts their level of performance.

One piece I directed, a documentary play — Fires in the Mirror<sup>3</sup> — generated a larger response within the school. Then Superintendent Sue Bridge requested that it be remounted as an Institute Day presentation. The themes of the play – understanding difference and conflict through a multitude of voices and perspectives – resonated with the school's mission of equal voice and equity. In addition, for all of the performances, I coordinated several guest facilitators (including Dr. David Stovall from UIC) to lead post show discussions on how the motifs the play confronts – power, race, difference – exist at OPRF. It was a powerful way to validate student voices (particularly of color) and engage us in a collective conversation about equity in the school.

Lastly, in terms of theater, for many years I facilitated a club – Documentary Theater Ensemble – that combined storytelling, improvisation, and movement. This club, since it did not require auditions, brought in different groups of students from the school with quite a range of talent and experience. At its height it drew 30 students a week and offered a safe setting chance for students to reflect meaningfully on their past experiences. The club later turned into the Young Visionaries whose mission was to effect positive change in the school. That group, using Open Space Technology, helped lead an Institute Day organized by Amy Hill that sought to engage faculty and students in community wide conversation about issues of concern and opportunities for change in the school.

<sup>&</sup>lt;sup>2</sup> Daniel Goleman's Social Intelligence is a particularly good resource

<sup>&</sup>lt;sup>3</sup> A quick summary: the play composed entirely of material derived from interviews explored the causes and aftereffects of the 2001 Crown Heights Race Riots.

I have consistently sought to champion student voices and deepen their sense of voice and access at the school. For example, through a class assembly I helped lead with Jason Dennis last year (he produced, I directed) a group of students assembled every day at lunch for four weeks discussing their experiences of high school, and generating stories from their life that would connect with the student body. Together we crafted, edited, and rehearsed their pieces. We then inter-spliced and staged them. The students memorized the pieces before eventually performing them for their peers. Most importantly they led the discussion afterwards. This approach marked a shift from the way class assemblies had been run. The students courageously spoke about taboo subjects – race and class, sexual orientation, bullying, and teacher student conflict. This year, Cindy Milojevic and others continued with a similar format. We will meet in January to discuss how to continue the format and institutionalize our efforts.

In addition, with the help of a grant from the school and coordination of Steve Gevinson and Don Vogel, I visited New Trier several time over a year in order to help implement the Student Instructional Leadership Corps (SILC) program and to help bring portfolio design and assessment to our English department. The SILC program has been highly successful at our school, offering more opportunities for one-on-one tutoring and mentoring, while forging greater connections between older and younger students. The portfolio design has offered a way to measure student progress in writing as well as promote uniform assessment across the department. It has been an effective way of meeting Response to Intervention (RTI) standards for ongoing assessment that measures change and integrates students into the feedback loop.

My focus, since I arrived at the school, has been to foster community and create equity by making connections with students, colleagues, parents, and administrators. I brought together parents (including Francis Kraft, whose mentoring program has been implemented), teachers, administrators, school counselors, and principals to talk about how to integrate the work of SEL into the core curriculum. On many occasions I have created opportunities in my classes for students to present in authentic rhetorical situations such as the *Odyssey Project* and the *Parent Project* (in both projects students created monologues based on interviews that they performed in front of their interviewees). They have been successful examples of how SEL and skill development can happen simultaneously.

I have also been a part of an ongoing effort to bring training to students on issues around mental health in conjunction with the non-profit Erasing the Distance. We need only recall recent local and national events to know that mental wellness in the school is vital. I have presented on that very topic — and how to integrate SEL into curriculum — at the Progressive Education Network conference, at the IFS conference in Lake Bluff, at the CASE conference, at the Theater of Oppressed conference in Omaha, and with James Bell at NCTE, in New York.

Finally, I have served on several summer curriculum projects – vocabulary, SEL, SILC, and most recently the RTI planning committee, in which Ginger Brent and I worked on speaking and listening goals. After considering each state goal and writing our own, we realized that we must teach listening as a discernible skill, separate from reading and writing. Most recently, Mike Soffer and I gave a

presentation to the faculty on experimenting with RTI evaluative techniques in the classroom.

As a teacher at OPRF, I have tried to advocate for and create spaces, opportunities, and curricula that make students' experiences at the school positive and enriching. This sabbatical will increase my capacity to meet the challenge of understanding how to strengthen the connection between student wellness and academic outcomes to create positive change in the school. That work has often been complex, and has never been easy, but I think it has benefited the school, its students, and without question, me, as a teacher and human being.

6. Will you receive any type of outside grant or remuneration as a result of your sabbatical activities? If so, please indicate the amount.

No remuneration exists for my planned sabbatical activities.

#### Works Cited

CASEL | Welcome to CASEL. Web. 15 Jan. 2011. < http://www.casel.org/sel/impacts.php>.

Goleman, Daniel. Social Intelligence: the New Science of Human Relationships. New York: Bantam, 2006.

## Oak Park and River Forest High School District 200

201 North Scoville Avenue • Oak Park, IL 60302-2296

DATE: February 24, 2011

TO: Board of Education

FROM: Dr. Steven T. Isoye

Subj: Approval of the Des Plaines Valley Education for Employment Regional Delivery System (DVR) Intergovernmental Agreement

ACTION

Attached is a draft copy of the Des Plaines Valley Education for Employment Regional Delivery System (DVR) Intergovernmental Agreement. Below are the recommended changes.

Page 2 – Section 1.3 – Membership Triton College added

Page 2 – Section 1.3.2 – Accepting

New language added to 2<sup>nd</sup> paragraph:

"...or to alternative ongoing contributions in lieu of assessment and program development expenses identified by unanimous Board agreement at the time of admittance."

Page 2 - Section 2.1.1 - Board Membership

New language added to 1<sup>st</sup> paragraph:

"...and the Triton College District 504 President."

Deleted language: "...One Triton College, (District #504) administrator and the superintendent of the Cook County Education Services Region will be ex-officio non-voting member of the BOARD."

#### Page 3 - Section 2.1.3 - Officers

New language: "The Board establishes and annually reviews the cycle of appointment of Board officers." Deleted language: "The BOARD will elect its own chair, vice chairperson, and secretary annually. An election will take place in May of each school year."

<u>Page 3 – Section 2.1.4 – Terms of Office</u> Deleted language: "Elections will be held at the regular May-June annual meeting each year."

#### Page 6 - Section 3.1.5 - Evaluation

New language: "Members agree to make data available to DVR for the purpose of program development, evaluation and improvement. Data will include but not be limited to information on student enrollments in courses as required for State reporting purposes, aggregate numbers of participation in approved programs, demographic information, and in the case of the college will include but not be limited to, numbers captured for the A1 and/or Pod reports that may contribute to baseline benchmarks and evaluation of program participation over time."

Page 8 – Section 4.1 – Program Approval

New language: "The Board reserves the right to approve programs independently of Administrative Council recommendations."

TEL: (708) 383-0700 WEB: www.oprfhs.org TTY/TDD: (708) 524-5500 FAX: (708) 434-3910

## Oak Park and River Forest High School District 200

201 North Scoville Avenue • Oak Park, IL 60302-2296

#### Page 8 - Section 4.3 - Independent Programs

Changed language that programs "will" be eligible for financial support, to language that programs "may or may not" be eligible for financial support.

#### Page 9 – Section 5.3 – Tuition Costs

Deleted language that tuition costs "will be computed by the actual costs of all DVR SYSTEM courses divided by the number of students participating regionally in such courses."

I recommend that the Board of Education approve this agreement.

Motion:

Move to approve the DVR Intergovernmental Operating Agreement, as presented.

Roll Call Vote

Agenda Item No. XII. A.

#### Des Plaines Valley Education for Employment Regional Delivery System

#### INTERGOVERNMENTAL AGREEMENT

This Intergovernmental Agreement is executed pursuant to the Intergovernmental Cooperation Clause of the Illinois Constitution. (Ill. Const. 1970, Art. VII, Sect. 10), as well as the provisions of the Intergovernmental Cooperation Act of 1973. (Ill. Rev. Stat. 1985, Ch. 127, Par. 741, et. seq.).

#### ARTICLE I - GENERAL

#### Section 1.1 Name of Agreement

<u>Name</u> The name of the regional delivery system formed hereunder shall be the Des Plaines Valley Education for Employment Regional Delivery System hereinafter referred to in this document as the DVR SYSTEM.

#### Section 1.2 Purpose

Purpose The purpose of the DVR SYSTEM formed under this agreement is:

- To provide a framework for offering quality Education for Employment opportunities (programs) for all students who reside within the region or are approved by the DVR SYSTEM Governing Board to ensure that students attain a satisfactory level of achievement appropriate for moving directly into employment or into advanced education or training to prepare for later employment;

- To provide for the administration of programs and services conducted under this Agreement in accordance with all provisions specified in the by-laws, policies and procedures of the \*SYSTEM Governing Board and applicable provisions of the Intergovernmental Cooperation Act of 1973;

- To provide a structure for supplying the resources needed to operate such programs and services with maximum efficiency;

- To participate in cooperative efforts with appropriate agencies as needed in order to plan, implement, evaluate, and improve programs that serve adults and youths as well as the area's economic needs;

- To provide a systematic program of curriculum renewal, staff development, and equipment/facilities modernization to ensure student access to high quality programs;

2/17/04 FY11.dee 1InterGovAgreement FY11InterGovAgreement

#### Section 1.3 Membership

Membership

1.3.1 The membership of this system includes:

Oak Park/River Forest High School-District #200Riverside-Brookfield Township High School-District #208Proviso Township High Schools-District #209Leyden Community High Schools-District #212Ridgewood Community High School-District #234Elmwood Park Community High School-District #401Triton College-District #504

Accepting

1.3.2 Nonmember districts who wish to join may be admitted only by unanimous vote of the Governing Board and by meeting the following conditions:

- They agree to an initial assessment established by the Governing Board; including, but not necessarily limited to, a fair and equitable portion of previous program development expenses; they satisfy the same conditions which were met by present member districts as established by the Governing Board, or to alternative ongoing contributions in lieu of assessment and program development expenses identified by unanimous Board agreement at the time of admittance.
- Membership in the DVR SYSTEM as defined by this Agreement, shall be continuous from year to year. Withdrawal must be executed according to provisions in Article VII.

ARTICLE II - ORGANIZATION AND OPERATION

#### Section 2.1 Governance

Board Membership 2.1.1 The Governing Board, hereinafter referred to as the BOARD shall be composed of the superintendent of each member district and the Triton College District 504 President. One Triton College, (District #504), administrator and the superintendent of the Cook County Education Services Region will be ex-officio nonvoting member of the BOARD.

> In July of each school year, each member District Superintendent shall appoint one individual to represent the District, at scheduled DVR/Partnership Board meetings, when an emergency precludes the Superintendent from attending the scheduled meeting. Only one delegated representative will be allowed to make up the quorum/vote at any one meeting. A Superintendent planning to send a delegate to a DVR/Partnership Board meeting will notify the other members of the Board prior to the meeting.

> Other rights accorded to the non-voting ex-officio representatives to the BOARD will be at the discretion of the voting members, with a quorum required to approve

2/17/04 <del>FY11.doc</del>

Inter@ovAgreement FY11InterGovAgreement

Comment [DY3]: New language
Formatted: Strikethrough

Comment [DY1]: New language

Comment [DY2]: New language

## INTERGOVERNMENTAL AGREEMENT DVR 030 such rights.

Powers

Officers

legal powers to act in behalf of the DVR SYSTEM, with Board members and their appointed delegates, indemnified via resolution signed by the presidents of their respective school boards.

2.1.2 The BOARD will serve as the policy-making body with full

cers 2.1.3 The Board establishes and annually reviews the cycle of appointment of Board officers. The BOARD will elect its own chair, vice chairperson, and secretary annually. An election will take place in May of each school year.

Terms of Office 2.1.4 Elections will be held at the regular May June annual meeting each year. The slate of officers will serve a term commencing with July 1 and ending June 30 of the following year.

Resignation 2.1.5 If an officer resigns during his/her term, the BOARD shall replace that officer at the next meeting of the BOARD.

Meetings 2.1.6 The BOARD will conduct its meetings in accordance with the accompanying by-laws, Roberts' Rules of Order, and in accordance with the requirements of the Illinois Open Meetings Act. (Ill. Rev. Stat. 1985, Ch. 102, Par. 42, et seq.).

<u>Voting Procedures</u> 2.1.7 Voting procedures will follow Roberts Rules of Order. The Board honors a one vote per district policy.

Quorum 2.1.8 A quorum will consist of four member districts voting. Any official action or decision of the BOARD will require a quorum. No proxies will be allowed.

Board Duties 2.1.9 Duties will be:

- to periodically evaluate the performance of the Administrative Agent;
- to submit recommendations to the Administrative Agent for the employment of personnel;
- to provide direction and assistance to the DVR SYSTEM staff pursuant to BOARD policies;
- to make recommendations to the Administrative Agent regarding budget, personnel policies and decisions, and other such matters as the BOARD may delegate;
- to approve the annual Regional Education for Employment

InterGovAgreement FY11InterGovAgreement

2/17/04

Comment [DY4]: New language Formatted: Strikethrough

Formatted: Strikethrough

**Comment [DY5]:** Note: Current policy statement (Policy 8300) is included at the the end of this document.

plan for the DVR SYSTEM;

- to adopt all necessary rules for the management and governance of the DVR SYSTEM including by-laws, personnel policies, and an annual calendar of meetings;
- to indemnify, insure, and protect the DVR SYSTEM, Board members, Administrative Agent, the DVR SYSTEM employees, and authorized DVR SYSTEM volunteers against civil and constitutional rights, damage claims and suits, and bodily injury and property damage claims and suits; and workers compensation and employers liability insurance;
- to submit information and reports as deemed necessary by the Illinois State Board of Education;
- to approve all programs and services offered by the DVR SYSTEM;
- to approve all facilities utilized by the DVR SYSTEM; and
- to authorize a schedule for regular evaluation of programs, procedures, and costs.

#### Section 2.2 Administrative Agent

Administrative<br/>Agent2.2.1 The BOARD will designate an Administrative Agent subject<br/>to that designee's approval. The Administrative Agent<br/>will act as fiscal and legal agent for the DVR SYSTEM.

- <u>Changes</u> 2.2.3 Any change in the designation of the Administrative Agent will require BOARD action as outlined in Article II, Section 2.2.1, and will not be considered as an amendment to this Intergovernmental Agreement.
- Responsibilities 2.2.4 The Administrative Agent will:
  - file the annual budget as authorized by the BOARD;
  - transmit reports and implement BOARD authorized procedures;
  - receive and disburse all income credited to the DVR SYSTEM;
  - and perform other services as delegated by the BOARD. In the event the Administrative Agent makes an expenditure at the direction of the BOARD or a member district, which is later determined to be illegal or unauthorized, the BOARD or member district, as the case may be (and not the Administrative Agent) shall be responsible for any required fund reimbursement. (Article II, Section 2.2.4 revised 2/26/87).

#### Section 2.3 Administrative Structure

2/17/04 FY11.doe Inter&ovAgreement FY11InterGovAgreement

		AGREEMENT DVR 030
<u>Administrative</u> <u>Council</u>	2.3.1	The Administrative Council will be comprised of one or more, but not more than two, staff members from each member district, as designated by the superintendent of the respective district. Each district will be allowed only one vote on the Administrative Council.
Responsibilities	2.3.2	The Administrative Council responsibilities will include:
	-	Organizing itself to carry out its responsibilities.
	-	Developing DVR SYSTEM procedures for BOARD approval.
	-	Preparing recommendations for the BOARD concerning all matters related to planning, delivery and evaluation of DVR SYSTEM programs.
	-	Performing all other duties assigned by the BOARD.
<u>Administrative</u> <u>Staff</u>	2.3.3	All decisions regarding employment, evaluation, and/or termination of staff must be authorized by the BOARD. DVR SYSTEM employment policies, procedures, and practices shall be in accordance with state and federal law.
Employee Benefits	2.3.4	The BOARD will provide for employee insurance coverage and other benefits in the DVR SYSTEM annual budget.
	<u>Secti</u>	on 2.4 Duties of Administrative Personnel
<u>Responsibilities</u>		Duties of DVR SYSTEM personnel will be assigned by the BOARD in accordance with DVR SYSTEM policy and procedures.
	<u>Secti</u>	on 2.5 Advisory Council
Advisory Council		A general ADVISORY COUNCIL will be appointed to advise the BOARD and program staff on effective planning for and evaluation of programs. Advisory Council Members will be appointed by the BOARD in accordance with DVR SYSTEM policy and procedures.
Advisory Committe	<u>e</u>	OCCUPATIONAL PROGRAM ADVISORY COMMITTEES will be appointed on an ad hoc basis with the length of term

#### Section 2.6 Staffing

All staffing decisions will be determined by the BOARD Determination and any staff hired by the DVR SYSTEM will be approved by the BOARD.

#### ARTICLE III - PROGRAM

#### Section 3.1 Programs and Services

determined by need.

2/17/04 FY11.doc InterSovAgreement FY11InterGovAgreement

- Procedure3.1.1 DVR SYSTEM programs and services will be authorized by<br/>the BOARD with input from the appropriate internal and<br/>external advisory bodies as described in DVR SYSTEM<br/>purpose, policies, and procedures and in accordance with<br/>state and federal laws. DVR SYSTEM instructional<br/>programs will be delivered to students via regional<br/>programs which will be operated by local districts<br/>and/or other contracted agencies.
- <u>Selection</u> 3.1.2 Selection of instructional programs will be based upon sound planning procedures, as specified in DVR SYSTEM policies and procedures, as authorized by the BOARD and in accordance with state and federal laws.

<u>Features</u> 3.1.3 Instructional programs delivered through the DVR SYSTEM will manifest the following quality features:

- An outcome oriented curriculum, based upon clearly stated learning objectives derived from worker task lists as validated by the local program advisory committee;
- A clearly defined sequence of activity options that lead to accomplishment of the program learning objectives;
- Clear learning objectives to be accomplished within each course and a description of how the course relates to the sequence of the program (or programs) to which the course is related.
- Approval
   3.1.4 All DVR SYSTEM programs will be reviewed by the DVR

   SYSTEM Administrative Council and presented to the EOARD

   for approval. Each district will determine its level of

   participation in each approved program.

Each district will devise an internal procedure for its approval of the DVR SYSTEM curriculum.

**Evaluation** 3.1.5 Members agree to make data available to DVR for the purpose of program development, evaluation and improvement. Data will include but not be limited to information on student enrollments in courses as required for State reporting purposes, aggregate numbers of participation in approved programs, demographic information, and in the case of the college will include but not be limited to, numbers captured for the Al and/or Pod reports that may contribute to baseline benchmarks and evaluation of program participation over time:

#### Section 3.2 Eligibility of Students for Programs

<u>Student</u> Eligibility 3.2.1 Each district shall determine the eligibility of its own resident students for programs provided, coordinated or approved by the BOARD as well as for independent or nonapproved programs.

InterGovAgreement FY11InterGovAgreement

2/17/04 FY11.doe Comment [DY6]: New language

Level of Participation

- 3.2.2 In order to provide effective administration for staff employment and budget matters, each member district will present an annual report on or before March 1 which provides minimum levels of participation within each shared program for the following year.
- private and<br/>Parochial3.2.3 To be eligible for participation in a DVR SYSTEM<br/>program, a private or parochial school student mustSchool Studentsfirst document legal residence within one of the member<br/>districts. To be enrolled, an eligible private or<br/>parochial school student must register for the program<br/>as a part-time student at the public school district<br/>within which he/she resides. The public school district<br/>then becomes responsible for tuition and other approved<br/>costs resulting from the private school student's<br/>participation.
  - 3.2.4 Under these circumstances the private or parochial school assumes no responsibility for any portion of administrative or other system costs.

#### Section 3.3 Program Management

- <u>Program Management</u> 3.3.1 The BOARD will delegate program management to authorized providers of programs, whether school districts and/or other agencies.
- <u>Responsibilities</u> 3.3.2 The responsibilities of providers by formal agreement shall include:
  - Employment, supervision, and evaluation of qualified staff;
  - Delivery of approved curriculum and instruction;
  - Providing suitable and adequate facilities, equipment, and supplies;
  - Periodically reviewing the effectiveness of each program;
  - Insurance coverage for above mentioned facilities and equipment.
  - 3.3.3 The authorized program provider will comply with Title VI of the Civil Rights Act of 1964; Section 504 of the Rehabilitation Act of 1973; Title IX of the Educational Amendments of 1972; Section 10-21.1, 10-22.7 of The School Code of Illinois; Article I, Section 18 of the Illinois Constitution of 1970; and all subsequent amendments to these laws, and federal and state rules and regulations which prohibit discrimination on the basis of race, color, national origin, sex, or handicap.

#### ARTICLE IV - HOUSING

2/17/04 <del>FY-11.doc</del>

InterGiovAgreement FY11InterGovAgreement

#### INTERGOVERNMENTAL AGREEMENT DVR 030 Section 4.1 Approved Local and Shared Programs/Courses

Program ApprovalAll DVR SYSTEM programs will be recommended by the<br/>Administrative Council for BOARD approval. The Board<br/>reserves the right to approve programs independently of<br/>Administrative Council recommendations. Each district<br/>will determine its level of student participation in<br/>each approved program. Courses approved to serve<br/>students from two or more member districts shall be<br/>referred to as shared courses. Courses approved to serve<br/>students from only one member district shall be referred<br/>to as a local course.

#### Section 4.2 Program Sites

- Determination 4.2.1 The BOARD will determine sites for the DVR SYSTEM programs keeping maximum accessibility and efficiency as major determinants in the selection process.
- <u>Utilization</u> 4.2.2 Member districts may elect to utilize applicable provisions of The School Code including, but not limited to, Section 10-22.31a, 17-2.4 and 19-30 acquire, establish, to remodel and/or maintain equipment and buildings for DVR SYSTEM programs.
- Recommendation 4.2.3 Recommendations concerning the location of DVR SYSTEM courses which are not to be offered in each member school district will be made to the BOARD by the Administrative Council. Such recommendations will be based on criteria previously agreed on by the BOARD, and must receive BOARD approval before course locations are established.
- Termination 4.2.4 In the event of termination of this agreement or of withdrawal of one or more of its members, any sites, facilities or equipment purchased by a member district will remain the property of that district. Any DVR SYSTEM sites, facilities or equipment acquired through the Administrative Agent for the DVR SYSTEM or any assets derived there from will be distributed proportionately among member districts as recommended by the BOARD and approved by joint action of member BOARDS.

#### Section 4.3 Independent Programs

Independent Programs Nothing in this agreement shall prohibit or restrict the right of a member school district or combination of districts to provide any vocational education course or program independently without approval or coordination by the BOARD. Such programs <u>shall</u> may or may not be eligible for State Board vocational education financial support.

Comment [DY8]: New language	
Formatted: Strikethrough	S.I.

#### Section 4.4 Official Office

Official Office

Inter&ovAgreement FY11InterGovAgreement

The DVR SYSTEM official office will be determined by the

2/17/04 FY11.doc Formatted: Strikethrough

#### Comment [DY7]: New language

BOARD, in accordance with DVR SYSTEM policy, and state and federal law. Change of the official office will not constitute an amendment to this agreement.

#### ARTICLE V - FINANCE

#### Section 5.1 Administrative Costs Assessment

- Administrative
   5.1.1 Administrative costs will include the salary and

   Cost
   benefits of any SYSTEM central staff, supplies,

   printing, and other costs associated with operation of

   the DVR SYSTEM administrative function which have been

   authorized and budgeted by the BOARD.
- Funding 5.1.2 Administrative cost will be funded by the contribution of member districts in the same percentage as the current funding policy.

#### Section 5.2 Equipment Costs Assessment

Equipment Costs The BOARD may assess an instructional equipment cost that prorated to all member districts as a percentage of the current funding policy. Ownership of equipment purchased under this section will remain the property of the DVR SYSTEM and an inventory will be maintained.

#### Section 5.3 Tuition Costs

Tuition CostsAll student tuition costs will be considered as non-<br/>administrative costs and non-equipment outlay. The Board<br/>will determine allocation of tuition costs for<br/>individual programs and will be computed by the actual<br/>costs of all DVR SYSTEM courses divided by the number of<br/>students participating regionally in such courses.

#### - { Comment [DY9]: New Janguage Formatted: Strikethrough

#### Section 5.4 Method of Payments

- Procedures5.4.1 Each member district will pay such sums of money asFor Paymentare due under the terms of this agreement based upon the<br/>operating budget adopted by the BOARD. The billing of<br/>each fiscal year is payable in August.
- Shortfall/5.4.2 In the event of a budget shortfall or deficit, a year-<br/>end assessment shall be made to balance the budget. Such<br/>assessment will be made according to the financial<br/>procedures for that portion of the budget that generated

Inter@ovAgreement FY11InterGovAgreement

2/17/04 FY11.doe

the shortfall. If this cannot be determined, then a prorated assessment will be based upon the percentage of the current funding policy.

#### Section 5.5 Charges to Non-members

Establishing Charges

Charges made to non-member school districts, individuals and other governmental bodies for educational and training services provided by the DVR SYSTEM will be established by the BOARD, and in no case will such charges be less than charges made to member districts.

#### Section 5.6 Distribution of Income

Income

Reimbursement received by the DVR SYSTEM will be distributed to member districts as follows:

- All State Board of Education reimbursement earmarked for equipment shall be distributed to the member districts in accordance with Article IV, Sec. 2.
- All State Board of Education reimbursement designated specifically for DVR SYSTEM administrative costs will be dispersed in accordance with Article V, Sec. 1.
- All State Board of Education program reimbursements will be distributed to member districts according to the costs incurred by each district in accordance with Article V, Sec. 2 and Sec. 3.

All monies received by the DVR SYSTEM will be distributed to the individual schools based on the current funding procedure. Any changes to the current funding procedure will be determined and approved by the BOARD.

#### Section 5.7 Accounting Procedures

Financial	5.7.1	All member boards shall fully assume their respective
<u>Obligation</u>		financial obligations, without the imposition of financial responsibility of any other district. Each member district shall promptly do all things necessary to legally commit the district to the timely payment of its cost with respect to any other legal financial obligation.
<u>Contributions</u>	5.7.2	Contribution from private businesses, governmental and foundation sources for the benefit of the DVR SYSTEM shall be made to the Administrative Agent for the use of the contribution subject to approval of the BOARD.
Accounting Procedures	5.7.3	Accounting procedures shall conform to all applicable rules and regulations of the Illinois State Board of Education.
Reporting Expenses	5,7,4	The Administrative Agent shall maintain accounts of the DVR SYSTEM'S operational expenses and shall make these
2/17/04		Inter100vAgreement FY11InterGovAgreement

2/17/04 FY11.doe

available to participating districts on a monthly basis.

Annual Audit 5.7.5 A minimum of one audit per year shall be conducted in accordance with Section 3-15.1 of the School Code of Illinois, as amended.

#### ARTICLE VI - TRANSPORTION

#### Section 6.1 Responsibility

Responsibility The BOARD shall specify in DVR SYSTEM Policy and Procedures the responsibilities for transporting students in accordance with Article 29, Section 5 of the Illinois School Code.

#### Section 6.2 Coordination

Coordination DVR SYSTEM Policy and Procedure will delineate the manner in which coordinating of transportation, if needed, will occur.

#### ARTICLE VII - TERMINATION, WITHDRAWAL, AND REMOVAL

#### Section 7.1 Termination

#### Termination

This agreement may be terminated in the event that all the member districts so agree. The vote to terminate must occur twelve months prior to the July 1 termination date and notice shall be given to the Illinois State Board of Education. DVR SYSTEM assets shall be distributed in accordance with the interests vested in each asset. All the member Boards of Education must vote to terminate this agreement.

#### Section 7.2 Withdrawal

Withdrawal

Member districts may withdraw from participation in the DVR SYSTEM provided they give written notice 12 months preceding the beginning of the fiscal year (July) in which they plan to withdraw. Notification shall be provided to the State Board of Education and each member district upon notice of withdrawal of a member district.

If a member district gives written notice of withdrawal, that district is to continue participation and financial obligation until withdrawal is effective.

#### Section 7.3 Removal

Removal

A district may be removed from membership by two-thirds (2/3) vote of the BOARD. Notification of a proposed

2/17/04 FY1Ldoe InterflovAgreement FY11InterGovAgreement

removal shall be given to the member district and the Illinois State Board of Education at least 90 days prior to the end of the fiscal year. A hearing shall be held within 60 days of the notification to remove a member district. The Illinois State Board of Education shall be notified of the call for a hearing and any BOARD action to remove, after the hearing. A removed district shall be responsible for all financial obligations incurred while a member.

#### ARTICLE VIII - AMENDMENTS

#### Section 8.1 Amendment Procedures

Amendment Procedure The conditions of this Agreement may be amended by the affirmative vote of the Board of Education of all districts that are parties to this Intergovernmental Agreement, as set forth below.

Proposed amendments must be submitted in writing to the BOARD. All amendments must come before the BOARD to be approved. If approved by two-thirds (2/3) majority of the voting members of the BOARD, the proposed amendment will be submitted, along with a resolution, to each voting member Board of Education.

Member Boards of Education shall act on a proposed amendment within 60 days. Ratification of the amendment will be deemed to take place when two-thirds (2/3) of the districts have voted passage. Failure of a district to act within 60 days shall be deemed to be a vote against the amendment. The amendment will take effect upon ratification unless the amendment provides otherwise.

#### Section 8.2 Ratification

Ratification Ratification will take place by resolution of the individual participating Boards of Education. Each district whose Board of Education adopts said resolution shall become a participating member of the DVR SYSTEM effective July 1, 2004, or upon the date of direct approval, whichever occurs last.

Original Document Approved June 11, 1986 Section 2.2.4 Revised March 26, 1987 Section 2.2.2 Deleted February, 2004 Entire document reviewed/revised February 2004, effective July 1, 2004

2/17/04 FY11.doc Interf2ovAgreement FY114nterGovAgreement

The regular Board meeting of the Board of Education of the Oak Park and River Forest High School was held on Thursday evening, January 27, 2011, in the Board Room.

Call to Order	Vice President Allen called the meeting to order at 7:04 p.m. The following Board
	of Education members were present: John C. Allen, IV, Jacques A. Conway, Terry
	Finnegan, Dr. Ralph H. Lee, Dr. Dietra D. Millard (attended electronically until
	8:00 p.m.), Amy McCormack, and Sharon Patchak-Layman. Also present were:
	Dr. Steven T. Isoye, Superintendent; Michael Carioscio, Chief Technology Officer;
	Amy Hill, Director of Assessment and Research; Philip M. Prale, Assistant
	Superintendent for Curriculum & Instruction; Nathaniel L. Rouse, Principal;
	Lauren M. Smith, Director of Human Resources; Cheryl L. Witham, Chief
	Financial Officer and Treasurer; James Paul Hunter, FSEC Executive Committee
	Chair; Katrina Vogel, Student Council Liaison; and Gail Kalmerton, Executive
	Assistant/Clerk of the Board.

VisitorsThe Board of Education welcomed the following visitors: Kay Foran,<br/>Communications and Community Relations Coordinator; Dr. Tina Halliman,<br/>Director of Special Education; David Zimmerman of the Trapeze, Peggy Swanson<br/>of the League of Voters, Liz Robinson, Co-chair of Citizens' Council; John Bokum,<br/>John Phelan, and Maggie Skiver, Board of Education candidates; Terry Dean of the<br/>Wednesday Journal.

Changes to theChanges included: 1) public comments were moved head of the Board of Education'sAgendacomments, and 2) Approval of Policy 4015 for first reading was removed from the<br/>agenda.

**Status of FOIA** Mr. Allen reported that one FOIA request had been received and resolved.

Requests

Visitor Comments No comments.

**Board of Education** Ms. Patchak-Layman spoke about three issues:

 Communication to parents about changes made to a student's schedule mid-year, including non-core classes. She had received feedback from parents about changes made without notification to them or to the students.

2) Do course recommendations reflect the Board of Education's goal of allowing all students to work at their highest and best? Colleges no longer look as intently at a student's scores but instead at whether the student is a "good fit" for the school. OPRFHS' selection process does not include direct contact with students prior to making the recommendation. While there is an over-ride system in place, the placement is made before the school knows who the student really is.

3) More student involvement and participation in civics was important. None of the issues that come before the Board of Education is discussed with the Student Council Board Liaison Representative. Because the goals are to peak student curiosity and have students intellectually explore

	things, i.e., through groups or teams, adults should model what is already being done for students.
	Mr. Allen attended a meeting with the parents of the girls on "the list," representatives of the high school, and the Oak Park and River Forest Police Departments. He was struck by the parents' civility and the support that the students gave to each other and using this event as a way to grow.
Student Council	<ol> <li>Student Council representative Katrina Vogel reported:</li> <li>In response to the comments made by the administration about "the list," on January 28, clubs, sports, faculty, and staff would be wearing T-shirts printed with the word RESPECT on them. In addition, the Star Club would write anonymous notes of encouragement to post on a bulletin board.</li> <li>Student Council is seeking one representative from each club in order to be as diverse as possible.</li> <li>Student Council will also have conversations about current issues, e.g., closed campus, etc.</li> <li>The theme for the King of hearts Dane is Superheroes. Students will help donate to the Heart Association Found by purchasing the delivery of "crushed cans" to someone.</li> </ol>
Principal Report	<ul> <li>Principal Rouse reported that second semester started relatively smoothly and he congratulated the faculty on successfully closing first semester. Grades are accessible online and grade reports will be mailed home on January 28, 2011.</li> <li>Regarding eighth-grade transition, Principal Rouse stated that residency verification for eighth-grade families was due the following week and that the Class of 2015 Open House was scheduled for Wednesday, February 2, from 6-8 pm in the Fieldhouse. He thanked Cindy Milojevic for organizing this event. Incoming students will have the opportunity to take a self-guided tour and all academic departments will be represented.</li> <li>He also congratulations the OPRFHS Huskie Speech Team for its performance the week of January 15 at the Rolling Meadows High School Invitational Speech</li> </ul>
District Reports	<ul> <li>Tournament for taking 6<sup>th</sup> place out of 26 schools. It will also participate on Saturday, January 29, at Warren Township High School in Gurnee, Ill.</li> <li>In an effort to establish more positive meaningful relationships with students, Mr. Rouse will begin hosting "Pizza with the Principal" once a month during lunch periods so that students have the opportunity to discuss topics relative to learning environment issues.</li> <li>Citizens' Council— Liz Robinson, co-chair of Citizens' Council reported the following: <ol> <li>Ron Orzel will be the Council's new chair.</li> <li>The PAC subcommittee was composed of a broad group of people.</li> <li>The proposed energy policy was presented;</li> <li>The communication strategy for the school and the vehicles that are available; and</li> </ol> </li> <li>BOOSTER—Ms. McCormack reported 1) 650 Student Directories sill available for</li> </ul>
	<u>BOOSTER</u> —Ms. McConnack reported 1) 650 Student Directories sin available for purchase, planning for its annual dinner dance at Concordia University was underway

2

and 3) Cathy Yen was the new performing arts chairperson who was putting together a committee of parents from various groups.

<u>Alumni Association</u>—The Alumni Association has two new board members, Scott Dunnell (a 1971 Graduate) and Claire Walter (a 2003 grade and faculty member). Its board members are to take part in the conversations about closed campus. A lengthy discussion about summer grants occurred and the Alumni Association was hopeful that enough funds to provide at least of portion of summer programs for approximately fifty students and Alumni Association membership renewal.

<u>Concert Tour Association</u>—Ms. Patchak-Layman reported that the CTA spoke about procedures for granting special monetary requests for programs for students as its treasury has additional money because it did not need to put deposit downs for the Orchestra and Marching Band trips. CTA is also beginning a strategic planning process for the organization to develop direction and ideas for the coming years.

<u>Faculty Senate</u>—Mr. Hunter appreciated Mr. Rouse's comments as the faculty was working hard to close the semester and he thanked Mr. Allen for his comments.

Dr. Isoye made the following comments.

He too acknowledged acknowledge that our community was hit hard with the coverage of the unfortunate incident targeted at the District's female students. The community is working to move forward and work towards a better way of having the students see the ramifications of this behavior. He thanked the faculty and staff, students, and parents for their as the administration moves forward.

Dr. Isoye attended the Oak Park Council of Governments meeting.

Dr. Isoye attended a West 40 meeting for area superintendents where he learned about the most recent pieces of legislation passed by the state legislators from the executive director of the Illinois Association of School Administrators.

At a recent Rotary meeting, Dr. Isoye sat with Jan Pate of the YMCA and he met her special guests from Kosovo who were leaders studying Oak Park's local governance model through the YMCA and many other multiple agencies.

Dr. Isoye attended the Illinois State Board of Education Performance Evaluation Advisory Committee as an appointed member. This committee is working to advise the state on ways to support the new legislation in regards to evaluation of teachers and principals and a state model to assist boards when they begin to discuss adopting these evaluation systems.

The OPRFHS Administration continued its ongoing discussions with the Village of Oak Park about emergency plans.

Dr. Isoye attended the High School/College Partnership with area high schools that have students who matriculate to Triton. Triton also attended. Discussion included the various programs supported by this partnership and the work of the Des Plaines Valley Region, our partners in education, employment, and lifelong learning.

Superintendent Report On January 29, Dr. Isoye will attend a law conference hosted by Franczek, Radelet. On the same day, many of the other administrators will attend the Northwest Personnel Administrators Association teacher recruitment fair at the Rosemont Conference Center.

Dr. Isoye attended a special meeting at the Boeing Company as it collaborates with the US Naval Academy on its outreach efforts to attract minority youth from the Chicago Metropolitan Area. Opportunities available to students through the Naval Academy are numerous and include a tuition-free education at the fifth best undergraduate engineering school in the country. One of the greatest challenges facing the Academy is the ability to identify, attract, mentor, and retain qualified minority candidates. He will participate in the conversation as it looks to broaden their scope.

The Board of Education's Committee meetings will be Monday and Tuesday, February 14 and 15, as Dr. Isoye will be attending the AASA Conference later that week.

**Consent Items** 

Mr. Finnegan moved to approve the consent items as follows:

- the Check Disbursements and Financial Resolutions dated January 27, 2011;
- the Treasurer's Report for December 2010;
- the Monthly Financials for December 2010;
- Authorization to Prepare Amended FY 2011 Budget;
- Authorization to Prepare FY 2012 Tentative Budget;
- the CDARS Deposit Placement Agreement;
- the Comcast Contract for increased bandwidth;
- the Food Service Bid for Commodities; and
- the agreement with Konica;

seconded by Ms. McCormack. A roll call vote resulted in five ayes. Motion carried. Dr. Millard had departed.

Ms. Patchak-Layman made a request to receive copies of legal bills received by the District.

Policy 4113Mr. Allen moved to amend Policy 4113, Certified Personnel-Certification, as<br/>presented; seconded by Dr. Lee. A roll call vote resulted in five ayes. Motion<br/>carried. Ms. McCormack was absent from this vote and Dr. Millard had<br/>departed.

- Policy 4122Mr. Conway moved to approve Policy 4122, Substitute Teachers, for First<br/>Reading, as presented; seconded by Dr. Lee. A roll call vote resulted in five<br/>ayes. Motion carried. Ms. McCormack was absent from this vote and Dr.<br/>Millard had departed.
- **Gifts & Donations** Mr. Finnegan moved to accept with gratitude the gifts and donations as presented (attached to and made a part of the minutes of this meeting); seconded by Dr. Lee. A roll call vote resulted in six ayes. Motion carried. Dr. Millard had departed.

#### Architectural Services

Ms. McCormack moved to approve the contract with Legat Architect for architectural services; seconded by Mr. Finnegan. A roll call vote resulted in three ayes and three nayes. Motion failed. Ms. Patchak-Layman, Mr. Allen, and Dr. Lee voted nay.

Last month the District approved Legat Architect as the architect of record. Ken Florey of Robbins Schwartz Nicholas Lifton and Taylor, Ltd. is the legal counsel who is negotiating a contract, which will define the duties and responsibilities between the architect and a future construction manager, as well as strengthen the District's position. Approval would have allowed the negotiating process to continue in order to prepare for summer 2011 projects. The attorney made the recommended amendments.

Several Board members felt it was more prudent to wait until the fees for the extra services could be included in the contract before approving it, understanding there would be some time delay. Ms. Witham will also ask the attorney to strengthen the contract language regarding the use of environmentally designed alternatives and soliciting a diverse workforce per Board of Education member requests.

Ms. Witham suggested that the costs would be less, between 7% to 8.5%, and that negotiations were occurring on the 2011 projects and planning for those in 2012, which will be brought to the Board of Education soon.

Ms. Witham gave the example of a public versus a nonpublic bid as referenced in the contract. Should asbestos be found and in need of removal, a proposal would be provided but it did not have to be bid out. Legat would help the District to understand the proposal and, if appropriate, test for services, and monitor the work.

A complete contract will come back to the Board of Education for approval at its Special Board meeting on February 15, 2011.

Mr. Finnegan moved to approve the application for the School Energy Efficiency Project Grant; seconded by Ms. McCormack. A roll call vote resulted in six ayes. Motion carried. Dr. Millard had departed.

Legat Architects assisted in the submission of this ISBE School Energy Efficiency Proposal Project Grant. The caveat is that the ISBE may not have enough funds to meet all of the application requests and, thus, the District may not be selected as a recipient.

Mr. Finnegan moved to approve the Summer School Stipends, Tuition and Dates (Schedule C); seconded by Ms. McCormack. A roll call vote resulted in three ayes and three nayes. Motion failed. Ms. Patchak-Layman, Dr. Lee, and Mr. Conway voted nay. Dr. Millard had departed.

Mr. Allen moved to approve the Summer School Stipends, Tuition and Dates (Schedule A); seconded by Mr. Conway. A roll call vote resulted in four ayes and two nays. Mr. Finnegan and Ms. McCormack voted nay. Motion carried. Dr. Millard had departed.

School Energy Efficiency Project Grant

2011 Summer School

Stipends:	No increase
Dates:	June 15 through July 22, 2011
Tuition:	\$180.00

# Team

Advisory Leadership Ms. Patchak-Layman moved to approve the moved to accept DLT's recommendations to the Board of Education concerning the ALT with the amendment that rather than have four community members, they would be representatives of the Boosters, APPLE, Citizens' Council, and the Alumni Association; seconded by Dr. Lee. A roll call vote resulted in four naves and two ayes. Ms. Patchak-Layman and Dr. Lee voted aye. Dr. Millard had departed.

> Ms. Patchak-Layman wanted parents to be part of this process, as they understood the school, had the financial expertise, and were willing to do the work. Citizens' Council was always available to be part of things and it would positive for the community to be included. She also opposed it being a closed process. While understanding her purpose, Mr. Allen felt her concerns would be met through presentations to the Board of Education. Ms. Patchak-Layman asked if the Board of Education intended to amend its original motion last month approving the FAC Report because it said this committee would advise and report to the Board of Education. This recommendation is a change from that which the Board of Education had approved. If its purpose is advisory to the Board of Education, then one cannot legislate that it be a closed meeting. Ms. Witham responded that the FAC would report monthly to the Board of Education via a committee meeting.

> Mr. Hunter appreciated the interest in specifying from where the committee members came. The reason that the committee has worked so well is that people from different arenas have been present. Mr. Isoye stated that the original intent of FAC was to look for committee members who had a deep understanding of future planning and this model allowed for flexibility. Ms. Patchak-Layman reiterated that the expertise already existed within all of the parent groups.

Mr. Conway moved to approve the Superintendent's recommendation for the Structure of the Advisory Leadership Team, as presented; seconded by Ms. McCormack. A roll call vote resulted in five ayes and one nay. Motion carried. Ms. Patchak-Layman voted nay and Dr. Millard had departed.

Personnel Recommendations Dr. Lee moved to approve the personnel recommendations, as presented; seconded by Ms. McCormack. A roll call vote resulted in five aves. Motion carried. Mr. Conway as absent from this vote and Dr. Millard had departed.

The tutor in the In-school Suspension Center is a non-certified position because it is similar to that of an aide who provides educational support. When the job description was written 10 years ago, it was classified as a CPA position. The present job description is aligned with the previous one.

**Ratification and** Adoption

Mr. Allen moved to approve the Ratification and Adoption of Written Assurances of Provisions, as amended; seconded by Dr. Lee. A roll call vote resulted in five ayes. Motion carried. Mr. Conway was absent from this vote.

	Page 126, Item B: Delete "and the Principal of the High School." Ms. Halliman reported that the District's attorney had recommended the
	amendments to the Written Assurances of Provisions so that it was aligned with the law. The last part of this document includes procedures that not only comply with the law but also go beyond it. Realignment will occur again when there is a change in the law or in the District's procedures. The ISBE in Springfield will keep a copy of this.
	Discussion ensued about aligning the Board of Education Policy 6500, Special Education, to these as well.
Minutes	Mr. Finnegan moved to approve the open and closed session minutes of December 16, 2010, and January 5, and 18, 2011 a Declaration that the Audiotapes of the closed sessions in April 2009 be destroyed; seconded by Dr. Lee. A roll call vote resulted in five ayes and one nay. Ms. Patchak-Layman voted nay and Dr. Millard had departed ayes. Motion carried.
	Ms. Patchak-Layman regularly votes no because the Board of Education does not have a process to review the closed session tapes.
School Calendar 2011-12	Mr. Finnegan moved to approve the school Calendar for the 2011-12 school year, as presented; seconded by Dr. Lee. A roll call vote resulted in five ayes. Mr. Conway was absent from this vote. Motion carried.
	Mr. Allen appreciated the fact that the District would honor Veterans' Day on the actual national holiday date.
Superintendent Contractual Goals	Dr. Lee moved to approve the indicators of measure to be included in Appendix B of the Contract between Dr. Steve T. Isoye and the Board of Education dated April 15, 2010 and amended September 23, 2010, as presented; seconded by Ms. McCormack. A roll call vote resulted in six ayes. Motion carried. Dr. Millard had departed.
	As this was the first year of the contract, Ms. Patchak-Layman stated that these transitional indicators. She hoped that they would be more specific in the future, so that the community can pinpoint the work being done through the indicators.
Closed Session	At 8:40 p.m., on Thursday, January 27, 2011, Mr. Allen moved to go into closed session to discuss The appointment, employment, compensation, discipline, performance, or dismissal of specific employees of the District or legal counsel for the District, including hearing testimony on a complaint lodged against an employee or against legal counsel for the District to
	determine its validity. $5 \text{ ILCS } 120/2(c)(1)$ , as amended by PA.93—57; Litigation, when an action against, affecting or on behalf of the particular District has been filed and is pending before a court or administrative tribunal, or when the District finds that an action is probably or imminent, in which
	case the basis for the finding shall be recorded and entered into the closed meeting minutes 5 ILCS $120/2(c)(11)$ ; Collective negotiating matters between the District and its employees or their representatives or deliberations concerning salary schedules for one or more classes of employees. The

	placement of individual students in special education programs and other matters relating to individual students 5 ILCS 120/2(c)(11) 5 ILCS 120/2(c)(2); Student disciplinary cases 5 ILCS 120/2(c)(10); seconded by Mr. Finnegan. A roll call vote resulted in six yes. Motion carried. Dr. Millard had departed.
	At 9:48 p.m. on Thursday, January 27, 2011, the Board of Education resumed its open session.
Student Discipline	Mr. Finnegan moved to withdraw student RES 01/18/11-01 and assess tuition charges in the amount of \$7681.59, representing tuition from the beginning of this school year through the end of semester one; seconded by Dr. Lee. A roll call vote resulted in five ayes and one nay. Motion carried. Ms. Patchak-Layman voted nay. Dr. Millard had departed.
B&G Letters of Agreement	No action was taken.
Rescission of Personnel Recommendations	Mr. Finnegan moved to rescind the personnel recommendations; seconded by Mr. Conway. A roll call vote resulted in six ayes. Motion carried. Dr. Millard had departed.
Personnel Recommendations	Mr. Finnegan moved to approve the personnel recommendations based on information presented in open and closed sessions; seconded by Mr. Conway. A roll call vote resulted in six ayes. Motion carried. Dr. Millard had departed.
Adjournment	At 9:48 p.m. on Thursday, January 27, 2011, Mr. Conway moved to adjourn the Board of Education meeting; seconded by Dr. Lee. A roll call vote resulted in six ayes. Motion carried. Dr. Millard had departed.

Terry Finnegan Secretary

by Gail Kalmerton Clerk of the Board A special meeting of the Board of Education of the Oak Park and River Forest High School was held on February 8, 2011, in the Board Room of the high school.

Call to OrderVice President Allen called the meeting to order at 7:38 a.m. A roll call<br/>indicated the following members were present: John C. Allen IV, Jacques A.<br/>Conway, Terry Finnegan, Dr. Ralph H. Lee, Amy Leafe McCormack, and<br/>Sharon Patchak-Layman. Also present was Dr. Steven T. Isoye,<br/>Superintendent; Nathaniel L. Rouse (arrived at 8:40 a.m.); Cheryl L. Witham,<br/>Chief Financial Officer, (attended telephonically and departed at 8:33 a.m.); and<br/>Gail Kalmerton, Executive Assistant/Clerk of the Board of Education and FOIA<br/>Officer.

Paul Keller of Ancel Glink, Valerie G. Fisher, consultant, and Lee Brown of Teska & Associates (departed at 8:33 a.m.).

**Closed Session** At 7:40 a.m., Dr. Lee moved to enter to closed session for the purpose of discussing Litigation, when an action against, affecting or on behalf of the particular District has been filed and is pending before a court or administrative tribunal, or when the District finds that an action is probably or imminent, in which case the basis for the finding shall be recorded and entered into the closed meeting minutes 5 ILCS 120/2(c)(11) and Student disciplinary cases 5 ILCS 120/2(c)(10); seconded by Mr. Finnegan. A roll call vote resulted in all ayes. Motion carried.

At 9:13 a.m., the Board of Education resumed its open session.

#### **Student Discipline**

Visitors

Mr. Allen moved to expel Student EXP 02/08/11-07 through August 2012 but to hold the expulsion in abeyance pending successful completion of an alternative educational setting with appropriate social and emotional counseling, and social probation, which includes being banned from all activities. A review of this expulsion will be held in June 2012; seconded by Mr. Finnegan. A roll call vote resulted in two ayes and four nays. Mr. Allen and Mr. Finnegan voted aye. Motion failed.

Dr. Lee moved to expel Student EXP 02/08/11-07 through August 2011 but to hold the expulsion in abeyance pending successful completion of an alternative educational setting with appropriate social and emotional counseling, and social probation, which includes being banned from all activities. A review of this expulsion will be held in June 2011; seconded by Ms. McCormack. A roll call vote resulted in six ayes and one nay. Ms. Patchak-Layman voted nay. Motion carried.

Mr. Allen felt the Board of Education should have supported the administration's original recommendation and while he disagreed with giving a lighter consequence, he supported it by voting yea.

Ms. Patchak-Layman did not want to use discipline as a way of finding the right academic environment for students.

### Adjournment

At 9:20 a.m. on Tuesday, February 08, 2011, Dr. Lee moved to adjourn the Special Board Meeting; seconded by Mr. Allen. A roll call vote resulted in all ayes. Motion carried.

> Terry Finnegan Secretary

By Gail Kalmerton Clerk of the Board A special meeting of the Board of Education of the Oak Park and River Forest High School was held on February 15, 2011, in the Board Room of the high school.

Call to Order President Millard called the meeting to order at 9:55 a.m. A roll call indicated the following members were present: John C. Allen IV, Jacques A. Conway, Terry Finnegan, Dr. Ralph H. Lee, Amy Leafe McCormack, and Sharon Patchak-Layman. Also present was Steven T. Isoye, Superintendent; Cheryl L. Witham, Chief Financial Officer); and Gail Kalmerton, Executive Assistant/Clerk of the Board of Education and FOIA Officer.

Visitors James Paul Hunter, Faculty Senate Executive Committee Chair

Check Distribution Dr. Millard moved to approve the Check Distribution List dated February 15, 2011 (attached to and made a part of the minutes of the meeting); seconded by Dr. Lee A roll call vote resulted in all ayes. Motion carried.

ConstructionDr. Millard moved to accept Henry Bros. Company as the ConstructionManagerManagement Firm as presented for 2011 Summer Construction Projects;Bidseconded by Mr. Finnegan. A roll call vote resulted in all ayes. Motion<br/>carried.

Closed Session At 9:58 a.m., Dr. Lee moved to enter to closed session for the purpose of discussing Litigation, when an action against, affecting or on behalf of the particular District has been filed and is pending before a court or administrative tribunal, or when the District finds that an action is probably or imminent, in which case the basis for the finding shall be recorded and entered into the closed meeting minutes 5 ILCS 120/2(c)(11); seconded by Mr. Finnegan. A roll call vote resulted in all ayes. Motion carried.

At 10:55 a.m., the Board of Education resumed its open session.

Adjournment At 10:56 a.m. on Tuesday, February 15, 2011, Mr. Allen moved to adjourn the Special Board Meeting; seconded by Dr. Lee. A roll call vote resulted in all ayes. Motion carried.

Terry Finnegan Secretary

By Gail Kalmerton Clerk of the Board